[Different first page setting changed from on in original to off in modified.].

CCS Network Code

Contents

Section A <u> </u>	<u>24</u>
Section BGovernance	7 <u>11</u>
Section CConnection	40 <u>67</u>
Section DNetwork Structure and Planning	4682
Section ENetwork Use and Capacity	4784
Section FNetwork Design and Specification	67 <u>117</u>
Section G Industrial Common Interface Procedures	77 138
Section HCharges, Invoicing and Payment	<u>82</u> 145
Section I <u>_</u> Data	97 169
Section J <u>–</u> General	<u>109</u> 184
Section K <u>_</u> Glossary	121 199
Exhibit A Code Accession Agreement	<u>143</u> 232
Exhibit B Construction Agreement	<u>148</u> 242
Exhibit C Connection Agreement	<u>165</u> 270
Exhibit D CDS Accession Agreement	<u>172</u> 284
Annexure A – Carbon Dioxide Specifications - General	<u>173</u> 289
Annexure B – Carbon Dioxide Specifications – T&S Networks	<u>175</u> 291
Annexure C – Measurement Requirements	<u>178</u> 292
Annexure D – CO ₂ Quality Monitoring Procedure Terms of Reference	<u>180</u> 293
Annexure E – Technical Parameters – T&S Networks	<u>181</u> 297
Annexure F – Emergency Procedure Terms of Reference	182 298
Annexure G – Start-up and Shut-down Procedure Terms of Reference	183 299
Annexure H – Isolation Procedure Terms of Reference	<u>185300</u>
Annexure I – Minimum Documentation Requirements	<u>186303</u>
Annexure J – Information Publication Requirements	<u>187304</u>
Annexure K – Retained Data	188 <u>305</u>
Annexure L – CDS Terms of Reference	<u>189</u> 306
Annexure M – Independent Verifier Terms of Reference	<u>190</u> 313
Annexure N – T&S Network Portal Minimum Specification	191 324



CCS Network Code

<u>Section A Section A – Introduction, structure and interpretation</u>

1. Introduction

1.1 This document is the CCS Network Code prepared and maintained by each T&SCo pursuant to Standard Licence Condition [x] of itsB5 (CCS Network Code) of the Licence (the "Code").

Note: the conditions of T&SCo's Licence will require each T&SCo to have a network code—with anacknowledgement that this single uniform Code will satisfy that requirement, unless otherwise specified in the T&SCo's licence.

- 1.2 The Code <u>is divided into different sections and</u> sets out the commercial and technical rules and arrangements between <u>the</u> T&SCos and Users, and between individual T&SCos, in relation to:
 - (a) the connection by Users to a T&S Network, or part of a T&S Network;
 - (b) the delivery of carbon dioxide by Users into the T&S Network at a Delivery Point;
 - (c) the transportation and storage of carbon dioxide delivered by Users at Delivery Points;
 - (d) the operation and maintenance of each T&S Network; and
 - (e) the interface between the T&SCos in relation to different T&S Networks or parts of a T&S Network.
- 1.3 This Section A (Introduction, Structure and Interpretation) also deals with the requirement for each User to enter into a Connection Agreement, a Construction Agreement and a Code Accession Agreement.

2. Parties

- 2.1 For the purposes of the Code:
 - (a) **"T&SCo"** means, in relation to each T&S Network or part of a T&S Network, the entity that is the holder of a Licence authorising it to operate that T&S Network or part of a T&S Network; and
 - (b) <u>"User"</u> means <u>a personan entity</u> other than <u>a T&SCo</u> who is <u>for the time being</u> bound by the <u>provisions of this Code and delivers carbon dioxide into the T&S Network at a Delivery Point.Code;</u>

2.2 For the purposes of the Code:

- (c) (a) a "Party" means a T&SCo or a User;
- (d) (b) in the context of any Ancillary Agreement, references to "Parties" are to the T&SCo or a User who is party to that agreement;

- (e) where a right or obligation is imposed on a T&SCo, then, unless otherwise specified, that right or obligation relates to the T&S Network, or part of a T&S Network, to which the Licence held by that T&SCo relates;
- (f) (d) a reference to the T&SCos is a reference to all T&SCos collectively; and
- (g) anything required to be done jointly by the T&SCos which is done by or on behalf of all of them shall be treated as having been done by all of them, provided that any T&SCo's failure to comply with such joint obligation will not place any other T&SCo in breach of the Code where such other T&SCo has co-operated in relation to the joint obligation.

3. Contractual effect

3.1 The Code is given legal effect by and made binding on a T&SCo or a User pursuant to the Code Agreement or the Code Accession Agreement, as the case may be.

T&SCo

3.2 Each T&SCo shall enter into:

- 4. Additional Code Parties
- 4.1 The Parties must admit as an additional party to the Code Agreement any person who:
 - (a) accepts a Connection Offer from the T&SCo in accordance with Section C
 (Connection) and who has satisfied the additional requirements set out in paragraph
 4.7 (a "New User"); or
 - (b) holds a Licence and satisfies the additional requirements set out in paragraph 4.6 (a "New T&SCo"),

(each a "New Code Party") and who is not at the time a Party.

4.2 (a) the Admission of a New Code Agreement or Party shall take effect by way of a Code Accession Agreement; and prepared by:

(b) the CDS Contract or a CDS Accession Agreement.

User accession

- (a) in the case of a New T&SCo, an existing T&SCo nominated by the Regulator; and
- (b) in the case of a New User, the T&SCo that owns and operates the T&S Network to which the New User will connect,

at the expense and cost of the New Code Party and to be executed by the New Code Party and the relevant existing T&SCo (being the T&SCo referred to in paragraph 4.2(a) or 4.2(b), as the case may be) for itself and on behalf of all other Parties.

4.3 Upon execution of the Code Accession Agreement by the relevant T&SCo under paragraph 4.2 and the New Code Party, subject to and in accordance with the terms and

- conditions of that Code Accession Agreement, the New Code Party shall become a Party for all purposes of the Code Agreement.
- 4.4 Each Party hereby authorises and instructs the relevant T&SCo under paragraph 4.2 to sign any such Code Accession Agreement on its behalf and undertakes not to withdraw, qualify or remove any such authority or instruction at any time.
- 4.5 The relevant T&SCo under paragraph 4.2 must promptly notify the Secretary under Section B

 (Governance) that the New Code Party has become a Party. Such notification must be by written notice (which may be sent electronically) and notify the name, registered address and capacities in which the New Code Party will, or intends to, be connected to or use the T&S Network.
- 4.6 In order to become a New T&SCo in relation to a T&S Network, a person must enter into a CDS Accession Agreement.
- 4.7 3.3 In order to become a New User in relation to a T&S Network, a person must satisfy or secure satisfaction of the following requirements:
 - (a) the applicant shallmust have entered into:
 - (i) the Code Agreement or a Code Accession Agreement;
 - (i) (ii) a Construction Agreement with the T&SCo; and
 - (ii) (iii) a Connection Agreement with the T&SCo,

and each document shall have become effective in accordance with its terms;

- (b) the applicant shallmust have provided to the T&SCo, in the form prescribed by the T&SCo from time to time, the applicant's name, legal nature and address, and any other information that is required to be provided as part of the connection process set out in Section C (Connection); and
- (c) the applicant shallmust have provided the emergency contact details required under Section G (Industrial procedures Section G (Common Interface Procedures).
- 4.8 Unless otherwise expressly stated or unless the context otherwise requires:
 - (a) <u>a User must comply with the obligations under the Code from the date it accedes to the Code, including during any period prior to that User's Commencement Date; and</u>
 - (b) <u>a T&SCo must comply with the obligations under the Code from the date it accedes to the Code, including any period prior to that T&SCo's Commercial Operations Date.</u>
- 5. 4. Structure
- 5.1 4.1 The Code comprises:
 - (a) the following sections of the main body:

- (i) this Section ASection A: Introduction, structure and interpretationStructure and Interpretation;
- (ii) Section BSection B: Governance;
- (iii) Section C: Connection;
- (iv) Section D: Network structure Structure and planning Planning;
- (v) Section ESection E: Network use Use and capacity Capacity;
- (vi) Section F Section F: Network design Design and specification Specification;
- (vii) Section G: Industrial procedures Section G: Common Interface Procedures;
- (viii) Section HSection H: Charges, invoicing Invoicing and payment Payment;
- (ix) Section I: Data;
- (x) Section JSection J: General; and
- (xi) Section K: Glossary.;
- (b) the following exhibits:
 - (i) Exhibit A Exhibit A: Code Accession Agreement;
 - (ii) Exhibit B Exhibit B: Construction Agreement;
 - (iii) Exhibit CExhibit C: Connection Agreement; and
 - (iv) Exhibit DExhibit D: CDS Accession Agreement and

Note: the Code Agreement, which will be signed by the Original T&SCos and the Initial Users, is included as an exhibit to the Code Accession Agreement.

- (c) the following annexures:
 - (i) Annexure AAnnexure A: Carbon Dioxide Specifications General;
 - (ii) Annexure BAnnexure B: Carbon Dioxide Specifications (Northern Endurance Partnership T&S Network) Networks;
 - (iii) Annexure C: Carbon Dioxide Specifications (HyNet T&S Network);
 - (iii) (iv) Annexure DAnnexure C: Measurement Requirements (Northern Endurance Partnership T&S Network) Networks;
 - (iv) Annexure D: CO2 Quality Monitoring Procedure Terms of Reference;
 - (v) Annexure E: Measurement Requirements (HyNet Annexure E: Technical Parameters T&S Network) Networks;

- (vi) Annexure FAnnexure F: Emergency Procedure Terms of Reference;
- (vii) Annexure G: Commissioning Annexure G: Start-up and Shut-down Procedure Terms of Reference;
- (viii) Annexure HAnnexure H: Isolation Procedure Terms of Reference;
- (ix) Annexure I: Minimum Documentation Requirements;
- (x) Annexure JAnnexure J: Information Publication Requirements;
- (xi) Annexure KAnnexure K: Retained Data;
- (xii) Annexure LAnnexure L: CDS Terms of Reference;
- (xiii) Annexure MAnnexure M: Independent Verifier Terms of Reference; and
- (xiv) Annexure NAnnexure N: T&S Network Portal Terms of Reference Minimum Specification.
- 4.2 Section A to Section J, Exhibit A to Exhibit D and Annexure A, Annexure F, Annexure G, Annexure H, Annexure I, Annexure J, Annexure K, Annexure L, Annexure M are referred to as The following are the "Uniform Provisions" and apply equally to all T&S Networks, while Annexure B to Annexure E, referred to as: Section A to Section K, the prescribed forms set out in Exhibit A to Exhibit D and Annexure A, Annexure D, Annexure F, Annexure G, Annexure H, Annexure I, Annexure J, Annexure K, Annexure L, Annexure M, and Annexure N.
- 5.3 The following are the "Specific Provisions", and are specific to each individual T&S Network: Annexure B, Annexure C and Annexure E.
- 4.3 The Exhibits include the prescribed forms for the Ancillary Agreements (the "Ancillary Agreement Templates").
- 4.4 Where applicable, parties are required to enter into Ancillary Agreements must be in or substantially in the form of the template set out in the relevant Exhibit and Ancillary Agreement Templates and must be prepared in accordance with any "Drafting instruction" set out in the template (save Ancillary Agreement Templates (provided that where the contents of the schedules in the Ancillary Agreements require further population, any information to be populated shallmust be specific to each User and/or connection). Any substantial departures from the template of an Ancillary Agreement as set out in the relevant Exhibit Templates (including to the terms of any schedule and any "Drafting instruction") shall be dealt with via the modification procedure under Section B are only permitted where such substantial departures are the subject of an approved Modification under Section B (Governance) or where they have been approved by the Regulator in accordance with the T&SCo's Licence.
- Each and every Ancillary Agreement entered into by a User and a T&SCo and in force from time to time shall constitute a separate agreement governed by the terms of the Code and will be read and construed accordingly.

- 5.7 For the purposes of any Ancillary Agreement:
 - (a) no T&SCo shall enjoy any rights nor incur any obligations against any other T&SCo; and
 - (b) <u>no User shall enjoy any rights nor incur any obligations against any other User.</u>
- 6. 5. Interpretation general
- 6.1 <u>In the Code and in each Ancillary Agreement:</u>
 - (a) the interpretation rules in this paragraph 6; and
 - (b) the words and expressions defined in paragraph 7,

shall, unless the subject matter or context otherwise requires or is inconsistent therewith, apply.

- Save as otherwise expressly provided in the Code, in the event of any inconsistency between the provisions of any Ancillary Agreement and the Code, the provisions of the Ancillary Agreement shall prevail in relation to the Delivery Point which is the subject thereof to the extent that the rights and obligations of Users not party to that Ancillary Agreement are not affected.
- 6.3 5.1-Unless the context otherwise requires, words and expressions defined in the Act or the Standard-Licence Conditions and not otherwise defined in the Code shall have the meaning given to them in the Act or the Standard-Licence Conditions.
- 5.2 References to the Code (or any part thereof) are to the Code (or such part thereof) as from time to time modified in accordance with the Modification Rules.
- 6.5 5.3 A reference in the context of any provision of the Code to a "representative" of any person is a reference to any director, officer or employee of that person or any agent, consultant or contractor appointed or engaged by that person for purposes connected with the subject matter of the relevant provision of the Code.
- 6.6 5.4 An "Affiliate", in relation to a partyParty, is any entity that directly or indirectly controls, is controlled by, or is under common control with that partyParty from time to time and "control" has the meaning given in section 1124 of the Corporation Tax Act 2010 and controls, controlled and the expression "change of control" shall be interpreted accordingly.
- 6.7 <u>In the Code and in each Ancillary Agreement:</u>
 - (a) unless the context otherwise requires all references to a particular paragraph, Part,
 Section, Schedule, Exhibit or Annexure shall be a reference to that paragraph, Part,
 Section, Schedule, Exhibit or Annexure in or to the Code and all references to a
 particular Appendix shall be a reference to that Appendix to the relevant Ancillary
 Agreement;
 - (b) unless the context otherwise requires, all references to a particular paragraph within a Section shall be a reference to that paragraph within that Section;

- <u>a table of contents and headings are inserted for convenience only and shall be</u> ignored in construing the Code or any Ancillary Agreement;
- (d) references to the words "include" or "including" are to be construed without limitation to the generality of the preceding words;
- (e) unless the context otherwise requires, any reference to an Act of Parliament or any part or section or other provision of or schedule to an Act of Parliament shall be construed, at the particular time, as including a reference to any modification, extension or re-enactment thereof then in force and to all instruments, orders or regulations then in force and made under or deriving validity from the relevant Act of Parliament; and
- references to the masculine shall include the feminine and references in the singular shall include the plural and vice versa and words denoting persons shall include any individual, partnership, firm, company, corporation, joint venture, trust, association, organisation or other entity, in each case whether or not having separate legal personality.

7. 6. Interpretation – times-and, dates and measures

For the purposes of the Code:

- (a) "Day" means the 24 hour period from 00:00 hours (midnight) until 2324:5900 hours;
- (b) "Business Day" means a Day other than:
 - (i) a Saturday or a Sunday; or
 - (ii) a Day which begins at 00:00 hours on a bank holiday in England, Wales or Scotland the nation where the T&S Network is located;
- (c) <u>"Calendar Week"</u> means a period of 7 <u>consecutive</u> Days starting <u>at 00:00 hours</u> on <u>a</u> Monday;
- (d) <u>"Charging Year"</u> means the period from <u>00:00 on</u> 1 April in any year until and including <u>24:00 hours on</u> 31 March in the following year;
- (e) unless the context otherwise requires, "carbon dioxide" means a stream consisting primarily of carbon dioxide (CO₂) molecules and references to carbon dioxide being delivered by a User to the T&S Network assume that the gaseous stream being delivered is carbon dioxide;
- (e) (f)-references to:
 - (i) a week is to the period from 00:00 hours on a day until 2324:5900 hours on the 7th day following;
 - (ii) a month (or a number of months) is to the period from 00:00 hours on a dayDay in one (1) month until 00:00 hours on the same day of the month which

- follows (or follows by the relevant number of months), or if there is no such dayDay in such month 00:00 on the first dayDay of the next following month;
- (iii) a calendar month is to the period from 00:00 hours on the first day Day of a month until 00:00 hours on the first day Day of the following month, and references to a particular calendar month (such as January) shall be construed accordingly;
- (iv) a year is to the period from 00:00 hours on a dayDay in one <a href="mailto:(1) year until 00:00 hours on the same dayDay (or where the day in the first year was 29 February, on 1 March) in the following year;
- (v) a calendar year (such as 2022) is to be construed accordingly; and
- (vi) each reference in this the Code to any time of the day shall, unless otherwise stated, be construed as a reference to Londonofficial time. in the United Kingdom;
- (f) <u>in relation to any Charging Year ("Charging Year t"):</u>
 - (i) Charging Year t-1 means the Charging Year before it; and
 - (ii) Charging Year t+1 means the Charging Year after it;
- (c) <u>a reference to t or tonne is a reference to a metric tonne; and</u>
- (g) unless the context otherwise requires, capitalised terms have the meanings set out in Section K (Glossary).

CCS Network Code

Section B - Governance

1. Modification Panel

1.1 Each T&SCo, together with the other T&SCos, is required to operate a procedure for the Modification of the CCS Network Code so as to better facilitate the achievement of the Relevant Objectives in accordance with the terms of its Licence.

1.2 The procedure is to provide for:

- (a) a mechanism by which the Code and each Specific Provision may be modified and/or reviewed;
- (b) the making of proposals for the modification of the Code either by a T&SCo, a User, a Third Party Participant, the Secretary of State or, in certain circumstances, the Regulator:
- where a proposal is made for the modification of the Code, the making of alternative proposals for the modification of the Code by any T&SCo, User, Third Party

 Participant or Secretary of State and, in certain circumstances, the Regulator, other than the person who raised the original modification proposal;
- determination by the Modification Panel whether a proposal for modification of the Code (other than a proposal made by the Regulator in the circumstances set out in paragraph 23.3) is a Self-Governance Modification Proposal;
- (e) determination by the Modification Panel whether a proposal for modification of the Code (other than a proposal made by the Regulator in the circumstances set out in paragraph 23.3) satisfies the Fast Track Self-Governance Criteria;
- <u>a right of appeal in respect of any determination by the Modification Panel to implement or not to implement a Self-Governance Modification Proposal;</u>
- (g) the giving of adequate publicity to any such proposals;
- (h) the seeking of the view of the Regulator on any matter connected with any such proposals other than Self-Governance Modification Proposals, Significant Code

 Review Modification Proposals made by the Regulator in the circumstances specified at paragraph 23.3 and SoS Modification Proposals;
- (i) the consideration of any representations relating to such proposals (excluding Significant Code Review Modification Proposals made by the Regulator in the circumstances specified at paragraph 23.3 and SoS Modification Proposals) made (and not withdrawn) by a T&SCo, a User, a Third-Party Participant or Non-Code Parties invited to make such representations by the Modification Panel;

where the Regulator accepts that the Code may require modification as a matter of urgency, the exclusion, acceleration or other variation, subject to the Regulator's approval, of any particular procedural steps which would otherwise be applicable;

- (k) the making of Significant Code Review Modification Proposals by the Regulator and the co-ordination of the processes for modification of the Code with Significant Code Reviews;
- (I) the making of SoS Modification Proposals by the Secretary of State during the Interim
 Period and the co-ordination of the processes for modification of the Code with SoS
 Modifications; and
- (m) the making of determinations regarding the Resource Plan and Budget Forecast and Secretary Expenses Claims to be submitted under paragraph 7.
- 1.3 4.2 The T&SCos shallmust ensure that a Modification Panel is established within three (3) months from the Code Implementation Date. The Modification Panel will be comprised of:
 - (a) the Panel Chairperson, being a non-voting Member;
 - (b) <u>subject to paragraph 11.8, a maximum of six</u> (6) T&SCo Representatives, being Voting Members;
 - (c) <u>subject to paragraph 11.7, a maximum of six</u> (6) User Representatives, being Voting Members, with at least one <u>(1)</u> representative of each User Type;
 - (d) one (1) representative of the Authority Regulator, being a non-voting Member; and
 - (e) one (1) representative of the Secretary of State, being a non-voting Member-

the persons appointing the Members under paragraphs 1.3(b) and 1.3(c) being the "Panel Constituency" in respect of such Members.

- Subject to paragraph 1.5, it is expected that each Voting Member must, as appropriate, represent and inform the Modification Panel of the views of that Member's appointor (or appointors) in relation to Modification Proposals.
- Member must represent and inform the Modification Panel of the views of the T&SCo, User or User Type (as applicable) of which that Voting Member is a part, in relation to Modification Proposals.
- 1.6 Any appointment of a Voting Member must expressly include provisions which require that Voting Member to carry out the duties of the Voting Members as set out in this Section B (Governance).

2. Panel Chairperson

2.1 The Members must appoint (and as the case may be remove and reappoint) by Panel Majority and by notice to the Secretary, one (1) person from time to time as the:

- (a) Panel Chairperson; and
- (b) the Deputy Panel Chairperson.
- 2.2 1.3 The Panel Chairperson and Deputy Panel Chairperson shall be appointed by majority vote of for a two (2) year period commencing on 1 October and ending with effect from the second anniversary of such date provided that, the first appointment period shall commence on the date which the Modification Panel and shall serve as Panel Chairperson until such time as their appointment expires in accordance with paragraph 1.31 is established and end on 30 September the following year.
- 2.3 The Panel Chairperson and Deputy Panel Chairperson must at all times act and take decisions in an impartial, objective and balanced manner.
- 2.4 The Panel Chairperson or, in the absence of the Panel Chairperson, the Deputy Panel
 Chairperson shall preside at meetings of the Modification Panel. If, in the period directly
 following the Code Implementation Date only, a Panel Chairperson has not yet been
 appointed under paragraph 2.1, the Secretary shall preside at the first and any subsequent
 meeting of the Modification Panel until the Panel Chairperson has been appointed.
- 2.5 The Panel Chairperson must conduct all meetings of the Modification Panel in compliance with the Code.

3. T&SCo Representatives

- 1.4 Subject to paragraph 1.13, the T&SCos shall appoint (and as the case may be re-appoint) up tosix (6) T&SCo Representatives to the Modification Panel.
- 3.1 Subject to paragraphs 3.3 and 11.2, each T&SCo must appoint one (1) T&SCo Representative to the Modification Panel.
- 3.2 1.5 Subject to paragraph 1.6 and 1.7, each T&SCo will have an equal representation of T&SCo Representatives and in such circumstances 11.2(b), each T&SCo shall have absolute discretion over the individuals individual that it appoints as the T&SCo Representative(s) representing that T&SCo, provided that such individual is required at all times to be an employee, representative, officer or contractor of the T&SCo and, subject to paragraph 3.4, does not hold a Conflicting Appointment.
- 3.3 1.6-Subject to paragraph 1.7, where 11.8, the number of T&SCos may appoint (and as the limit case may be re-appoint) a maximum of six (6) T&SCo Representatives prevents equal representation in accordance with paragraph 1.5, each T&SCo shall have at least one T&SCo-Representative (the identify of whom shall be determined in that T&SCo's absolute discretion) and the T&SCos shall collectively by agreement appoint any additional T&SCo-Representatives to reach 6 T&SCo Representatives to the Modification Panel.
- 1.7 Where the number of T&SCos and the limit of six (6) T&SCo Representatives prevents each T&SCo having at least one T&SCo Representative, the T&SCos shall collectively by agreement appoint the T&SCo Representatives.

1.8 The Authority shall (itself or through a designated independent person appointed by the Authority) appoint the T&SCo Representatives where the T&SCos are unable to collectively agree upon-the identity of any T&SCo Representatives under paragraph 1.6, 1.7 or 1.39.

- 3.4 Where a T&SCo proposes to appoint an individual under paragraph 3.1 who holds a Conflicting Appointment, the T&SCo must:
 - (a) notify the Secretary and the Secretary must notify the Regulator of the nature of any such appointment;
 - (b) provide satisfactory evidence of the confidentiality and conflict management processes which are in place to prevent a conflict of interest arising; and
 - not give notice to the Secretary of such appointment until such time as the Regulator has provided its approval in writing of such proposed appointment, which may be subject to conditions.

4. Users' Representatives

- 4.1 1.9 Subject to paragraph 1.13, the Users shall paragraphs 4.3 and 11.5, each User must appoint one (and as the case may be re-appoint) up to six (6) User Representatives 1) User Representative to the Modification Panel.
- 4.2 1.10-Subject to paragraph 1.11 and 1.12, each User will have an equal representation of User Representatives and in such circumstances each 11.5(b), each User shall have absolute discretion over the individualsindividual that it appoints as the User Representative(s) representing that User, provided that such individual is required at all times to be an employee, representative, officer or contractor of the User and, subject to paragraph 4.4, does not hold a Conflicting Appointment.
- 4.3

 1.11-Subject to paragraph 1.12, where 11.7, the number of Users may appoint (and as the limit case may be re-appoint) a maximum of six (6) User Representatives prevents equal representation in accordance with paragraph 1.10, each User shall have at least one User Representative (the identity of whom shall be determined in that User's absolute discretion) and the Users shall collectively by agreement appoint any additional User Representatives to reach 6 User Representatives to the Modification Panel.
- 1.12 Where the number of Users and the limit of 6 User Representatives prevents each User having at least one User Representative, the Users shall collectively by agreement appoint the User-Representatives, provided that there is at least one User Representative for each User Type.
- 1.13 Where there are more than six (6) User Types, from the date upon which any such additional User Types accede to the Code, the number of User Representatives and T&SCo-Representatives on the Modification Panel shall be increased to the number of User Types (and where such accession occurs after the Modification Panel has been established in accordance with paragraph 1.2, the process in paragraphs 1.4 to 1.14 shall apply to the appointment of any additional User Representatives and/or T&SCo Representatives).
- 1.14 The Authority shall (itself or through a designated independent person appointed by the Authority) appoint the relevant User Representatives where the Users (or Users of a User Type) are

unable to collectively agree upon the identity of any User Representatives under paragraph 1.11, 1.12 or 1.40.

- 4.4 Where a User proposes to appoint an individual under paragraph 4.1 who holds a Conflicting Appointment, the User must:
 - (a) notify the Secretary and the Secretary must notify the Regulator of the nature of any such appointment;
 - (b) <u>provide satisfactory evidence of the confidentiality and conflict management</u> processes which are in place to prevent a conflict of interest arising; and
 - (c) not give notice to the Secretary of such appointment until such time as the Regulator has provided its approval in writing of such proposed appointment which may be subject to conditions.

5. **Regulator Representative**

- 5.1 The Regulator may, from time to time, by notice to the Secretary identify the individual to be appointed as the Regulator's representative on the Modification Panel. The Regulator shall have absolute discretion over the individual that it appoints as the Regulator's representative.
- 5.2 The Regulator may, from time to time, revoke the appointment of an individual appointed as the Regulator's representative pursuant to paragraph 5.1 and, in its absolute discretion, identify a new individual to take over the role as the Regulator's representative.

6. Secretary of State Representative

- 6.1 The Secretary of State may, from time to time, by notice to the Secretary identify the individual to be appointed as the Secretary of State's representative on the Modification Panel. The Secretary of State shall have absolute discretion over the individual that it appoints as the Secretary of State's representative.
- 6.2 The Secretary of State may, from time to time, revoke the appointment of an individual appointed as the Secretary of State's representative pursuant to paragraph 6.1 and, in its absolute discretion, identify a new individual to take over the role as Secretary of State's representative.

7. Secretary

- 1.15 The T&SCos shall collectively appoint an individual representative to act as the Secretary. The Secretary shall:
- 7.1 Subject to paragraph 7.3, the T&SCos must jointly appoint an individual representative to act as the Secretary within two (2) months from the Code Implementation Date. The Secretary must:
 - (a) attend meetings of the Modification Panel;
 - (b) carry out the various tasks and duties assigned to it in this <u>Section BSection B</u>
 (Governance) in an efficient, effective and economic manner; and

(c) at all times act and take decisions in an impartial, objective and balanced manner and so as to act on behalf of all Parties.

- 1.16 The Authority shall (itself or through a designated independent person appointed by the Authority) appoint the Secretary where the T&SCos are unable to collectively agree upon the identity of the Secretary under paragraph 1.15.
- The Secretary shall be appointed for a two (2) year period commencing on 1 April and ending with effect from the second anniversary of such date, provided that the first appointment period shall commence on the date which the Secretary is appointed under paragraph 7.1 and end on the following 31 March.
- <u>7.3</u> Where, at the time of appointment under paragraph 7.1 there is only one (1) T&SCo, that T&SCo must solely appoint the Secretary for the first appointment period.
- 7.4 Any appointment of the Secretary under paragraph 7.1 must expressly include provisions which:
 - <u>require the Secretary to carry out the duties of the Secretary as set out in this Section</u>
 B (Governance);
 - (b) make clear that any duties of care that the Secretary owes to Users will be owed to
 Users whether or not they are a party to any agreement governing such appointment;
 - (c) make clear that any duties of care that the Secretary owes to the T&SCos will be owed to the T&SCos whether or not they are a party to any agreement governing such appointment; and
 - (d) require the Secretary to enable future T&SCos to accede to any agreement governing such appointment.
- 7.5 The T&SCos may at any time jointly revoke the appointment of an individual as the Secretary and appoint a new individual to the role.
- 7.6 The Secretary must notify the Regulator of any changes in the composition of the Modification Panel.
- 7.7 <u>The Secretary must:</u>
 - (a) develop a CCS Network Code Website within three (3) months of appointment under paragraph 7.1 for the purposes of publishing various Code related documents:
 - (b) notify each User, each T&SCo, each Third Party Participant, the Regulator and the Secretary of State in writing of the domain name of the CCS Network Code Website upon the CCS Network Code Website going live;
 - (c) <u>maintain a list on the CCS Network Code Website of:</u>
 - <u>(i)</u> <u>all Parties to the Code and the dates upon which such Parties acceded to the Code; and</u>

- <u>(ii)</u> <u>all Third Party Participants and Designated Non-Code Parties and the dates of designation of them as such;</u>
- (d) upon written notice by a T&SCo under paragraph 4.5 of Section A (Introduction,
 Structure and Interpretation), update the list of Parties to the Code under paragraph
 7.7(c) to include any New Code Party; and
- (e) otherwise maintain and update the CCS Network Code Website.
- 1.17 Each T&SCo shall be responsible for funding an equal proportion of the costs and expenditure incurred by the Secretary in the carrying out of the various tasks and duties assigned to it in this Section B (Governance), provided that where paragraph 7.3 applies, the costs of the Secretary shall solely be borne for the first appointment period by the appointing T&SCo.
- 7.9 1.18 By:
 - (a) the date that is one (1) month after the Code Implementation Date date upon which the Secretary is first appointed under paragraph 7.1 in relation to the first Charging Year; and
 - (b) 30th October each Charging Year thereafter,

the Secretary shallmust prepare and submit to the Modification Panel and the AuthorityRegulator for the Modification Panel's approval:

- (c) a Resource Plan; and
- (d) Budget Forecast,

(the "Resource Plan and Budget Forecast") in accordance with the requirements of paragraphs 1.197.10 and 1.207.11.

- 7.10 1.19 The "Resource Plan shall" must include provision for the level of resources that the Secretary considers necessary to undertake its functions, including any:
 - (a) suitable facilities for business purposes;
 - (b) office equipment, including telephones and other communication devices or computer workstations;
 - (c) office furniture; and
 - (d) any applications developed for internal or external business purposes, including the CCS Network Code Website to be established and maintained under paragraph 7.7.
- 7.11 1.20 The "Budget Forecast shall" must set out on a quarterly basis:
 - (a) the itemised budget forecast expenditure for each line item of resource to be provided;

(b) the estimated aggregate funds required to implement the resource plan Resource Plan; and

- (c) such other costs and expenses to be incurred or likely to be incurred by the Secretary.
- 7.12 1.21 The Modification Panel shallmust, as quickly as practicable and in any event within ten (10) Business Days of receipt of the Resource Plan and Budget Forecast, determine whether or not to approve the proposed Resource Plan and Budget Forecast and shallmust within such ten (10) Business Day period either notify the Secretary of its approval or, in the event that the Modification Panel considers changes or variations are required and reasonably withholds its approval, notify the Secretary of any changes or variations to the proposed Resource Plan and Budget Forecast recommended by the Modification Panel. If the Secretary does not accept such changes or variations submitted by the Modification Panel, any such dispute or difference shallmust be referred to the AuthorityRegulator to be determined by the AuthorityRegulator.
- 7.13 1.22 Within one (1) month of the end of each Quarter within a Charging Year, the Secretary shallmust prepare and submit to the Modification Panel and the Authority an Expenses Claim Regulator an expenses claim setting out the details of any Secretary Expenditure (net of VAT) during that Charging Year Quarter, together with reasonable supporting evidence in relation to such expenses ("Secretary's Expenses Claim").
- 7.14 1.23 On receipt of the Secretary's Expenses Claim, the Modification Panel shallmust, within ten (10) Business Days prepare and circulate to the Parties a statement setting out for the relevant Quarter within the Charging Year its determination of:
 - (a) the budgeted expenditure under the Resource Plan;
 - (b) taxes or duties chargeable, if any;
 - (c) the proportion of the Secretary Expenditure payable by, each PartyT&SCo in accordance with the principle of equal apportionment as set out in paragraph 1.177.8; and
 - (d) any other information the Modification Panel, in its reasonable discretion, considers relevant,

(the "Draft Statement of Expenses").

- 7.15 1.24 If a Party disputes the Draft Statement of Expenses (or any part of it), it shallmust, within five (5) Business Days of the date of the Draft Statement of Expenses notify the Modification Panel and the Secretary with details of the dispute (an "Expenses Dispute Notification") and, within a further five (5) Business Days, provide such supporting evidence as the Modification Panel may reasonably require.
- 7.16 4.25 Within five (5) Business Days of the notice period for disputes under paragraph 4.247.15 expiring, the Modification Panel shallmust:

(a) (where no Expenses Dispute Notification has been made) notify the Parties and the Secretary that the Draft Statement of Expenses is final; or

(b) (where one (1) or more Expenses Dispute Notifications have been made), circulate to the Parties and the Secretary a revised statement of expenses which shallmust reflect the undisputed part of the SecretarySecretary's Expenses Claim only,

(the "Final Statement of Expenses").

- 7.17 4.26 The Modification Panel shallmust consider all disputes under paragraph 4.247.15 and shallmust make a determination, which shallmust be notified to the Parties and the Secretary within fifteen (15) Business Days of the notice period for disputes under paragraph 4.247.15 expiring.
- 7.18 1.27 Determinations under paragraph 1.26 shall 7.17 must be reflected in the next following Draft Statement of Expenses to be issued under paragraph 1.23 7.14, and any Party that made an Expenses Dispute Notification may not make an Expenses Dispute Notification in respect of that part or parts of the Draft Statement of Expenses that were was subject to the earlier dispute.
- 7.19 4.28 Following receipt by the Modification Panel of the Final Statement of Expenses, the Secretary may invoice each of the T&SCos for its proportion of the Secretary Expenditure, together with VAT, for each Quarter within the Charging Year, as set out in the Final Statement of Expenses.
- 7.20 4.29 Each T&SCo shallmust pay any amount due under paragraph 4.287.19 to the Secretary within thirty (30) days Days of the amount being invoiced to it.
- 1.30 The allocation of funding between the Parties under paragraph 1.17 shall be reviewed by the Authority once every two years from the Code Implementation Date.

8. Appointments and retirements

- 8.1 1.31 Each Voting Member shall, by notice to the Secretary, be appointed for a two (2) year period commencing on 1 October and ending with effect from the second anniversary of such date ("Appointment Period"), provided that the first Appointment Period shall commence on the date of such appointment as indicated in the notice to the Secretary and end on the following 30 September.
- 8.2 Each Voting Member shall, prior to attending their first meeting of the Modification Panel, confirm by notice to the Secretary (in such form as the Secretary may reasonably request) that the Voting Member either:
 - (a) does not hold a Conflicting Appointment; or
 - (b) holds a Conflicting Appointment but such Conflicting Appointment has been approved by the Regulator in accordance with paragraph 3.4 or 4.4,

and until such time as the Voting Member has submitted such notice, the Voting Member shall not be entitled to exercise any voting rights.

- 8.3 4.32 Each Voting Member shall retire at end of their Appointment Period.
- 8.4 1.33-Each retiring Member shall be eligible for re-appointment, save that where additional Voting Members have been appointed pursuant to paragraphs 1.6, 1.7, 1.8 (in the case of T&SCo Representatives), or paragraphs 1.11, 1.12 or 1.14 (in the case of User Representatives) and subject always to paragraphs 1.39 and 1.40, such Voting Members will be appointed for a maximum of one Appointment Period.
- 4.34-Any Voting Member who is, pursuant to paragraph 4.328.3 to retire and who is not to be re-appointed (and consequently in respect of whom no notice shall have been received pursuant to paragraphs 4.358.6 or paragraph 1.368.7) shall not be entitled to receive notice of any meeting of the Modification Panel which is to take place after the end of their Appointment Period.
- 8.6 1.35 In respect of each individual who is a T&SCo Representative, the T&SCos shallmust, not later than 1 September in the year following the year in respect of which the T&SCo's Representative's six (6) months prior to the expiry of the T&SCo Representative's Appointment Period commences, notify the Secretary (copying the notice to the Regulator):
 - (a) that such individual is to be re-appointed as a T&SCo Representative; or
 - (b) that such individual is not to be so re-appointed and the identity of the individual to be appointed as a T&SCo Representative in accordance with paragraph 3.3.
- 8.7 1.36 In respect of each individual who is a User Representative, each User shall the Users must, not later than 1st September in the year following the year in respect of which the Users' Representative's six (6) months prior to the expiry of the User Representative's Appointment Period commences, notify the Secretary (copying the notice to the AuthorityRegulator):
 - (a) that such individual is to be re-appointed as a User Representative; or
 - (b) that such individual is not to be so re-appointed at the relevant meeting and the identity of the individual to be appointed as a User Representative in accordance with paragraph 4.3.
- 4.37 Where any notice is received pursuant to paragraphs 1.358.6 or 1.368.7, the appointment or, as the case may be, the re-appointment to which such notice relates shall have effect from the 1 October specified in such notice. Each individual who is to be appointed or, as the case may be, reappointed as a Member at that meeting shall be entitled to receive notice of any meeting of the Modification Panel which is to take place after the commencement of their Appointment Period.
- 4.38 If notices shall be received in accordance with paragraph 4.358.6 or 4.368.7 are in respect of less than six (6) individuals to be appointed (or, as the case may be, reappointed) as User Representatives or T&SCo Representatives (as applicable), all individuals identified in such notices shall, pursuant to paragraph 4.358.6 or 4.368.7 (as applicable) be appointed (or, as the case may be, reappointed) and the individuals who are, pursuant to paragraph 4.328.3, to retire and in respect of whom no notice shall have has been received pursuant to

paragraph 1.35(a) or paragraph 1.36(a) 8.6(a) or 8.7(a), shall retire at the end of their Appointment Period.

- 1.39 If less than six (6) individuals are appointed (or, as the case may be, reappointed) as T&SCo-Representatives the T&SCos shall collectively by agreement appoint the individuals to act as additional T&SCo Representatives (and paragraph 1.32 shall not apply to any such appointment).
- 1.40 If less than six (6) individuals are appointed (or, as the case may be, reappointed) as User-Representatives the Users shall collectively by agreement appoint the individuals to act as additional User Representatives (and paragraph 1.32 shall not apply to any such appointment).

9. Alternates

- 9.1 <u>Each Voting Member shall appoint two (2) individuals (including another Member other than</u> the Panel Chairperson) to be the Voting Member's alternate.
- 9.2 Each non-Voting Member may appoint an individual (which may be another Member, but not the Panel Chairperson) to be the Member's alternate.
- 9.3 The appointment (and revocation of the appointment) of an alternate must be conditional upon and shall only be effective upon:
 - (a) <u>in respect of an appointment, receipt of notice by the Secretary (in such form as the Secretary may reasonably request):</u>
 - (i) from the Member identifying the individual to be appointed as the alternate; and
 - (ii) from the alternate to confirm a willingness to act as the Member's alternate and (in the case of an alternate to a Voting Member) to confirm the alternate either (A) does not hold a Conflicting Appointment or (B) holds a Conflicting Appointment but such Conflicting Appointment has been approved by the Regulator in accordance with paragraph 3.4 or 4.4; and
 - (b) in respect of a revocation of appointment, receipt of notice by the Secretary (in such form as the Secretary may reasonably request):
 - (i) of the Member's notice of revocation of the alternate's appointment; or
 - (ii) <u>from the alternates confirming they no longer wish to act as an alternate.</u>
- 9.4 A Member who is, by reason of also being an alternate of a Voting Member, entitled to exercise more than one (1) vote shall not be required to exercise all the votes which that Member is entitled to exercise, or to exercise all of the votes which that Member is entitled to exercise in the same way.
- 9.5 In addition to notices sent to Members, each alternate for the time being shall be entitled to be sent notices.

An alternate may attend any meeting of the Modification Panel which is not also attended by the Member (in their capacity as a Member) who appointed them. If that alternate is the alternate of a Voting Member, they may also vote and generally at any such meeting shall have and shall be able to exercise and discharge any and all of the functions, powers and duties of the Member who shall have appointed that alternate; provided that in the event both alternates of a Voting Member attend only the first to attend the meeting shall be able to vote, and the alternate second to attend shall have observer status only. Alternates of Voting Members may sign written resolutions pursuant to paragraph 18.3, provided that if an alternate of a Voting Member and the Voting Member who appointed them shall sign a written resolution, the signature of the Voting Member shall be effective and the signature of the alternate shall be disregarded.

- 9.7 If a Member ceases, for whatever reason, to be a Member, the appointment of any alternate of the Member shall lapse with immediate effect; provided that if any Member retires but is reappointed, any appointment made by that Member pursuant to paragraph 9.1 which is, and continues to be, effective prior to the retirement of that Member, shall continue to be effective after that Member's re-appointment.
- 9.8 Where an alternate to a Voting Member ceases to be an alternate in accordance with paragraph 10.5 or 10.6 or notifies the Voting Member that they no longer wish to be an alternate or has their appointment revoked, the Voting Member must appoint a replacement alternate as soon as reasonably practicable.
- 10. Ceasing to be a Member
- 10.1 A Member shall, other than by way of retirement pursuant to paragraph 1.328.3, cease to be a Member where:
 - (a) the Member notifies the Secretary that they wish to cease to be a Member; or
 - (b) in the case of a Voting Member:
 - (i) (b) in the case of a Voting Member, the Member (or its alternate) is absent from three (3) consecutive monthly meetings (convened in accordance with paragraph 1.48 paragraphs 13.1 and 14.1) of the Modification Panel.
- 1.42 If at any time after the Modification Panel has been established in accordance with paragraph 1.2:
 - (a) an existing User has more than one Voting Member;
 - (ii) the Member notifies the Panel Chairperson that they hold (or will hold) a Conflicting Appointment in accordance with paragraph 10.5 or
 - <u>the Panel Chairperson determines the Member holds (or will hold) a Conflicting Appointment in accordance with paragraph 10.6,</u>

and the Secretary must notify the other Members and such other relevant persons as the Secretary determines, on receipt of a notice under paragraph 10.1(a) or on the occurrence of any of the circumstances referred to in paragraph 10.1(b).

Where a Member ceases to be employed or engaged by the same specified entity as at the time of nomination, the appointing T&SCo or User must procure that such Member is required (as soon as is practicable) to notify the Secretary of such change in circumstances and the date from which such change took, or will, take effect.

- <u>Where a notification is received by the Secretary under paragraph 10.2, that Member shall cease to be a Member with effect from the later of:</u>
 - (a) the date from which that Member will no longer be employed or engaged by the relevant entity; or
 - (b) and/or a new User accedes to the Code, the day following the date the Secretary is informed, or made aware, of such change in circumstances.

such existing User or Users shall, by notice to the Secretary in accordance with paragraph 1.41, remove one or more Voting Members from their membership as is necessary to maintain the principle of equal representation set out in paragraph 1.10, based on the new number of Users (and the principles set out in paragraphs 1.4 to 1.14 shall apply to any replacement Voting Members).

- 1.43 If at any time after the Modification Panel has been established in accordance with paragraph1.2:
 - (a) an existing T&SCo has more than one Voting Member; and
 - (b) a new T&SCo accedes to the Code,

such existing T&SCos shall by notice to the Secretary in accordance with paragraph 1.41, remove one or more Voting Members from their membership as is necessary to maintain the principle of equal representation set out in paragraph 1.10, based on the new number of T&SCos (and the principles set out in paragraphs 1.4 to 1.14 shall apply to any replacement Voting Members).

- 10.4 A Voting Member or its alternate holds a "Conflicting Appointment" where:
 - (a) the Voting Member or alternate is employed or engaged by, or represents in any capacity an organisation (or an Associate or representative of such organisation); and
 - (b) another Voting Member or its alternate appointed to represent a different Panel

 Constituency is employed or engaged by, or represents in any capacity that same organisation (or an Associate or representative of that same organisation).
- Unless the appointment has been approved by the Regulator pursuant to paragraph 3.4 or
 4.4, where a Voting Member or an alternate holds a Conflicting Appointment, the Voting

 Member or alternate must promptly notify the Panel Chairperson and the Secretary, and the

 Secretary must notify the other Members and such other relevant persons as the Secretary

 determines, and the Voting Member or alternate shall cease to be a Member or an alternate
 with immediate effect.

Where a Member has bona fide grounds for believing a Voting Member or alternate holds (or will hold) a Conflicting Appointment, the Member may notify the Panel Chairperson, and the Panel Chairperson may investigate the matter; and where the Panel Chairperson determines that (in its reasonable opinion) the Voting Member or alternate holds (or will hold) a Conflicting Appointment the Panel Chairperson must notify the Secretary and the Secretary must notify the Voting Member or the alternate, the other Members (and such other relevant persons as the Secretary determines) and the Voting Member or the alternate shall cease to be a Member or an alternate with immediate effect.

- 10.7 For the purposes of an investigation in accordance with paragraph 10.6, the Panel Chairperson may engage a suitably qualified third party to assist with the investigation.
- 10.8 Where a Voting Member or an alternate ceases to be a Member in accordance with paragraph 10.5 or 10.6, the relevant individual may not:
 - (a) act as a Member or as an alternate until such time as there is no longer a Conflicting

 Appointment or such Conflicting Appointment has been approved by the Regulator in accordance with paragraph 3.4 or 4.4; or
 - (b) be appointed to fill the vacancy arising as a result of the Voting Member or alternate ceasing to be a Voting Member or alternate.
- 10.9 1.44 Where pursuant to paragraph 1.41 10.1 or otherwise a vacancy shall arise, subject always to the principles set out in paragraphs 1.4 to 1.14 3, 4 and 11, in the case of:
 - (a) a T&SCo Representative, the T&SCos shallmust, by notice to the Secretary, identify another individual to be appointed as a T&SCo Representative in accordance with paragraph 3.3; or
 - (b) a User Representative, the Users shallmust, by notice to the Secretary, identify another individual to be appointed as a User Representative in accordance with paragraph 4.3.

11. New Code Parties

- At any time after the Modification Panel has been established in accordance with paragraph

 1.3 where a new T&SCo accedes to the Code:
 - (a) subject to paragraph 11.2, such new T&SCo shall be entitled to appoint a T&SCo Representative in accordance with paragraph 3.1; and
 - (b) the Appointment Period for any such new T&SCo Representative shall commence on the date notice is provided to the Secretary of such appointment under paragraph 8.1 and end on the following 30 September.
- At any time after the Modification Panel has been established in accordance with paragraph

 1.3, where a new T&SCo accedes to the Code and the limit of six (6) T&SCo Representatives

 prevents each T&SCo from having one (1) T&SCo Representative in accordance with

 paragraph 3.1:

<u>each of the existing T&SCo Representatives must give notice to the Secretary in</u> accordance with paragraph 10.1 that it wishes to cease to be a Member;

- (b) the T&SCos (including the T&SCo that has recently acceded to the Code) shall collectively by ballot, vote to appoint the six (6) T&SCo Representatives; and
- (c) the Appointment Period for any such newly appointed Voting Members shall commence on the date notice is provided to the Secretary of such appointment under paragraph 8.1 and end on 30 September the following year.
- Where upon completion of the voting process under paragraph 11.2(b), two (2) or more individuals have received the same number of votes, the ballot process will be repeated. A new ballot paper will be circulated to each T&SCo containing only the names of the individuals who received the same number of votes.
- 11.4 At any time after the Modification Panel has been established in accordance with paragraph 1.3, where a new User accedes to the Code:
 - (a) <u>subject to paragraph 11.5, such new User shall be entitled to appoint a User Representative in accordance with paragraph 4.1; and the subject to paragraph 11.5, such new User shall be entitled to appoint a User Representative in accordance with paragraph 4.1; and</u>
 - (b) the Appointment Period for any such new User Representative shall commence on the date notice is provided to the Secretary of such appointment under paragraph 8.1 and end on 30 September the following year.
- At any time after the Modification Panel has been established in accordance with paragraph

 1.3, where a new User accedes to the Code and the limit of six (6) User Representatives

 prevents each User from having one (1) User Representative in accordance with paragraph

 4.1:
 - (a) each of the existing User Representatives must give notice to the Secretary in accordance with paragraph 10.1 that it wishes to cease to be a Member;
 - (b) provided that there is at least one (1) representative of each User Type, the Users (including the User that has recently acceded to the Code) shall collectively by ballot, vote to appoint the six (6) User Representatives; and
 - (c) the Appointment Period for any such newly appointed Voting Members shall commence on the date notice is provided to the Secretary of such appointment under paragraph 8.1 and end on 30 September the following year.
- Where upon completion of the voting process under paragraph 11.5(b), two (2) or more individuals have received the same number of votes, the ballot process will be repeated. A new ballot paper will be circulated to each User containing only the names of the individuals who received the same number of votes.
- Where there are more than six (6) User Types, from the date upon which any such additional

 User Types accede to the Code, the maximum number of User Representatives on the

 Modification Panel shall be increased to the number of User Types (and where such accession occurs after the Modification Panel has been established in accordance with

paragraph 1.3, the process in paragraph 4.1 shall apply to the appointment of any additional <u>User Representatives</u> where there is one (1) <u>User of each User Type and in all other cases</u>, paragraph 11.5 shall apply to the appointment of <u>User Representatives</u>).

11.8 If the number User Representatives on the Modification Panel is increased pursuant to paragraph 11.7, the maximum number of T&SCo Representatives on the Modification Panel shall be increased to equal the increased number of User Representatives (and where such accession occurs after the Modification Panel has been established in accordance with paragraph 1.3, the process in paragraph 11.2 shall apply to the appointment of T&SCo Representatives).

12. Meetings of the Panel

- 12.1 4.45 Meetings of the Modification Panel will provide a forum in which Modifications Proposals can be discussed in accordance with the Modification Procedures.
- 12.2 1.46-Except as otherwise permitted in the Code:
 - (a) determinations of the Modification Panel shallmust be made by Panel Majority; and
 - (b) Workgroups may be created or dissolved by Panel Majority,

where "Panel Majority" shall meanmeans:

- (c) a majority (in number) of the votes exercised by the Voting Members present at that meeting and voting in favour of such matter; or
- (d) if there is an equal number of votes by the Voting Members present at that meeting in favour of and against such matter, where the representative of the Authority Panel Chairperson elects to exercise a casting vote Casting Vote in favour of such matter.
- 12.3 Other than as expressly provided in the Modification Rules, the Modification Panel shall have no ability to determine any matter and no competence to discharge any function or to exercise any power.

13. Frequency of meetings

13.1 1.47 The Secretary shallmust convene meetings of the Modification Panel by notice to the Members and Panel Chairperson not less frequently than once each month unless there is no matter as an agenda Agenda item for the Modification Panel to discuss. In any event, a meeting of the Modification Panel willmust be convened once every three (3) calendar months.

14. Notice convening meetings

1.48 The Secretary shall be responsible for providing notices of meetings with agendas and materials in relation to the matters to be considered at the meeting.

Subject to paragraphs 15.1, 15.2 and 41.2 and 42.1, the Secretary must not convene meetings of the Modification Panel pursuant to paragraph 13.1 on less than ten (10) Business Days notice.

- Every notice convening a meeting of the Modification Panel pursuant to paragraph 13.1 must specify the place, date and time of the meeting and enclose an Agenda. All relevant materials in respect of a meeting of the Modification Panel must be circulated by the Secretary not less than five (5) Business Days prior to the meeting to which they relate, subject to paragraph 41.2 and 42.1 or paragraph 14.4 where in the opinion of the Secretary a shorter period would better facilitate the exercise by the Modification Panel of its powers, within such shorter period as the Secretary shall determine.
- Each Member and the Panel Chairperson shall be entitled to receive each notice and the relevant materials. At the same time as any notice is despatched to Members and the Panel Chairperson, a copy of such notice must be despatched by the Secretary to each User, each T&SCo, each Third Party Participant and each Designated Non-Code Party. There may be circumstances where materials to be despatched with a notice have, pursuant to the Modification Rules, already been sent to Users, the T&SCos, Third Party Participants and Designated Non-Code Parties. In any such circumstance the relevant materials may be, but do not have to be, sent with the notice.
- Notwithstanding where the Secretary has determined a shorter period to circulate the relevant materials in respect of a meeting of the Modification Panel pursuant to paragraph 14.2, no new items to the Agenda shall be considered except by determination of the Modification Panel.

15. **Short notice**

- The Modification Panel may at any meeting of the Modification Panel determine that the next following meeting of the Modification Panel be convened on shorter notice than specified in paragraph 14.1 and where the Modification Panel has so determined, the Secretary must convene a meeting of the Modification Panel in accordance with that determination.
- Without prejudice to paragraph 14.1, if a Panel Majority agree in writing, the Secretary must convene a meeting of the Modification Panel on shorter notice than specified in paragraph 14.1.
- Any meeting of the Modification Panel convened pursuant to paragraph 15.1 or paragraph 15.2 shall, notwithstanding that such meeting is convened on shorter notice than that specified in paragraph 14.1, be duly convened.

16. Location and Form of Meetings

- 16.1 4.49 Meetings of the Modification Panel may take place by any audio, audio-visual or interactive communication, provided that each Member shall be able to communicate to each of the other Members and be heard by each of the other Members simultaneously.
- 16.2 1.50 Any Member who shall be able to participate in the manner envisaged by paragraph
 1.49 16.1 in any meeting of the Modification Panel shall be treated as being present at such meeting, notwithstanding that the Members may not all be meeting in the same place and

accordingly shall, if such Member is a <u>votingVoting</u> Member, be entitled to vote and shall count towards a quorum.

1.51 Where any Member is unable to attend a meeting of the Modification Panel such Member shall-be entitled to appoint another person as their proxy to exercise all or any of their rights to attend and to speak and vote at such meeting, provided that notice of such appointment is given to the Secretary no later than twenty-four (24) hours prior to such meeting taking place.

17. Quorum

- 1.52 Eight (8) Members, of whom four (4) shall be T&SCo Representatives and four (4) shall be User-Representatives present at a meeting of the Modification Panel shall be a quorum.
- The persons required to be present at a meeting of the Modification Panel for the meeting to be quorate are:
 - (a) a majority of the number of T&SCo Representatives appointed at the time of the meeting; and
 - (b) <u>a majority of the number of User Representatives appointed at the time of the meeting, including at least one (1) User Representative for each User Type.</u>
- If a quorum is not present at the time for the holding of a meeting (specified in the notice convening the meeting) or at any time during the hour following that time or shall at any time during the meeting cease to be present for more than fifteen (15) minutes, the meeting shall stand adjourned to the same place and at the time specified in the notice convening the meeting five (5) Business Days later and the Secretary must notify each Member and (for information purposes only) each User, T&SCo, Third Party Participant and Designated

 Non-Code Party that such is the case. If at such place and time the meeting so adjourned is not quorate in accordance with paragraph 17.1 the Voting Members (if any) present shall be a quorum.
- Any meeting of the Modification Panel at which a quorum is and remains present shall be competent to discharge any and all of the functions within the competence of the Modification Panel.

18. Voting

- 18.1 4.53 Any matter expressed to require a determination of the Modification Panel shallmust be determined by a vote.
- 18.2 1.54 On any vote, each Voting Member present shall be entitled to exercise at least one (1) vote. Where there is:
 - (a) an equal number of T&SCo Representatives and User Representatives present, each Voting member Member shall be entitled to exercise one (1) vote; and
 - (b) an unequal number of T&SCo Representatives and User Representatives present, votes shall be weighted such that:

(i) there shall be an equal number of voting rights between T&SCo Representatives and User Representatives; and

- (ii) <u>between</u> each T&SCo Representative <u>there shall be equally weighted voting</u> <u>rights;</u> and
- <u>between each</u> User Representative <u>has</u>there shall be equally weighted voting rights.
- 1.55 Subject to paragraph 1.56, it is expected that each Voting Member shall, as appropriate, represent and inform the Modification Panel of the views of that Member's appointor (or appointors) in relation to Modification Proposals.
- 1.56 Where a Voting Member has been appointed by the Authority pursuant to paragraph 1.8 or 1.14, such Voting Member shall represent and inform the Modification Panel of the views of the T&SCo, User or User Type (as applicable) of which that Voting Member is a part, in relation to Modification Proposals.
- 18.3

 1.57-A resolution in writing signed in wet ink or electronically by Voting Members shall be valid and effective for the purposes of discharging any function requiring a determination of the Modification Panel as if such vote were conducted in accordance with paragraph 1.5318.1 at a duly convened meeting of the Modification Panel, provided that votes are received by the Secretary from such Members as would, if present, form a quorum in accordance with paragraph 1.5217.1 not later than three (3) Business Days (or such shorter period as the Secretary may reasonably notify) following receipt by Voting Members of such documents as are necessary for the purposes of such written resolution.

19. Observers and Invitees

- 19.1 1.58 Any Third Party Participant and any Designated Non-Code Party is entitled to have one
 (1) representative attending a meeting of the Modification Panel, subject to the terms of their
 designation by the Regulator. The Panel Chairperson may also at its discretion, authorise any
 additional individual (who is not a Member) to attend a meeting of the Modification Panel asan observer on behalf of a User-or, a T&SCo, a Third Party Participant, the Secretary of State
 or a Designated Non-Code Party and to participate in the business of the meeting.
- In exercising its right to authorise any individual to attend a meeting of the Modification Panel in accordance with paragraph 19.1, the Panel Chairperson must have regard to the principle of maintaining (where reasonably practicable) equality of representation in any one (1) meeting.
- 19.3 1.59 The Modification Panel and/or the Authority may, from time to time, determine to invite any individual to attend all or part of a meeting of the Modification Panel.
- 19.4 1.60 The Secretary may, at the written request of the Regulator, invite any individual to a meeting or meetings of the Modification Panel. Where such an invitation is made, the Secretary shallmust give as much notice as reasonably practicable to Members of the Modification Panel of the name of the individual invited, the organisation that the individual represents and the date of the relevant meeting(s). Where the AuthorityRegulator requests

the Secretary to invite any individual, the Secretary shallmust set out in writing to the AuthorityRegulator details of any objections that the Parties may have to such attendance.

19.5 1.61 Any invitee to a meeting of the Modification Panel pursuant to paragraph 1.60 19.3 or 19.4 shall be entitled to receive copies of any relevant minutes, agendas, notices and Modification Proposals due to be discussed at that meeting (but shall not and shall, at the discretion of the Panel Chairperson, be entitled to participate in the business of the meeting).

20. Minutes

20.1 The Secretary must:

- (a) ensure that all meetings of the Modification Panel and all determinations of the Modification Panel (at meetings of the Modification Panel) are minuted and, as regards such determinations, that the minutes record the manner in which each Voting Member cast their vote in respect of each matter determined by the Modification Panel; and
- (b) issue a report of determinations of the Modification Panel to each Member, each

 T&SCo, each User and each Third Party Participant within three (3) Business Days of the meeting of the Modification Panel at which such determinations were made.
- 20.2 The Secretary must, within five (5) Business Days following the date of the relevant meeting, send each Member, each T&SCo, each User and each Third Party Participant a copy of any minutes of that meeting made pursuant to paragraph 20.1.

21. **Provision of information**

- 21.1 The Modification Panel may, from time to time, determine to provide to any person a document derived from the application of these Modification Rules (including any minutes made pursuant to paragraph 20.1), or considered in accordance with the Modification Rules.
- 21.2 If the Modification Panel determines, the Secretary must ensure that within a reasonable period of time the document which is the subject of such determination shall be sent to such person.

22. 2. Third Party Participants

- 22.1 The Authority shall have Regulator has the right (but not the obligation) to designate, from time to time, any person or body who is not a party Party to the Code but who is representative of interested third parties to participate in the Modification Procedure as a third party participant ("Third Party Participant"), in accordance with Standard Condition B5.7 (CCS Network Code) of the Licence.
- 22.2 Third Party Participants may include (but are not limited to), a future User and/or future

 T&SCo as identified under thea Selection Process.
- 22.3 Any such designation made under paragraph 2.1, shall 22.1 must be made in writing, and shall must be maintained on a register held by the AuthorityRegulator.

22.4 The Regulator shall be entitled to de-designate any persons designated under paragraph 22.1 at any time.

23. 3. Modification Procedure Proposals

Modification Proposals

- 23.1 3.1-A Modification Proposal in respect of the Code may be made from time to time by:
 - (a) a T&SCo;
 - (b) a User;
 - (c) a Third Party Participant;
 - (d) the AuthorityRegulator, where such Modification Proposal is a Significant Code Review Modification Proposal or an Urgent Modification Proposal; or
 - (e) the Secretary of State.
- 23.2 The AuthorityRegulator may direct a T&SCo to make a Significant Code Review Modification Proposal in respect of the Code and the T&SCo shallmust make a proposal in accordance with that direction and such a proposal shallmust proceed in accordance with the Modification Procedures.
- 23.3 Without prejudice to the generality of paragraph 23.1(d), where the Regulator has issued:
 - (a) a statement (that it will continue work on a Significant Code Review) in accordance with Standard Condition B5.17 (CCS Network Code) of the Licence; or
 - (b) <u>a direction (a backstop direction) in accordance with Standard Condition B5.23 (CCS Network Code) of the Licence,</u>

the Regulator may make a Significant Code Review Modification Proposal.

- 3.3-A T&SCo, a User, or a Third Party Participant may not make a Modification Proposal in respect of the Code during the relevant Significant Code Review Phase if the subject matter of such proposal relates to a matter which is the subject of a Significant Code Review, unless the AuthorityRegulator directs that it may do so, having taken into account, among other things, the urgency of the subject matter of such proposal.
- 24. Form of Modification Proposals
- 24.1 3.4 Each Modification Proposal made pursuant to paragraph 3.123.1:
 - (a) shallmust state the Proposer's view as to:
 - (i) whether it should be a Self-Governance Modification Proposal and the Proposer's reasons for such a view; and

 (ii) whether, if the Proposer's view is that it should be a Self-Governance Modification, it satisfies the Fast Track Self-GovernanceSelf-Governance Procedure and the Proposer's reasons for such a view;

- (b) shallmust, where it is made pursuant to a direction of the AuthorityRegulator, state that it is so made;
- (c) in the case of a Modification which proposes a timescale for the implementation of the Modification (for the purposes of enabling the AuthorityRegulator and any persons, including but not limited to Users, the-T&SCos and Third Party Participants to be aware of the potential benefits or constraints associated with such timing), except in the case where the AuthorityRegulator has directed a timetable in accordance with paragraph 3.7242.1 and/or 3.7342.2, where only one (1) Fixed Implementation Date may be included, shallmust include:
 - (i) two (2) or more Fixed Implementation Dates;
 - (ii) a Proposed <u>AuthorityRegulator</u> Decision Date in respect of each Fixed Implementation Date, for the purposes of enabling the Modification to be implemented by the Fixed Implementation Date;
 - (iii) a Backstop Lead Time; and
 - (iv) the reasons why it is proposing each date under paragraph (i), (ii) and (iii).24.1(c)(i), 24.1(c)(ii) and 24.1(c)(iii);
- (d) shallmust be in writing;
- (e) shallmust set out in reasonable but not excessive detail the case for change and the solution proposed;
- (f) <u>shallmust</u> detail the <u>sections</u> and paragraphs of the Code which are potentially impacted by the Modification Proposal;
- (g) shallmust, if the Proposer considers that the Modification Proposal should be treated as an Urgent Modification Proposal, identify the Modification Proposal as such and indicate the Proposer's justification for such belief;
- (h) shallmust state the name of the Proposer and their contact details;
- (i) shallmust, without prejudice to the Modification Panel's right of determination pursuant to paragraph 3.1428.2, state the Proposer's preference as to whether the Modification Proposal should:
 - (i) proceed to Consultation;
 - (ii) proceed to Workgroup Assessment; or
 - (iii) where paragraph 3.4(a)(ii)24.1(a)(ii) applies, be implemented;

 may state the Proposer's opinion of the likely impact of the implementation of the Modification Proposal upon User's computer systems and/or manual processes and procedures;

- (k) may include the Proposer's suggested text;
- (I) where it is a Significant Code Review made by the Authority shall Regulator must state that it is such; and
- <u>must set out on the basis upon which the Proposer considers that it would better</u>
 <u>facilitate the achievement of the Relevant Objectives; and</u>
- (n) (m) shall must state the proposed timeframe for the implementation of the Modification Proposal.
- 24.2 3.5 The Modification Panel may, from time to time, stipulate the form which Modification Proposals should take.
- 24.3 3.6-The Secretary shall be required to must establish and maintain an online register publishing the content of all Modification Proposals which shall be made publicly available on the CCS Network Code Website to be established under paragraph 7.7.
- 24.4 3.7 Each Modification Proposal shallmust be given to the Secretary who shallmust ensure that the information required pursuant to the Code has been provided, and the form of the Modification is as specified in the Code before accepting such Modification Proposal.
- 24.5 3.8 Where a purported Modification Proposal, does not comply with paragraph 3.2 24.1, the Secretary may reject such a Modification Proposal.
- 24.6 3.9 On receipt of a Modification Proposal, the Secretary shallmust:
 - (a) allocate a unique reference number to that proposal Modification Proposal;
 - (b) update the online register of Modification Proposals established pursuant to paragraph 3.524.3; and
 - (c) by the later of:
 - (i) the end of three (3) Business Days following receipt of a Modification Proposal; or
 - (ii) the end of the first Business Day following the date on which the Secretary receives notification of any decision of the <u>AuthorityRegulator</u> pursuant to paragraph <u>3.7141.2</u> as to whether the Modification Proposal should be treated as an Urgent Modification Proposal,

send a copy of that Modification Proposal to each T&SCo, each User Member, each Third Party Participant, the Secretary of State and the Authority and each Designated Non-Code Party (if any); and

(d) subject to paragraph 3.7141.2, put initial discussion of the Modification Proposal, on the Agenda for the next meeting of the Modification Panel (provided the Modification Proposal is received no later than [x]eight (8) Business Days prior to the date of the next meeting of the Modification Panel). Where there are fewer than eight (8)

Business Days before the next Modification Panel meeting, the discussion of the Modification Proposal shall be put on the Agenda for the following Modification Panel meeting.

25. **Proposer's Representative**

- Subject to paragraph 41.2, each Proposer must ensure the attendance of its representative at the meeting of the Modification Panel at which a Modification Proposal is to be discussed initially; at such meeting the representative may give a presentation in respect of the Modification Proposal and must endeavour to answer any questions which the Modification Panel may have in respect of the Modification Proposal and any presentation given.
- 25.2 If a representative of the Proposer does not for whatever reason attend the meeting of the

 Modification Panel at which the relevant Modification Proposal is to be discussed initially, the

 Modification Panel may determine that, notwithstanding such non-attendance of the
 representative, the Modification Panel shall proceed to discuss the Modification Proposal.
- 25.3 If the Modification Panel does not make a determination pursuant to paragraph 25.2, the Modification Panel will not discuss the Modification Proposal further unless and until the representative of the Proposer has attended a meeting of the Modification Panel pursuant to paragraph 25.1 or the Modification Panel has determined to discuss the Modification Proposal further.

26. **Non-Code Parties**

- 26.1 The Secretary must:
 - (a) send copies of any Modification Proposal and any Modification Report prepared in respect of such Modification Proposal to and invite representations in respect thereof from Designated Non-Code Parties; and
 - (b) publish on the CCS Network Code Website copies of the documents referred to in paragraph 26.1(a) so that they are accessible to any other interested Non-Code Parties.

27. Modification Proposals made during a Significant Code Review Phase

- 27.1 3.10-Where the AuthorityRegulator has received a written assessment of the Modification Panel under paragraph 3.1528.4 in respect of a Modification Proposal, the Secretary shall informmust notify the Proposer and each T&SCo, each User, and each Third Party Participant and each Designated Non-Code Party (if any), of the direction or re-direction it has received from the AuthorityRegulator.
- 27.2
 3.11 Where the direction or re-direction received by the Secretary from the

 Authority Regulator is not to proceed with the Modification Proposal, that Modification

 Proposal shall become a Significant Code Review Suspended Modification Proposal and shall continue to be so until either the end of the Significant Code Review Phase or unless as

the <u>AuthorityRegulator</u> directs otherwise (having taken into account, among other things not limited to, the urgency of the subject matter of such a proposal).

3.12-Where the AuthorityRegulator has not made a direction or has not made a re-direction not to proceed with a Modification Proposal in respect of which the AuthorityRegulator has received a written assessment under paragraph 3.1528.4, such proposal shall not be or shall cease to be a Significant Code Review Suspended Modification Proposal and that Modification Proposal shall proceed in accordance with the Modification Procedures.

28. Discussion of Modification Proposal

- 28.1 3.13 The Subject to paragraph 32.1 and 42.1, the Modification Panel shallmust discuss each new Modification Proposal at a meeting of the Modification Panel. The Modification Panel shallmust make determinations having:
 - (a) discussed the Modification Proposal;
 - (b) heard the presentation of the Proposer's representative (provided that the Modification Panel shall be entitled to determine to proceed where the Proposer's representative fails to attend one and such non-attendance shall not be taken into account in any further determination); and
 - (c) <u>had an opportunity to ask the Proposer's representative questions in respect of the Modification Proposal and the presentation of the Proposer's representative;</u>
 - (d) (c)-considered whether there are any other persons from whom representations should be invited and considered; and
 - (e) determined whether it could be reasonably expected that the Modification Proposal would, if implemented, have a material impact having regard to any statements made by the Proposer pursuant to paragraph 24.1(a) as to whether and why the Modification Proposal should be a Self-Governance Modification Proposal.
- 28.2 3.14 The Modification Panel may determine that a Modification Proposal:
 - (a) either satisfies the Self-Governance Criteria or does not;
 - (b) where the Self-Governance Criteria has been determined to be satisfied, either satisfies the Fast Track Self-Governance Procedure (which must be determined Unanimously by the Modification Panel) or does not;
 - (c) where the Fast-Track Self-Governance Criteria has been determined to be satisfied, the Modification be implemented (which must be determined Unanimously by the Modification Panel);
 - (d) should be referred to a Workgroup for a Workgroup Assessment (and the Modification Panel may determine the <u>Modification</u> Terms of Reference for such work (including terms as to the identity of any third parties to be consulted, which may include Third Party Participants <u>and Non-Code Parties</u>) and the date upon which it requires the Workgroup to submit its Workgroup Report);

- (e) should proceed to Consultation;
- (f) should be deferred to a subsequent meeting of the Modification Panel for further discussions; or
- (g) <u>should</u> be referred back to the Proposer for further development (provided this can only be determined once per Modification Proposal).
- 28.3 3.15 The Modification Panel shall must assess whether the subject matter of a Modification Proposal made by a Proposer other than the Secretary of State during a Significant Code Review Phase relates to a matter that is the subject of an ongoing Significant Code Review and for the purposes of paragraph 23.4.
- 28.4 <u>The Modification Panel must</u> instruct the Secretary to submit to the <u>AuthorityRegulator</u> as soon as is reasonably practicable, a written assessment <u>of a Modification Proposal</u> including:
 - (a) representations received as to whether such proposal relates to the matter which is the subject of a Significant Code Review;
 - (b) its determination (including where no determination can be reached due to a tie in the number of votes for and the number of votes against) as to whether such proposal relates to the matter which is the subject of Significant Code Review;
 - (c) its reasons for making such determination; and
 - (d) a copy of the relevant Modification Proposal.
- 28.5 If the Modification Panel determine that a Modification Proposal satisfies the Fast Track

 Self-Governance Criteria but fails to vote Unanimously to implement that Modification

 Proposal under paragraph 28.2(c), the Modification shall become a Self-Governance

 Modification and be further determined by the Modification Panel under paragraph 28.2.
- 28.6 Where the Modification Panel determines that paragraph 28.2(g) applies they:
 - <u>must set out the questions to be answered and specify any areas for further</u> development by the Proposer; and
 - (b) <u>shall only be entitled to exercise this vote once per Modification Proposal.</u>
- 28.7 Paragraphs 28.1 to 28.6 shall not apply to any Significant Code Review Modification
 Proposals made by the Regulator in the circumstances specified under paragraph 23.3.
- 29. Workgroup Assessment
- 3.16-If the Modification Panel determines that thea Modification Proposal should proceed to Workgroup Assessment then the Secretary shallmust within ten (10) Business Days of such determination finalise the Modification Terms of Reference for that Workgroup Assessment in accordance with paragraph 31 and notify each T&SCo-and, each User, each Third Party Participant and each Designated Non-Code Party (if any) and invite them to make representations in respect of the Modification Proposal within fifteen (15) Business Days.

29.2 3.17 Once the Modification Terms of Reference for the Workgroup Assessment have been finalised:

- (a) the Workgroup shall must meet and consider the Modification Proposal in accordance with the details and timescales set out in the Modification Terms of Reference; Each Workgroup must conduct its business:
 - <u>(i)</u> <u>in such manner as to allow any person to attend and participate in meetings of the Workgroup; and</u>
 - (ii) in accordance with the Code;
- (b) the Secretary shallmust ensure that the AuthorityRegulator is notified of all meetings of the Workgroup and invited to attend;
- (c) the Secretary shallmust ensure that the Secretary of State is notified of all meetings of the Workgroup and invited to attend; and
- the Workgroup shall Secretary must prepare a Workgroup Report containing a recommendation based on the Workgroup's consideration of the Modification Proposal within the timescale set out in the Modification Terms of Reference and send a copy to all Members, Third Party Participants and Designated Non-Code Parties (if any) as soon as reasonably practicable but in any event not less than eight (8) Business Days prior to the meeting of the Modification Panel at which that report is to be discussed.
- 29.3 3.18 The Workgroup Report will need to must comply with the requirements for a Modification Report set out in paragraph 3.4036.1.
- 29.4 If two (2) or more Modification Proposals shall have been considered together by the Workgroup, the Workgroup Report must report on each Modification Proposal.
- Each Workgroup Report must (subject to paragraphs 15.1 and 15.2) be sent to all Members,

 Third Party Participants and Designated Non-Code Parties (if any) as soon as reasonably

 practicable but in any event not less than eight (8) Business Days prior to the meeting of the

 Modification Panel at which that report is to be discussed.
- 30. Discussion of Workgroup Report
- 30.1 <u>Having:</u>
 - (a) <u>discussed the Workgroup Report; and</u>
 - (b) allowed any person who attended and participated in the relevant Workgroup (in attendance at the meeting of the Modification Panel) to express any views on the substance of the Workgroup Report or the conduct of the consideration of the Modification Proposal by the Workgroup,

the Modification Panel must determine:

<u>(c)</u> <u>that the Modification Proposal shall proceed to Consultation in accordance with paragraph 35.1;</u>

- (d) 3.19 Having considered to refer the Workgroup Report, the Modification Panel shall determine if the Modification Proposal shall proceed to Consultation or whether it should be referred back to the Workgroup for revision or further work.; or
- (e) to consider further the Workgroup Report at a subsequent meeting of the Modification Panel.

31. **Modification Terms of Reference**

- 31.1 The Modification Terms of Reference for each Modification Proposal referred to a Workgroup must:
 - (a) <u>detail the Modification Proposal, including;</u>
 - (i) <u>detailing the work to be undertaken by the Workgroup, to enable the Workgroup to prepare its report; and</u>
 - (ii) specifying any matters, in addition to those referred to in the Code, which the Workgroup, should address in its report;
 - (b) <u>detail other matters (if any) to be considered or reviewed by the Workgroup;</u>
 - state whether the Workgroup, should consult with any other person for the purposes
 of its report, and if so, detail the extent to which and identify which person (or persons)
 it should consult with, but this shall not require the Secretary to engage or remunerate
 any person so consulted; and
 - (d) set a timetable in accordance with which the work of the Workgroup is to be done and its report prepared.
- Unless the Modification Panel shall otherwise determine, the timetable referred to in paragraph 31.1(d) shall consist of a period of up to six (6) months. Where the Modification Panel determines that such timetable should be extended so that it exceeds a period of six (6) months in aggregate, it must notify the Regulator and the timetable shall be so extended unless the Regulator objects. Where the Regulator objects to such extension, the Modification Panel shall make a determination under paragraph 28.2 in respect of the relevant Modification Proposal at the next meeting of the Modification Panel.
- 31.3 <u>The Modification Panel may, from time to time, determine:</u>
 - (a) to change the Modification Terms of Reference of any Workgroup; or
 - (b) if the Modification Terms of Reference of the relevant Workgroup so anticipate, that the Workgroup should undertake new or further work or consider new matters (whether or not related to any earlier work undertaken by that Workgroup).

32. Alternative Proposals

3.20 In respect of a Modification Proposal which the Modification Panel pursuant to paragraph 3.14(d)28.2(d) has determined should be referred to a Workgroup, a Party or a Third Party Participant, the Regulator (in the case of Urgent Modification Proposals or a Significant Code Review Modification Proposal) or the Secretary of State, being a person other than the Proposer, may make an alternative Modification Proposal in accordance with paragraph 3.424.1 and the Workgroup shallmust consider an alternative Modification Proposal made under this paragraph 3.2032.1 if it is made no less than five (5) Business Days before the next meeting of the Workgroup. Where there are fewer than five (5) Business Days before such Workgroup meeting, the alternative Modification Proposal will be included on the agenda for the following Workgroup meeting.

- 32.2 3.21 Where a Modification Proposal has been referred to a Workgroup and the Workgroup requests that the Modification Proposal should be amended but the Proposer of the Modification Proposal does not agree to amend the Modification Proposal, any person (other than the Proposer) who is eligible to make an alternative Modification Proposal under paragraph 3.2032.1 may, in accordance with paragraph 3.424.1 make an alternative Modification Proposal under this paragraph 3.2132.2 which shallmust include the amendment.
- 32.3 3.22 Where the Modification Panel has determined a Modification Proposal should be referred to a Workgroup and:
 - (a) the Workgroup Report in respect of such Modification Proposal has been sent to all Members in accordance with paragraph 3.1729.2; or
 - (b) the Modification Panel has made a determination to refer the Workgroup Report in respect of such Modification Proposal back to the Workgroup for revision or further work under paragraph 3.1930.1(d) and such Workgroup Report has been sent to all Members in accordance with paragraph 3.2032.1 after such revision or further work,

an alternative Modification Proposal shall not be made in respect of Modification Proposal or be considered by the Workgroup under paragraph 2.1932.1.

- 32.4 3-23 Where the Modification Panel has determined a Modification Proposal should be referred to a Workgroup under paragraph 3.14(d)28.2(d) and the Workgroup Report in respect of such Modification Proposal has been sent to the Modification Panel, an alternative Modification Proposal shall not be made in respect of such Modification Proposal or be considered by the Workgroup under paragraph 3.2032.1.
- 32.5 3.24 An alternative Modification Proposal may not be made:
 - (a) in respect of a Modification Proposal after the Modification Panel pursuant to paragraph 3.14(e)28.2(e) has determined such Modification Proposal should proceed to Consultation; or
 - (b) where the Authority Regulator directs a T&SCo in writing that an alternative Modification Proposal shall not be made in respect of a Modification Proposal; or

(c) by the Authority Regulator, unless such alternative Modification Proposal is in respect of a Significant Code Review,

in case of limb (b), the T&SCo shallmust notify the Secretary about the direction as soon as reasonably practicable after it has been received by the T&SCo, and the Modification Panel shallmust ensure that a Modification Proposal will not be considered as an alternative Modification Proposal to the relevant Modification Proposal.

3.25 In respect of any Modification Proposal which is withdrawn, any of the parties (except for the Proposer) may, but shall not be required to, either raise an alternative Modification Proposal in accordance with paragraph 3.424.1) or adopt the withdrawn proposal (in which case the adopted proposal shall continue through the Modification Procedures from the point at which it was withdrawn).

33. Withdrawal or variation of Modification Proposals

33.1 3.26 A Proposer may:

- (a) subject to paragraph 3.3233.7, withdraw a Modification Proposal, at any time before the final Modification Report is circulated to the AuthorityRegulator pursuant to paragraph 3.3935.7 or in respect of a Self-Governance Modification Proposal at any time before the Modification Panel makes a determination under paragraph 3.68(a)39.8(a), by notice to the Secretary, and subject to paragraph 3.2032.1, any Modification Proposal so withdrawn shall lapse; or
- (b) subject to paragraph 3.3233.7, vary a Modification Proposal, at any time before the Modification Panel has determined to proceed to Consultation by notice to the Secretary, and subject to paragraphs 3.2032.1 to 3.2532.6, any Modification Proposal so varied shall replace the original Modification Proposal; or
- (c) subject to paragraphs 3.20 to 3.25, 3.32, 3.3332.1 to 32.6, 33.7 and 33.8, request a variation to a Modification Proposal ("Variation Request"), at any time after the Modification Panel has determined to proceed to Consultation pursuant to paragraph 3.14(e)28.2(e) and before the Modification Panel has made a determination in respect of such Modification Proposal pursuant to paragraph 3.38(a)35.6(a), by notice to the Secretary, and any such variation request Variation Request shall contain a description of the nature of the variation.
- 33.2 3.27 The Secretary shallmust, within a reasonable period of time following any withdrawal (including any withdrawal directed by the AuthorityRegulator pursuant to paragraph 3.3333.8), variation or variation requestVariation Request (except where such variation requestVariation Request is made at the Modification Panel meeting) notify each T&SCo, each Member, each User, and each Third Party Participant and each Designated Non-Code Party of such withdrawal, variation or variation requestVariation Request (as the case may be) and update the online register of Modification Proposals to indicate that this is the case.
- 33.3 3.28 Following receipt of the notice given pursuant to paragraph 3.26(c) 33.1(c), the Secretary shallmust submit such variation request Variation Request to the appropriate

Modification Panel which the Proposer may attend for the purpose of explaining the variation request Variation Request.

- 33.4 3.29 Subject to paragraphs 3.2032.1 to 3.25,32.6 and paragraph 3.3333.8, the Modification Proposal shall be varied to replace the original Modification Proposal which shall be deemed withdrawn where the Modification Panel:
 - (a) determines by a <u>unanimous Unanimous</u> vote that the <u>variation request Variation</u>

 <u>Request</u> is immaterial, and in such case the varied Modification Proposal shall continue through the Modification Procedures from the point at which the original Modification Proposal was deemed withdrawn; or
 - (b) does not so determine in accordance with (a)paragraph 33.4(a), and in such case the Modification Panel shall make a determination in respect of the varied Modification Proposal in accordance with paragraph 3.1428.2.
- 33.5 3.30-The Proposer of a variation request Variation Request may withdraw it at any time before the Modification Panel votes in accordance with paragraph 3.2933.4.
- 33.6 3.31 A Modification Proposal made by a User shall be deemed withdrawn where the User ceases to be a User.
- 33.7 3.32 For the purposes of paragraphs 3.2633.1 to 3.3333.8, the Authority Regulator, upon request by a Proposer of a Significant Code Review Modification Proposal, may direct:
 - (a) that the Proposer may vary, withdraw or make a variation request Variation Request in respect of the relevant Significant Code Review Modification Proposal; or
 - (b) that the Proposer may not vary, withdraw or make a variation request Variation

 Request in respect of the relevant Significant Code Review Modification Proposal and that such proposal shall proceed in accordance with the Modification Procedures;
- 33.8 3.33 Any Significant Code Review Modification Proposal and any alternative to such Significant Code Review Modification Proposal shall be withdrawn where the Authority Regulator issues a direction to that effect.
- 34. Significant Code Review Modification Proposals made by the Regulator during Significant Code Reviews
- <u>Upon receipt of a Significant Code Review Modification Proposal made by the Regulator in the circumstances specified in paragraph 23.3, the Modification Panel must:</u>
 - (a) determine the factors which (in its opinion) justify the making or not making of the proposed modification, including whether, and if so, how, the proposed modification would better facilitate the achievement of the Relevant Objectives;
 - (b) <u>determine whether or not to recommend the implementation of the Modification</u>

 <u>Proposal to the Regulator; and</u>
 - <u>require the Secretary to prepare a report on the factors on which that recommendation is based, or where no recommendation can be reached, state in such report the state in such report the</u>

- number of Voting Members in favour of, and the number of Voting Members present and not voting in favour of, the implementation of the Modification Proposal.
- 34.2 The Secretary must, within five (5) Business Days of receipt of a Modification Panel recommendation pursuant to paragraph 34.1 (or within such other period as may be determined by the Regulator under paragraph 42.1):
 - (a) prepare a Modification Report setting out in detail the factors which (in the opinion of the Modification Panel) justify the making or not making of the proposed modification; and
 - <u>submit copies of the report and the Modification Panel's recommendation to the Regulator and provide copies to each T&SCo, each User, each Member, each Third Party Participant and each Designated Non-Code Party (if any).</u>
- The contents of any Significant Code Review Modification Proposal made by the Regulator and any conclusions published by the Regulator in respect of matters relating thereto shall not fetter the discretion of the Modification Panel or any Panel Member in connection with the Modification Panel's determination pursuant to paragraph 34.1(b) in respect of a Significant Code Review Modification Proposal made by the Regulator or in relation to any factor taken into account in making any such determination.

35. Consultation

- 35.1 3.34-If the Modification Panel determines that the Modification Proposal should proceed to Consultation (including in circumstances where the Modification Proposal has already been the subject of a Workgroup Report), the Secretary shall:
 - (a) the Modification Panel must notify the Secretary if it determines that the time periods set out in the remainder of this paragraph 35 for Consultation should, in its opinion, be deviated from in relation to the relevant Modification Proposal; and
 - where the Modification Proposal is a Self-Governance Modification Proposal, the
 Secretary may invite each T&SCo, each User, the Secretary of State, each Third Party
 Participant and each Designated Non-Code Party (if any) to make representations in
 respect of whether such Modification Proposal should be a Self-Governance
 Modification Proposal.
- 35.2 Following determination of the Modification Panel that a Modification Proposal should proceed to Consultation under paragraph 28.2, the Secretary must prepare a draft Modification Report in accordance with paragraph 3.4036.1 and circulate the draft Modification Report to each T&SCo, each User, each Third Party Participant, each Designated Non-Code Party (if any) and the Secretary of State within:
 - (a) fifteen (15) Business Days, where the Modification Panel has determined that legal text is required;
 - (b) three (3) Business Days, where the Modification Panel has determined that legal text is not required; or

(c) such other time period as the Modification Panel shall determine, inviting them to make representations to the Secretary within fifteen (15) Business Days following the date of that invitation.

- 35.3 3.35 The Secretary shallmust update the online register of Modification Proposals established pursuant to paragraph 3.524.3 to include publication in full of all representations received during Consultation in respect of such Modification Proposal.
- 35.4 3.36-Within ten (10) Business Days following the last day for making representations to the Secretary in respect of the draft Modification Report, the Secretary shallmust:
 - (a) prepare a final Modification Report in accordance with paragraph <u>3.40</u>36.1 for approval of the Modification Panel;
 - (b) submit a copy of the final Modification Report to each T&SCo, each User and each Member;
 - <u>(i)</u> <u>each T&SCo, each User, each Third Party Participant and each Designated</u>
 <u>Non-Code Party (if any) that submitted (and did not withdraw) a representation</u>
 <u>with regard to the draft Modification Report pursuant to paragraph 35.2; and</u>
 - (ii) <u>each Member,</u>

and must attach to that report all representations (if any) received and not withdrawn;

- (c) add the Modification Proposal as an agenda item for the Modification Panel to discuss in the next meeting of the Modification Panel; and
- (d) where the Secretary considers that any representation made identifies issues that may arise through implementation of the <u>proposal then Modification Proposal</u>, the Secretary <u>shallmust</u> record such issues and include them in the final Modification Report and the Secretary <u>shallmust</u> submit it to the next Modification Panel for determination pursuant to paragraph 3.3735.5.
- 35.5 3.37 Where the Modification Panel determines that views on such issues identified under paragraph 3.36(d)35.4(d) should be obtained by the relevant Workgroup then:
 - (a) the Secretary shallmust submit the final Modification Report to the relevant Workgroup and obtain a report containing their views;
 - (b) within three (3) Business Days following receipt of such report, the Secretary shallmust submit a copy of the final Modification Report to each T&SCo, each User and each Member; and:
 - <u>each T&SCo, each User, each Third Party Participant and each Designated</u>

 <u>Non-Code Party (if any); and</u>
 - (ii) <u>each Member;</u>

- and must attach to that report all representations (if any) received and not withdrawn and the report (if any) received pursuant to paragraph 35.5(a); and
- (c) the Secretary shallmust add the Modification Proposal as an Agenda item for the Modification Panel to discuss in the next meeting of the Modification Panel.
- 35.6 3.38-Subject to paragraph 3.6739.3, upon receipt of the final Modification Report under paragraphs 3.3635.4 or 3.3735.5, the Modification Panel shallmust assess whether the final Modification Report complies with the requirements of paragraph 3.4036.1, and if it is compliant, shallmust:
 - (a) determine whether or not to recommend the implementation of the Modification Proposal to the <u>AuthorityRegulator</u>;
 - (b) submit to the Authority Regulator its determination under paragraph (a35.6(a) (including where no determination can be reached due to a tie in the number of votes for and the number of votes against) and the factors which (in its opinion), justify its determination (including howand which shall include details of the Modification Panel's reasoning for determining whether or not the Modification Proposal meets T&SCo's objectives as set out in its Licence) better facilitates the achievement of the Relevant Objectives; and
 - (c) instruct the Secretary to send the final Modification Report, together with its recommendation to the AuthorityRegulator.
- 3.39-The Secretary shallmust, within one (1) Business Day of an instruction under paragraph 3.38(c), circulate 35.6(c), send to the Authority Regulator, each T&SCo, each User, each Third Party Participant, each Designated Non-Code Party (if any) and the Secretary of State the following:
 - (a) a copy of the final Modification Report and any other attachments; and
 - (b) a copy of the Modification Panel's recommendation to the <u>AuthorityRegulator</u> seeking a determination of the <u>AuthorityRegulator</u> as to whether the Modification <u>Proposal</u> should be implemented or not.
- 36. Content of Modification Report
- 36.1 3.40 Each Modification Report shall must include the following information:
 - (a) set out or append in full:
 - (i) the relevant Modification Proposal including, in the case of a Modification—

 Proposal, any suggested text provided by the proposer; and
 - (ii) where the Modification Panel has determined that legal text is required, the legal text; and
 - <u>(iii)</u> <u>all representations (if any) received and not withdrawn in respect of the relevant</u>

 Modification Proposal;

(b) shall-in relation to a recommendation of the Modification Panel under paragraph 3.38(a), include 35.6(a), details of the Modification Panel's reasoning for determining whether or not the Modification Proposal better facilitates the achievement (for the purposes of each T&SCo's Licence) of the Relevant Objectives;

- (c) state whether or not a determination has been made by the Modification Panel under paragraph 3.38(a)35.6(a) and the number of Voting Members in favour of, and the number of Voting Members present and not voting in favour of, the implementation of the Modification Proposal;
- (d) where it relates to a Self-Governance Modification Proposal, state that fact and whether the Modification Panel or the <u>AuthorityRegulator</u> determined that such proposal satisfied the Self-Governance Criteria;
- (e) state whether the proposal is made in respect of a Significant Code Review by, or at the direction of, the <u>AuthorityRegulator</u>;
- (f) where it relates to a Modification Proposal other than a Self-Governance Modification Proposal and where the Proposer has proposed a timescale for the implementation of the Modification in accordance with paragraph 3.4(c), shall 24.1(c), must include:
 - (i) two (2) or more Fixed Implementation Dates;
 - (ii) a Proposed <u>AuthorityRegulator</u> Decision Date in respect of each Fixed Implementation Date, for the purposes of enabling the Modification to be implemented by the Fixed Implementation Date;
 - (iii) a Backstop Lead Time; and
 - (iv) the reasons why it is proposing each date under paragraph (i), (ii) and (iii) 36.1(f)(i), 36.1(f)(ii) and 36.1(f)(iii);
- (g) address the implications (if any) of the implementation of such Modification Proposal:
 - (i) on the operation of the T&S Network; and
 - (ii) for the T&SCos and each T&SCo, including:
 - (A) the implications for the operation of the Onshore Transportation System and/or the Offshore Transportation and Storage System; and
 - (B) the development and capital cost and operating cost implications (if any) for each T&SCo of implementing the Modification Proposal and if there are any such cost implications, the extent to which it is appropriate for each T&SCo and/or the T&SCos to recover the costs, and (to such extent) a proposal for the most appropriate way for the costs to be recovered; and
- (h) address the implications (if any) of the implementation of the Modification for Users, including:

(i) the administrative and operational implications (if any) of the implementation of the proposal on Users; and

- (ii) the capital cost and operating cost implications (if any) for Users of implementing the Modification Proposal;
- (i) address the implications (if any) for other relevant persons, including future users who may connect to a T&S Network;
- (j) provide a summary of the representations and/or responses of those persons (if any) which the Workgroup consulted, pursuant to the Modification Terms of Reference and paragraph 3.1629.1 and (to the extent that representations are not addressed or otherwise commented upon in the Modification Report) comment, in the context of the Modification Report as a whole, on such representations;
- (k) detail the programme for works (if any) required as a consequence of the implementation of the Modification Proposal;
- set out a proposed timetable to be followed in respect of the implementation of the Modification (including, but without limitation, the date upon which the Modification should become effective);
- (m) state whether or not any matter detailed in the Workgroup Report (pursuant to paragraph 3.17(d)29.2(d)) has been addressed or otherwise dealt with and:
 - (i) if such matter has been so addressed or dealt with, the result; or
 - (ii) if such matter has not been so addressed or dealt with, an explanation as to why such matter has not been addressed or dealt with; and
- (n) state the number of Voting Members in favour of, and the number of Voting Members present and not voting in favour of, the implementation of the Modification Proposal pursuant to the determination made under paragraph 3.38;
- (o) have attached to it copies of all the representations received by the Secretary; and
- (n) (p) set out a high level summary of views expressed in the representations (to the extent that representations are not addressed or otherwise commented upon in the Modification Report), which may be satisfied by stating the number of representatives in favour of, and the number of representatives against the implementation of the Modification Proposal.
- 3.41-Where two (2) or more Modification Proposals have proceeded through the Modification Procedures together (and neither proposal Modification Proposal has been withdrawn and all work has not been discontinued following a decision of the Modification Panel in respect of either proposal Modification Proposal) the Modification Report shallmust, in addition to the analysis referred to in paragraph 3.40(b)36.1(b), provide an analysis as to which of the Modification Proposals would in the opinion of the Modification Panel better facilitate the achievement of the Relevant Objectives. Where two (2) Modification Proposals were proceeding through the Modification Procedures together and the Modification Panel shall-

have has determined that only one (1) of the Modification Proposals should proceed, the Modification Report shall must provide a commentary as to the circumstances in which the Modification Panel so determined made such determination.

- 36.3 3.42-Each Modification Report shallmust be addressed and furnished provided to the Authority Regulator and none of the facts contained or opinions stated in any Modification Report should be relied upon by any other person.
- 36.4 3.43 In preparing any Modification Report, the Secretary shall must:
 - (a) do so on the basis set out in these Modification Rules; and
 - (b) not be required to have regard (other than as expressly provided in these Modification Rules) to the consequences of any Modification on any person-or persons.

37. Implementation

- 37.1 3.44-Where the AuthorityRegulator receives a final Modification Report pursuant to paragraph 3.3935.7, the AuthorityRegulator may determine:
 - (a) that the final Modification Report provides insufficient detail in order to make a decision, or the Regulator considers that it cannot properly form an opinion on the approval of the Modification Proposal, in which case the Secretary shallmust make the necessary arrangements to address the deficiency and re-submit it to the Modification Panel and paragraph 3.3835.6 shall re-apply; or
 - (b) whether the proposed Modification shall be implemented and may give notice of its decision together with reasons, to the Secretary, in which case on receipt of such notice from the AuthorityRegulator:
 - (i) if the notice confirms the Authority's Regulator's determination not to implement the proposed Modification, the Secretary shallmust circulate to each T&SCo, each User, each Third Party Participant, each Designated Non-Code Party (if any) and each Member a non-implementation notice; and
 - (ii) if the notice confirms the <u>Authority's Regulator's</u> determination to implement the Modification, the Secretary <u>shallmust</u> circulate to each T&SCo, each User, <u>each Third Party Participant</u>, <u>each Designated Non-Code Party (if any)</u>, and each Member an implementation notice.
- 37.2 3.45-If, in respect of a Modification Proposal which is the subject of a Modification Report previously submitted to the <u>AuthorityRegulator</u> by the Secretary in accordance with paragraph 3.3935.7:
 - (a) the AuthorityRegulator has not given notice of its decision in respect of that Modification Report within two (2) calendar months (in the case of Urgent Modification Proposals), or four (4) calendar months (in the case of non-Urgent Modification Proposals) from the date upon which the relevant Modification Report was submitted to it; or

(b) the <u>AuthorityRegulator</u>, or any Voting Member, by notice to the Secretary expresses the reasonable opinion that the circumstances relating to that Modification have materially changed,

the Secretary shallmust place that Modification Proposal on the Agenda for consideration at the next Modification Panel meeting.

37.3 3.46-Having considered the circumstances relating to the Modification Proposal which is subject to paragraph 3.4537.2, the Modification Panel may determine that the Secretary should request the AuthorityRegulator to give an indication of the likely date by which the Authority's Regulator's decision shall be made, it being recognised at all times that the Regulator is not obliged, as a result of a request pursuant to this paragraph 37.3, to provide any response or indication.

38. Modification

- 38.1 3.47 Subject to paragraph 3.4838.2, the Secretary shall must modify the Code in accordance with each consent given and each direction made by the Authority Regulator.
- 38.2 3.48-The Secretary shall modify the Code without the consent of the Authority Regulator in accordance with a determination made by the Modification Panel in respect of a Self-Governance Modification Proposal under paragraph 3.68(a)39.2 no less than fifteen (15) Business Days after the date of such determination, provided that no Appeal has been made or the Authority has not directed that its approval is required in accordance with paragraph 3.51(b).:
 - (a) either:
 - (i) paragraph 39.3 applies and the Modification Panel has not instructed the Secretary to withdraw the Self-Governance Statement submitted to the Regulator under paragraph 39.7; or
 - (ii) paragraph 39.10 applies;
 - (b) the Regulator has not directed that its approval is required in respect of such Modification Proposal under paragraph 39.5(b);
 - subject to paragraph 38.2(d), no Appeal has been made in accordance with paragraph
 40.1 which is still outstanding in respect of such Modification Proposal; or
 - where an Appeal has been made, the Secretary has sent an implementation notice under paragraph 40.4(a)) or 40.4(b) and the Appealing Party has not made a Regulator Appeal in respect of such Modification Proposal by the end of the period specified in paragraph 40.5.
- 38.3 The Secretary must, as soon as reasonably practicable:
 - (a) 3.49 The Secretary shall, as soon as reasonably practicable, notify each T&SCo, each User, each Third Party Participant, each Designated Non-Code Party (if any) and each Member of each Modification...; and

- (b) <u>publish by notice each Modification on the CCS Network Code Website.</u>
- 38.4 Each such notice shall under paragraph 38.3 must specify the legal text of the Modification and the date upon which the Modification shall become effective and may provide (for the purposes of information only) an explanatory note (which note should not be relied upon) in respect of the Modification. Each Modification shall become effective upon the date specified in the relevant notice.

39. Self-Governance Modifications

- 39.1 Where the Modification Proposal is a Self-Governance Modification Proposal, paragraphs
 35.6, 35.7 and 37.1 shall not apply to such Modification Proposal.
- 39.2 The Modification Panel must upon receipt of the final Modification Report under paragraph
 35.4 or 35.5, in respect of a Self-Governance Modification Proposal:
 - (a) assess whether the final Modification Report complies with the requirements of the Code, and if it is compliant, must make a determination as to whether or not the Self-Governance Modification Proposal should be implemented, having regard to whether or not the Self-Governance Modification Proposal better facilitates the achievement of the Relevant Objectives than either the status quo or any alternative Self-Governance Modification Proposal;
 - (b) <u>instruct the Secretary to include such determination and the Modification Panel's</u> reasoning for such determination in the final Modification Report; and
 - instruct the Secretary to circulate an implementation notice or a non-implementation notice (as the case may be) in respect of such proposal to each T&SCo, each User, each Third Party Participant and each Member.
- 39.3 3.50 Where Subject to paragraph 39.5, where the Modification Panel under paragraph 28.2 determines that thea Modification Proposal satisfies the Self-Governance Criteria and paragraph 3.14(a) applies, the Secretary shall must on behalf of the Modification Panel submit to the Regulator a Self-Governance Statement to the Authority with a proposed such Modification Proposal as soon as reasonably practicable and set a Proposed Self-Governance Determination Date.
- 39.4 The Secretary shall not submit to the Regulator a Self-Governance Statement in respect of a

 Modification Proposal (or set a Proposed Self-Governance Determination Date in relation to
 it) pursuant to paragraph 39.3 where such Modification Proposal is an alternative Modification
 Proposal made under paragraph 32.1 or 32.2, in respect of an original Modification Proposal
 which is not itself a Self-Governance Modification Proposal unless that original Modification
 Proposal has been withdrawn.
- 39.5 3.51 The Authority Regulator may, at any time before the proposed Self-Governance Determination Date:
 - (a) reject the Modification Panel's determination that the Self-Governance Criteria have been satisfied: or

(b) direct that its approval is required to implement the Self-Governance Modification Proposal.

- 39.6 Where the Regulator rejects a Self-Governance Statement in respect of a Modification

 Proposal or the Secretary is prohibited from submitting to the Regulator a Self-Governance

 Statement in respect of a Modification Proposal by paragraph 39.3, such Modification

 Proposal must:
 - (a) not be a Self-Governance Modification Proposal; and
 - (b) be subject to the Modification Procedures.
- 39.7 The Modification Panel may withdraw a Self-Governance Statement that it submits under paragraph 39.3 at any time before the Self-Governance Modification Proposal Determination Date in respect of the related Self-Governance Modification Proposal.
- 39.8 The Modification Panel shall withdraw a Self-Governance Statement in respect of a

 Self-Governance Modification Proposal where, prior to the Self-Governance Modification

 Proposal Determination Date, an alternative Modification Proposal is made in respect of such

 Self-Governance Modification Proposal under paragraph 32.1 or 32.2 and:
 - (a) 3.52 Notwithstanding any determination to the contrary the Modification Panel determines under paragraph 3.14(a), the Authority may determine that a28.2(a) that such alternative Modification Proposal satisfies does not satisfy the Self-Governance Criteria.
 - (b) the Regulator rejects a Self-Governance Statement in respect of such alternative Modification Proposal; or
 - (c) the Modification Panel, pursuant to paragraph 39.7, withdraws a Self-Governance Statement in respect of such alternative Modification Proposal.
- Where a Self-Governance Statement in respect of a Modification Proposal is withdrawn pursuant to paragraph 39.7, such Modification Proposal shall, subject to paragraph 39.11, cease to be a Self-Governance Modification Proposal.
- Where the Modification Panel has determined under paragraph 28.2(a) that a Modification

 Proposal does not satisfy the Self-Governance Criteria or the Modification Panel has

 withdrawn a Self-Governance Statement in respect of a Modification Proposal under

 paragraph 39.7, the Regulator may determine that the relevant Modification Proposal satisfies

 the Self-Governance Criteria and is a Self-Governance Modification Proposal, and, where the

 Regulator so determines:
 - (a) <u>its determination shall be effective upon the Regulator giving notice of the same to the Secretary; and</u>
 - (b) any Self-Governance Statement in respect of any other Modification Proposal that was withdrawn pursuant to paragraph 39.8 as a consequence of the Modification Panel's determination with respect to, or withdrawal of, such relevant Modification Proposal shall be reinstated, and the Secretary must, on behalf of the Modification Panel,

resubmit such Self-Governance Statement(s) to the Regulator, and the Modification Proposals to which such Self-Governance Statement(s) relate shall be reinstated as Self-Governance Modification Proposal(s) accordingly.

- Where the Modification Panel determines under paragraph 28.2(a), that the Modification

 Proposal does not satisfy the Self-Governance Criteria, the Secretary must as soon as

 reasonably practicable submit to the Regulator a statement on behalf of the Modification

 Panel in respect of such Modification Proposal setting out why in the opinion of the

 Modification Panel it could be reasonably expected that the Modification Proposal would have a material impact if implemented.
- 3.53-Where paragraph 3.14(b)28.2(b) applies but paragraph 3.14(e)28.2(c) does not (i.e. the Modification Proposal satisfies the Fast Track Self Governance Self-Governance Procedure but the Modification Panel does not determine that it should be implemented immediately), the Modification shall become a Self-Governance Modification Proposal and be further determined by the Modification Panel under paragraph 3.1428.2.
- 39.13
 3.54-Where the Modification Panel makes a determination in respect of a Self-Governance Modification Proposal under paragraph 3.14(e)28.2(c) (i.e. in respect of a fast tracked Modification) a Party or the AuthorityRegulator may object to such determination and the Self-Governance Modification Proposal shall not be implemented. Any such objection shall be received within fifteen (15) Business Days of the implementation notice under paragraph 3.44(b)(ii)37.1(b)(ii) being issued.

40. Appeal Procedures

- 40.1 3.55 No later than fifteen (15) Business Days after the Self-Governance Determination Date, a User, the T&SCo or Third Party Participant may make an Appeal by giving written notice of the Appeal to the Secretary, such notice to identify:
 - (a) the Self-Governance Modification Proposal which is the subject of the Appeal;
 - (b) any representations which the Appealing Party wishes to make in support of the Appeal; and
 - (c) evidence supporting the Appealing Party's view that it will be unfairly prejudiced by the modification proposed in the relevant Self-Governance Modification Proposal.
- 40.2 3.56-Subject to the Appeal being made within the period referred to in paragraph 3.5540.1, the Secretary shallmust, on receipt of a notice under paragraph 3.5540.1 notify each T&SCo, each User, each Third Party Participant and the Authority, each Designated Non-Code Party (if any) and the Regulator that an Appeal has been made and the Secretary shallmust put discussion of the Appeal on the Agenda of the next meeting of the Modification Panel.
- 40.3 3.57 The Modification Panel shallmust decide, at the meeting referred to in paragraph 3.5640.2, whether to confirm or reverse its determination under paragraph 3.38(a)39.2(a) having regard to any representations made by the Appealing Party in support of the Appeal.
- 40.4 3.58 The Modification Panel shallmust, upon deciding whether to confirm or reverse its determination under paragraph 3.38(a)39.2, notify the Secretary and the Appealing Party of

its decision as soon as reasonably practicable following receipt of the Appeal, and where the Modification Panel:

- (a) confirms its determination under paragraph 3.38(a),40.2 the Secretary shallmust notify each T&SCo, each User, each Third Party Participant and the Authority, each Designated Non-Code Party (if any) and the Regulator of the Modification Panel's confirmation of its determination and send each such person an implementation or (as the case may be) non-implementation notice under paragraph 3.44(b)37.1(b) confirming the implementation or (as the case may be) non-implementation of the proposal; or
- (b) reverses its determination under paragraph 3.38(a)40.2, the Secretary shallmust notify each T&SCo, each User, each Third Party Participant and the Authority, each Designated Non-Code Party (if any) and the Regulator of the Modification Panel's determination and send each such person an implementation or (as the case may be) non-implementation notice, which shall have effect in place of the implementation or (as the case may be) non-implementation notice issued pursuant to the Modification Panel's initial determination under paragraph 3.38(a)40.2.
- 3.59-The Appealing Party may no later than fifteen (15) Business Days after the Self-Governance Modification Proposal Determination Date make an appeal to the AuthorityRegulator; such appeal to identify the Self-Governance Modification Proposal which is the subject of the appeal and include any representations which the Appealing Party wishes to make in support of the appeal.
- 40.6 3.60 If the AuthorityRegulator determines the AuthorityRegulator Appeal satisfies the Appeal Criteria, the AuthorityRegulator may consider the AuthorityRegulator Appeal having regard (without limitation) to:
 - (a) the Self-Governance Modification Proposal;
 - (b) the final Modification Report in respect of the Self-Governance Modification Proposal;
 - (c) any representations made by the Appealing Party in support of the AuthorityRegulator Appeal; and
 - (d) the determination made by the Modification Panel under paragraph 3.58(a)40.4(a).
- 40.7 3.61 Following such determination in accordance with paragraph 3.60 40.6, the Authority Regulator may:
 - (a) confirm the Modification Panel's Panel's determination under paragraph 3.58(a)40.4(a); or
 - (b) quash the Modification Panel's Panel's determination under paragraph 3.58(a)40.4(a) and rule that the Modification Panel's Panel's determination should have no further effect for the purposes of the Modification Rules.
- 40.8 3.62 The Authority shall Regulator must notify the Secretary and the Appealing Party of its decision under paragraph 3.6140.7 in respect of the Authority Regulator Appeal as soon as

reasonably practicable following receipt of the <u>AuthorityRegulator</u> Appeal, and the decision of the <u>AuthorityRegulator</u> shall be final and binding.

- 40.9 3.63 Where paragraph 3.61(b)40.7(b) applies, the Authority Regulator may:
 - (a) remit the Self-Governance Modification Proposal which is the subject of the Modification Panel's Panel's quashed determination under paragraph 3.58(a)40.4(a) back to the Modification Panel for reconsideration and redetermination; or
 - (b) determine that the proposed modification to the Code should be made; or
 - (c) determine that the proposed modification to the Code should not be made.

40.10 3.64-Where the Authority Regulator:

- (a) confirms the Modification Panel's determination under paragraph 3.58(a)40.4(a), the Secretary shallmust notify each T&SCo, each User-and, each Third Party Participant of the Authority's and each Designated Non-Code Party (if any) of the Regulator's confirmation of the Modification Panel's determination under paragraph 3.58(a)40.4(a) confirming the implementation or (as the case may be) non-implementation of the proposal; or
- (b) quashes the Modification Panel's determination under paragraph 3.58(a)40.4(a), the Secretary shallmust notify each T&SCo, each User, each Member and, each Third Party Participant of the Authority's and each Designated Non-Code Party (if any) of the Regulator's decision and send each such person an implementation notice or (as the case may be) non-implementation notice, which shall have effect in place of the Modification Panel's determination under paragraph 3.58(a);40.4(a).
- 3.65-Where the Secretary, under paragraph 3.61(a)40.7(a) gives notice of the Authority's Regulator's confirmation of the Modification Panel's determination under paragraph 3.58(a)40.4(a) confirming the implementation of a proposal or under paragraph 3.63(b)40.9(b) sends an implementation notice, paragraph 3.4738.1 shall apply.
- 40.12 3.66-Where a (fast tracked) Self-Governance Modification Proposal is objected to under paragraph 3.5439.13 then the Self-Governance Modification Proposal shall return to the Modification Panel and be considered as a Modification Proposal under the provisions of paragraph 3.1428.2.
- 3.67 Where the Modification Proposal is a Self-Governance Modification Proposal paragraphs 3.38, 3.39, and 3.44 shall not apply to such proposal.
- 3.68 The Modification Panel shall upon receipt of the final Modification Report under paragraph 3.36 or 3.37, in respect of a Self-Governance Modification Proposal:
 - (a) assess whether the final Modification Report complies with the requirements of the Code, and if it is compliant, shall make a determination as to whether or not the Self-Governance Modification Proposal should be implemented;

(b) instruct the Secretary to include such determination and the Modification Panel's reasoning for such determination in the Final Modification Report; and

- (c) instruct the Secretary to circulate an implementation notice or a non-implementation notice (as the case may be) in respect of such proposal to each T&SCo, each User and each Member.
- 3.69-Where the Modification Panel makes a determination in respect of a Self-Governance Modification Proposal under paragraph 3.68(a)28.2(c), a T&SCo-or, a User or a Third Party Participant may appeal such determination and the Self-Governance Modification Proposal shall not be implemented until the Authority Modification Panel has confirmed made a determination in respect of an appeal (and where no subsequent appeal has been made to the Regulator) and an implementation notice has been sent.

41. Urgent Modification Proposals

- 3.70-If the Proposer considers that a Modification Proposal should be treated as an Urgent Modification Proposal it shall state so in its Modification Proposal and include reasons as to why the Modification Proposal should be treated as an Urgent Modification Proposal, and the Secretary shallmust as soon as possible send a copy of the Modification Proposal to the Authority.Regulator and:
 - (a) following consultation with the Regulator, the Secretary must recommend the procedure and timetable to be followed in respect of each Urgent Modification Proposal; and
 - (b) where the Regulator requests the Modification Panel to provide the Regulator with the opinion of the Modification Panel on whether or not the Modification Proposal should be considered as an Urgent Modification:
 - (i) the Secretary must convene a meeting of the Modification Panel within five (5)

 Business Days of the Regulator requesting the opinion of the Modification

 Panel and place such request on the Agenda for that meeting; and
 - (ii) the Modification Panel must provide the Regulator with its opinion.

41.2 3.71 Where the Authority Regulator:

- (a) agrees that the Modification Proposal should be treated as an Urgent Modification Proposal,:
 - (i) the Secretary must notify each T&SCo, each User, each Third Party
 Participant, each Member and each Designated Non-Code Party (if any);
 - <u>to the extent that the Regulator agrees with the recommendation made in the procedure and timetable submitted by the Secretary.</u> all or any of the Modification Rules (including, but without limitation, consulting with the Modification Panel and seeking representations from each T&SCo-and. Users. Third-Party Participants and any Designated Non-Code Party or any other

- person) may be deviated from and any other procedure accepted by the AuthorityRegulator may be followed; orand
- (iii) the Secretary must prepare and submit to the Regulator, a Modification Report in a format and in accordance with a timetable accepted by the Regulator; or
- (b) does not agree that the Modification Proposal should be treated as an Urgent Modification Proposal, the Secretary shallmust notify the Modification Panel and the standard Modification Rules will apply.

42. Time Periods

- 3.72-Notwithstanding any time periods specified in this paragraph 3, paragraphs 23 to 42 in respect of any Significant Code Review Modification Proposal, the AuthorityRegulator may, by direction in writing to a T&SCo, specify and/or amend the timetable to apply to each stage of such Modification Proposal and/or the implementation date of such Modification—Proposal determine to shorten or lengthen the time period within which any (or all) of the Modification Procedures or any other procedure is (or are) to be undertaken. If the Modification Panel shall so determine, the Secretary must, so far as consistent with the Modification Rules and Licence, do all acts and things which these Modification Rules specify as acts and things to be done the Secretary within such shorter or longer period of time so as to give effect to such determination.
- 42.2 3.73 Where the Authority Regulator makes a direction referred to in paragraph 3.7242.1 and such direction specifies:
 - (a) a timetable in respect of all of the steps specified in this paragraph 3 paragraphs 23 to 42 as applying to such Modification Proposal, the Modification Proposal shall progress in accordance with the timetable specified in such direction (and not the timetable set out in this paragraph 3 paragraphs 23 to 42);
 - (b) a timetable in respect of some (but not all) of the steps specified in these Modification Rules as applying to such Modification Proposal (and does not direct that the steps in respect of which no timetable is directed are not to apply to such Modification Proposal), then such Modification Proposal shall proceed on the basis of the timetable specified in the direction (in respect of the steps specified in the direction) and the timetable set out in this paragraph 3 paragraphs 23 to 42 (in respect of all other steps specified in these Modification Rules as applying to such Modification Proposal); or
 - (c) an implementation date in respect of such Modification Proposal but specifies no timetable to apply to such Modification Proposal in relation to these Modification Rules, then the T&SCo to whom the direction is made by the Authority shall Regulator must notify the Secretary of the timetable that the T&SCo reasonably believes ought to apply to such Modification Proposal in order to achieve the directed implementation date, and such notified timetable shall apply for the purposes of this paragraph 3paragraphs 23 to 42 in relation to such Modification Proposal.

42.3 3.74 Until such time as the Authority Regulator makes a direction referred to in paragraph
 3.7342.2, the Modification Proposal shall proceed in accordance with the Modification Procedures set out in this Section B (Governance).

43. **Representations**

- All representations (and any withdrawal of the same) made by each User, each T&SCo, each Third Party Participant or any Non-Code Party pursuant to these Modification Rules must be readily identifiable as representations (or, as the case may be, a withdrawal thereof), must identify the unique reference designation on the Modification Proposal to which they relate and must be sent to the Secretary at an address to be notified on the CCS Network Code Website. The T&SCo, User, Third Party Participant or Designated Non-Code Party may at any time prior to the submission of the Modification Report by notice to the Secretary withdraw any representation made by it, whether made in writing or at a meeting of Users or Non-Code Parties.
- The Secretary must send to the Regulator copies of all representations, but the Secretary shall not be required to take account of representations received after the expiry of the period established in accordance with these Modification Rules, or of any representations which in the Secretary's reasonable opinion are of no relevance to the subject matter to which they are intended to relate, or from persons who are not likely to be materially affected.
- 43.3 In submitting representations pursuant to any provision of these Modification Rules, the person making such submission consents to the publication and circulation of such representations by the Secretary for the purposes of these Modification Rules.

44. 4. Specific Provisions

- 44.1 A Modification Proposal in respect of a Specific Provision may only be initiated by:
 - (a) the T&SCo that owns and operates the T&S Network to which those Specific Provisions apply;
 - (b) <u>a User that is the holder of a Connection Agreement relating to the T&S Network, or relevant part of the T&S Network, to which those Specific Provisions apply;</u>
 - (c) the Secretary of State; and
 - (d) the Regulator where the Modification Proposal is in respect of a Significant Code Review.

provided that where there is no User holding a Connection Agreement for a T&S Network, T&SCo may only initiate a Self-Governance Modification Proposal in respect of a Specific Provision for that T&S Network.

- <u>Where a Modification Proposal in respect of a Specific Provision is initiated, the Secretary must convene a panel comprising:</u>
 - (a) the T&SCo that owns and operates the T&S Network to which those Specific Provisions apply, being a voting member;

(b) each User that is the holder of a Connection Agreement relating to the T&S Network, or relevant part of the T&S Network, to which those Specific Provisions apply, being voting members;

- (c) the Secretary of State, being a non-voting Member;
- (d) the Regulator, being a non-voting Member; and
- (e) the Panel Chairperson.
- 4.1 Subject to paragraph 545, Modification Proposals in respect of Specific Provisions shallmust be addressed using an equivalent procedure determined by the Panel Chairperson based on the principles of the Modification Procedure, save that only but modified as appropriate to take account of:
 - (a) the make-up of the panel specified in paragraph 44.2;
 - (b) the Specific Provisions which are under review; and
 - (c) the principles for voting rights set out below:
 - (i) (a) the T&SCo that owns and operates the T&S Network to which those the Specific Provisions apply is to be entitled to one (1) vote; and
 - (ii) (b) aeach User that is thea holder of a Connection Agreement relating to the T&S Network, (or relevant part of the T&S Network,) to which those Specific Provisions apply, is to be entitled to one (1) vote; and

shall be deemed a "T&SCo or a "User" for the purposes of the Modification Rules.

- votes shall be weighted such that there shall be an equal number of voting rights as between (1) the T&SCo referenced in paragraph 44.3(c)(i) and (2) the User(s) referenced in paragraph 44.3(c)(ii) (in aggregate).
- 45. 5. Ancillary Agreement Agreements
- 45.1 The Ancillary Agreements, once entered into, may only be amended:
 - (a) 5.1 The "User-specific provisions" (i.e. those that are contemplated as being completed in the Schedules) of an Ancillary Agreement may be amended by agreement of the parties to that Agreement and not otherwise, and accordingly "User-specific provisions" of an Ancillary Agreement shall not be subject to modification (unless otherwise specified in the relevant Ancillary Agreement); or
 - (a) pursuant to thea Modification Rules (but without prejudice to any modification of any provisions of the Code which is expressly stated to apply to or are and be incorporated into such Agreement) the Ancillary Agreements.
- 45.2 The Uniform Provisions of an Ancillary Agreement Templates may only be amended pursuant to the Modification Rules paragraph 5.5 of Section A (Introduction, Structure and Interpretation).

46. 6. Modifications – General

The Secretary may decide to seek a view from the Authority Regulator on any matter arising (under these Modification Rules or otherwise) from a Modification Proposal that is not a Self-Governance Modification Proposal at any stage prior to the date on which the Secretary finalises the Modification Report.

- 46.2 If the Secretary decides to seek a view pursuant to paragraph 46.1, the Secretary must:
 - (a) forward an application to the Regulator setting out the Secretary's views and reasons for seeking a view from the Regulator;
 - (b) if the Modification Panel shall so determine, suspend (in whole or in part) the consideration of the Modification Proposal by any Workgroup; and
 - (c) notify each T&SCo, each User, each Third Party Participant and each Designated
 Non-Code Party (if any) of such decision to seek a view.
- 46.3 If, in respect of a view sought pursuant to this paragraph 46, the Regulator expresses the view that the Modification Proposal should not proceed, the Modification Proposal shall lapse and the Secretary must, within five (5) Business Days following receipt by the Secretary of the view, prepare and circulate notification of such lapse.
- 46.4 If, in respect of a view sought pursuant to this paragraph 46, the Regulator does not express any view as to how the Modification Proposal should proceed or expresses the view that the Modification Proposal should proceed to Workgroup Assessment, the Modification Proposal shall proceed to Workgroup Assessment.
- 46.5 If the Regulator expresses any other view, the Modification Proposal shall proceed in accordance with such view.
- 47. 7. Secretary of State Modification Right Modifications during Interim Period
- 7.1 Where the Secretary of State makes a Modification Proposal during the Interim Period, then in addition to making a Modification Proposal pursuant to paragraph 3.123.1, the Secretary of State may choose to exercise powers in relation to that Modification Proposal in accordance with paragraph 7.247.3, by way of a direction pursuant to this paragraph 47.
- Where the Secretary of State elects to exercise the powers in paragraph 47.3 in respect of any Modification Proposal, in addition to the requirements of paragraph 24.1, the Secretary shall ensure that any such Modification Proposal clearly indicates that it is a SoS Modification Proposal for the purposes of paragraph 47.3.
- 47.3

 7.2 In respect of any Modification Proposal which is made by the Secretary of State during the Interim Period under this paragraph 47.3, (a "SoS Modification Proposal"), paragraph 3isparagraphs 23 to 42.3 are to have effect with the following modifications until the end of the Interim Period:

 in paragraph 3.223.2 for "AuthorityRegulator" substitute "Secretary of State" and for "Significant Code Review Modification Proposal" substitute "SoS Modification Proposal";

- (b) in paragraph 3.323.4, for "AuthorityRegulator" substitute "Secretary of State", for "Significant Code Review" substitute "SoS Modification Proposal", and for "relevant Significant Code Review Phase" substitute "period within which a SoS Modification Proposal Phase has been introduced under paragraph 47.3 but which has not yet been finally determined";
- (c) in paragraph 3.4(I)24.1(I), for "AuthorityRegulator" substitute "Secretary of State" and for "Significant Code Review " substitute "SoS Modification Proposal";
- (d) in paragraphs 3.10, 3.11, 3.12-27.1, 27.2 and 3.1527.3 for "AuthorityRegulator" substitute "Secretary of State", for "Significant Code Review Modification Proposal" substitute "SoS Modification Proposal", for "Significant Code Review Phase" substitute "SoS Modification Proposal Phase", for "Significant Code Review Phase" substitute "period within which a SoS Modification Proposal has been finally determined", and for "Significant Code Review Suspended Modification Proposal" substitute "SoS Modification Suspended Proposal"; and
- (e) in paragraph 28.3 delete "made by a Proposer other than the Secretary of State during a Significant Code Review Phase" and for "Significant Code Review Phase" substitute "made during the period within which a SoS Modification Proposal has been finally determined", and for "Significant Code Review" substitute "SoS Modification Proposal";
- (f) in paragraph 28.4 for "Regulator" substitute "Secretary of State" and for "Significant Code Review" substitute "SoS Modification Proposal";
- (g) in paragraph 38.1 delete "subject to paragraph 38.2" and for "Regulator" substitute "Secretary of State";
- (h) (e)-in paragraphs 3.4(b), 3.24(c), 3.32, 3.33, 3.38, 3.39(b), 3.40(e), 3.42, 3.44, 3.47-and 3.72 for "Authority24.1(b), 33.7, 33.8, 35.6, 35.7(b), 36.1(e), 36.3, 37.1, 42.1, 42.2 and 42.3 for "Regulator" substitute "Secretary of State" and for "Significant Code Review Modification Proposal" and "Significant Code Review" substitute "SoS Modification Proposal-"; and

Note: while the Secretary of State's powers during the Interim Period are based on the Ofgem SCR process, some further defined terms (e.g. "SoS Modification Proposal Phase") will need to be further considered and developed.

- in paragraph 24.1(c), in the sentence "except in the case where the Regulator has directed a timetable in accordance with paragraph 42.1 and/or 42.2", substitute "Regulator" for "Secretary of State".
- 47.4 7.3 Where the Secretary of State chooses to exercise powers rights in relation to an SoS Modification Proposal in accordance with this paragraph 7 then 47, the Regulator shall not be

- <u>required to approve</u> the SoS Modification Proposal does not need to be approved by the <u>Authorityin order</u> for it to be implemented.
- The Parties acknowledge and agree that notwithstanding paragraph 17.1 of Section J

 (General), the Contracts (Rights of Third Parties) Act 1999 shall apply to the rights conferred on the Secretary of State under this Section B (Governance).
- 48. 8. Dispute resolution
- 48.1 The Dispute avoidance
- 8.1 The Parties shallmust at all times deal proactively with any issues through Code governance processes to reduce the scope for disputes Disputes to arise. The Dispute Parties shallmust take all reasonable steps to identify solutions and mitigations that address the concerns of other parties Parties and at all times to act reasonably to avoid or minimise the scope for disputes Disputes.
- 48.2 If a Dispute arises, the objective of the Dispute Parties shall be to seek to ensure that the
 Dispute is resolved as quickly, as efficiently and as cost-effectively as possible. Each Dispute
 Party shall, at each stage of the Dispute Resolution Procedure, endeavour in good faith to
 resolve all Disputes through negotiation.
- The Dispute Parties shall continue to observe and perform all of their respective duties, responsibilities and obligations under the Code, the Code Agreement, any Code Accession Agreement, any CDS Accession Agreement and any Ancillary Agreement notwithstanding any Dispute which falls to be resolved in accordance with this paragraph 48.
- 48.4 8.2 Where the relevant <u>Dispute</u> Parties are unable to avoid or resolve a <u>dispute Dispute</u>, the <u>dispute Dispute</u> shall be resolved in accordance with the procedures <u>summarised in this paragraph 8 set out in the Code</u>.
- <u>Unless the Code expressly provides otherwise or the Dispute Parties otherwise agree, the Dispute Parties may refer any and all Disputes to the courts of England for resolution in accordance with paragraph 16 of Section J (General).</u>

Introduction

8.3 For the purposes of this section:

- (a) a "dispute" is any dispute or difference arising between the parties under or in connection with the Code, the Code Agreement or any Ancillary Agreement; and
- (b) in respect of any dispute "parties" means the T&SCo(s) and/or the User(s) party to such dispute, and "party" shall be construed accordingly.
- 48.6 8.4 The Secretary appointed in accordance with paragraph 1.15 shall 7.1 must perform the administrative functions set out in this paragraph 848.
- 48.7 If a Dispute arises either Dispute Party may (except where paragraph 51.1 applies, in which event either Party may instead proceed directly to an Expert Determination Notice) give a notice to the other Dispute Party to initiate the Dispute Resolution Procedure (a "Dispute")

Notice"). For the avoidance of doubt, in the event of an Expert Determination commenced pursuant to paragraph 51.1(a), a Dispute Notice and an Expert Determination Notice will be required. A Dispute Notice:

- (a) must specify that it is a Dispute Notice;
- <u>(b)</u> <u>must include a description of the subject matter of the Dispute and the issues to beresolved;</u>
- (c) must include a statement identifying the provision of the Code, Code Agreement,

 Code Accession Agreement, CDS Accession Agreement or Ancillary Agreement to

 which the Dispute relates or pursuant to which the Dispute arises;
- <u>must include a description of the position the referring Dispute Party considers is</u>
 correct and the referring Dispute Party's reasons for that position;
- (e) must include details of any other dispute or claim known to the referring Dispute Party between:
 - (i) a T&SCo and a different User;
 - (ii) <u>a User and a different T&SCo; and/or</u>
 - (iii) two (2) T&SCos,

which the referring Dispute Party considers should be consolidated with or joined to the Dispute;

- <u>may, where the referring Dispute Party considers it appropriate, include copies of anydocuments on which the referring Dispute Party intends to rely:</u>
- <u>must include a statement outlining the relief, determination, remedy or recourse</u> whichthe referring Dispute Party seeks in relation to the Dispute;
- (h) must include a statement as to whether the referring Dispute Party considers that the Dispute should (without a Senior Representatives Settlement being reached) be referred for determination in accordance with the Expert Determination Procedure; and
- (i) <u>must include the identity of the referring Party's Senior Representative.</u>
- 48.8 Following the service by either Dispute Party of a Dispute Notice:
 - (a) the Dispute Parties shall seek to resolve the Dispute by convening a meeting of the Senior Representatives of the Dispute Parties in accordance with paragraph 50; and
 - (b) <u>if the Senior Representatives are unable to settle, compromise or resolve the Dispute</u> in accordance with paragraph 50, either Dispute Party may:
 - <u>(i)</u> <u>if the Dispute Parties agree within the Resolution Period (or such longer period</u> as they may agree) that such Dispute is amenable to determination by an

- Expert, refer the Dispute to an Expert for determination in accordance with the Expert Determination Procedure pursuant to paragraph 51; or
- (ii) refer the Dispute to the courts of England for resolution in accordance with paragraph 16 of Section J (General).
- 48.9 8.5 Where any relevant provision provides or the parties Dispute Parties have agreed that a dispute Dispute is to be referred to or resolved by Expert Determination:
 - (a) paragraphs9.1 to 9.30 paragraph 51.1 shall apply; and
 - (b) subject to paragraph 8.748.11, no party Dispute Party shall commence proceedings in any court of competent jurisdiction in respect of or otherwise in connection with such dispute Dispute.
- 48.10
 8.6-The parties Dispute Parties may agree to refer any dispute Dispute (including such a dispute Dispute as is referred to in paragraph 8.448.9) to mediation in accordance with paragraph 10.152.1.
- 48.11 8.7 Nothing in this section B (Governance) shall prevent any party Dispute Party from seeking interim or interlocutory relief in any court of competent jurisdiction.
- 49. Related Connected Disputes
- 49.1 With respect to any Dispute to be resolved by Expert Determination, where:
 - (a) 8.8 If, acting reasonably, a T&SCo believes that a dispute Dispute between a T&SCo and a User raises the same or similar issues to a dispute Dispute between that T&SCo and another User, T&SCo may require that the related disputes are addressed as a single dispute under this paragraph 8. (a "Connected Dispute");
 - (b) <u>the Dispute Resolution Procedure has been commenced in relation to the Dispute;</u> and
 - (c) the Dispute Resolution Procedure has been commenced in relation to the Connected Dispute,
 - the T&SCo may request consolidation of the Dispute and any Connected Disputes at any time so that the Dispute and Connected Disputes shall be determined together.
- Where the T&SCo wishes to consolidate Connected Disputes pursuant to paragraph 49.1, that T&SCo must give notice in writing to all of the parties to the Connected Disputes (a "Consolidation Request").
- 49.3 A Consolidation Request must:
 - (a) specify that it is a Consolidation Request;
 - (b) include a description of the subject matter of the Dispute;
 - (c) include a description of the Connected Disputes to which the Dispute is connected;

(d) include the reasons why the T&SCo believes the Dispute should be consolidated with the Connected Disputes; and

- (e) request that each User to which it is given respond within ten (10) Business Days of receipt, stating whether it agrees to consolidate the Dispute with the Connected Dispute to which it is party.
- Where in response to a Consolidation Notice relating to a Dispute not less than fifty per cent (50%) in number of Users in receipt of such Consolidation Notice agree that the Dispute should be consolidated with the Connected Disputes to which that Consolidation Notice relates, the Dispute shall be consolidated with all the Connected Disputes to which such Consolidation Notice relates.
- 49.5

 If different Experts have been appointed (as relevant) in respect of any Dispute and its

 Connected Disputes prior to their being consolidated in accordance with the Dispute

 Resolution Procedure, the Dispute Parties must use reasonable endeavours to agree in

 writing with each other and the parties to any relevant Connected Dispute, no later than five

 (5) Business Days after the giving of that notice, which of the Experts shall be the Expert for
 the consolidated Connected Disputes. If no such agreement can be reached, the parties to
 the Dispute shall request that the president or vice-president of the LCIA select, no later than
 five (5) Business Days after such request, which of those Experts shall be the Expert for the
 consolidated Connected Disputes.
- 49.6
 8.9 If the Secretary identifies that a dispute Dispute between a T&SCo and a User(s) raises the same or similar issues to a dispute Dispute between another T&SCo(s) and another User(s), the Secretary may notify the relevant T&SCos and:
 - (a) notify the relevant T&SCos; and
 - (b) if agreed by those T&SCos, the T&SCos may require that the related disputes are addressed as a single dispute under the Code. issue a Consolidation Request to each affected User in accordance with paragraph 49.2, and the provisions of paragraphs 49.3 to 49.5 shall apply to any such connected dispute as though such dispute was a Connected Dispute.
- Where two (2) or more Users or T&SCos are parties to a dispute, any notice, submission, statement or communication to be given by them may be given by them jointly (and any reference to the party or parties by whom and to whom it is to be given shall be construed accordingly).
- 50. 9. Expert determination Resolution by Senior Representatives
- The Dispute Parties must procure that their respective Senior Representatives shall meet within ten (10) Business Days of the date of service of the Dispute Notice or Expert Determination Notice (as applicable). If the Senior Representatives of the Dispute Parties:
 - are able to resolve the Dispute within thirty (30) Business Days of the date of service of the Dispute Notice or Expert Determination Notice (or within such longer period as the Senior Representatives of the Dispute Parties may agree in writing) (the "Resolution Period"), the terms of the agreement, compromise, resolution or

settlement reached between the Senior Representatives in respect of the Dispute (a "Senior Representatives Settlement") must be documented in writing and must be signed by the Senior Representative of each Dispute Party; or

- (b) <u>are unable to resolve the Dispute within the Resolution Period (where the Resolution Period follows the issue of a Dispute Notice), either Dispute Party may:</u>
 - if the Dispute Parties agree within the Resolution Period (or such longer period as they may agree) that such Dispute is amenable to determination by an Expert, refer the Dispute to an Expert for determination in accordance with the Expert Determination Procedure pursuant to paragraph 51; or
 - (ii) refer the Dispute to the courts of England for resolution in accordance with paragraph 16 of Section J (General); or
- (b) are unable to resolve the Dispute within the Resolution Period (where the Resolution Period follows the issue of an Expert Determination Notice), either Dispute Party may refer the Dispute to an Expert for determination in accordance with the Expert Determination Procedure pursuant to paragraph 51.
- 50.2 If, at any time during the Resolution Period, both Dispute Parties agree in writing that the Senior Representatives of the Dispute Parties will not be able to resolve the Dispute:

Expert Determination

- 9.1 A dispute which is to be referred to or resolved by Expert Determination shall be determined by an individual appointed as an expert in accordance with the relevant provisions of the Code.
- 9.2 Within one month of the Code Implementation Date, the Parties shall develop a list of experts comprising 12 persons available to act as experts in relation to:
 - (a) CDS Data Disputes; and
 - (b) Invoice Disputes.

in accordance with paragraph 9.1 and listed in alphabetical order by surname (**Listed Experts**).

- 9.3 The lists under paragraph 9.2 shall comprise six (6) persons nominated collectively by agreement of the T&SCos and six (6) persons nominated collectively by agreement of Users.
- 9.4 No person shall be nominated as expert under this paragraph 9 unless they have first confirmed in writing to the Secretary that they accept their nomination.
- 9.5 If any Listed Expert indicates that they no longer wish to be included in the list they shall beremoved from the list and replaced by a person nominated collectively by agreement of theoriginal nominating parties (i.e. either T&SCos or Users).

9.6 Any party to a dispute which is to be resolved by or referred to Expert Determination may givenotice of the dispute in accordance with this paragraph. The notice shall be given to each other party and shall:

- (a) state whether the party believes the dispute to be a CDSeither Dispute or an Invoice Party may refer the Dispute either to the courts of England or to Expert Determination (as applicable); and
- (b) there shall be no requirement for the Dispute Parties to wait until the expiry of the Resolution Period before making any such reference.
- Neither Dispute Party may commence the Expert Determination Procedure or (subject to paragraph 48.5) refer a Dispute to the courts of England prior to the expiry of the Resolution Period except in the circumstances specified in paragraph 50.2.
- Any statement, concession, waiver or agreement (other than a Senior Representative Settlement) made by a Dispute Party in the course of discussions and meetings pursuant to paragraph 50.1 shall be "without prejudice" to the Dispute unless expressly stated to be "without prejudice save as to costs" or to be made on an open basis, and the details of the discussions and any meetings, minutes and/or statements relating to such meetings shall be inadmissible in any proceedings that may follow (including pursuant to the Expert Determination Procedure), except that those expressly stated to be "without prejudice save as to costs" shall be admissible in respect of the matter of such costs.

51. **Expert determination**

- <u>Either Dispute Party may, subject to paragraph 50, refer a Dispute to be determined by an Expert if either:</u>
 - (a) the Dispute Parties have agreed in writing that a Dispute is amenable to determination by an Expert pursuant to paragraph 50.1(b) or 50.2(a); or
 - (b) provide brief details of the issues to be resolved; and the Code expressly provides for the relevant Dispute to be determined by an Expert,

and such referral shall be effected by either Dispute Party giving a notice (an "Expert Determination Notice") to the other Dispute Party.

- 51.2 <u>An Expert Determination Notice must:</u>
 - (c) nominate four persons as proposed experts.
- 9.7 Within five (5) Business Days after receiving a notice under paragraph 9.6, each party (other than the party giving such notice) shall by notice to each other party nominate four persons as proposed experts based on the nature of the dispute.
- 9.8 The parties shall endeavour within ten (10) Business Days after the notice under paragraph 9.6being given to agree upon the selection of an expert and may meet for this purpose.
- 9.9 If within ten (10) Business Days after the notice under paragraph 9.6 was given the parties shall not have agreed upon the selection of an expert, each shall within a further five (5) Business

Days give to the others a notice (**Preference Notice**) specifying the names of seven (7)-Listed Experts in order of preference and assigning to each a number (**Preference Number**) from 7 (the most preferred) to 1 (the least preferred).

9.10 The Listed Expert with the greatest aggregate the Preference Numbers assigned to them undereach Preference Notice shall be selected as the expert.

Appointment of the Expert

- 9.11 Once the expert is selected, the parties shall notify the expert of the selection and request that the expert confirm within five (5) Business days whether they are willing to accept the appointment.
- 9.12 The notification to the expert shall include the following:
 - (a) the names of the parties and a summary of the dispute specify that it is an Expert Determination Notice;
 - (b) a request that the expert provides a confirmation include the information required underto be included in a Dispute Notice pursuant to paragraph 9.2748.7; and
 - <u>include a proposal as to the identity, and terms of appointment, of the Expert and the relevant expertise that the referring Dispute Party considers qualifies the Expert to determine the relevant Expert Dispute.</u>
- <u>Any Expert appointed to determine any Expert Dispute shall be required to have an appropriate level of experience in relation to matters of the same general description as the matter in Dispute.</u>
- The Dispute Party receiving the Expert Determination Notice (the "Respondent") must, within ten (10) Business Days of the earlier of (i) the conclusion of the Resolution Period and (ii) the date on which the Dispute Parties agree in writing that the Senior Representatives of the Dispute Parties will not be able to resolve the Dispute pursuant to paragraph 50.2 either:
 - give notice to the other Dispute Party (the "Claimant") in writing that it is willing to appoint the Expert proposed by the Claimant; or
 - (b) give notice to the Claimant in writing that it is not willing to appoint the Expert proposed by the Claimant, in which case, the Respondent must, in such notice, propose an alternative Expert for consideration by the Claimant.
- 51.5 If the Dispute Parties have failed to agree on the identity of the Expert within twenty (20)

 Business Days of the earlier of (i) the conclusion of the Resolution Period and (ii) the date on which the Dispute Parties agree in writing that the Senior Representatives of the Dispute Parties will not be able to resolve the Dispute pursuant to paragraph 50.2, either Dispute Party may request that the Expert be nominated by the LCIA, which shall be requested to choose a suitably qualified and experienced Expert for the Expert Dispute in question. The LCIA's nomination shall be binding on the Dispute Parties.
- 51.6 The Dispute Parties must:

(a) use reasonable endeavours to procure that within ten (10) Business Days of the
Dispute Parties agreeing the identity of the Expert to be appointed (or the LCIA having nominated an Expert in accordance with paragraph 51.5):

- (i) the Expert confirms in writing to the Dispute Parties that:
 - (A) the Expert is willing and available to act in relation to the Expert Dispute; and
 - (B) the Expert has no conflict of interest which prevents the Expert from determining the Expert Dispute; and
- (ii) the terms of appointment of the Expert are agreed between the Dispute Parties and the Expert (and an appointment letter entered into among them), such terms:
 - (A) to include an undertaking that the Expert shall not disclose to any person any information or documentation disclosed or delivered by a Dispute Party to the Expert in consequence of, or in connection with, their appointment as the Expert; and
 - (B) to exempt the Expert (and any employee, agent or adviser of or to the
 Expert) from liability for anything done or omitted in the discharge or
 purported discharge of the Expert's functions unless such act or
 omission is fraudulent, negligent or in bad faith;
- (b) <u>instruct the Expert:</u>
 - (c) a request for confirmation of the expert's scale of feesto act fairly, independently and impartially;
- (d) a statement that the expert's fees and expenses will be paid as provided in paragraph-9.30;
- (e) a statement that the information disclosed in the notification is confidential and that itshould not be disclosed, copied or revealed whether the appointment is accepted or not;
 - (ii) to take the initiative in ascertaining the facts and the law, including by:
 - (A) considering any written representations, statements and experts' reports submitted to the Expert by the Dispute Parties;
 - (B) instructing an expert and/or taking counsel's opinion as to any matter raised in the Expert Determination, provided that the Expert shall not be entitled to delegate any decision to such expert or Counsel; and
 - <u>requiring the Dispute Parties to produce any information, evidence,</u> supporting documentation or explanations in written or documentary

form (except any documents which would be privileged from production in court proceedings);

- (iii) to reach their decision in accordance with the applicable laws in relation to the Expert Dispute referred to them; and
- <u>(iv)</u> <u>if requested by either Dispute Party in writing, to provide reasons for their</u> decision, which shall be communicated to the Dispute Parties;
- <u>afford the Expert the discretion to establish the procedure (including the timetable) for the determination of the Expert Dispute, it being agreed by the Dispute Parties that:</u>
 - the Expert shall be requested to confirm to the Dispute Parties the proposed procedure for the relevant Expert Dispute as soon as reasonably practicable after the Expert Appointment Date and, in any event, within ten (10) Business Days of the Expert Referral Date and, in so doing, the Dispute Parties agree that:
 - the Expert shall be requested to afford the Dispute Parties the opportunity to address the Expert in a meeting at which both Dispute Parties shall have the right to be present, either where a Dispute Party requests such a meeting in writing or the Expert otherwise considers it to be necessary or desirable to reach a determination in respect of the relevant Expert Dispute, with the format and procedure applicable to any such meeting being a matter for the Expert to decide in their sole and absolute discretion; and
 - (B) the Expert may (acting reasonably) modify the time periods provided for in paragraph 51.7 and otherwise modify the procedure contemplated by such paragraph;
 - (ii) all submissions made by a Dispute Party to the Expert (including all information, evidence, supporting documentation and explanations provided to them) shall be provided to the other Dispute Party contemporaneously with such submissions being made to the Expert; and
 - (iii) the Dispute Parties must (without prejudice to paragraph 51.6(c)(i)) request the Expert to determine the Expert Dispute within thirty (30) Business Days of the earlier of:
 - (A) (f) a copy of this paragraph 9the date on which a Response Submission has been provided by each Dispute Party; and
 - (B) the date falling thirty (30) Business Days after the First Submission

 Deadline; and
- afford the Expert all information, evidence, supporting documentation, explanations and assistance which the Expert requires to determine the Expert Dispute (and, if a Dispute Party fails to produce any such information, evidence, supporting documentation, explanation or assistance, the Expert may continue the determination

process in the absence of that information, evidence, supporting documentation, explanation or assistance).

51.7 Subject to paragraph 51.6(c):

- (a) the Claimant must provide the Expert with a copy of the Expert Determination Notice
 no later than ten (10) Business Days after the Expert Appointment Date (the date on
 which the Expert receives the copy of the Expert Determination Notice being the
 "Expert Referral Date");
- each Dispute Party may, but is not obliged to, provide a written statement of its case to the Expert (the "First Submission") within twenty (20) Business Days of the Expert Referral Date (the "First Submission Deadline") and, without limitation, the First Submission may cover any of the matters required to be contained in the relevant Expert Determination Notice pursuant to paragraphs 48.7(c) to 48.7(h) (inclusive) and a copy of such First Submission must be provided to the other Dispute Party at the same time as it is provided to the Expert; and
- (c) each Dispute Party may submit a reply to the other Dispute Party's First Submission (a "Response Submission") within thirty (30) Business Days of receipt of the First Submission.
- 51.8 If after their appointment the Expert becomes aware of any interest or duty which conflicts or potentially conflicts with the performance of their duties under their contract with the Dispute Parties, the Expert must notify the parties forthwith of such conflict giving full details thereof.
- Any Dispute Party may within five (5) Business Days of the disclosure of any such conflict or potential conflict pursuant to paragraph 51.8 object to the appointment or continued appointment of an Expert, in which case the Expert shall not be or shall cease to be appointed and a new Expert shall be selected and appointed in accordance with this Section B (Governance) (and the rejected Expert shall not be nominated for such selection).
 - (g) a request for confirmation that the expert is able and willing to act in accordance with the procedure set out herein.
- 9.13 If the selected expert is unwilling or unable to accept the appointment, or shall not have confirmed their willingness and ability to accept such appointment within the period required, the parties shall endeavour to agree upon the selection of another expert within three (3)

 Business Days, failing which another expert shall be selected in accordance with paragraph 9.11.

Timetable and Procedure

- 9.14 No later than five (5) Business Days following their appointment, the expert shall by giving reasonable notice to each party convene a meeting with the parties at which the expert shall-raise any matters upon which they requires clarification and discuss with the parties any additional procedural requirements the expert or the parties may have.
- 9.15 The parties shall, not later than ten (10) Business Days after the appointment of the expert, submit to the expert and to each other party written submissions of not more than ten (10)

pages in length together with all supporting documentation, information and data which they wish to submit in respect of the dispute; and the parties may also submit a statement of facts which they have agreed between themselves to the expert.

- 9.16 Each party may, not later than twenty (20) Business Days after the appointment of the expert, submit to the expert and each other party written submissions of not more than ten (10) pages in length, together with any additional supporting documentation, information and data, in reply to the submissions made under paragraph 9.15.
- 9.17 The expert shall disregard any documentation, information, data or submissions supplied or made (other than pursuant to paragraph 9.22) by any party later than twenty (20) Business-Days after their appointment unless the same are provided in response to a request from the expert.
- 9.18 If the expert shall wish to obtain independent professional and/or technical advice in connection with the dispute:
 - (a) they shall first provide the parties with details of the name, organisation and estimated fees of the professional or technical adviser; and
 - (b) they may engage such adviser with the consent of the parties (which consent shall not be unreasonably withheld) for the purposes of obtaining such professional and/or technical advice as they may reasonably require.
- 9.19 The expert may at their discretion and at any time request information from any of the parties orally but shall only do so in the presence of the other parties.
- 9.20 At any time after the period referred to in paragraph 9.16 expires, with the written consent of the parties, the expert may (but shall not be required to) convene a hearing upon giving the parties reasonable notice.
- 9.21 The expert shall provide a draft of their determination, which shall be a report in writing giving reasons for the determination, to the parties not later than thirty-five (35) Business Days-following the expert's appointment.
- 9.22 Each party may, within ten (10) Business Days following delivery of the draft determination, submit to the expert any documentation, information, data, submissions or comments not exceeding 5 pages in length on or in respect of the draft determination.
- 9.23 The expert shall submit their final determination, which shall be a report in writing giving reasons for their determination of the dispute, to the parties not later than fifty (50) Business Daysfollowing the expert's appointment.
- 9.24 If the expert fails to submit the final determination by the time required under paragraph 9.23, at the request of any party another expert may be appointed in accordance with the provisions of this paragraph 8 and the appointment of the previous expert shall cease unless, before the appointment of the new expert, the previous expert shall have submitted their final determination hereunder, in which case the new expert shall be forthwith informed that their services will not be required.

- 51.10 Effect of No Expert determination shall have the effect of amending the Code.
- 51.11 The Expert shall not act as an arbitrator nor a mediator and the Arbitration Act 1996 and the law relating to arbitrators and arbitrations shall not apply to the Expert or their determination or the procedure by which they reach their determination.
- The Expert shall be an independent contractor and the relationship of the Dispute Parties and the Expert shall in no event be construed to be that of principal and agent or master and servant.
- 51.13 <u>If:</u>
 - <u>the Parties are unable to agree on the terms of the appointment of the Expert under paragraph 51.6(a)(ii);</u>
 - (b) the Expert is at any time or unable or unwilling act (whether under paragraph 51.6(a)(i) or otherwise); or
 - (c) <u>the Expert fails to render a determination in accordance with paragraph 51.6(b),</u>
 - either Dispute Party may proceed to seek the appointment of a replacement Expert as if the Expert Determination Notice had just been served.
- The provisions of the Expert Determination Procedure shall apply mutatis mutandis to any replacement Expert and the replacement Expert shall be authorised to determine any Expert Dispute which was submitted to their predecessor but which their predecessor had not determined at the time when their predecessor became unable or unwilling to act.
- <u>9.25</u> The expert's Expert's final determination shall be final and binding on the parties Dispute Parties except in the event of fraud, error of law (including an error of interpretation of the Code) or where it is so clearly erroneous on its face that it would be unconscionable for it to stand, in which case another expert may be appointed in accordance with the provisions of this paragraph 951.
- 9.26-Except in respect of the application of paragraph 9.2551.15, no party Dispute Party shall commence proceedings in respect of or refer to any court any finding by the expert Expert, whether made at any time after the expert's Expert's appointment or in his their determination, as to the dispute Dispute or the construction of or otherwise in respect of the Code, the Code Agreement, any Code Accession Agreement, any CDS Accession Agreement or any Ancillary Agreement.
- 51.17 Notwithstanding paragraph 51.16, either Dispute Party can seek enforcement of an Expert Determination in the courts of competent jurisdiction.
- The Expert may, in their determination, provide that one (1) or other or both of the Dispute

 Parties pay the Expert's fees and each other's legal costs in such proportions as they may

 specify on the general principle that costs should reflect the Dispute Parties' relative success
 and failure in the Expert Determination Procedure. Without such a direction, each Dispute

Section B – Governance CCS Network Code

Party shall bear its own legal costs and the fees and expenses of the Expert shall be paid in equal shares by the Dispute Parties.

Conflict of interest

9.27 The expert shall confirm to the parties before the expert's appointment that they do not hold any interest or duty which would or potentially would conflict with the performance of their duties under their contract with the parties.

Confidentiality

9.28 Subject to paragraphs 8.8 and 8.9 parties and the expert shall keep the fact that the Expert Determination is taking place and its outcome confidential.

Costs

- 9.29 Each party shall bear its own costs including, without limitation, costs of providing documentation, information, data, submissions or comments and all costs of and expenses of all witnesses and other persons retained by it.
- 9.30 The expert's fees and expenses shall be payable by the parties in equal amounts, unless the expert (having regard to the conduct of the parties with respect to the dispute in question) shall direct in their final determination that such costs and expenses should be borne by one or some only of the parties, in which case the parties shall pay such fees and expenses in accordance with such direction.
- 52. 40. Mediation
- 52.1 40.1 A dispute <u>Dispute</u> which is to be referred to mediation shallmust be referred to a single mediator who shall explore the interests of the parties to the dispute <u>Dispute Parties</u> and encourage the parties <u>Dispute Parties</u> to resolve the dispute <u>Dispute</u> in light of such interests.

Appointment of mediator

- 52.2 40.2-Within five (5) Business Days after agreeing to refer a dispute Dispute to mediation, the parties shall Dispute Parties must meet and use their best endeavours to agree upon a person to act as mediator.
- 52.3 Here the parties Dispute Parties agree upon a mediator they shall request such person to accept appointment as mediator, and shall must use all reasonable endeavours to agree (between themselves and with the mediator) upon the terms of their appointment.
- 52.4 10.4-A person shall be treated as appointed as a mediator f-when they have confirmed their acceptance to act as mediator in accordance herewith.

Procedure and timetable

52.5 Hithin five (5) Business Days following their appointment, the mediator shall require each party Dispute Party to provide them with a written summary of the dispute, which written summary shallmust not exceed five (5) pages.

Section B – Governance CCS Network Code

- 52.6 10.6 The mediator may in their discretion:
 - (a) request any party Dispute Party to provide them with copies of any documentation or information which the mediator believes will assist to explain any such summary; and
 - (b) provide any such written summary and/or any information or copy documentation received under paragraph (a)52.6(a) to the other party(ies) to the dispute Dispute Parties.
- 52.7 Within ten (10) Business Days following their appointment, the mediator shallmust contact the parties and shall Dispute Parties and must arrange to meet them.
- 52.8 10.8 Each party shall Dispute Party must attend the meeting with the mediator with a maximum of three (3) representatives, one (1) of whom shall must be a person with decision making authority in relation to the subject matter of the dispute Dispute and one (1) of whom may be the legal adviser of that party Dispute Party.
- 52.9 40.9 No additional persons shall attend without the prior written consent of the mediator.
- 52.10 10.10-The mediator may convene more than one (1) meeting with the parties Dispute Parties but shallmust not convene any meetings later than forty (40) Business Days following their appointment, unless the parties Dispute Parties agree otherwise.
- 52.11 10.11 The mediator may at their discretion meet each party Dispute Party on their own whether during a meeting attended by the other parties Dispute Parties or otherwise, but they shall must not disclose to any other party Dispute Party matters disclosed to the mediator in such circumstances without the consent of the disclosing party Dispute Party.
- 52.12 10.12 At any meeting attended by the party_Dispute Party to make a brief presentation of its case and the mediator may also require the other <a href="mailto:party_spute_Party
- 52.13 10.13 The mediator shall not act as an arbitrator (and accordingly the provisions of the Arbitration Acts Act 1996 and the law relating to arbitrators and arbitrations shall not apply to any mediation).

Result of mediation

- 52.14 10.14 No partyNo Dispute Party shall be bound to adopt the views or advice expressed or provided by the mediator.
- 52.15

 10.15 If the dispute Dispute is resolved or the parties accept the views and advice of the mediator, the parties shall Dispute Parties must use their best endeavours, within five (5) Business Days after such resolution or acceptance, to enter into a settlement agreement which shall must:
 - (a) set out the terms accepted by the parties Dispute Parties or on which the dispute Dispute was resolved; and
 - (b) contain provisions of confidentiality.

Section B – Governance CCS Network Code

Confidentiality

10.16 The parties shall keep confidential the fact that the mediation is taking place.

The mediator and the Dispute Parties, their representatives and advisers and any person connected in any way with the mediation must keep confidential the fact that the mediation is taking place, and its outcome, and all documents, submissions, statements, information and data including anything revealed orally or otherwise during the mediation and any settlement agreement except as may be necessary for implementation or enforcement of the settlement agreement.

All documents and information prepared by a Dispute Party for and disclosed in the mediation, and all discussions which take place with a Dispute Party during the course of the mediation, shall be treated as confidential, provided that this shall not preclude any document, which may have been disclosed during the mediation but was not prepared solely for use in the mediation, from being discoverable in any proceedings.

Costs

- 52.18 10.17 The parties shall Dispute Parties must, unless they agree otherwise, bear their own costs and expenses of whatsoever nature of the mediation.
- The parties shall Dispute Parties must bear the fees and expenses of the mediator and all administrative costs arising from the mediation equally.

Further proceedings

- 52.20 The mediator shall not act in any subsequent Expert Determination, court proceedings, arbitration or similar proceedings in respect of the Dispute in which they acted as mediator.
- 52.21 The mediator shall not be held liable for any act or omission unless it shall be shown that they have acted fraudulently or in bad faith.

Section C – Connection CCS Network Code

CCS Network Code

Section C - Connection

1. Introduction

1.1 This Section C:

- 1.1 (a) describes how Initial Users connected This Section C (Connection) sets out the process for connection to a T&S Network; by:
 - (b) confirms the process for Prospective T&SCos to accede to the Code; and
 - (c) sets out the process for connection to a T&S Network by prospective Users who seek to apply to connect to a T&S Network; or to modify an existing Delivery Point
 - (b) <u>existing Users who seek to modify an existing Delivery Point,</u>

after the Code Implementation Date, where:

- (c) (i) a T&SCo is already a party to the Code at the time a prospective User becomes an Eligible Applicant; and or
- (d) (ii) a T&SCo is a Prospective T&SCo at the time a prospective User becomes an Eligible Applicant.
- 1.2 Notwithstanding any other provision of this Section C (Connection), the T&SCo's obligations under this Section C (Connection) are subject to the provisions of Standard Condition B3 (Access to T&S Network) of the Licence.
- 1.3 The Parties acknowledge and agree that notwithstanding paragraph 17.1 of Section J

 (General), the Contracts (Rights of Third Parties) Act 1999 shall apply to the rights conferred on Eligible Applicants and Prospective T&SCos under this Section C (Connection).
- 2. Application of and rationale for Section C
- 2.1 This Section C (Connection) reflects the fact that, as at the Code Implementation Date, HM

 Government is leading the process for:
 - (a) the selection of T&SCos to develop and operate new T&S Networks;
 - (b) the selection of new Users to connect to T&S Networks; and
 - (c) the allocation of Network Capacity to existing and new Users.
- 2.2 For the reasons described in paragraph 2.1:
 - (a) the connection process set out in this Section C (Connection); and
 - (b) the process for allocation of Network Capacity described in Section E (Network Use and Capacity),

are linked to the Selection Process.

- 2.3 1.2 This Section C (Connection) (including the connection dispute provisions in paragraph 1015) is intended to apply to applications for connection to a T&S Network (or modification of an existing Delivery Point) until:
 - (a) the UKHM Government discontinues the Selection Process; or
 - (b) a new connection process is developed under the Code (whether in response to new regulations being made under the Act or otherwise),

whichever is the earlier.

Note: in the early stages of the development of the CCUS industry the Government has been leading the process for the allocation of new connections and network capacity to different types of users. It is expected that this will change over time, as the industry matures. The definition of "Selection Process" remains under review and development.

- 2.4 It is further intended that where the provisions of this Section C (Connection) are revised (as contemplated by paragraph 2.3) then any new provisions of this Section C (Connection) will include transitional provisions to apply to any applications already commenced under the current version of this Section C (Connection), subject to any Legal Requirements.
- 3. 2. Initial Users
- 3.1 2.1 It is acknowledged that as at the Code Implementation Date:
 - (a) <u>in accordance with the Track-1 Cluster Sequencing Process,</u> the Initial Users have agreed to connect to a T&S Network and a T&SCo has agreed to allow these Users been selected to connect to a T&S Network; and
 - (b) each of thesethe Initial Users has entered into will accede to the Code Agreement, and will enter into a Construction Agreement and a Connection Agreement, or will do so as soon as it has reach financial investment decision; and in accordance with and subject to paragraph 13.
 - (c) each of these Initial Users holds Registered Capacity or has been provisionally allocated. Network Capacity (pending entry into the documents referred to in paragraph 2.1(b) and becoming a User).
- 2.2 For the avoidance of doubt, it is acknowledged that as these Initial Users applied for a connection to a T&S Network before the Code Implementation Date, the provisions of the balance of Section C do not apply to these Initial Users, except for paragraph 10 (Connection disputes).
- It is acknowledged that where an Initial User has entered into the Code Agreement, and has entered into a Construction Agreement and a Connection Agreement as at the Code Implementation Date, the provisions of this Section C (Connection) do not apply to that Initial User in relation to the Delivery Point to which that Connection Agreement relates, except for paragraph 15.

- 3.3 Where an Initial User has not entered into the Code Agreement and/or has not entered into a Construction Agreement and a Connection Agreement as at the Code Implementation Date:
 - (a) this Section C (Connection) will apply to that Initial User, except that the Initial User and the T&SCo shall not be required to re-start, after the Code Implementation Date, any steps that form part of the connection process set out in this Section C (Connection), to the extent that those steps have already been completed; and
 - (b) without limiting the generality of paragraph 3.3(a), the T&SCo's obligation to issue a

 Connection Offer to the Initial User (where a Connection Offer has not yet been issued as at the Code Implementation Date) will be conditional on the Initial User giving the T&SCo a copy of the Final Confirmation Notice in accordance with paragraph 11.
- 4. 3. Compliance with Section C where a T&SCo is a Prospective T&SCosSCo
- 4.1 From the Code Implementation Date, where as part of a Selection Process, HM Government indicates that it is intended that a Prospective T&SCo will become a party to the Code in accordance with the Licence that will be granted to that T&SCo:
 - (a) this Section C (Connection) will apply as if the Prospective T&SCo was a party to the Code; and
 - (b) in paragraphs 5 to 15, a reference to a T&SCo includes a reference to a Prospective T&SCo, unless otherwise specified.
- <u>Where paragraph 4.1 applies, notwithstanding that the Prospective T&SCo is not a party to the Code, it is intended that:</u>
 - <u>a prospective User seeking a connection to the Prospective T&SCo's T&S Network</u>

 must make its connection application in accordance with this Section C (Connection);
 - (b) the Prospective T&SCo must comply with Section C (Connection) in relation to the prospective User's application for a connection; and
 - (c) 3.1 From the Code Implementation Date a Connection Application may be made to a Prospective T&SCo₁ notwithstanding the fact that a Prospective T&SCo is not a party to the Code.
- 4.3 3.2-A Prospective T&SCo shall have the right <u>and obligation</u> to accede to the Code, in accordance with the provisions of the Code Agreement, <u>Section A (Introduction, Structure and Interpretation) and its Licence</u>, upon being granted a Licence.
- 3.3 In paragraphs 4 to 10 of this Section C a reference to a T&SCo includes a reference to a Prospective T&SCo, unless otherwise specified.

4. Pre-Selection due diligence

Note: further consideration is being given to any due diligence input needed from T&SCo-(under the CCS Network Code/Licence) as part of any Selection Process.

5. Eligibility

- 5. Eligibility for application for connection
- 5.1 A prospective User <u>(or an existing User applying to modify an existing Delivery Point)</u> is an "Eligible Applicant" for the purposes of this Section C where the prospective User has been Selected and has not been Deselected. Section C (Connection) where that party:
 - (a) has been Selected; and
 - (b) <u>has not been Deselected.</u>
- 5.2 An Eligible Applicant may apply for:
 - (a) a connection to a T&S Network at a new (as yet unbuilt) Delivery Point; or
 - (b) the modification of an existing Delivery Point; in relation to which the Eligible Applicant holds a Connection Agreement.

as the case may be, in accordance with this Section C (Connection).

- 5.3 An Eligible Applicant's eligibility for a connection to a T&S Network or modification of an existing Delivery Point (as the case may be) relates to the T&S Network or Delivery Point that the Eligible Applicant specified in its application under the Selection Process.
- 6. Connection Application process

Connection Application

- 6.1 Where an Eligible Applicant wishes to request:
 - (a) a connection to the T&S Network at a new (as yet unbuilt) Delivery Point; or
 - (b) the modification of an existing Delivery Point;

then the Eligible Applicant shallmust complete and submit to the T&SCo an application (in the form prescribed by the T&SCo from time to time) (the "Connection Application") and comply with the terms of it.

- 6.2 Where the Connection Application relates to a new Delivery Point, it shall specifythe T&SCo may require the Eligible Applicant to provide the following information in the Connection Application:
 - (a) the nature and location of the User Facility to be connected to the T&S Network at the Delivery Point;
 - (b) the proposed size and location of the Delivery Point being applied for;
 - (c) the date when the Eligible Applicant proposes to start using the Delivery Point to deliver its carbon dioxide into the T&S Network;

- (d) the quantity of Network Capacity provisionally allocated to the Eligible Applicant through the Selection Process; and
- (e) <u>a forecast of:</u>
 - (i) the quantity of carbon dioxide (expressed in tCO₂) to be delivered to the T&S

 Network through the Delivery Point; and
 - (ii) the Eligible Applicant's capacity requirements,

for the following twenty (20) Charging Years;

- (f) any Local Requirements relating to the proposed connection to the T&S Network; and
- (g) (e) any other information that the T&SCo may reasonably require.
- 6.3 Where the Connection Application relates to a modification of an existing Delivery Point, it—
 shall specifythe T&SCo may require the Eligible Applicant to provide the following information
 in the Connection Application:
 - (a) the nature and location of the User Facility at the Delivery Point;
 - (b) details of the existing Delivery Point and the modifications being applied for;
 - (c) the date when the Eligible Applicant proposes to start using the modified Delivery Point to deliver its carbon dioxide into the T&S Network;
 - (d) the quantity of additional Network Capacity (if any) provisionally allocated to the Eligible Applicant through the Selection Process; and
 - (e) <u>a forecast of:</u>
 - (iii) the quantity of carbon dioxide (expressed in tCO₂) to be delivered to the T&S

 Network through the Delivery Point; and
 - (iv) the Eligible Applicant's capacity requirements,

for the following twenty (20) Charging Years;

- <u>(f)</u> <u>any Local Requirements relating to the proposed modification of an existing Delivery</u> Point; and
- (g) (e) any other information that the T&SCo may reasonably require.
- 6.4 The An Eligible Applicant must provide the following when submitting its Connection Application shall also be accompanied by:
 - (a) evidence that the Eligible Applicant has been Selected; and
 - (b) the Original Selection Details;

- (c) <u>details of any Local Requirements relating to the proposed connection to the T&S</u>

 Network or modification of an existing Delivery Point (as the case may be); and
- (d) (b) copies of any documents/information relating to the proposed connection to the T&S Network or modification of an existing Delivery Point (as the case may be) that were submitted by the Eligible Applicant as part of its application through the Selection Process.

7. ValidValidation of Connection Application

- 7.1 6.5 A Connection Application shall be a "Valid Connection Application" where:
 - (a) the application form referred to in paragraph 6.1:
 - (i) <u>has been correctly and fully completed, and</u>
 - (ii) (a) the application form has been correctly and fully completed, and is consistent in all respects with the details which formed the basis of the Eligible Applicant's application through the Original Selection Process and subsequent Selection Details; and
 - (b) the Eligible Applicant has provided the evidence documents referred to in paragraph 6.4(a)6.4.

7.2 6.6 The T&SCo shallmust:

- (a) as soon as reasonably practicable, and in any case no more than five (5) Business
 Days from the date of receipt of the Connection Application, acknowledge receipt of such Connection Application;
- (b) within twenty (20) Business Days of receipt of the Connection Application:
 - (i) confirm that it provide notice to the Eligible Applicant that the Connection Application is a Valid Connection Application; or
 - <u>where the T&SCo considers that the Connection Application is not a Valid Connection Application:</u>
 - (A) (ii) advise the Eligible Applicant that the Connection Application is not a Valid Connection Application and specify the reasons that the Connection Application is not a Valid Connection Application; and
 - (B) where it the T&SCo considers that further information is required in order for it to consider or progress the Connection Application, request such further information from the Eligible Applicant.
- 7.3 6.7 The Eligible Applicant shallmust provide any additional requested information as soon as reasonably practicable and in any case no more than fifteen (15) Business Days after the date of receipt of the request from the T&SCo.
- 7.4 6.8 Where a submitted Connection Application is not a Valid Connection Application, the Eligible Applicant may resubmit, amend or supplement the Connection Application and the

provisions of paragraph 7.2 shall apply in relation to such resubmitted, amended or supplemented Connection Application.

7. Post-Selection due diligence

Note: further consideration is being given to any due diligence input needed from T&SCo-(under the CCS Network Code/Licence) as part of any Selection Process.

8. T&SCo to issue Initial Offer

Issue of Initial Offer

- 8.1 Within three (3) months of the date on which the T&SCo notifies the Eligible Applicant that the relevant Connection Application is a Valid Connection Application, the T&SCo shallmust either:
 - subject to paragraph 8.2, advise the Eligible Applicant that the Connection Applicant's
 Valid Connection Application has been rejected; or
 - (b) issue an Initial Offer to the Eligible Applicant.
- 8.2 <u>The T&SCo can onlymay</u> reject an Eligible Applicant's Valid Connection Application where the Eligible Applicant has been Deselected or refuse to issue an Initial Offer only where:
 - (a) the Eligible Applicant has been Deselected; or
 - (b) the new connection or modification of an existing Delivery Point (as the case may be) is conditional upon:
 - (i) the T&SCo obtaining a Change in Scope Approval and the Regulator has confirmed that it will not be giving such Change in Scope Approval; or
 - (ii) any other regulatory approvals or permits being granted to the T&SCo and the T&SCo (acting reasonably) considers that there is no reasonable prospect of such approvals or permits being granted (or such approvals or permits have already been refused); or
 - (c) there are any other circumstances that fall within the scope of Part C (Exceptions to obligation to offer access) of Standard Condition B3 (Access to T&S Network) of the T&SCo's Licence.

9. Contents of Initial Offer

- 9.1 8.3 An Initial Offer shall The T&SCo must include the following documents with the Initial Offer it issues to an Eligible Applicant pursuant to paragraph 8.1:
 - (a) a draft Construction Agreement, including:
 - (i) a draft Construction Programme; and
 - (ii) a draft Commissioning Programme (except where Schedule 6 of the Construction Agreement will apply);

- (b) a draft Connection Agreement; and
- (c) a draft <u>Code</u> Accession Agreement (if the Eligible Applicant is not already a User under the Code).
- 9.2 In providing the Initial Offer, the T&SCo must advise the Eligible Applicant whether the Eligible Applicant is an Onshore User or an Offshore User (for the purposes of Section H (Charges, Invoicing and Payment)).
- 9.3 The T&SCo must ensure that that the Initial Offer and the draft Construction Agreement and draft Connection Agreement issued with it are consistent with the Original Selection Details and take account of any Local Requirements set out in a Connection Application.
- 9.4 Where relevant, the Initial Offer will be conditional upon a Change in Scope Approval (and any other regulatory approvals or permits being granted to the T&SCo) and where this is the case, this will be stated in the Initial Offer.
- 9.5 Where the T&SCo is a Prospective T&SCo, it is intended that the Initial Offer will be conditional upon the grant of a Licence to the Prospective T&SCo.
- 9.6 The Initial Offer must be substantially in the form of the template letter set out in Exhibit A to this Section C (Connection).
- 10. Acceptance or refusal of Initial Offer
- 10.1 8.4 The Subject to paragraph 10.3, the Eligible Applicant will have one (1) month to must accept or refuse the Initial Offer within one (1) month of the date of the Initial Offer.
- 8.5 If the Eligible Applicant decides to accept the Initial Offer then, the Eligible Applicant shallmust return to the T&SCo a signed copy of the Initial Offer.
- Where the Eligible Applicant has not accepted the Initial Offer within the time period referred to in paragraph 10.1, the Initial Offer will lapse, unless the T&SCo and Eligible Applicant agree to extend the time for acceptance of it by the Eligible Applicant.
- <u>8.6.</u> An Eligible Applicant's acceptance of the Initial Offer shall not give rise to a legally binding contract between the Eligible Applicant and the T&SCo.
- 11. Final Confirmation Notice
- 11.1 The Secretary of State may issue a Final Confirmation Notice to an Eligible Applicant and the Eligible Applicant must give the T&SCo a copy of the Final Confirmation Notice as soon as practicable after it has received it in accordance with the Selection Process.
- 11.2 The T&SCo agrees with the Secretary of State that it will only make a Connection Offer to an Eligible Applicant in accordance with paragraph 12.1.
- 11.3 The Parties acknowledge and agree that notwithstanding para 17.1 of Section J (General), the Contracts (Rights of Third Parties) Act 1999 shall apply to the rights conferred on the Secretary of State under paragraph 11 of Section C (Connection).

12. 9. Connection Offer

Issuing of a Connection Offer

- 9.1-Where an Eligible Applicant has accepted an Initial Offer, the T&SCo shall, subject to paragraph 9.4, must issue a Connection Offer to the Eligible Applicant as soon as it reasonably practicable having regard to the work required to be undertaken by T&SCo to prepare the Connection Offer. within thirty (30) Business Days (subject to paragraph 12.2) of the later of:
 - (a) receipt of the Eligible Applicant's acceptance of the Initial Offer pursuant to paragraph 10; and
 - (b) receipt of a copy of the Final Confirmation Notice from the Eligible Applicant pursuant to paragraph 11.
- Where, in the reasonable opinion of the T&SCo, having regard to the scope of work that is required to prepare the Connection Offer, more time is required to prepare and issue the Connection Offer to the Eligible Applicant, the T&SCo may extend the time period referred to in paragraph 12.1 by notice, advising the Eligible Applicant why and how much additional time is required for the T&SCo to be able to issue the Connection Offer, provided that:
 - (a) the time period may be extended by any period of delay in the Eligible Applicant providing information requested pursuant to paragraphs 12.3 and 12.4; and
 - (b) <u>for any other reason of delay, the T&SCo may only extend on one (1) occasion.</u>
- 12.3 9.2 As part of the process to prepare a Connection Offer, the T&SCo and the Eligible Applicant shall cooperate with each other and must:
 - (a) <u>cooperate with each other;</u>
 - (b) provide each other with any additional information or clarification as may be reasonably required, and within such timeframes as may be reasonable in the circumstances; and

Content of Connection Offer

- (c) <u>keep each other updated at regular intervals in relation to matters that are relevant to the Connection Application.</u>
- so that the T&SCo can prepare the Connection Offer, including all the information that must be included in the documents referred to in paragraph 12.6.
- Where it is intended that Schedule 6 of the Construction Agreement will apply (such that no Commissioning Programme can be agreed at the time the Construction Agreement is entered into), the Eligible Applicant must provide to the T&SCo information about when the Eligible Applicant is likely to be available to have its connection commissioned and any specific requirements relating to such commissioning.

- Each Party must pay its own costs relating to the preparation of the Connection Offer, including the costs of any technical studies or reports.
- 12.6 9.3 A Connection Offer shall The T&SCo must include the following documents with the Connection Offer it issues to an Eligible Applicant pursuant to paragraph 12.1:
 - (a) a final Construction Agreement, including:
 - (i) the Construction Programme; and
 - (ii) the Commissioning Programme (except where Schedule 6 of the Construction Agreement will apply);
 - (b) a final Connection Agreement; and
 - (c) a final <u>Code</u> Accession Agreement (if the Eligible Applicant is not already a User under the Code).

Obligation to issue Connection Offer

- The T&SCo must ensure that that the Connection Offer and the Construction Agreement and
 Connection Agreement issued with it are consistent with the Original Selection Details (as
 may be amended by the Final Confirmation Notice) and take account of any Local
 Requirements set out in a Connection Application, subject to any amendments discussed as
 part of the process to prepare the Connection Offer pursuant to paragraph 12.3.
- Where, in order to be able to connect the User Facility to the T&S Network or modify an existing Delivery Point (as the case may be), the T&SCo is required to obtain a Change in Scope Approval, the agreements referred to in paragraph 12.6 will include this requirement as a condition precedent (including any conditions associated with that Change in Scope Approval), and this will be set out in the Connection Offer.
- Where either the T&SCo or the Eligible Applicant require the agreements referred to in paragraph 12.6 to be subject to any other conditions precedent to a final investment decision by that party (other than the condition precedent referred to in paragraph 12.8) then:
 - (a) that requirement must be communicated to the other party as part of the Connection

 Offer preparation process pursuant to paragraph 12.3; and
 - (b) the agreements referred to in paragraph 12.6 will include any such conditions precedent, and they will be set out in the Connection Offer.
- 12.10 9.4 A<u>The</u> T&SCo is required may refuse to issue a Connection Offer to an Eligible Applicant unless the Eligible Applicant has been Deselected only where:
 - (a) the applicant has been Deselected; or
 - (b) the new connection or modification of an existing Delivery Point (as the case may be) is conditional upon:

- <u>(i)</u> <u>the T&SCo obtaining a Change in Scope Approval and the Regulator has</u> confirmed that it will not be giving such Change in Scope Approval; or
- (ii) any other regulatory approvals or permits being granted to the T&SCo and the T&SCo (acting reasonably) considers that there is no reasonable prospect of such approvals or permits being granted (or such approvals or permits have already been refused); or
- (c) there are any other circumstances that fall within the scope of Part C (Exceptions to obligation to offer access) of Standard Condition B3 (Access to T&S Network) of the T&SCo's Licence.
- 12.11 Where the T&SCo is a Prospective T&SCo, at the time that it makes a Connection Offer, it is intended that the Connection Offer shall be conditional on the Prospective T&SCo becoming a party to the Code upon being awarded a Licence and acceding to the Code Agreement.
- 12.12 The Connection Offer must be substantially in the form of the template letter set out in Exhibit B to this Section C (Connection).
- 13. Acceptance or refusal of Full-Connection Offer
- 13.1 9.5 The Subject to paragraph 13.3, the Eligible Applicant will have three (3) months to must accept or refuse the Connection Offer within three (3) months of the date of it.
- 13.2 9.6 If the Eligible Applicant decides to accept the Connection Offer then, the Eligible Applicant shallmust return to T&SCo duly executed copies of:
 - (a) return to T&SCo a duly executed copy of the Connection Offer; and
 - (b) at the same time, commit to entering into:
 - (a) (i) the Code Agreement or Code Accession Agreement (as applicable);
 - (b) (ii) the Construction Agreement; and
 - (d) (iii) the Connection Agreement,

in accordance with the instructions set out in the Connection Offer, and subject to:

- (e) subject to any conditions precedent to the financial investment decision of both T&SCoand the Eligible Applicant, as referenced, as set out in the Connection Offer- in accordance with paragraphs 12.8 and 12.9; and
- 9.7 Where T&SCo is a Prospective T&SCo at the time it makes a Connection Offer, then that Connection Offer shall be conditional on the Prospective T&SCo becoming a party to the Code upon being awarded a Licence and acceding to the Code Agreement.
 - (f) <u>any conditions precedent set out in the Final Confirmation Notice.</u>
- 13.3 Where the Eligible Applicant has not accepted the Connection Offer within the time period referred to in paragraph 13.1, the Connection Offer will lapse, unless the T&SCo and Eligible Applicant agree to extend the time for acceptance of it by the Eligible Applicant.

14. <u>Information publication requirements</u>

- 14.1 For the purposes of the Connection Application process, the T&SCo must publish the following information on the T&S Network Portal:
 - (a) the T&SCo's Connection Application template; and
 - (b) a succinct guide to the information that must be provided with the Connection Application.

15. 40. Connection disputes

Mediation

- 15.1 Where a dispute arises between a User or a prospective User and a T&SCo in relation to a Connection Application, the directors or other senior representatives of the parties with authority to settle the dispute shall, following a written request from one party to the other, attempt in good faith to resolve the dispute.
- 15.2 16.2 If the dispute cannot be resolved within 10thirty (30) Business Days of the request referred to in paragraph 10.115.1, then the parties agree to refer the dispute to mediation in accordance with the mediation procedure set out in paragraph 10 of Section B of Section B (Governance), as if all the parties had acceded to the Code.

The Storage of Carbon Dioxide (Access to Infrastructure) Regulations

15.3 The process for the parties seeking to resolve a dispute in accordance with paragraphs 10.115.1 and 10.215.2 shall be without prejudice to the right of a User or a prospective Userthe parties under the Storage of Carbon Dioxide (Access to Infrastructure) Regulations 2011 or the Storage of Carbon Dioxide (Access to Infrastructure) Regulations (Northern Ireland) 2015.

Exhibit A - Initial Offer

Date [•]

Dear [•]

INITIAL OFFER - [USER FACILITY] [REFERENCE]

This is an Initial Offer in relation to your application to [connect to the [insert name] T&S Network]/[modify your existing Delivery Point at the [insert name] T&S Network].

Please note that words and expressions used in this document have the same meaning as the meaning given to them in Section C (Connection) of the CCS Network Code.

Enclosed documents

We enclose the following draft documents with this Initial Offer:

- (a) <u>a draft Construction Agreement, including:</u>
 - (i) <u>a draft Construction Programme; and</u>
 - <u>a draft Commissioning Programme (except where Schedule 6 of the Construction Agreement will apply);</u>
- (b) <u>a draft Connection Agreement; and</u>
- (c) [a draft Code Accession Agreement].

These draft documents have been prepared on the basis of the information available to us, as at the date of this Initial Offer, based on your Connection Application.

Onshore/Offshore Users

Based on your Connection Application, you will be an [Onshore]/[Offshore] User for the purposes of Section H (Charges, Invoicing and Payment) of the CCS Network Code.

[Conditions precedent]

[The work required to [connect your facility]/[modify your Delivery Point] will involve certain [expansion/enhancement] works, which are subject to the approval of the Regulator pursuant to the Change in Scope mechanism under our Licence (referred to in the CCS Network Code as the Change in Scope Approval). Therefore this Initial Offer is subject to that Change in Scope Approval (including any conditions associated with that Change in Scope Approval).]

[As at the date of this Initial Offer, we have not yet been awarded a Licence. Therefore this Initial Offer is subject to us being awarded a Licence in relation to the T&S Network and acceding to the CCS Network Code.]

Acceptance or refusal of Initial Offer

You have one (1) month from the date of this Initial Offer to accept it. If this Initial Offer is not accepted within the aforementioned one (1) month period then, unless we both agree to extend the time for acceptance, it will lapse. If you would like to proceed, please sign the duplicate copy of this Initial Offer and return it to us. We will then proceed to progress your application in accordance with Section C (Connection) of the CCS Network Code.

Yours faithfully	
for and on behalf of	
[insert name of T&SCo]	
for and on behalf of	
[insert name of User]	

Exhibit B - Connection Offer

Date [•]

Dear [•]

CONNECTION OFFER - [USER FACILITY] [REFERENCE]

This is a Connection Offer in relation to your application to [connect to the [insert name] T&S Network]/[modify your existing Delivery Point at the [insert name] T&S Network].

Please note that words and expressions used in this document have the same meaning as the meaning given to them in Section C (Connection) of the CCS Network Code.

Enclosed documents

We enclose final, execution versions of the following documents with this Connection Offer:

- (a) the Construction Agreement (in duplicate), including:
 - <u>(i)</u> <u>the Construction Programme (except where Schedule 6 of the Construction</u>

 <u>Agreement will apply); and</u>
 - (ii) <u>the Commissioning Programme;</u>
- (b) the Connection Agreement (in duplicate); and
- (c) the [Code Accession Agreement (in duplicate)].

[Conditions precedent]

[The work required to [connect your facility]/[modify your Delivery Point] will involve certain [expansion/enhancement] works, which are subject to the approval of the Regulator pursuant to the Change in Scope mechanism under our Licence (referred to in the CCS Network Code as the Change in Scope Approval). Therefore the agreements enclosed with this Connection Offer are subject to that approval (including any conditions associated with that Change in Scope Approval), and a condition precedent addressing this for our benefit has been included in the agreements to that effect.]

[As at the date of this Connection Offer, we have not yet been awarded a Licence. Therefore the agreements enclosed with this Connection Offer are subject to us being awarded a Licence and acceding to the CCS Network Code, and a condition precedent addressing this for the benefit of us both has been included in the agreements to that effect.]

[As notified by us to you, and you to us, pursuant to the provisions of Section C (Connection) of the CCS Network Code, the agreements are also subject to the following conditions precedent, which have been included in the agreements:]

[insert any other conditions precedent]

[It is intended that each party will keep the other updated about the status of these conditions precedent, and the party that has the benefit of the relevant condition precedent can decide to waive it.]

Acceptance or refusal of Connection Offer

You have three (3) months from the date of this Connection Offer to accept it. If you do not accept this Connection Offer within the aforementioned three (3) month period then, unless we both agree to extend the time for acceptance, it will lapse. If you would like to proceed, please sign the agreements enclosed (in duplicate) and return the originals to us. We will then countersign these agreements and one (1) original of each will be returned to you for your retention. The agreements are only effective in accordance with their terms once have been countersigned by us, subject to any conditions precedent.

[insert name of T&SCo]
for and on behalf of
Tours laitifiully

Vours faithfully

CCS Network Code

Section D - Network Structure and Planning

1. Network Classification

Overview of T&S Network

- 1.1 Each T&S Network owned and operated by each T&SCo is made up of the following:
 - (a) the Onshore Transportation System;
 - (b) the Offshore Pipeline Infrastructure; and
 - (c) the Storage ComplexSite.

Note: the relevant T&S Network and Delivery Points etc. will be specifically defined in the relevant Connection Agreement.

2. Network Planning

- 2.1 Subject to paragraph 2.3, for the purposes of assisting the T&SCo in planning the operation and future development of the T&S Network, all Users shallmust, by 1 May of each Charging Year provide to the T&SCo a forecast of:
 - (a) the massquantity of carbon dioxide (expressed in tCO₂) to be delivered to the T&S Network through the User's Delivery Point; and
 - (b) the User's capacity requirements,

for the following twenty (20) twenty Charging Years.

- 2.2 The data provided pursuant to paragraph 2.1 shall be presented must be provided in a form tobe-specified by the T&SCo, including and include a periodic breakdown of period to be
 specified monthly forecast figures. T&SCo must specify the form for such data to be provided
 in by no later than 1 February in each Charging Year.
- 2.3 Subject to paragraph 2.4, any forecast provided by a User to <u>the</u> T&SCo pursuant to paragraph 2.1 shall not be binding on the User and each T&SCo acknowledges that any such forecast shall be used by the T&SCo for planning purposes only.
- 2.4 Users shallmust use reasonable endeavours to provide accurate forecasts when providing information to the T&SCo under this paragraph 2 and all such forecasts must be provided on a bona fide basis.
- 2.5 Users shallmust respond promptly to any queries from the T&SCo regarding such forecasts.
- 3. Interface between Onshore System and Offshore System Forum for sharing industry best practice

Note: if the onshore and offshore components of the T&S Network are, in the future, under separate ownership and control, then the CCS Network Code will need to include some

- provisions dealing with the interface between the two systems and overall operation of the total system. It is not envisaged that this will apply under the initial CCS Network Code.
- 3.1 The T&SCos must develop and share industry best practice to facilitate the effective development and operation of existing and future carbon dioxide transport and storage networks.
- 3.2 To achieve the objective set out in paragraph 3.1, the T&SCos must jointly establish a forum for the sharing of such industry best practice (the "CCS Networks Industry Best Practice Forum"), which must:
 - (a) be open to all existing T&SCos and Users, as well as prospective T&SCos and prospective Users, and the wider carbon capture and storage industry; and
 - meet at least once every twelve (12) months (but with the flexibility to meet more frequently, if it is considered appropriate), with the first meeting to be held within twelve (12) months of the Code Implementation Date.
- 3.3 The arrangements for meetings of the CCS Networks Industry Best Practice Forum shall be as follows:
 - (a) the existing T&SCos will take turns to administer all meetings, arranging locations, rooms and remote communication facilities and publish details of agendas, with at least three (3) months notice to be provided of the date of the meeting to give any interested parties the opportunity to propose items for the agenda;
 - (b) in all cases, the option to join remotely should be provided to prospective attendees;
 - (c) the existing T&SCos will take turns to chair each meeting and take appropriate minutes which will be published on the CCS Network Code Website; and
 - any documents and presentations relevant to the CCS Networks Industry Best Practice Forum will be published on the CCS Network Code Website.
- In convening the CCS Networks Industry Best Practice Forum, the T&SCos must ensure that any discussions do not contravene the Competition Act 1998 or any other applicable competition laws. In particular, the T&SCos must take steps to ensure that a reminder on applicable competition law is read out at the start of every meeting, reminding attendees that they should not disclose to any competitors any competitively sensitive information regarding their respective activities.
- 3.5 Where there is only one (1) T&SCo, references in this paragraph 3 to "the T&SCos" shall be treated as references to the sole T&SCo that is a Party to the Code until such time as another T&SCo accedes to the Code.

CCS Network Code

Section E - Network Use and Capacity

1. <u>Use of T&S</u> Network <u>useand passing of title and risk</u>

Use of Network

1.1 <u>Subject to the provisions of the Code,</u> Users may use the T&S Network by delivering carbon dioxide to the Onshore Transportation System or <u>the Offshore Transportation and Storage System at a Delivery Point for permanent storage in <u>thea</u> Storage <u>ComplexSite</u>.</u>

Note: each User's Connection Agreement will identify (by description or a diagram or both) the Delivery Point to which that User will deliver carbon dioxide.

Title and risk

- 1.2 Title and risk in carbon dioxide delivered to the T&S Network at a Delivery Point in accordance with paragraph 1.1 shall pass to the T&SCo at that Delivery Point, subject to the provisions of Section F (Network Design and Specification) dealing with Non-Compliant CO₂.
- 2. Network Capacity
- 2.1 1.3 Users may apply for and register as holding The capacity in the T&S Network (Network Capacity) at for a particular quantity of carbon dioxide to be delivered at one (1) or more Delivery Points.
 - (a) referred to as "Network Capacity"; and
 - (b) 1.4 Network Capacity is expressed in tCO₂/hour.
- 2.2 For the purposes of allocation of Network Capacity to Users, Network Capacity is allocated for individual Delivery Periods, where a "Delivery Period" is a period of 0.5 hours, with the first Delivery Period of any Day ending at 00:30 hours of that Day and the next Delivery Period of that Day ending at 01:00 hours of that Day.
- 2.3 There are forty eight (48) Delivery Periods in each Day, except on the following Days:
 - (a) the last Sunday of March, when there are forty six (46) Delivery Periods, because of the change to British Summer Time; and
 - (b) the last Sunday of October, when there are fifty (50) Delivery Periods, because of the change to Greenwich Mean Time,

and all provisions of the Code shall apply subject to these exceptions.

2.4 1.5 The Network Capacity that the T&SCo is required to must make available a minimum mass of Network Capacity (to Users in aggregate is equal to the Obligated Network Capacity) in accordance with the terms of its Licence. and subject to the provisions of the Code, being the "Obligated Network Capacity" set out in the T&SCo's Licence (or established pursuant to the Licence), expressed as:

- (a) <u>a maximum instantaneous flow rate;</u>
- (b) <u>a minimum instantaneous flow rate;</u>
- (c) an ONC Maximum Annual Cumulative Flow; and
- (d) a overall store capacity.

2.5 Registered Capacity Users:

- <u>must not deliver any carbon dioxide at a Delivery Point, in any Delivery Period, unless</u>

 <u>they have been allocated and registered as holding Network Capacity at that Delivery Point and for that Delivery Period; and</u>
- (b) may apply to be allocated and registered as holding Network Capacity in relation to a particular Delivery Point and for particular Delivery Periods, in accordance with the provisions of the Code.
- 2.6 1.6-The maximum Network Capacity that can be allocated to a User and registered as being held by that User at a Delivery Point is subject to can be no greater than the physical limits established by the size of the User's connection at the User's Delivery Point, as defined characterised by the Delivery Point Size, and the value of which, for each individual User, is set out in the User's Connection Agreement).
- 2.7 Any Network Capacity that has been allocated and registered to a particular User becomes that User's Registered Capacity, the nature of which is defined in more detail in paragraph 3.
- 3. Nature and scope of Registered Capacity
- 3.1 4.7 A User's "Registered Capacity" is the Network Capacity which:
 - (a) <u>has been allocated to a User in accordance with the Selection Process and provisions</u> of the Code;
 - (a) is the Network Capacity which upon allocation, the User is registered (in accordance with this Section E (Network Use and Capacity)) as holding at a Delivery Point for a particular Delivery Period or Periods; and
 - (c) is held at the quantity and for the term set out in the User's Connection Agreement, as adjusted in accordance with the Code and subject to paragraphs 12 to 14 of Section J (General).

3.2 <u>A User's Registered Capacity:</u>

- (a) (b)-constitutes a User's entitlement (but not an obligation) to deliver carbon dioxide at a Delivery Point measured in in a Delivery Period, subject to the provisions of the Code; and
- (b) <u>is expressed in tCO</u>₂/hour-and, constituting the <u>peak</u> maximum instantaneous flow rate of carbon dioxide.

Scope of Registered Capacity

- 3.3 Consistent with the provisions of paragraph 3.2, each User's peak maximum instantaneous flow rate of carbon dioxide must not exceed an instantaneous flow rate that can be accommodated by the User's Registered Capacity, such that:
 - where a User's peak maximum instantaneous flow rate of carbon dioxide in a particular Delivery Period is expected to be xtCO₂/hour, then
 - (b) the User's Registered Capacity for that Delivery Period must be no less than xtCO₂/hour.
- 3.4 Where any provision of the Code states that the quantity of carbon dioxide (expressed in tCO₂) to be nominated, renominated for, or flowed in a Delivery Period must not exceed the User's Registered Capacity or Available Registered Capacity (as the case may be) for that Delivery Period, this means that the relevant quantity must be capable of being delivered in that Delivery Period at the rate (expressed in tCO₂/hour) corresponding to the User's Registered Capacity or Available Registered Capacity (as the case may be) for that Delivery Period.
- 3.5 4.8 Where an Onshore User holds Registered Capacity, then (unless otherwise specified), that Registered Capacity constitutes Network Capacity in the following parts of the T&S Network:
 - (a) (a) the Onshore Transportation System; and
 - (b) (b) the Offshore Transportation and Storage System.
- 3.6 1.9-Where an Offshore User holds Registered Capacity, then (unless otherwise specified), that Registered Capacity constitutes Network Capacity in the Offshore Transportation and Storage System.

2. Categories of Network Capacity

- 3.7 <u>The Registered Capacity of a User must be recorded:</u>
 - (a) <u>in the User's Connection Agreement, reflecting the quantity and term of Registered</u>
 Capacity held by that User as at the date of the Connection Agreement; and
 - (b) on the T&S Network Portal, reflecting the quantity and term of Registered Capacity
 held by that User as at the date of the Connection Agreement and as updated from
 time to time.

Categories of Network Capacity

In accordance with paragraph 2.2, the Network Capacity which may be allocated to Users and registered as held by Users (as Registered Capacity) is Network Capacity for each Delivery Period of each Day, but such Network Capacity will be allocated as yearly products, for a term of one (1) or more years ("Long-term Network Capacity"), as determined through the Selection Process.

Network Capacity may be allocated to and registered as held by Users as:

- (a) Long-term Network Capacity which may be applied for by a User and become Registered-Capacity for each Hour of each Day in a period of between one (1) Charging Year andfifteen (15) Charging Years; or
- (b) Annual Network Capacity which may be applied for by a User and become Registered-Capacity by a User for each Hour of each Day in a particular Charging Year.

Note: it is expected that shorter-term capacity products and interruptible capacity will be introduced by modification when there is demand.

- 5. 3. Allocation of Network Capacity Initial Users
 - **Initial allocations of Network Capacity**
- 3.1 It is acknowledged that certain each of the Initial Users have been allocated is seeking allocation of Network Capacity outside of the process set out in this Section E, where that allocation has taken place prior to the Code Implementation Date, as described in paragraph 3 of Section C. in accordance with the Track-1 Cluster Sequencing Process (pending entry into the documents referred to in paragraph 3.1(b) of Section C (Connection) and becoming a User).
- 6. Allocation of Network Capacity publication requirements enduring provisions
- 6.1 3.2 Each T&SCo shall must:
 - (a) publish on its T&S Network Portal (once the T&S Network Portal is established in accordance with the Code):
 - (i) information about the amount quantity of Network Capacity available to Userson the T&S Network on the basis of the T&SCo's Obligated Network Capacity and, in particular;
 - (ii) this link to the CCS Network Code Website, as well as any relevant procedures prepared and published pursuant to the provisions of the Code;
 - (iii) any other technical and operating requirements relating to connection to and use of the T&S Network; and
 - (iv) the Use of System Charging Statement for the current Charging Year and for preceding Charging Years; and
 - (v) (iv) a statement advising Users and prospective Users that if they should wish to apply for Network Capacity, they should make that application through the Selection Process; and
 - (b) keep the information referred to in paragraph $\frac{(a)6.1(a)}{(a)}$ updated at all times.

Capacity allocation

- 3.3 Users and prospective Users (<u>"Capacity Applicants"</u>) shall be invited to apply for Network Capacity (or additional Network Capacity) through thea Selection Process.
- 6.3 3.4 The Network Capacity that may be allocated to Capacity Applicants may be through a Selection Process is Long-term Network Capacity or Annual Network Capacity as set out at paragraph 4.1.

Note: in the future Network Capacity will be allocated by T&SCo through different mechanisms, such as set capacity allocation windows. During the early stages of the CCUS industry, the Government has been taking the lead on the allocation of network capacity. The definition of "Selection Process" remains under review and development.

3.5 Where a Capacity Applicant already holds Registered Capacity, the Capacity Applicant can is entitled to apply for additional Network Capacity provided that by acquiring the addition Network Capacity, the Capacity Applicant's total Registered Capacity does not extend for a period of time beyond fifteen (15) Charging Years (not including any Registered Capacity that has already been used up before the application for additional Network Capacity). through a Selection Process.

Recording of Registered Capacity

- 3.6-Where Network Capacity has been allocated to a Capacity Applicant in accordance with the CCUS Cluster Sequencing Process or another Selection Process, then, subject to paragraph 3.76.6, upon the offer and acceptance of that Network Capacity it shall become that Capacity Applicant's Registered Capacity in accordance with the User's Connection Agreement.
- 3.7 Where a Capacity Applicant is a prospective User then then any Network Capacity that has been allocated to and then offered and accepted by the prospective User shall become Registered Capacity upon that prospective User becoming a User, in accordance with the requirements of the Code.
- 7. 3.8 Registered Capacity shall be recorded in: Financial Security

 (a) the T&S Network Portal; and
- 7.1 Any User that:
 - (a) applies for and is allocated Network Capacity of a duration of more than one (1)
 Charging Year; or
 - (b) for will, if allocated additional Network Capacity Applicant's acquiring, hold total Registered Capacity for the first time in relation to their Delivery Point, in their Connection Agreement. of a duration of more than one (1) Charging Year,
 - 4. <u>must provide Registered Capacity</u> Financial Security for the amount determined in accordance with paragraph 7.2 and in the form set out in paragraph 7.3.
- 7.2 Note: Until the Code is modified in the future Users may be required to provide financial security (in addition to the financial security required under Section H) to secure post-termination liabilities for Long-term Network Capacity. However, the level of financial

security that has been set for the time being is in accordance with Section B (Governance), to protect the T&SCo's ability to fund the operation, maintenance and development of the T&S Network, the amount for which the Registered Capacity Financial Security must be provided shall be equal to zero (0).

- 7.3 The Registered Capacity Financial Security may be provided in one (1) or more of the forms set out below:
 - (a) <u>Letter of Credit; and/or</u>
 - (b) Deposit Deed,

(the "Registered Capacity Financial Security").

8. 5. Forecasting procedure

Purpose of forecasts

8.1 5.1 Users will Each User must forecast quantities of carbon dioxide (expressed in tCO₂) for delivery at athat User's Delivery Point for each Day or for each Hour Delivery Period of each Day in accordance with this paragraph 58 for the purposes of enabling the T&SCo to carry out both longer term and shorter term operational planning.

Types of Forecasts

- 8.2 5.2 The Forecasts forecasts that Users are required to make provide to the T&SCo in accordance with this paragraph 8 are as follows:
 - (a) Annual Forecasts;
 - (b) Rolling QuarterlyThree-month Forecasts; and
 - (c) Monthly Forecasts; and
 - (c) (d)-Weekly Forecasts,

collectively referred to as **"Forecasts"**.

- 8.3 5.3 AWhere a User shall has more than one (1) Delivery Point, the User must submit separate Forecasts in respect of each of the User's Delivery PointPoints.
- 8.4 5.4 A User's Forecast must not exceed the lower of:
 - (a) the User's Registered Capacity in relation to the Delivery Point and time period that the Forecast relates to; and
 - (b) the maximum amount quantity of carbon dioxide that can be delivered to that Delivery Point, for that Day or any Hour Delivery Period of that Day, under the prevailing Maintenance Programme or because of some other notified Capacity Constraint (being the User's Available Registered Capacity).

- 8.5 Where a User's Forecast specifies the amount quantity of carbon dioxide being forecast for any Day within the Forecast, that amount must not exceed the User's Registered Capacity totalled over the number of hours in that Day, and the forecast must forecast quantity must:
 - (a) be based on estimates for each Hour Delivery Period within that Day which do not exceed the User's Available Registered Capacity in tCO₂/Hour for that Hour. those Delivery Periods; and
 - (b) not exceed the User's Available Registered Capacity totalled over the number of Delivery Periods in that Day.
- 8.6 She where a User's Forecast specifies the amount quantity of carbon dioxide being forecast for any Hour Delivery Period within the Forecast, that amount forecast quantity must not exceed the User's Available Registered Capacity in tCO₂/Hourhour for that Hour Delivery Period.
- 8.7 5.7 A User shallmust use reasonable endeavours to ensure that:-
- (a) each Forecast constitutes an accurate representation of the User's expected delivery of carbon dioxide during the time periods to which are the subject of the Forecast relates to; and.
 - (b) as the time period between the provision of each relevant Forecast and the Day or Hour ina Day to which the Forecast relates is progressively reduced, the relevant Forecastsalso become progressively more accurate.
- 8.8 A User must provide its:
 - (a) <u>first Annual Forecast (pursuant to paragraph 9) by 31 October in the Charging Year</u>
 prior to the Charging Year in which that User's Scheduled Commencement Date falls;
 - (b) <u>first Rolling Three-month Forecast (pursuant to paragraph 10) in the month before the</u> month within which the User's Scheduled Commencement Date falls; and
 - (c) <u>first Weekly Forecast (pursuant to paragraph 11) in the week before the week within</u> which the User's Scheduled Commencement Date falls.
- 9. Annual Forecasts
- 9.1 5.8 AOn an annual basis, a User shallmust provide an Annual Forecast (as partwhich may use the same data as is used for the purposes of the forecasting process under paragraph 5-of Section H17 of Section H (Charges, invoicing Invoicing and payment Payment)) for each Charging Year t by the 31 October of Charging Year t-1.
- 9.2 <u>5.9 Each Annual Forecast shall provided by a User must specify:</u>
 - (a) the Charging Year that it relates to;
 - (b) the identity of the User;
 - (c) the Delivery Point; and

(d) the quantity of carbon dioxide (expressed in tCO₂) forecast for delivery for each and every Day of that Charging Year-t.,

(the "Annual Forecast").

- 10. Rolling Quarterly Three-month Forecasts
- 5.10-A User shallmust provide a Rolling QuarterlyThree-month Forecast covering a period of three calendar months (quarter)(3) consecutive months starting on the first day of each month, five (5) Business Days before the start of that quarter. those three (3) consecutive months, such that, for example:

5.11 Each Rolling Quarterly Forecast shall specify:

- (a) <u>for</u> the three <u>calendar months that it relates to; (3) consecutive months starting on 1</u>
 <u>January and ending on 31 March, the Rolling Three-month Forecast must be provided</u>
 five (5) Business Days before 1 January; and
- (b) for the three (3) consecutive months starting on 1 February and ending on 30 April, the

 Rolling Three-month Forecast must be provided five (5) Business Days before 1

 February.
- (b) the identity of the User;
- (c) the Delivery Point; and
- (d) the quantity of carbon dioxide forecast for delivery for each Day of that quarter.

Monthly Forecast

- 5.12 A User shall provide a Monthly Forecast for each calendar month five (5) Business Days before the start of that month.
- 10.2 <u>5.13-Each MonthlyRolling Three-month</u> Forecast shall provided by a User must specify:
 - (a) the calendar month three (3) consecutive months that it relates to;
 - (b) the identity of the User;
 - (c) the Delivery Point; and
 - (d) the quantity of carbon dioxide (expressed in tCO₂) forecast for delivery for each and every Day of that calendar month. those three (3) consecutive months,

(the "Rolling Three-month Forecast").

- 11. Weekly Forecast
- 11.1 5.14 A User shallmust provide a Weekly Forecast for each Calendar Week by 12.00 o'clock 12:00 hours on Thursday of the preceding Calendar Week.
- 11.2 5.15 Each Weekly Forecast shall provided by a User must specify:

- (a) the Calendar Week that it relates to;
- (b) the identity of the User;
- (c) the Delivery Point; and
- the quantity of carbon dioxide (expressed in tCO₂) forecast for delivery for each Hourand every Delivery Period of that Calendar Week.

(the "Weekly Forecast").

12. 6. Nominations procedure

Purpose of Nominations

- 12.1 6.1 Users will A User must nominate quantities of carbon dioxide (expressed in tCO₂) for delivery at a Delivery Point for each Hour Delivery Period of each Day in accordance with this paragraph 6the requirements of this Section E (Network Use and Capacity) for the purposes of enabling the T&SCo to:
 - (a) optimise the use of the T&S Network; and
 - (b) operate the T&S Network in a safe and efficient manner.
- The procedures for Nominations and Renominations set out in this Section E (Network Use and Capacity) shall not apply to any deliveries of carbon dioxide made by a User prior to that User's Commencement Date and any such deliveries shall instead be governed by any commissioning procedures agreed under that User's Construction Agreement.
- 13. Daily Nominations
- 13.1 6.2 A User shall For each Day in relation to which a User holds Registered Capacity, the User must provide a Daily Nomination for each Delivery Period in that Day by the Nominations Nomination Close Time, where the.
- 13.2 <u>The "Nomination Close Time"</u> is <u>13.00 hours</u> on the Day before the Day to which the Daily Nomination relates.
- 13.3 A Daily Nomination may be submitted no earlier than seventy two (72) hours before the Day to which the Daily Nomination relates.
- 6.3 At any time before the Nomination Close Time (but not after), a User may change an earlier Daily Nomination already made.submitted, and where a User does this then:
 - the last Daily Nomination that it submitted before the Nomination Close Time shall be treated as the User's Daily Nomination for the purposes of this Section E (Network Use and Capacity); and
 - (b) any earlier Daily Nomination for that Day shall be superseded by that later Daily Nomination.
- 13.5 6.4 Each Daily Nomination shall made by a User must specify:

- (a) the Day that it relates to;
- (b) the identity of the User;
- (c) the Delivery Point; and
- (d) the quantity of carbon dioxide <u>(expressed in tCO₂)</u> nominated for delivery for each Hourand every Delivery Period of the Day (the "Nominated Quantity")-.

(the "Daily Nomination").

14. General principles applying to Daily Nominations and Renominations

- 14.1 6.5 A User's Daily Nomination (or Renomination) for each Delivery Period must be the User's best estimate, made in good faith, of the amountquantity of carbon dioxide (expressed in tCO₂) that the User intends to deliver at the Delivery Point for each Hour of the Day Delivery Period to which the Daily Nomination (or Renomination) relates.
- 14.2 6.6 A User's Nominated Quantity must not exceed:
- (a) the User's <u>Available</u> Registered Capacity for the <u>HourDelivery Periods</u> and Delivery Point to which the Nominated Quantity relates;
 - (b) the maximum amount of carbon dioxide that can be delivered to that Delivery Point, for that Hour, under the prevailing Maintenance Programme or because of some other notified Capacity Constraint.
- 14.3 Where a User is given notice of a Capacity Constraint in circumstances where:
 - (a) the Capacity Constraint impacts the User's ability to deliver carbon dioxide at the
 User's Delivery Point, such that the User's Daily Nomination for any Delivery Period no
 longer complies with paragraph 14.2; and
 - (b) the notice has been given less than one (1) hour before Nomination Close Time,

that Daily Nomination shall be deemed to have been amended (through a reduction in the relevant Nominated Quantity) so as to comply with paragraph 14.2.

15. Failure to nominate

- 45.1 6.7 If in respect of any Delivery Point and each Delivery Period of a Day a User has not submitted a Daily Nomination that complies with the requirements of paragraphs 6.2, 6.4 and 6.6 by the Nomination Close Time then:
 - (a), the User shall be deemed to have failed to make a Daily Nomination; and
- (b) the Nominated Quantity for each Hour of the Day in relation to which the User has failed to submitate a Daily Nomination that Delivery Period (or Delivery Periods) of that Day shall be deemed to be zero (0).

- 16. Confirmation
- 16.1 6.8 The T&SCo shall must confirm to the User by no later than the Confirmation Close Time whether the User's Daily Nomination for each Delivery Period has been accepted or rejected.
- 16.2 The "Confirmation Close Time" is 15:00 hours on the Day before the Day to which the Daily Nomination relates.
- 16.3 Where:
 - (a) 6.9 Where the T&SCo has not notified the User by the Confirmation Close Time whether the User's Daily Nomination has been accepted or rejected, then; or
 - (b) the Nominated Quantities for every Delivery Period in a User's Daily Nomination are zero (0),

the Daily Nomination shall be deemed to have been accepted.

- 6.10 The Confirmation Close Time is 15.00 on the Day before the Day to which the Daily Nomination relates.
- 16.4 6.11 Where the T&SCo has accepted (or is deemed to have accepted) the User's Daily Nomination then for any Delivery Period, each Nominated Quantity included in the Daily Nomination for that Delivery Period shall be the "Confirmed Nominated Quantity".
- 17. Rejection of Daily Nomination
- 17.1 6.12 The T&SCo may reject a User's Daily Nomination for any Delivery Period only where:
 - (a) as at Nomination Close Time, the Daily Nomination does not comply with the requirements of paragraphs 6.2, 6.4 13.1, 13.5 and 6.6 or 14.2; or
 - (b) after Nomination Close Time, a Capacity Constraint affects the ability of the T&SCo to accept the whole or part of any Nominated Quantity, in which case the provisions of paragraph 12.12 of Section E shall apply, for that Delivery Period; or
 - (c) <u>after Nomination Close Time, a Minimum Flow Deficit affects the ability of the T&SCo</u> to accept any Nominated Quantity for that Delivery Period.
- 17.2 6.13-Where the T&SCo rejects has rejected the whole or part of a User's Daily Nomination:
 - (a) because of a Capacity Constraint (which is not an ONC Maximum Annual Cumulative Flow Constraint) pursuant to paragraph 17.1(b) (or contrary to the provisions of the Code) because of a Capacity Constraint pursuant to paragraph 6.12(b); or
 - (b) for some other reason, other than pursuant to paragraph 6.12(a),

then:

(e) where any whole Nominated Quantity included in the User's Daily Nomination has been rejected, then such a rejected Nominated Quantity is deemed to be a "Curtailed Nomination" for the purposes of the Availability Adjustment; and

- (d) where any part of a Nominated Quantity included in the User's Daily Nomination has been rejected and is subject to reduction pursuant to paragraphs 12.12 of this Section E, then, the Curtailed Nomination (for the purposes of the Availability Adjustment) shall be the difference between the Nominated Quantity included in the User's Daily Nomination and the greater of:
 - (i) any Renominated Constrained Quantity (that becomes the Final Confirmed Nominated Quantity).; and

Renominations

- (ii) the actual quantity of carbon dioxide delivered by the User (expressed in tCO₂) in that Delivery Period (recognising that a User may flow more carbon dioxide than it has Renominated).
- Where the T&SCo has rejected (or partially rejected) a User's Daily Nomination pursuant to paragraph 17.1(c), the T&SCo must keep the User updated about the extent to which the Nominated Quantities of the User or other Users would be required to increase to meet the Minimum Flow Rate in any Delivery Period affected by a Minimum Flow Deficit, to enable the User to respond to the Minimum Flow Deficit by increasing its Nominated Quantity or Nominated Quantities through a Renomination.
- Where a User has submitted a Daily Nomination that does not comply with the requirements of paragraphs 13.1, 13.5 and/ or 14.2 and that Daily Nomination has not been rejected by Nomination Close Time for reasons outside of the T&SCo's reasonable control, the T&SCo may reject that Daily Nomination after Nomination Close Time, but must do so as soon as reasonably practicable after identifying that the Daily Nomination does not comply with the relevant requirements.
- 18. **Renominations Standard Response Service**
- 6.14 Following the Confirmation Close Time, a User may vary the Nominated Quantities set out in its Daily Nomination by submitting a <u>renomination in accordance with paragraphs 18.3 and 18.4 (a "Renomination")</u>.
- 18.2 The "Standard Response Service" is the right of a User to be able to Renominate in accordance with this paragraph 18.
- 18.3 (a) A Renomination can vary one (1) or more of the Nominated Quantifies Quantities set out in the User's Daily Nomination, provided that:
 - (a) (b) subject to paragraphs 20 and 21, the Renomination is submitted by the User at least 30 minutes three (3) hours prior to the Hour Delivery Period in relation to which a Nominated Quantity is being amended; and
 - (b) 6.15 any Nominated Quantity included in the Renomination does not exceed the restrictions that applied to the User's Daily Nomination pursuant to clause 6.6 as at Nomination Close Time.:
 - (i) <u>pursuant to paragraph 14.2 as at Nomination Close Time; or</u>

- (ii) <u>subsequently, pursuant to paragraph 43.1.</u>
- 18.4 <u>Each Renomination made by a User must specify:</u>
 - (a) the Delivery Period that it relates to;
 - (b) the identity of the User;
 - (c) the Delivery Point;
 - (d) the revised Nominated Quantity; and
 - (e) whether the User may also be making a Flow Tolerance Request in accordance with paragraph 21 and if so, the reason why the User is making the Flow Tolerance Request.
- 18.5 6.16 Where the T&SCo has received a Renomination from a User in accordance with paragraph 6.14, that Renomination shall be deemed to have been accepted by T&SCo unless T&SCo has notified the User that it has been 18.1, the T&SCo must confirm to the User within thirty (30) minutes whether it has accepted or rejected the Renomination.
- Subject to the User complying with paragraphs 18.3 and 18.4, where the T&SCo has not confirmed to the User whether it has accepted or rejected the User's Renomination, within the time specified in paragraph 18.5, that Renomination shall be deemed to have been accepted.
- 18.7 6.17 Where the T&SCo has accepted (or is deemed to have accepted) the User's Renomination then, each revised Nominated Quantity included in the Renomination shall be the new Confirmed Nominated Quantity.
- 18.8 6.18 A User may make more than one (1) Renomination in relation to the same Day (or part of a Day) provided that the Renomination complies with the requirements of this paragraph-6Section E (Network Use and Capacity).
- 18.9 A User may make a Renomination in relation to a Delivery Period even where:
 - (a) <u>it failed to make a Daily Nomination in relation to that Delivery Period; or</u>
 - (b) the Daily Nomination in relation to that Delivery Period was rejected or partially rejected,

provided that the Renomination complies in all other respects with this Section E (Network Use and Capacity).

- 19. Rejection of Renomination
- 19.1 6.19-T&SCo may reject a User's Renomination enlyfor any Delivery Period where:
 - (a) the Renomination does not comply with the requirements of paragraph

 6.15 paragraphs 18.3 or 18.4; or

- (b) a Capacity Constraint affects the ability of T&SCo to accept the whole or part of any new Nominated Quantity, in which case the provisions of paragraph 12.12 of Section E shall apply. for that Delivery Period; or
- (c) <u>a Minimum Flow Deficit affects the ability of the T&SCo to accept any Nominated</u>

 Quantity for that Delivery Period.
- 19.2 6.20 When Where the T&SCo rejects has rejected a Renomination on the basis of pursuant to paragraph 6.19(a) 19.1(a), the prevailing Daily Nomination or Renomination shall remain in place.
- 19.3 6.21 Where the T&SCo rejects has rejected the whole or part of a Renomination:
 - (a) because of a Capacity Constraint (which is not an ONC Maximum Annual Cumulative
 Flow Constraint) pursuant to paragraph 19.1(b) or contrary to the provisions of the
 Codebecause of a Capacity Constraint pursuant to paragraph 6.19(b); or
 - (b) for some other reason, other than pursuant to paragraph 6.19(a),

then:

- (a) (c) where the whole Renomination has been rejected, then such a rejected Renomination is deemed to be a Curtailed Nomination (for the purposes of the Availability Adjustment); and
- (d) where any part of a Nominated Quantity included in the User's Renomination has been rejected and is subject to reduction pursuant to paragraph 12.12 of this Section E, then, the Curtailed Nomination (for the purposes of the Availability Adjustment) shall be the difference between the Nominated Quantity included in the User's Renomination and any Renominated Constrained Quantity (that becomes the Final Confirmed Nominated Quantity) the greater of:
 - (i) <u>any Renominated Constrained Quantity; and</u>
 - (ii) the actual quantity of carbon dioxide delivered by the User (expressed in tCO₂) in that Delivery Period (recognising that a User may flow more carbon dioxide than it has Renominated).
- Where the T&SCo has rejected a Renomination for a Delivery Period pursuant to paragraph 19.1(c), the T&SCo must keep the User updated about the extent to which the Nominated Quantities of the User or other Users must increase to meet the Minimum Flow Rate in any Delivery Period affected by a Minimum Flow Deficit, to enable the User to respond to the Minimum Flow Deficit by increasing its Nominated Quantity through a Renomination.
- 20. Renominations Enhanced Response Service
- 20.1 Each User may request that the T&SCo provide the Enhanced Response Service in relation to any one (1) or more Delivery Periods, in accordance with this paragraph 20.

- 20.2 The "Enhanced Response Service" is the right for a User to be able to Renominate by submitting a Renomination up to thirty (30) minutes before the start of a Delivery Period, instead of the time referred to in paragraph 18.3(a), in accordance with this paragraph 20.
- 20.3 Each individual request for the Enhanced Response Service can only relate to one (1) or more Delivery Periods in a particular Day. If the Enhanced Response Service is required in relation to one (1) or more Delivery Periods in different Days, these must be the subject of separate requests for each Day.
- 20.4 Where a User wishes to request the Enhanced Response Service, the User must:
 - (a) only request the Enhanced Response Service where it is reasonably required in order for the User to be able to operate its User Facility in a way that a Reasonable and Prudent Operator would;
 - (b) notify the T&SCo of the reason why the User is requesting the Enhanced Response

 Service (to allow the T&SCo to keep a record of this for the purposes of paragraphs

 36 and 48);
 - (c) notify the T&SCo if the User may also be making a Flow Tolerance Request in accordance with paragraph 21 and if so, the reason why the User is making the Flow Tolerance Request (with this notification requirement applying at this point in time, instead of pursuant to paragraph 18.4(e));
 - (d) without limiting the generality of paragraphs 20.4(b) and 20.4(c), notify the T&SCo if the reason the User is seeking a greater degree of responsiveness through the Enhanced Response Service or Flow Tolerance Request is because the User is participating in the Balancing Mechanism;
 - (e) specify in relation to which Delivery Periods the Enhanced Response Service is being requested:
 - <u>specify the maximum Nominated Quantity up to which the User may wish to Renominate (the "Enhanced Response Service Cap") when utilising the Enhanced Response Service in relation to any Delivery Period (within the Day to which the request relates);</u>
 - (g) make the request:
 - no earlier than twenty four (24) hours before the Nomination Close Time for the
 Day of the Delivery Periods in relation to which the Enhanced Response
 Service is being requested; and
 - (ii) no later than three (3) hours before the Delivery Period in relation to which the Enhanced Response Service is being requested; and
 - (h) notify the T&SCo as soon as practicable if the User no longer needs the Enhanced Response Service in relation to any Delivery Period, having previously requested it, including where the requirement in paragraph 20.4(a) is no longer satisfied.

- 20.5 Where the Enhanced Response Service is being provided to a User in relation to a Delivery Period:
 - (a) the time period for making a Renomination for the Delivery Period, pursuant to paragraph 18.3(a), is reduced to thirty (30) minutes;
 - (b) the T&SCo can only reject that Renomination in accordance with paragraph 19; and
 - (c) the time period for confirmation by the T&SCo whether a Renomination is accepted or rejected, pursuant to paragraph 18.5, is fifteen (15) minutes, including for the purpose of paragraph 18.6,

to the extent that the amended Nominated Quantity included in any such Renomination does not exceed the Enhanced Response Service Cap for that Delivery Period.

20.6 Save as expressly provided in this paragraph 20, paragraphs 18 and 19 apply to a Renomination under this paragraph 20.

21. Flow Tolerance Requests

- 21.1 In addition to the T&SCo's obligation to accept Renominations in accordance with the

 Standard Response Service pursuant to paragraph 18 and the Enhanced Response Service

 pursuant to paragraph 20, the T&SCo must use reasonable endeavours to accept requests

 from Users to vary the quantity of carbon dioxide to be delivered (expressed in tCO₂) in a

 Delivery Period on shorter notice (as set out in paragraph 21.2) before the Delivery Period to

 which the request relates ("Flow Tolerance Requests"), to accommodate:
 - (a) <u>the requirements of Users who, for commercial or operational reasons, need to:</u>
 - <u>(i)</u> <u>start or increase their operations (resulting in an increased quantity of carbon dioxide) at short notice; or </u>
 - <u>stop or decrease their operations (resulting in a decreased quantity of carbon dioxide) at short notice;</u>
 - (b) the requirements of any Users who, in accordance with paragraph 20.4(d) have notified the T&SCo of an intention to participate in the Balancing Mechanism; or
 - <u>circumstances where a Renomination has been rejected because of a Minimum Flow Deficit and the User wishes to respond to that Minimum Flow Deficit by increasing its Nominated Quantity.</u>
- 21.2 The shorter notice period referred to in paragraph 21.1 is:
 - (a) less than three (3) hours, in the case of Users who are Renominating pursuant to the Standard Response Service; and
 - (b) less than thirty (30) minutes, in the case of Users who are Renominating pursuant to the Enhanced Response Service.

- 21.3 When exercising reasonable endeavours to accommodate Flow Tolerance Requests pursuant to paragraph 21.1, the T&SCo:
 - (a) must respond to the initial request within thirty (30) minutes;
 - (b) <u>must keep the User updated; and</u>
 - (c) may reject but then subsequently accept a Flow Tolerance Request (and vice versa), or partially accept or reject it.
- Where a User has requested the Enhanced Response Service pursuant to paragraph 20, but in making a Renomination for a Delivery Period the User is seeking to increase the Nominated Quantity beyond the Enhanced Response Service Cap for that Delivery Period, then the T&SCo's obligation to accommodate the increase beyond the Enhanced Response Service Cap is the same as if it was a Flow Tolerance Request.
- 21.5 For the purposes of assisting Users to assess whether their Flow Tolerance Request is likely to be accommodated by the T&SCo, the T&SCo must publish on the T&S Network Portal information about the quantity of residual Network Capacity that is available in the current and next six (6) Delivery Periods, where residual Network Capacity means, in relation to that time period:
 - (a) the Network Capacity available under the operating conditions deployed to accommodate the flows of carbon dioxide expected by the T&SCo; less
 - (b) the quantity of Network Capacity already committed to be used, based on any existing
 Final Confirmed Nominated Quantities and any Nominated Quantities that the T&SCo
 considers are likely to be increased through the Enhanced Response Service.
- 21.6 Subject to the provisions of this paragraph 21, all other provisions applying to Renominations shall also apply to Flow Tolerance Requests.
- 21.7 Where the T&SCo rejects (or partially rejects) any Flow Tolerance Request (whether or not it was first accepted), this shall not be considered to be a Curtailed Nomination for the purposes of the Availability Adjustment.

22. Minimum Flow Deficits

- 22.1 Notwithstanding any other provision of this Section E (Network Use and Capacity), the T&SCo may:
 - (a) reject a Daily Nomination or Renomination in relation to any Delivery Period; or
 - (b) where the User is already delivering carbon dioxide during a Delivery Period, request that the User ceases to deliver carbon dioxide within the time period specified by the T&SCo.

where the aggregate of Nominated Quantities or actual flow of carbon dioxide (as the case may be) of all Users for that Delivery Period is below the relevant Minimum Flow Rate for all

or part of the T&S Network, and the occurrence of such an event is referred to as a "Minimum Flow Deficit".

- <u>Where a Minimum Flow Deficit occurs and the T&SCo is required to take the action referred to in paragraph 22.1, the T&SCo must:</u>
 - (a) notify all Users of the occurrence of the Minimum Flow Deficit as soon as the T&SCo becomes aware of it, together with details of when and for how long the Minimum Flow Deficit is likely to continue; and
 - (b) use reasonable endeavours to:
 - <u>(i)</u> <u>mitigate the impact on Users, by declaring a Minimum Flow Deficit only where</u> necessary; and
 - (ii) notify Users as soon as it becomes apparent that a Minimum Flow Deficit is no longer impacting a Delivery Period.
- Where a Minimum Flow Deficit occurs, Users must keep submitting Daily Nominations and Renominations in accordance with this Section E (Network Use and Capacity) to allow the T&SCo to determine when a Minimum Flow Deficit will no longer be impacting a Delivery Period.
- Where a Minimum Flow Deficit has been caused by a Capacity Constraint (that is, where the Minimum Flow Deficit results from the fact that some Users are not flowing or nominating quantities of carbon dioxide because an insufficient quantity of Registered Capacity is available to them), the Minimum Flow Deficit shall be treated as a Capacity Constraint for the purposes of calculating the Curtailed Nomination (for the purposes of the Availability Adjustment) under paragraph 17.2 or 19.3 (as applicable).
- 23. **Maximum Annual Cumulative Flow Constraints**
- 23.1 Acting as a Reasonable and Prudent Operator, the T&SCo must at all times comply with the provisions of any Storage Permit issued in relation to any Storage Site that forms part of the T&S Network.
- Without limiting the generality of paragraph 23.1, the T&SCo's obligation to accept delivery of carbon dioxide from Users in accordance with the provisions of this Section E (Network Use and Capacity) are subject to the T&SCo's obligation not to exceed the Maximum Annual Cumulative Flow that applies to any Storage Site.
- 23.3 The T&SCo must publish details of each Maximum Annual Cumulative Flow that applies to each individual Storage Site, as may be updated from time to time in response to any requirements of the NSTA, on its T&S Network Portal.
- 23.4 The T&SCo must, on an ongoing basis, monitor if and when it is likely that it may exceed the

 Maximum Annual Cumulative Flow for any individual Storage Site, using the existing and
 estimated data and information available to it, including:

- (a) the quantity of carbon dioxide already delivered into the T&S Network, as measured using Flow Meters in accordance with Section F (Network Design and Specification);
- (b) <u>any periods of Programmed Maintenance of the T&S Network;</u>
- (c) the Registered Capacity of Users; and
- (d) the Forecasts provided by Users pursuant to this Section E (Network Use and Capacity).
- 23.5 The T&SCo must, on a monthly basis, publish information on the T&S Network Portal, based on the monitoring activities carried out in accordance with paragraph 23.4, about how likely it is that the T&SCo may need to take action pursuant to paragraph 23.6.
- 23.6 If and when it becomes apparent to the T&SCo, based on the monitoring referred to in paragraph 23.4, and acting as a Reasonable and Prudent Operator, that it will exceed the Maximum Annual Cumulative Flow (or one (1) or more Maximum Annual Cumulative Flows, where the T&S Network includes more than one (1) Storage Site) unless it takes action to reduce the quantity of carbon dioxide being delivered into the T&S Network (such event being referred to as a "Maximum Annual Cumulative Flow Constraint"), the T&SCo:
 - (a) must publish on the T&S Network Portal details of how it has determined that the Maximum Annual Cumulative Flow will be exceeded; and
 - (b) has the right to reduce the quantity of Registered Capacity available to Users in accordance with the Maximum Annual Cumulative Flow Mitigation Measures to reduce the risk of the Maximum Annual Cumulative Flow being exceeded.
- 23.7 The "Maximum Annual Cumulative Flow Mitigation Measures" are as follows:
 - (a) the T&SCo must give Users as much notice as reasonably possible of its intention to reduce the quantity of Registered Capacity available to Users because of a Maximum Annual Cumulative Flow Constraint, including details of the Days and Delivery Periods being impacted;
 - (b) in reducing the Registered Capacity available to be used by Users, the T&SCo and
 Users shall discuss, in good faith, how to apply the principles set out in paragraph 38
 that apply to other Capacity Constraints, recognising that a Maximum Annual
 Cumulative Flow Constraint can be foreseen with greater notice than other Capacity
 Constraints and may impact Users over a longer period of time; and
 - (c) the T&SCo must continue to carry out monitoring activities in accordance with paragraph 23.4 and restore the Registered Capacity available to each User to its full level as soon as reasonably possible.
- Where the T&S Network includes more than one (1) Storage Site, the T&SCo, acting as a

 Reasonable and Prudent Operator, must use reasonable endeavours to allocate the injection of carbon dioxide between its different Storage Sites to avoid a Maximum Annual Cumulative Flow Constraint arising or, where this is not possible, to reduce the impact on Users.

- 24. Delivery of carbon dioxide in accordance with nomination
- 24.1
 6.22-A User's final Confirmed Nominated Quantity, as varied in accordance with any Renomination, shall be the (including any Renomination made, or deemed to be made, in accordance with the requirements of paragraph 43.1), shall be the "Final Confirmed Nominated Quantity".
- 24.2 A User must use reasonable endeavours to ensure that for every Delivery Period:
 - (a) 6.23 A User shall use reasonably endeavours to ensure that the amount the quantity of carbon dioxide (expressed in tCO₂) it delivers at a Delivery Point is equal to the Final Confirmed Nominated Quantity. (as may be adjusted through a Flow Tolerance Request); and
 - within the Delivery Period, the rate of flow is not subject to any decreases or increases except where any material variations in the rate of flow are due to unavoidable operational processes which have been notified by the User to the T&SCo in advance, including any ramping up or ramping down of the User Facility.
- Where the T&SCo has accepted the whole or part of a Daily Nomination or Renomination but

 (subject to paragraphs 4 and 12 of Section F (Network Design and Specification))

 subsequently constrains the delivery of part of the relevant Final Confirmed Nominated

 Quantity during the relevant Delivery Period, the Curtailed Nomination for the purposes of the Availability Adjustment shall be the difference between:
 - (a) the lower of:
 - (i) the User's pro-rated pre-constraint flow of carbon dioxide in the Delivery Period
 (defined as the number of tonnes of carbon dioxide delivered by the User
 before it was notified of a constraint, pro-rated for the full Delivery Period); and
 - (ii) the Final Confirmed Nominated Quantity plus the Over Variation Limit of the Final Confirmed Nominated Quantity; and
 - (b) the User's actual flow of carbon dioxide (being the number of tonnes of carbon dioxide flowed) in that Delivery Period.
- 25. Non-conformity with Final Confirmed Nominated Quantity
- 6.24 Where a User has Materially or Persistently failed to comply with the requirements of paragraph 6.23 then:
 - (a) T&SCo may service a notice on the User (Nominations Non-conformity Notice) which:
- Where, notwithstanding the obligation on the User pursuant to paragraph 24.2(a), a User is delivering a quantity of carbon dioxide that is not equal to its Final Confirmed Nominated Quantity (as adjusted through any Flow Tolerance Request) for a Delivery Period, subject to paragraph 25.3, the User must use reasonable endeavours to ensure that the quantity of carbon dioxide being delivered:

- (a) <u>is not lower than the Final Confirmed Nominated Quantity by more than the Under Variation Limit; and</u>
- (b) <u>is not higher than the Final Confirmed Nominated Quantity by more than the Over Variation Limit.</u>
- Where a User is delivering a quantity of carbon dioxide in a Delivery Period that is outside of the bounds set out in paragraph 25.1 (including at a rate that is higher or lower than the instantaneous flow rate that corresponds to those bounds) (referred to as an "Exceedance Event"):
 - (a) the User must notify the T&SCo of the Exceedance Event as soon as reasonably practicable, together with the reasons for such Exceedance Event, and use reasonable endeavours to adjust its delivery of carbon dioxide so as to be within those limits; and
 - (b) the T&SCo must use reasonable endeavours to accommodate the User's delivery of carbon dioxide, except where the User has not complied with paragraph 25.3.
- 25.3 The flow of carbon dioxide being delivered by a User during a Delivery Period must not exceed the User's Available Registered Capacity for that Delivery Period.

25.4 Where:

- (a) a User falls within the scope of paragraph 25.2 and, notwithstanding the T&SCo's obligation under paragraph 25.2(b), the T&SCo considers that accommodating the User's flow of carbon dioxide will have an adverse effect on the safe and reliable operation of the T&S Network and/or result in a Capacity Constraint affecting other Users; or
- (i) requires the User to desist from failing to comply with the requirements is in breach of paragraph 6.23; and 25.3,
 - (ii) includes details of how the User has Materially or Persistently failed to comply with the requirements of paragraph 6.23;
- (b) the User shall be permitted a "grace period" of 14 Days to implement measures to assist the User in complying with paragraph 6.23;
- (c) for the avoidance of doubt, any non-compliance with paragraph 6.23 by the User during the grace period referred to in paragraph (b) shall not entitle T&SCo to serve another Nominations Non-conformity Notice on the User.
- 6.25 Where a User has Materially or Persistently failed to comply with the requirements of paragraph-6.23 having received two Nominations Non-conformity Notices within the preceding 12months, then:
 - (a) T&SCo may service a notice on the User (Non-Delivery Notice) which:

- (i) requires the T&SCo can require the User to cease delivering reduce or stop its flow of carbon dioxide at the Delivery Point within 24 Hours of receipt of the Non-Delivery Notice; and to align with its Final Confirmed Nominated Quantity.
- 25.5 In deciding what action to take pursuant to this paragraph 25, the T&SCo must actively communicate with the User.
- 25.6 Other than where the Final Confirmed Nominated Quantity is zero (0) (in which case paragraph 25.7 applies), where a User falls within the scope of paragraph 25.2 but:
 - (ii) includes details of how the User has Materially or Persistently failed to comply with the requirements of paragraph 6.23;
 - (b) the User shall not be entitled to delivery any carbon dioxide at the Delivery Point until it has reasonably satisfied T&SCo that it will cease to Materially or Persistently breach paragraph 6.23; and
 - (c) where the User is unable to remedy the breach as contemplated under paragraph (b) within 14 Days, then T&SCo may issue a Default Notice to the User in accordance with Section J.

6.26 For the purposes of paragraphs 6.24 and 6.25:

Materially means a variation of more than [5%] over the course of a Day, calculated by:

- (a) allocating a percentage figure (the "Percentage Variation") for each Hour in the Day (other than Zero Hours), which percentage figure represents the absolute variation (i.e. whether over or under) by which the User has deviated from its Final Confirmed Nominated Quantity; and
- (b) adding up all the Percentage Variations for that Day and dividing them by the number of Hours in the Day (other than Zero Hours),

where:

- (a) (c) Zero Hour means any Hour in that Day in relation to which the User's Final Confirmed Nominated Quantity is zero and in which the User has not delivered any carbon dioxide for a Delivery Period was less than fifteen per cent (15%) of the User's Registered Capacity for that Delivery Period; and
- (b) the quantity of carbon dioxide being delivered in that Delivery Period is equivalent to less than fifteen per cent (15%) of the User's Registered Capacity for that Delivery Period.

for the purposes of the T&SCo's record keeping obligations under paragraph 36, this event shall be recorded as a "Low Nomination and Flow Exceedance Event".

(d) Percentage Variation for an Hour is calculated according to this formula:

$$\frac{\text{(Del)} Q - Actual Amount}{FCNQ} \times 100$$

where:

- 25.7 (i) FCNQ is the Where a Final Confirmed Nominated Quantity for that Hour (expressed in tCO₂); was zero (0) for a Delivery Period, but the User is delivering carbon dioxide in that Delivery Period:
 - (ii) **Actual Amount** is the actual amount of carbon dioxide delivered in that Hour-(expressed in tCO₂); and
 - (iii) the numerator in the formula is an absolute value.

Note: where both the FCNQ and the Actual Amount have a relatively low value, equally small variations could result in a high Percentage Variation. This may need further consideration. In addition, further thinking will be required in relation to situations where the FCNQ is zero and the Actual Amount is not.

Persistently means a variation of more than [3%] for any Day (calculated using the methodology equivalent to that referenced in the definition of Materially), where that variation happens more than three times or more in any period of 14 Days.

6.27 The User shall be required to continue paying any Capacity Charges or Network Charges relating to the time period during which it has been required to desist from delivering carbon dioxide pursuant to paragraph 6.25.

Impact of Capacity Constraints on User ability to conform with Final Confirmed Nominated Quantity

- (a) where the quantity of carbon dioxide being delivered in that Delivery Period is equivalent to more than five per cent (5%) of the User's Registered Capacity for that Delivery Period:
 - (i) this is deemed to be an Exceedance Event for the purposes of this paragraph 25; and
 - (ii) for the purposes of the T&SCo's record keeping obligations under paragraph
 36, it shall be recorded as a "Zero Nomination Exceedance Event"; and
- (b) where the quantity of carbon dioxide being delivered in that Delivery Period is equivalent to five per cent (5%) or less of the User's Registered Capacity for that Delivery Period:
 - <u>(i)</u> <u>this is deemed to not be an Exceedance Event for the purposes of this paragraph 25; and</u>
 - (ii) for the purposes of the T&SCo's record keeping obligations under paragraph
 36, it shall be recorded as a "Zero Nomination Low Flow Event".
- Without limiting any other rights of the T&SCo under other Sections of the Code or any Legal Requirement, the T&SCo, acting as a Reasonable and Prudent Operator, may take action to stop the flow of a User's carbon dioxide into the T&S Network where the T&SCo has notified the User of any requirement to reduce or stop its flow pursuant to paragraph 25.4 and the

<u>User has failed to respond to such a requirement within a reasonable time having regard to</u> all the circumstances.

- 25.9 6.28 Where a User is unable to complythe T&SCo has taken action in accordance with paragraph 6.23 as a result of 25.8:
 - (a) the circumstances set out in paragraph 12.13 (relating to where a User has insufficient notice of its Reduced Capacity); or
 - (b) other circumstances where T&SCo has issued a direction to the User at short notice, requiring the User to change the amount of carbon dioxide it is delivering.

then such deviation from the requirements of paragraph 6.23 shall not be taken intoconsideration when applying the "Materially" and "Persistently" triggers set out in paragraph 6.26.

- (a) the T&SCo must restore the User's ability to flow carbon dioxide as soon as is reasonably practicable; and
- (b) the T&SCo and the relevant User must meet within one (1) month to discuss and prepare a joint report which sets out the facts that led to the T&SCo exercising its right under paragraph 25.8, actions taken by the T&SCo, actions taken in response by the User and a detailed summary of lessons learned and the steps to be taken in the future to avoid the T&SCo being required to act pursuant to paragraph 25.8.
- 25.10 Once the T&SCo and the relevant User have finalised the joint report pursuant to paragraph 25.9, the T&SCo must:
 - (a) send it to the Regulator; and
 - (b) publish it on the T&S Network Portal.
- 25.11 This paragraph 25 sets out the rights and remedies available to the T&SCo where a User is not flowing carbon dioxide in conformity with the requirement to flow carbon dioxide in any Delivery Period in conformity with the Final Confirmed Nominated Quantity for that Delivery Period. The T&SCo shall not be entitled to issue a Warning Notice or a Default Notice to a User in accordance with Section J (General) where a User is not in compliance with paragraph 24 or paragraph 25 (other than where a User is not in compliance with paragraph 25.4).

26. 7. Communication of Forecasts and Nominations

Users shall make all Forecasts and Nominations through the T&S Network Portal.

Subject to paragraph 2 of Section J (General) and any requirements for real-time "operator to operator" communications, unless otherwise specified, all other communications and notifications between a User and a T&SCo, including all Forecasts and Nominations, requests for the Enhanced Response Service and Flow Tolerance Requests, shall be made through the T&S Network Portal.

27. 8. Surrender of Registered Capacity

- 27.1 8.1 Users may offer to surrender Registered Capacity:
 - (a) for a Charging Year; or
 - (b) for the remainder of the period of the User's Registered <u>Capacity where that Registered Capacity is Long-term Network Capacity.</u>
- 27.2 8.2-An offer (a "Capacity Offer") to surrender Registered Capacity pursuant to paragraph 8.1-shall 27.1 must specify:
 - (a) the identity of the User;
 - (b) the Delivery Point;
 - (c) the time period for which the Registered Capacity is offered for surrender;
 - (d) the amount quantity of Registered Capacity offered for surrender; and
 - (e) the minimum amount quantity of Registered Capacity for which the User is willing to have the offer accepted.
- 27.3 8.3 A User canmay make a Capacity Offer at any time, but that Capacity Offer shall only be:
 - (a) considered as part of Network Capacity allocation during the <u>first_next</u> Selection
 Process relating to that T&S Network that follows the making of the Capacity Offer by the User; and
 - (b) accepted where such Registered Capacity can be re-allocated to another User or prospective User.
- 8.4-A User shall remain liable for Capacity Charges for any surrendered Registered Capacity until such capacity Registered Capacity is transferred and becomes registered to another User.
- 28. 9. Maintenance Planning general principles

Decisions on maintenance planning

- 28.1 9.1 The T&SCo willmust plan the maintenance of the T&S Network and prepare Maintenance Programmes in good faith with a view to, acting as a Reasonable and Prudent Operator, in order that, so far as is reasonably practicable and having regard to:
 - (a) the nature and urgency of the requirements for such maintenance;
 - (b) any Legal Requirements applying to the operation of the T&S Network; and
 - (c) the T&SCo's normal working practices,

coordinatingit coordinates the timing of such maintenance to maximise the quantity of carbon dioxide, produced by Users, that can be transported and stored.

Maintenance

- 28.2 9.2 References in the Code to maintenance (of a T&S Network or any part of it) include:
 - (a) maintenance, inspection, repair, replacement, reinstatement and recommissioning of a T&S Network or such part of it;
 - (b) Network Expansion Works; and
 - (c) any inspection, testing and commissioning of works within paragraphs (a)28.2(a) and (b)28.2(b), and works preparatory thereto, and any works required for bringing any new or existing part of the T&S Network into or back into service.

29. Users' planning and forecasting information

- 29.1 9.3 In each Charging Year, each User shallmust provide to the T&SCo, not later than 1 May of Charging Year t-1, for the purposes of the Maintenance Programme, an estimate in respect of each week in the Planning Period commencing in Charging Year t, of the following:
 - the quantities of carbon dioxide (expressed in tCO₂) expected to be delivered to the
 T&S Network on a Day (in each such week) at each Delivery Point; and
 - (b) details of the times and periods for which maintenance of relevant User Facilities is planned during the relevant Planning Period which may affect the delivery of carbon dioxide to the Delivery Point; and
 - (c) such further information (if any) as the T&SCo may reasonably require for the purposes of this Section E (Network Use and Capacity).
- 29.2 A User must provide the information referred to in paragraph 29.1 by the earlier of 1 May or six (6) months before that User's Scheduled Commencement Date.
- 29.3 9.4 A User must promptly, as soon as reasonably practicable upon becoming aware of any requirement to perform maintenance of relevant-User Facilities (where such maintenance will have an impact on the User's delivery of carbon dioxide to the T&S Network), notify the T&SCo of such maintenance, together with details of the times and periods of such maintenance and the anticipated impact on its delivery of carbon dioxide to the T&S Network, where such requirement was not known and therefore not notified by the User at the time that the information was provided by the User under paragraph 9.329.1.
- 29.4 User shallmust use reasonable endeavours to co-ordinate the maintenance of relevant User Facilities with any Programmed Maintenance of the T&S Network which may affect the delivery of carbon dioxide teat the Delivery Point in order to maximise the quantity of carbon dioxide, produced by Users, that can be transported and stored.

10. Maintenance Programme

- 30. Scope of T&SCoSCo's Maintenance Programme
- 30.1 10.1 Subject to paragraph 10.230.2, the T&SCo shallmust prepare and publish on its T&S Network Portal an updated Maintenance Programme for the Planning Period ahead, by 16

- October of the Charging Year immediately before the start of the Planning Period (the "Relevant Year").
- 30.2 40.2 Where a T&SCo has acceded to the Code and has not yet commenced commercial operation of the reached the Commercial Operations Date for its T&S Network, it shall must publish its first (1st) Maintenance Programme within 24 weeks from the date on which it acceded to the Code. by the date which is twenty four (24) weeks before the Scheduled Commercial Operations Date of its T&S Network.
- 30.3 The Maintenance Programme referred to in paragraph 40.130.1 is intended to always cover the Planning Period ahead, but the information published for that Planning Period is to be updated when the next annual Maintenance Programme is published, such that, for example:
 - (a) a Maintenance Programme is published for Charging Years t, t+1, t+2, t+3 and t+4; and
 - (b) when the next Maintenance Programme is published for Charging Years t+1 to t+5, the information originally published for Charging Years t+1 to t+4 will be updated in accordance with the requirements of this Section E (Network Use and Capacity).

Content of Maintenance Programme

- 30.4 10.4 A Maintenance Programme will must identify:
 - the Delivery Points at which the ability of the T&SCo to accept delivery of carbon dioxide will be constrained by planned maintenance Programmed Maintenance of the T&S Network;
 - (b) the periods for which such Delivery Points will be so affected;
 - (c) whether the T&SCo expects that it will continue within such period (or part thereof) to be able to accept delivery of carbon dioxide at any such Delivery Point, but (by reason of such maintenance) on a restricted basis, and if so, an indicative estimate of the maximum rate at which the T&SCo expects to be able to accept delivery of carbon dioxide:
 - (d) relevant maintenance activities to be performed by Users in accordance with information provided by Users under paragraph 9.329.1; and
 - (e) such other information as the T&SCo shall reasonably decide to include.

31. Publication of draft Maintenance Programme

- 31.1 40.5 Before publishing the Maintenance Programme in accordance with paragraph 40.130.1, the T&SCo shallmust:
 - (a) publish on its T&S Network Portal a draft of the Maintenance Programme by 29 May of the Relevant Year; and

- (b) convene a meeting (an "Annual Maintenance Meeting") for review of the timing of planned maintenance of the T&S Network under such draft Maintenance Programme, the date of which:
 - (i) shallmust be no later than 18 September of the Relevant Year (to allow the T&SCo to first consider information provider provided by Users pursuant to paragraph 10.631.2); and
 - (ii) is communicated to Users with not less than eight (8) weeks' notice.
- 31.2 10.6 Users may, by 24 July of the Relevant Year, submit to the T&SCo details of the dates and periods of maintenance planned in respect of any relevant User Facilities and/or comments in relation to the timing of any planned maintenance of the User Facilities under the draft Maintenance Programme.
- 31.3 10.7 Before holding the Annual Maintenance Meeting(s) the T&SCo will must discuss with each User, to such extent as it deems appropriate, the details and comments submitted to it by that User under paragraph 10.631.2.
- 31.4 10.8-Subject to paragraphs 9.128.1 and 10.932.1, Maintenance Programme decisions shall be taken by T&SCo in its sole discretion, after liaising with Users in accordance with paragraphs 10.5, 10.6-31.1, 31.2 and 10.731.3.
- 32. Co-ordination of maintenance between different T&SCos
- 32.1 10.9 In preparing the their draft Maintenance Programme, the T&SCos shall of different T&S

 Networks must consult each other with a view, to minimising seeking to minimise any adverse impacts on the T&SCos and Users (collectively) of maintenance activities.
- 33. Carrying out maintenance
- 33.1 10.10 Subject to paragraphs 10.1133.2 and 10.1233.3, the T&SCo willmust, so far as is reasonably practicable, carry out all maintenance of the T&S Network in accordance with the dates and periods provided for in the applicable Maintenance Programme, as updated on an annual basis.
- 33.2 10.11 If the T&SCo identifies the need to carry out Reactive Maintenance then it shallmust:
 - (a) notify any Affected Users as soon as it identifies the need to carry out the Reactive Maintenance; and
 - (b) schedule the carrying out of the Reactive Maintenance with a view to giving the Affected Users as much notice as is reasonably possible in the circumstances.
- 33.3 40.12 The notice given by the T&SCo to Affected Users in accordance with paragraph 40.11 shall 33.2 must include the following information:
 - (a) the date on which the Reactive Maintenance is to commence and its likely duration; and
 - (b) the impact of the Reactive Maintenance on the Affected Users.

34. Changes to Programmed Maintenance

- 34.1 10.13-If the T&SCo becomes aware that the commencement date or duration of any Programmed Maintenance needs to be changed, outside of the annual review of the Maintenance Programme, thenthe T&SCo shall must:
 - (a) notify the Affected Users as soon as is reasonably practicable that the Programmed

 Maintenance is no longer going ahead at the previously scheduled time/date; and
 - (a) give the Affected Users not less than twenty (20) Business Days' notice (of the new time and date for the carrying out of the Programmed Maintenance, or such lesser period of notice as the Affected Users may agree) of the changes to the Programmed Maintenance.
- 34.2 10.14 The T&SCo shallmust use reasonable endeavours to avoid making inany changes to Programmed Maintenance under paragraph 10.1334.1.

Notification requirements

- 34.3 10.15 The T&SCo willmust notify Affected Users, by way of reminder, of the carrying out of Programmed Maintenance (as revised under paragraph 10.1334.1) not less than five (5) Business Days before the commencement of such maintenance, provided that no accidental failure to notify a User shall prejudice the T&SCo's rights to carry out maintenance.
- 34.4 10.16 The T&SCo willmust comply with its notification and reporting obligations to ensure that Affected Users have ongoing visibility of updates about the progress of any activities under the Maintenance Programme and when any Capacity Constraint arising from maintenance will conclude.
- 35.1 41.1 A "Capacity Constraint" is a constraint in or affecting any part of the T&S Network at any time, as a result of which the capacity to accept delivery of carbon dioxide flows in any part of the T&S Network are is less than the affected Affected Users' aggregate Registered Capacity.
- 41.2 A Delivery Point is "affected" by a Capacity Constraint where the T&SCo is unable to accept delivery of the Registered Capacitya quantity of carbon dioxide (expressed in tCO₂) at that Delivery Point, or its in any Delivery Period that is equal to a User's Registered Capacity at that Delivery Point for that Delivery Period, or the T&SCo's ability to do so is impaired, by reason of a Capacity Constraint.
- 35.3 41.3 A Capacity Constraint includes (but is not limited to) a constraint which arises by reason of:
 - (a) Programmed Maintenance;
 - (b) Reactive Maintenance; or

- (c) an Emergency-; or
- (d) <u>a Maximum Annual Cumulative Flow Constraint, as set out in paragraph 23.</u>
- 36. Record keeping and publication obligations
- 36.1 11.4 In addition to the T&SCo's notification requirements obligations under paragraphs
 12.137.1 and 12.237.2, the T&SCo shallmust keep written records of any Capacity
 Constraints that have taken place, after the fact, including information relating to:
 - (a) the nature and cause of the Capacity Constraint;
 - (b) the Delivery Points affected by the Capacity Constraint;
 - (c) the action which the T&SCo has taken to address the Capacity Constraint; and
 - (d) whether and how a Pro-rata Reduction and/or the Constrained Capacity Optimisation
 Principles were applied to reduce the Registered Capacity available to Constrained
 Users; and
 - (e) (d) the magnitude and duration of the Capacity Constraint.

Note: T&SCo's obligations to maintain records of Capacity Constraints will also be addressed in the Licence, as there is a direct link between Capacity Constraints and the availability adjustment under the Licence.

- 36.2 The T&SCo must publish on its T&S Network Portal the information referred to in paragraph 36.1 as soon as is reasonably practicable after the event to which is relates.
- 36.3 The T&SCo must also keep records, including for the purposes of any Section E Review, of the following:
 - (a) <u>any requests for the Enhanced Response Service and the reasons for them;</u>
 - (b) <u>any Flow Tolerance Requests, the reasons for them, and the T&SCo's ability to accommodate such Flow Tolerance Requests;</u>
 - (c) any Exceedance Events, including which Exceedance Events are recorded as Low Nomination and Flow Exceedance Events or Zero Nomination Exceedance Events, and the reasons for them;
 - (d) <u>any Zero Nomination Low Flow Events and the reasons for them;</u>
 - what action the T&SCo needed to take (if any) to respond to the events referred to in paragraphs 36.3(c) and 36.3(d); and
 - (f) any Minimum Flow Deficits.
- 36.4 The T&SCo must publish on its T&S Network Portal details of the matters referred to in paragraph 36.3 by 30 November each Charging Year.

37. 42. Response to Capacity Constraints

Notification

- 37.1 12.1 To the extent not already provided in accordance with paragraph 1030, as soon as is practicable after the T&SCo becomes aware that a Capacity Constraint has arisen or will arise, the T&SCo shallmust take the following actions:
 - (a) the T&SCo shallmust publish information about the Capacity Constraint on the T&S Network Portal;
 - (b) the T&SCo shallmust send notification of the Capacity Constraint to all Users of Delivery Points affected by the Capacity Constraint (and such affected Users are referred to as "Constrained Users"); and
 - (c) the T&SCo shallmust keep the information referred to in paragraphs (a)37.1(a) and (b)37.1(b) reasonably updated, including in respect of information about the anticipated conclusion of the Capacity Constraint.
- 37.2 12.2 The information about the Capacity Constraint, which T&SCo must publish and notify to Users in accordance with paragraph 12.137.1 includes:
 - (a) the nature and cause of the Capacity Constraint;
 - (b) the Delivery Points affected by the Capacity Constraint;
 - (c) the action which the T&SCo is intending to take to address the Capacity Constraint; and
 - (d) the magnitude and expected duration of the Capacity Constraint.

Response

- 37.3 42.3 Where a Capacity Constraint arises the T&SCo shallmust:
 - (a) identify the Users whose Delivery Points are affected by the Capacity Constraint (the Constrained Users); and
 - (b) unless the Capacity Constraint arises by reason of an Emergency (under paragraph 12.639.1) or where paragraph 12.840.1 applies, the T&SCo shallmust apply the Pro-rata Reduction to the Registered Capacity of those Constrained Users.
- 38. Pro-rata Reduction
- 38.1 12.4-Pro-rata Reduction, where it is triggered pursuant to paragraph 12.337.3, means that the quantity of Registered Capacity of a Constrained User is reduced on a pro-rata basis, such that the reduced amount of Available Registered Capacity allocated to the Constrained User (the Reduced Capacity) is calculated according to the following formula:

$$\frac{\text{(Mod)} \text{ $} \text{$\lambda$ble Capacity}}{\sum CURC} \times RURC$$$

where:

- <u>"Available Capacity"</u> is the <u>amounttotal quantity</u> of Network Capacity still available <u>to all</u> <u>Constrained Users</u> during the Capacity Constraint;
- "CURC" is the Registered Capacity of all the Constrained Users; and
- "RURC" is the Registered Capacity of the particular Constrained User.

Surplus Available Capacity after Pro-Rata Reduction

- Once a User is notified of the Available Registered Capacity allocated to the User in accordance with paragraph 38.1, the User must notify the T&SCo as soon as reasonably possible whether or not the User will be able to or intends to use that Available Registered Capacity.
- 38.3 12.5 Where:
 - (a) <u>the T&SCo has reduced applied a Pro-rata Reduction to</u> the Registered Capacity of Constrained Users using Pro-rata Reduction; and
 - (b) one (1) or more Constrained Users have notified the T&SCo that they are unable to utilise the Reduceduse the Available Registered Capacity (or part of it) allocated to them,

then the T&SCo shall must seek to allocate the Reduced Available Registered Capacity referred to in paragraph 12.5(b)38.3(b) (being the "Surplus Available Capacity") in accordance with the Constrained Capacity Optimisation Principles set out in paragraph 12.7. For the avoidance of doubt, such 41.2.

- <u>Any Surplus Available Capacity to be allocated as referred to in paragraph 38.3,</u> may be allocated to any Constrained User, even if it would mean that the proportion of Available Capacity allocated to that Constrained User is greater than that of other Constrained Users, provided the T&SCo has applied the Constrained Capacity Optimisation Principles.
- 39. Reduction of Registered Capacity during Emergencies
- 39.1 12.6-Where a Capacity Constraint is caused by an Emergency then, unless it is safe and practicable for the T&SCo, acting as a Reasonable and Prudent Operator, to immediately apply a Pro-Rata Reduction to the Registered Capacity of Constrained Users, them! T&SCo shallmay respond to the Emergency by applying:
 - (a) the Emergency Procedure (where relevant to the Emergency in question); and
 - (b) the Constrained Capacity Optimisation Principles set out in paragraph <u>12.741.2</u>.
- 40. Reduction of Registered Capacity in exceptional circumstances other than an Emergency
- 40.1 Where:
 - (a) a Capacity Constraint arises by reason other than an Emergency; and

(b) notwithstanding the fact that there is no Emergency, the circumstances are such that, as set out in the Emergency Procedure, responding to the Capacity Constraint by applying Pro-rata Reduction to the Registered Capacity of Constrained Users would be likely to jeopardise the safety, integrity or operability of the T&S Network,

the T&SCo may respond to the Capacity Constraint by applying the Constrained Capacity Optimisation Principles to reduce the Registered Capacity of Constrained Users.

41. Constrained Capacity Optimisation Principles

41.1 <u>Where:</u>

- (a) 12.7 Wherethe T&SCo is unable to apply Pro-rata Reduction indue to the circumstances set out in paragraph 12.6 or there is surplus Available Capacity as setout in 39.1 or paragraph 12.5, then 40.1; or
- (b) there is Surplus Available Capacity as set out in paragraph 38.3,

the T&SCo shallmust allocate the Available Capacity to Constrained Users, by applying the following principles (the Constrained Capacity Optimisation Principles) to determine the reduced Available Registered Capacity (the Reduced Capacity) that can be allocated to each Constrained User:

41.2 <u>The "Constrained Capacity Optimisation Principles" are as follows:</u>

- (a) the T&SCo shallmust act with the primary objective to:
 - (i) only reduce a Constrained <u>User's User's</u> ability to deliver carbon dioxide at the Constrained User's Delivery Point as a last resort; and
 - (ii) allow the maximum amount quantity of carbon dioxide to be stored;
- (b) in achieving the primary objective referred to in paragraph (a) above 41.2(a), the T&SCo shallmust:
 - endeavour to maximise the number of Constrained Users that are able to deliver carbon dioxide into the T&S Network;
 - (ii) take into consideration the operational features and requirements of individual Constrained Users, including (but not limited to) each Constrained User's Minimum Turndown Rate, Minimum and Maximum Ramp Rates, any requirements for a minimum amountquantity of Network Capacity over a minimum period of time, and any maintenance that will be carried out by the Constrained User; and
 - (iii) take into consideration the Minimum Flow Rate of the T&S Network; and
- (c) <u>the T&SCo shallmust</u> actively communicate with all Constrained Users, to allow <u>the</u> principles (a) to (b) set out in paragraphs 41.2(a) and 41.2(b) to be effectively implemented.

Note: T&SCo has a general obligation of non-discrimination under its Licence.

12.8 Where:

- (a) a Capacity Constraint arises by reason other than an Emergency; and
- (b) the circumstances are such that responding to the Capacity Constraint by applying-Pro-rata Reduction to the Registered Capacity of Constrained Users would be likely tojeopardise the safety, integrity or operability of the T&S Network,

then T&SCo can respond to the Capacity Constraint by applying the Constrained Capacity Optimisation Principles to reduce the Registered Capacity of Constrained Users.

- 41.3 42.9 Where the T&SCo applies the Constrained Capacity Optimisation Principles to reduce the Registered Capacity of Constrained Users, either pursuant to paragraph 12.6 (to respond to an Emergency) or:
 - (a) pursuant to paragraph 39.1 (to respond to an Emergency); or
 - (b) pursuant to paragraph <u>12.840.1</u> (other circumstances where <u>a</u> Pro-rata Reduction cannot be immediately utilised to respond to a Capacity Constraint) then:

then:

- (c) (a) the T&SCo shallmust transition to applying Pro-rata Reduction to continue to respond to the Capacity Constraint as soon as it is reasonably practicable; and
- (d) (b) after a Pro-rata Reduction is applied, the T&SCo shallmust allocate any Surplus Available Capacity in accordance with paragraph 12.538.3.
- 42. Constrained Registered Capacity
- 42.1 42.10 Where, in the period up to Nomination Close Time, a Constrained User's Registered Capacity has been reduced:
 - (a) by a Pro-Rata Reduction; in accordance with paragraph 38; and/or
 - (b) in accordance with the Constrained Capacity Optimisation Principles at paragraph 41,

then the difference between the Constrained User's Registered Capacity and the Reduced Available Registered Capacity is the "Constrained Registered Capacity", for the purposes of the Availability Adjustment, except to the extent this was done in response to an ONC Maximum Annual Cumulative Flow Constraint.

12.11 For the purposes of the availability adjustment under T&SCo's Licence, where there is a Curtailed Nomination in relation to a Delivery Point d and Hour h, then the Constrained Registered Capacity for that Delivery Point d and Hour h, to the extent it is the result of the same Capacity Constraint, shall be disregarded.

43. Renominated Constrained Quantity

- 12.12 Where a Constrained User's Registered Capacity is reduced to Reduced Capacity in relation to a time period for which the Constrained User has already made a Daily Nomination or a Renomination, then:
- 43.1 Subject to paragraph 43.2, where the quantity of Registered Capacity available to a Constrained User is reduced or increased after Nomination Close Time:
 - (a) the Constrained User shallmust amend any earlier Nominated Quantities included in a Daily Nomination or Renomination, through a new-Renomination, so that no Nominated Quantity exceeds the Constrained User's Reduced Available Registered Capacity (but no Renomination is required where no Nominated Quantity exceeds the Available Registered Capacity);
 - (b) <u>the T&SCo shallmust</u> accept or reject the Renomination in accordance with the principles <u>applying to Renominations</u> set out in <u>paragraph 6</u>this Section E (Network Use and Capacity); and
 - (c) any amended Daily Nominated Quantity nominated under paragraph (a)43.1(a) shall be the Constrained User's
 - (i) "Renominated Constrained Quantity; and".
 - (ii) where accepted or rejected by T&SCo (or deemed accepted) in accordance with the principles set out in paragraph 6, the Final Confirmed Nominated Quantity for the purposes of paragraphs 6.22 to 6.27.
- Where a Constrained User has failed to amend any Daily Nomination or Renomination, through a Renomination in accordance with paragraph 43.1(a), the Constrained User shall be deemed to have done so and the Constrained User's Renominated Constrained Quantity shall be deemed to be equal to the Constrained User's Available Registered Capacity for that Delivery Period.
- 43.3 42.13 Where a Constrained User has insufficient notice of its Reduced Available Registered Capacity in order to be able to comply with the requirements of paragraph 12.12, then 43.1:
 - (a) the T&SCo and the User shallmust in good faith communicate and cooperate, both acting as Reasonable and Prudent Operators, and the User shallmust endeavour to comply as soon as it is reasonably possible; and
 - (b) <u>for that Delivery Period</u>, the User's <u>actual flow of carbon dioxide for that Hour shall bedeemed to be the Renominated Constrained Quantity and Final Confirmed Nominated Quantity</u> for the purposes of calculating the Curtailed Nomination <u>(for the purposes of the Availability Adjustment)</u> under paragraph <u>6.13</u>17.2 or paragraph <u>6.21</u>19.3 (as <u>applicable</u>), shall be deemed to be the User's actual flow of carbon dioxide in that <u>Delivery Period</u>.

General obligations of Users to follow T&SCo directions

- 12.14 A User must stop delivering or reduce the quantity of carbon dioxide being delivered to a Delivery Point in accordance with:
 - (a) any Maintenance Programme; or
 - (b) any direction given by T&SCo to do so in accordance with the provisions of the Code.
- 12.15 If a User fails to comply with any direction given by T&SCo in accordance with paragraph 12.14,

 T&SCo may take steps to prevent the User from delivering carbon dioxide to the Delivery

 Point, acting as a Reasonable and Prudent Operator to minimise any adverse impact on User

 Facilities.
- 44. No obligation to accept delivery of carbon dioxide
- 44.1

 12.16-Without prejudice to any other right the T&SCo may have under the Code to reject the delivery of carbon dioxide (including under Section F Section F (Network Design and Specification)), the T&SCo will be relieved of its obligations obligation to accept delivery of any quantity of carbon dioxide of the understand that the corresponds to that quantity, to the extent that it is not feasible to do so by reason of a Capacity Constraint.
- 45. Payment of Charges during Capacity Constraints
- 45.1

 12.17-Subject to paragraph 12.1845.2, a Constrained User will remain liable to pay Capacity Charges and Network Charges in respect of its Registered Capacity. notwithstanding that the T&SCo is unable to take delivery of carbon dioxide at the quantity of its Registered Capacity at its Delivery Point by reason of a Capacity Constraint.
- 45.2 12.18 Where:
 - (a) a Capacity Constraint arises for reasons other than Planned Programmed
 Maintenance;
 - (b) a Constrained User is unable to utilise the whole or any part of its Registered Capacity by reason of the Capacity Constraint, for a continuous period of more than [x Days]the waiting period under the relevant insurances; and
 - (c) the event that caused the Capacity Constraint is [covered by T&SCo's Business Interruption Insurance for Asset Damage]results from an Insured Risk or property damage on the T&S Network (an "Insured Event") which has caused a Capacity Constraint,

Note: details of the mechanism for the suspension of Capacity Charges and Network Charges where covered by business interruption insurance remain subject to review.

then from that point in time until the Capacity Constraint is resolved, the Constrained User shall only be liable to pay Capacity Charges and Network Charges [in relation to the amount of Registered Capacity that the Constrained User is able to utilise.]

the Capacity Charges and Network Charges payable by the Constrained User become provisional and must be reduced for the period for which the relevant insurers are liable (which may commence prior to the time from which it is agreed in writing by the relevant

- insurers or it has been determined under the relevant insurances that the insurers are liable under the relevant insurance), by an amount equal to the Business Interruption Proceeds referrable to the Constrained User's Capacity Charges and Network Charges.
- The T&SCo must provide notice to the User as soon as reasonably practicable following the date when it has been agreed in writing by the relevant insurers or it has been determined under the relevant insurances that the insurers are liable under the relevant insurances, setting out the level of Business Interruption Proceeds that are referable to the Capacity Charges and Network Charges payable by the Constrained User.
- 46. Capacity Constraints caused or contributed to by Users
- Where the T&SCo considers that any Capacity Constraint has been caused or contributed to by any act or omission of a User, the T&SCo must notify the User within ten (10) Business

 Days of the alleged act or omission, with full details of the relevant Capacity Constraint and how the User is alleged to have caused it or contributed to it.
- Where a User has received a notification from the T&SCo pursuant to paragraph 46.1, any response that the User wishes to make to the T&SCo must be made within ten (10) Business Days of the date of the notification.
- The T&SCo must notify the Regulator of any alleged act or omission by a User, with full details of the relevant Capacity Constraint and how the User caused it or contributed to it together with, where the T&SCo has received a response from the User pursuant to paragraph 46.2, a copy of that response, within fifteen (15) Business Days of the date of T&SCo's notification to the User pursuant to paragraph 46.1.
- 47. **General obligations of Users**
- Where a User's flow of carbon dioxide is in any way not in conformity with the provisions of the Code (whether in terms of conformity with the quantity of Registered Capacity available to a User, conformity with nominations or otherwise):
 - (a) the User must notify the T&SCo as soon as reasonably practicable after becoming aware of such non-conformity;
 - (b) the User must engage in a dialogue with the T&SCo to determine what, if any, action the User must take in response; and
 - if a User fails to respond to any direction given by the T&SCo in response to the
 non-conformity, the T&SCo may take steps to prevent the User from delivering carbon
 dioxide to the Delivery Point, acting as a Reasonable and Prudent Operator, to
 minimise any adverse impact on the relevant User Facility.
- 47.2 In an Emergency, every User must follow any directions of the T&SCo in accordance with the Emergency Procedure.
- 48. Section E Reviews
- 48.1 Each T&SCo must carry out reviews of certain operational matters relating to its T&S

 Network (the "Section E Reviews") in accordance with Part L of Standard Condition B5

(CCS Network Code) of the Licence and this paragraph 48, once there is sufficient operational data available from use of the T&S Network by Users of each different User Type among the Users who were Selected under the Track-1 Cluster Sequencing Process and which shall be no earlier than the date that is twelve (12) months after the date of commercial operations of the last of the Users who were Selected under the Track-1 Cluster Sequencing Process to achieve commercial operations (provided that where there are multiple Users of a single User Type using a T&S Network, only the first User of that User Type is required to have achieved commercial operations for the purposes of the review under this paragraph 48.1).

- The Section E Reviews must take place no later than thirty-six (36) months following commencement of commercial operations of the T&S Network and, where it is reasonably apparent that the conditions under paragraph 48.1 will not be satisfied by this date due to the discontinuation or imminent discontinuation of a User who was Selected under the Track-1 Cluster Sequencing Process (or the fact that an Initial User has elected not to accede to the Code), the timeline for carrying out the Section E Reviews pursuant to paragraph 48.1 shall be adjusted to exclude such User such that the Section E Reviews may be carried out prior to the thirty six (36) month longstop.
- 48.3 Following the carrying out of the first round of Section E Reviews, the Section E Reviews must continue to be carried out on an annual basis.
- 48.4 The Section E Reviews are as follows:
 - (a) <u>a review of:</u>
 - (i) the appropriateness of the levels of the Over Variation Limit and the Under Variation Limit;
 - the frequency of and reasons for Exceedance Events, including Exceedance

 Events which are Low Nomination and Flow Exceedance Events or Zero

 Nomination Exceedance Events;
 - (iii) the ability of the T&SCo to accommodate such Exceedance Events;
 - (iv) the extent to which Exceedance Events impact on the cost of operating the

 T&S Network, and on the safe and reliable operation of the T&S Network and
 the needs of all Users; and
 - (v) the frequency of, and reasons for, Users making Flow Tolerance Requests and the ability of the T&SCo to accommodate such Flow Tolerance Requests,

(the "Exceedance Events and Flow Tolerance Requests Review");

- (b) <u>a review of:</u>
 - <u>(i)</u> <u>the frequency of and reasons for requests by Users for the provision of the Enhanced Response Service;</u>

- <u>(ii)</u> <u>the level of responsiveness being provided by the Enhanced Response</u> Service; and
- (iii) the impact of the provision of the Enhanced Response Service on the cost of operating the T&S Network, and on the safe and reliable operation of the T&S Network and the needs of all Users.

(the "Network Responsiveness Review"); and

- <u>a review of the efficacy of the provisions dealing with Nominations and Renominations</u>
 <u>(including the length of Delivery Periods and implementation timelines and deadlines)</u>
 <u>as set out in this Section E (Network Use and Capacity) in meeting the needs of Users</u>
 in an effective, economic and efficient manner (the "Nominations Review").
- 48.5 In carrying out each of the Section E Reviews (that is, the Exceedance Events and Flow Tolerance Requests Review, the Network Responsiveness Review and the Nominations Review), the T&SCo must:
 - (a) review all relevant operational data and records maintained in accordance with this

 Section E (Network Use and Capacity) and publish a summary of its findings based on that data and records on the T&S Network Portal;
 - (b) having made the summary referred to in paragraph 48.5(a) available to Users

 (together with the relevant operational data and records it is based on), invite the

 Users of its T&S Network to provide feedback on the subject matter of the relevant review; and
 - (c) engage with the Users of its T&S Network and use reasonable endeavours to agree with the Users on a set of recommendations in relation to the subject matter of each of the reviews.
- Within six (6) months of commencement of a Section E Review, the T&SCo must publish and provide to the Regulator and the Secretary of State a report setting out a summary of its findings and:
 - (a) where the T&SCo was able to agree on a set of recommendations with the Users of its T&S Network, the agreed recommendations; and
 - (b) where the T&SCo and the Users were not able to agree on a set of recommendations, the different views of the T&SCo and the Users of its T&S Network.
- <u>Each Section E Review must be carried out separately, with its subject matter clearly defined,</u> but the findings may be published in a single report.

CCS Network Code

Section F - Network Design and Specification

1. Introduction

Introduction to specifications

- 1.1 This Section F (Network Design and Specification) sets out requirements for:
 - (a) 1.1 The provisions of this Section F shall apply in respect of the delivery of carbon dioxide by Users to the T&S Network at Delivery Points.
 - (b) the installation, operation and maintenance of Measurement Equipment for the purposes of measuring carbon dioxide flows (including impurities) from a User at the Delivery Point to the T&S Network, and certain related issues; and
 - (c) <u>carbon dioxide re-use where the Re-use Service is being provided to a User.</u>
- 1.2 Users delivering carbon dioxide to any part of the T&S Network at a Delivery Point shall comply with the relevant requirements of this Section F (Network Design and Specification).

User Facilities

- 1.3 For the purposes of this Section F a User Facility" is a single facility or several facilities operated by a User or Eligible Applicant (where such facilities share a single Delivery Point) connected including, following connection to the T&S Network, the pipeline(s) and all related equipment connecting such facility or facilities to the T&S Network at a Delivery Point.
- 2. Existence of <u>a</u> Connection Agreement
- 2.1 1.4-A User may not deliver carbon dioxide into the T&S Network at any Delivery Point unless there is in force a Connection Agreement between the User and the T&SCo which, among other things:
 - (a) identifies the User Facility;
 - (b) identifies the Delivery Point; and
 - (c) specifies the Entry Provisions applicable to that Delivery Point.
- 3. 2. Entry Provisions

Content of Entry Provisions

- 3.1 **2.1** The **"Entry Provisions"** for each Delivery Point are:
 - (a) (a) the Carbon Dioxide Specifications, as defined in paragraph 2.43.2;
 - (b) (b) the Measurement Requirements, as defined in paragraph 2.53.5; and
 - (c) (e) the Local Requirements, as defined in paragraph 2.93.8.

- 3.2 <u>The "Carbon Dioxide Specifications" are specific to each individual T&S Network and are set out in parts for each T&S Network in Annexure B (Carbon Dioxide Specifications T&S Networks).</u>
- 3.3 2.2 The Carbon Dioxide Specifications shall must at all times conform to the following key principles:
 - (a) the primary purpose of the Carbon Dioxide Specification is to ensure the safety, protection and operability of the T&S System Network; and
 - (b) the Carbon Dioxide SpecificationSpecifications shall achieve the following aims:
 - (i) protection of health and safety;
 - (ii) corrosion management and avoidance;
 - (iii) environmental protection;
 - (iv) operational control assured by maintaining predictable flow conditions; and
 - (v) management of reservoir impacts.
- 3.4 2.3 In order to achieve the aims set out in paragraph 2.2(b) 3.3(b), the Carbon Dioxide Specifications shallmust at all times comply with the requirements set out in Annexure Annexure A (Carbon Dioxide Specifications (General)).
- 2.4 The Carbon Dioxide Specifications are specific to each individual T&S Network and are set out:
 - (a) in Annexure B for the Northern Endurance Partnership T&S Network; and
 - (b) in Annexure C for the HyNet T&S Network.

Measurement Requirements

- 3.5 2.5 In this section, The "Measurement Requirements" means the requirements for the Measurement Equipment, including the procedures methods and standards by which compositional analysis and flow rate, Flow Rate, temperature and pressure will be measured, sampled and analysed and parameters regarding accuracy/regularity of checks, control systems, operation data transmission and quality assurance (including validation of Measurement Equipment) and includes the requirements set out in the CO₂ Quality Monitoring Procedure.
- 3.6 The Measurement Requirements which are specific to each individual T&S Network are set out in Parts for each T&S Network in Annexure C (Measurement Requirements T&S Networks).
- 3.7 2.6-The Measurement Requirements shall at all times conform to the following principles:
 - (a) the Measurement Equipment must determine the quantity (measured in tCO₂) of all the carbon dioxide delivered by a User at a Delivery Point; and

- (b) the components within the carbon dioxide stream shall be:
 - (b) the impurities within the CO₂ stream shall be sampled and monitored to be representative of the carbon dioxide stream and ensure that compliance with the Carbon Dioxide Specifications is maintained; and
 - <u>measured regularly at a frequency to be agreed with the T&SCo and must follow a risk assessment process,</u>
 - (c) the pressure, temperature and water content should be measured continuously, together with other including impurities where an exceedance may result in an imminent downstream impact. This may include water, oxygen NOx, SOx, H₂Soxides of nitrogen, oxides of sulphur, hydrogen sulphide, ammonia and, for dense and multi-phase applications, hydrogen and nitrogen;
- (d) other impurities should be measured regularly at a frequency to be agreed with the T&SCo, which may be four times every hour, and could follow a risk assessment process;
- <u>(c)</u> <u>the pressure and temperature within the carbon dioxide stream shall be measured continuously;</u>
- (d) (e) the calibration range, accuracy and measurement uncertainty of the measuring equipment Measurement Equipment shall be appropriate to the:
 - (i) permissible levels of impurity; and
 - (ii) <u>stream operating condition and flowrates,</u>

with sufficient capacity to <u>provide</u><u>retain</u> information to enable subsequent investigation in the event of an exceedance; and

- (f) the CO₂-content within the stream should be measured continuously and to an accuracy consistent with any fiscal requirements;
- (e) (g) monitoring data should be available simultaneously to both the User and the T&SCo, and records retained for a sufficient length of time, which must be for a period no less than the period specified in paragraph 11.3, to enable trends to be identified and for historical information to be interrogated in the event of an impact on the T&S Network infrastructure being identified.
- 2.7 The Measurement Requirements which are specific to each individual T&S Network are set out:
 - (a) in Annexure D for the Northern Endurance Partnership T&S Network; and
 - (b) in Annexure E for the HyNet T&S Network.

Local Requirements

 Delivery Point and/or the Measurement Equipment at the User's Delivery Point which shall be specified in a User's Connection Agreement, and may include:

- (a) any User-specific CO₂ Specification Carbon Dioxide Specifications requirements; and
- (b) any User-specific measurement requirements-; and
- (c) <u>any User-specific carbon dioxide monitoring requirements.</u>
- 3.9 2.9 The Each T&SCos acknowledge SCo acknowledges that:
 - (a) Users are required to comply with the User Requirements; and
 - (b) the T&SCo shallmust adapt the Local Requirements in order to accommodate the User Requirements-, but not to otherwise change any provisions that would otherwise apply to the User under the Code; and

Compliance with Regulatory Requirements

- (c) the T&SCo must take into account the Local Requirements set out in a Connection Application when developing the CO₂ Quality Monitoring Procedure.
- 3.10 <u>2.10 The Users acknowledge Each User acknowledges</u> that:
 - (a) <u>the T&SCo</u> is required to comply with the Regulatory Requirements in relation to the T&S Network; and
 - (b) in developing the Entry Provisions, T&SCo has had regard to the Regulatory Requirements.
- 3.11 2.11 Notwithstanding any other provision in this Section F Section F (Network Design and Specification), if any aspect of the Entry Provisions cause the T&SCo to be in breach of the Regulatory Requirements, then:
 - (a) <u>the T&SCo</u> will give notice to each User specifying any changes required to the Entry Provisions to ensure that <u>the T&SCo</u> is compliant with the Regulatory Requirements; and
 - (b) as soon as reasonably practicable and in any event within ten (10) Business Days of receipt of the notice of Entry Provision changes, each User must notify the T&SCo of any comments on the proposed changes, including:
 - <u>information about the User (company, registered office, etc.) and about at least one (1) reference person (name, telephone number, fax number, e-mail address, etc.) who can be contacted with respect to that User's comments;</u>
 - (ii) the reasons why that User believes its comments should be incorporated; and
 - (iii) any documentation (analyses, reports, etc.) to support the request;

- (c) the T&SCo must give reasonable consideration to any comments provided by a User pursuant to paragraph 3.11(b); and
- (d) once all comments have been received and considered by the T&SCo in accordance with paragraph 3.11(c), the T&SCo must:
 - (b) deliver to each User must use best endeavours to comply with the notice received pursuant to paragraph (a) forthwith. the LCCC and the Regulator the updated changes to the Entry Provisions; and

3. Compliance

- <u>put forward the revised Entry Provisions as a Modification Proposal under the Modification Procedure in Section B (Governance).</u>
- 3.12 For the purposes of paragraphs 3.8 to 3.11, references to Users shall be treated as including references to Eligible Applicants.
- 3.13 The Parties acknowledge and agree that notwithstanding paragraph 17.1 of Section J

 (General), the Contracts (Rights of Third Parties) Act 1999 shall apply to the rights conferred on Eligible Applicants under paragraphs 3.8 to 3.11.

4. Compliance with Carbon Dioxide Specifications

- 3.1 Where carbon dioxide which does not comply with the Entry Provisions is tendered for delivery to the T&S Network at a Delivery Point then T&SCo shall promptly inform the relevant User about the non-compliance, and T&SCo may, from time to time until such time as the relevant non-compliance is rectified, in its discretion either:
- 4.1 <u>Each User must ensure that:</u>
 - (a) the carbon dioxide delivered at the Delivery Point by the User is compliant with the Entry Provisions; and
 - (b) <u>sufficient processes are established by the User to detect Non-Compliant CO₂ in any carbon dioxide flows tendered for delivery by the User.</u>
- 4.2 Notwithstanding paragraphs 4.8 and 4.9, if a User becomes aware of Non-Compliant CO₂, it must:
 - (a) immediately commence the procedure to cease delivery of Non-Compliant CO₂ (as set out in the Isolation Procedure and the CO₂ Quality Monitoring Procedure); and
 - (b) promptly notify the T&SCo thereof and include in its notification all relevant information relating to the Non-Compliant CO₂, including the specifications thereof.
- 4.3 Where, prior to a User becoming aware of Non-Compliant CO₂, a User has reason to believe that there will be a temporary deviation of a component of carbon dioxide that might result in Non-Compliant CO₂, such User:

- (a) will provide the T&SCo with advanced notification thereof and include in its notification all relevant information relating to such matter; and
- where such a notification has been made, the T&SCo shall consider at its discretion providing such User with advanced notice of relief from the requirement to immediately commence the procedure to cease delivery of Non-Compliant CO₂ upon receipt of a notice from a User under paragraph 4.2(b) in relation to the delivery of Non-Compliant CO₂ as notified under paragraph 4.3(a).
- 4.4 The T&SCo must notify the User no later than one (1) hour after receiving a notification under:
 - (a) paragraph 4.3(a) of its decision to provide advanced notice of relief or not from the requirement to immediately commence the procedure to cease delivery of Non-Compliant CO₂ under paragraph 4.2(a) where the User becomes aware of it; or
 - (b) paragraph 4.2 of its decision to either accept or reject the Non-Compliant CO₂ (on the basis of the specifications notified by the User to T&SCo in accordance with paragraph 4.2(b) or paragraph 4.3(a) (as applicable)) in whole or at a partially reduced flow,

it being understood that any such decision may be subject to certain conditions and will apply for the limited period of time specified by the T&SCo and any carbon dioxide flow that does not comply with such conditions and/or time limit shall be deemed to have been rejected.

- 4.5 Any Non-Compliant CO₂ notified to the T&SCo or temporary deviations of a component of carbon dioxide notified to the T&SCo under this paragraph 4 which is not acknowledged by the T&SCo within the required period set out in paragraph 4.4 will not be deemed to be accepted by the T&SCo.
- 4.6 If and to the extent that the T&SCo has accepted Non-Compliant CO₂ (whether or not subject to certain conditions), the User shall be entitled to resume the delivery of carbon dioxide at the Delivery Point to enable the accepted Non-Compliant CO₂ to enter the T&S Network for the period of time specified by the T&SCo in accordance with paragraph 4.4.
- 4.7 If, in relation to Non-Compliant CO₂ accepted by the T&SCo, the User:
 - (a) becomes aware of one (1) or more further deviations from the specifications of such Non-Compliant CO₂; or
 - (b) has reason to believe that there will be one (1) or more further temporary deviations of a component of carbon dioxide that might result in Non-Compliant CO₂,

to that notified by the User to the T&SCo in its latest notification in accordance with paragraph 4.2(b) or 4.3, the User must make a new notification under paragraphs 4.2(b) or 4.3 (as applicable) and paragraph 4.2 shall apply *mutatis mutandis* (i.e. the User must promptly notify the T&SCo of the deviation(s) following which the T&SCo must decide to either accept or reject the Non-Compliant CO₂) until the expiry of the period specified by the T&SCo under paragraph 4.4.

- <u>Where the T&SCo has reason to believe or becomes aware of Non-Compliant CO₂ being tendered for delivery to the T&S Network at a Delivery Point, the T&SCo:</u>
 - (a) must promptly notify the relevant User about the suspected or actual non-compliance; and
 - (b) until such time as the relevant non-compliance is rectified where it has been determined that Non-Compliant CO₂ has been tendered for delivery at a Delivery Point pursuant to the Measurement Requirements, retains the right in its discretion (but subject to the Measurement Requirements and Isolation Procedure) to:
 - (i) (a)-refuse to accept delivery or continued delivery of such carbon-dioxideNon-Compliant CO₂; or
 - (ii) (b)-accept such delivery in full or accept such delivery as toof part only of what is tendered for delivery-; and/or
 - (iii) apply any reasonable (additional) technical or operational conditions to the acceptance of any Non-Compliant CO₂ in accordance with the Measurement Requirements.
- 4.9 3.2 The T&SCo shall be entitled to take any <u>reasonable</u> steps available to it to limit the rate at which <u>non-compliant carbon dioxide</u> Non-Compliant CO₂ is delivered to the T&S Network or to secure that such <u>carbon dioxide</u> Non-Compliant CO₂ is not so delivered, <u>including closing a valve at the Delivery Point with</u>, where reasonably possible, prior warning by telephone to the User.
- Any Non-Compliant CO₂ accepted by the T&SCo under this paragraph 4 shall be deemed to comply with all relevant Entry Provisions and shall not be Non-Compliant CO₂ with effect from the time such carbon dioxide was delivered to the T&S Network, provided that where such acceptance was subject to certain technical or operational conditions, only that part of the carbon dioxide stream that complies with such conditions shall be deemed to comply with all relevant Entry Provisions and any carbon dioxide that is not compliant with such conditions shall be Non-Compliant CO₂.
- 4.11 If all relevant Entry Provisions in connection with the flow of carbon dioxide, in particular relating to the detection, monitoring, sampling and testing of carbon dioxide components and Flow Rate:
 - (a) have been complied with by the User, any Non-Compliant CO₂ delivered to the T&S

 Network prior to the time that the T&SCo and/or User became aware of it shall be

 deemed to comply with all relevant Entry Provisions and shall not be Non-Compliant

 CO₂ with effect from the time such carbon dioxide was delivered to the T&S Network

 until the time that such Party became aware of it; or
 - (b) have not been complied with by the User, Non-Compliant CO₂ shall be deemed to have been delivered by the User with effect from the time at which the User did not comply with the relevant Entry Provisions.

- 4.12 For the purposes of this paragraph 4, a Party is "aware" of Non-Compliant CO₂ on and from the time the results of the monitoring are received.
- 5. 4. General Measurement Provisions

Measurement Equipment

- 5.1 <u>In this Section F (Network Design and Specification), "Measurement Equipment" means:</u>
 - (a) 4.1 In this section, Measurement Equipment means all equipment and installations, (including associated metering, sensors, sampling and analysis equipment, inlet and outlet pipework, instruments, regulator, filters, valves, sample lines, seals, and structures within which such equipment is housed and related communication systems or mounted) to be installed by a User used to:
 - (i) (a) determine mass flow of CO₂carbon dioxide for fiscal purposes; and/or
 - (ii) (b)-measure the critical components, gasstream quality and characteristicscomposition of carbon dioxide delivered at a Delivery Point as against the Carbon Dioxide Specifications for integrity purposes.; and
 - <u>(iii)</u> measure temperature and pressure of the carbon dioxide stream delivered to the T&S Network; and
 - (b) any system (forming part of or connected to any of the foregoing) for converting or otherwise processing data or signals obtained from any such device or instrumentation so as to derive the quantities which are to be provided in accordance with this Section F (Network Design and Specification).

Installation of Measurement Equipment

- 4.2 A User may not deliver any carbon dioxide at a Delivery Point until and unless

 Measurement Equipment which complies with the requirements of this section Section F

 (Network Design and Specification) has been installed, commissioned, operated and maintained in proper working order by the User at the Delivery Point.
- 5.3 The User must:

Note: as part of the CCUS Cluster Sequencing Process or another Selection Process the User will be required to provide assurance as to its capabilities/competence in relation to its responsibilities under this Section F.

- 4.3 The User shall at its own cost install, commission, operate and maintain the Measurement-Equipment in accordance with the requirements of this Section F.
 - (a) 4.4 The User shall obtain T&SCo's prior approval in relation to the siting, specifications and installation of the Measurement Equipment and any such installation and commissioning shall, to the extent relevant, comply with the Measurement Requirements.;

- (b) ensure that the Measurement Equipment is installed, operated and maintained to the standard of a Reasonable and Prudent Operator;
- <u>ensure that any such installation and commissioning shall, to the extent relevant, comply with the Measurement Requirements and the Local Requirements;</u>
- ensure that the Measurement Equipment is installed so as to measure the relevant property or characteristic of carbon dioxide flowing at, or as nearly as practicable at, the Delivery Point; and
- ensure that the Measurement Equipment installed is designed, installed, operated and maintained so as to be capable of complying with the requirements of the Data
 Transfer Procedures (e.g. so as to provide telemetry signals in respect of identified points of telemetry).
- Prior to commissioning, all relevant components of the Measurement Equipment shall be calibrated in compliance with the requirements specified in the Measurement Requirements.
 Such calibration shall demonstrate conformity with relevant product standards applicable to the class index of each component of the Measurement Equipment (where available).
- The User must procure the calibration certificates which include the actual errors of each component of the Measurement Equipment across its operating range. Such certificates shall be retained by the User in accordance Annexure K (Retained Data) and shall be made available, on request, to the T&SCo.
- 5.6 Without prejudice to paragraph 5.5, the User must provide such evidence as the T&SCo may require to confirm that, following its commissioning, the Measurement Equipment meets the requirements of this Section F (Network Design and Specification). This evidence shall be traceable and dated.
- 5.7 Subject to paragraph 5.8 and unless otherwise specified in the Measurement Requirements,
 each component of the Measurement Equipment must comply with the Applicable Standards
 specified in the Measurement Requirements which are current at the Code Implementation

 Date.
- If, following the Code Implementation Date, any component of the Measurement Equipment is calibrated, tested, installed or commissioned, the calibration, testing, installation or commissioning of such component must be carried out in accordance with and comply with the latest version of the Applicable Standards specified in the Measurement Requirements prevailing at the time of such calibration, testing, installation or commissioning.

6. Validation of Measurement Equipment

- 6.1 4.5-The T&SCo must at its cost, ensure that the Measurement Equipment shall be by the Independent Verifier in accordance with, and for compliance with, the Measurement Requirements:
 - (a) prior to any carbon dioxide being allowed to be delivered at a Delivery Point to flow in the T&S Network; and

- (b) at least annually once per calendar year thereafter.
- The User shall be entitled, but shall not be obliged, at its own cost to attend and witness any validation carried out by the Independent Verifier in accordance with this paragraph 6.
- 6.3 The T&SCo must give the User reasonable prior notice of the date, time, place and nature of every validation carried out pursuant to paragraph 6.1.
- <u>An Independent Verifier shall be entitled to assume that all required consents have been obtained for the relevant validation until such time as it is notified to the contrary.</u>
- 6.5 The T&SCo must procure the preparation of a report (the "Validation Report") of each validation carried out pursuant to this paragraph 6, setting out:
 - (a) the methodology used for the validation;
 - (b) the results of the validation;
 - (c) the steps taken or which are to be taken in accordance with paragraph 6.8 or 6.9 for adjustment or replacement of Measurement Equipment as a result of such validation; and
 - (d) <u>any other matter required pursuant to the Measurement Requirements.</u>
- 6.6 The T&SCo must provide the Validation Report to the User no later than fourteen (14) Days after the delivery of the Validation Report to the T&SCo.
- 6.7 <u>Unless disputed by the User in accordance with paragraph 6.11, the results of the Validation Report shall be binding on both Parties.</u>
- 6.8 Upon or immediately following delivery of the Validation Report to the User pursuant to paragraph 6.6, if required under paragraph 6.5(c) the User must:
 - (a) at its own cost ensure that the Measurement Equipment (or the relevant component thereof) is adjusted or replaced as necessary so that the Measurement Equipment (or such component) complies with the Measurement Requirements; and
 - (b) notwithstanding any dispute commenced by a User relating to the results of a

 Validation Report, unless otherwise agreed by the T&SCo, immediately commence
 the procedure to cease delivery of carbon dioxide (as set out in the Isolation

 Procedure) until such adjustment or replacement has occurred.
- 6.9 Where the User is unable to ensure the adjustment or replacement of the Measurement Equipment (or any component thereof) upon or immediately following delivery of the Validation Report pursuant to paragraph 6.8, the User must:
 - (a) submit to the T&SCo proposals for initiating and carrying any required rectification works to address any faults or errors identified in the Validation Report (and which necessitated the recommendation for adjustment or replacement of the Measurement Equipment):

- (b) ensure such rectification as soon as reasonably practicable and in accordance with the User's proposals and to the T&SCo's reasonable satisfaction; and
- <u>unless otherwise agreed by the T&SCo, immediately commence the procedure to</u>
 <u>cease delivery of carbon dioxide (as set out in the Isolation Procedure) until the fault</u>
 has been rectified to the T&SCo's reasonable satisfaction.
- 6.10 All costs incurred in respect of the Measurement Equipment outside of the validation undertaken under paragraph 6.1 shall be payable by the User.
- 6.11 The User shall be entitled to dispute the accuracy of any Validation Report by giving notice to the T&SCo not later than fourteen (14) Days after receipt of the same.
- 6.12 The Parties must, as soon as reasonably practicable after the date of the User's notice in accordance with paragraph 6.11 and in any event within twenty (20) Business Days from the date of such notice, consult together in good faith and use their reasonable endeavours to settle the dispute.
- Where the Parties are unable to resolve the dispute within twenty (20) Business Days from the date of the notice in accordance with paragraph 6.12, either of them shall be entitled to refer the dispute to an Expert for determination in accordance with paragraph 51 of Section B (Governance).

7. Measuring Carbon Dioxide Specifications Characteristics

- 4.6 The User shallmust at its own cost install, commission, operate and maintain Measurement Equipment to determine the characteristics defined in the Carbon Dioxide Specifications of any carbon dioxide that is delivered at a Delivery Point. Such Measurement Equipment shallmust be of a type, standard of design and accuracy so as to comply with the Measurement Requirements and shallmust, as a minimum, meet the following criteria:
 - (a) all measurement biases shall must as far as is practicable be eliminated or compensated for;
 - (b) the uncertainty of measurement shall must be such that the risk is minimised to both the User and T&SCo that there may be carbon dioxide flowing of which either or both of them are unaware;
 - (c) the sampling system used to obtain the composition sample of carbon dioxide for quality measurements shallmust ensure that the sample is representative of the carbon dioxide delivered at a Delivery Point and that no change to the carbon dioxide composition occurs between the sample point and the analytical instrument or that compensation or correction for any such change is fully accommodated within the Measurement Equipment; and
 - (d) measurements and validation of equipment to make such measurement shallmust, where feasible, be traceable to national or international standards and be in accordance with the Measurement Requirements.

8. Measuring quantity of carbon dioxide

- 8.1 4.7 The User shallmust at its own cost install, commission, operate and maintain flow Measurement Equipment to determine the quantity of CO2 carbon dioxide (measured in tCO2) flowing into the T&S Network on (or in any period within) a Day at a Delivery Point such that:
 - (a) all mass flows of carbon dioxide into the T&S Network shallmust comply with this paragraph 8.1; and
 - (b) the measurement of quantity of mass shallmust be without bias and with an uncertainty of ≤+/-1.0% of the measured value (at 95% confidence interval) over the equal to or less than that specified flow range in the Measurement Requirements and any Local Requirements.

9. Operation of Measurement Equipment

- 9.1 4.8 The operation and maintenance of Measurement Equipment shall must include:
 - (a) all activities associated with meter-reading, including, reading, calibration, and filing of all data provided by the Measurement Equipment; and
 - (b) provision of access to real time data to the T&SCo in accordance with the Data Transfer Procedures.

9.2 4.9 The User shallmust:

- at its own cost and expense, ensure that the Measurement Equipment is kept in good working order, repair and condition in accordance with the Measurement
 Requirements to the extent necessary to allow the correct registration, recording and transmission of the Measurement Data by the relevant component of the Measurement Equipment;
- (b) <u>ensure that the Measurement Equipment is operated so as to:</u>
 - <u>enable the User to provide Measurement Data in accordance with any applicable Data Transfer Procedures; and</u>
 - enable the CDS to have access to the data from the Flow Meter and the
 Re-Use Service Meter (where relevant) to enable the CDS to perform its
 functions as contemplated under Section H (Charges, Invoicing and Payment)
 and Section I (Data), and in particular to prepare a Processed Flow Meter Data
 Statement for each Billing Period for each User;
- (c) (a) ensure that the T&SCo is provided with sufficient operator permissions to enable it to exercise its rights as set out under this the Code to reject or refuse the delivery of Carbon Dioxidecarbon dioxide at a User's Delivery Point and/or close the Delivery Point; and
- (d) (b) informseek the T&SCo promptly and accurately regardingSCo's consent before making any changes to the any installed Measurement Equipment (unless such change is a like for like instrument swap).

10. 5. Access to Measurement Equipment, <u>Delivery Point and pipeline</u> infrastructure

10.1 <u>Where:</u>

- (a) any part of the Measurement Equipment is located at a site owned or controlled by the User; or
- (b) in relation to any pipeline infrastructure owned by a User between the Metering Point and the Delivery Point:
 - (i) <u>a fault or damage or suspected fault or damage is identified by the T&SCo; or</u>
 - <u>the T&SCo wishes to check that the pipeline infrastructure is properly</u> maintained; or
- <u>5.1 Where</u> any part of the <u>pipeline infrastructure</u>, Measurement Equipment <u>or the Delivery Point</u> is located at a site owned or controlled by the <u>User, the User shall, upon-reasonable notice and subject to any site safety or other rules regarding access, grant-the T&SCo and its agents, subcontractors and representatives access to such <u>Measurement Equipment to inspect (any part of) such Measurement Equipment and obtain samples of the User's CO₂-T&SCo,</u></u>

the relevant site owner or controller must:

- (d) permit the other Party and its agents, subcontractors and representatives (including in the case of the T&SCo, the Independent Verifier) to access any part of the relevant property from time to time in accordance with this paragraph 10; and
- (e) grant the other Party and its agents, subcontractors and representatives (including in the case of the T&SCo, the Independent Verifier) full rights to carry out such tasks and do all such acts as are necessary for the purpose of performing audits, tests, reviews and checks from time to time under this Section F (Network Design and Specification), including full rights to carry out such tests (including to obtain samples of the User's carbon dioxide) on any component of the Measurement Equipment or any such pipeline infrastructure,

provided that:

- (f) reasonable notice is provided in writing to the other Party of any such access required;
- (g) the person responsible for carrying out such tests is or are competent in the operation of such components; and
- the Party requiring access takes reasonable steps to procure that such person at all times observes and performs any such arrangements and provisions (or directions or regulations issued pursuant thereto) as may be made by the other Party in accordance with paragraph 10.3.
- 10.2 <u>The rights and permissions referred to in paragraph 10.1 are:</u>

- (a) <u>full rights to enter upon and through and remain upon the relevant property or do any</u> other act contemplated by this Section F (Network Design and Specification); and
- (b) <u>full rights to undertake on-site tests and checks in relation to the Measurement Equipment and/or pipeline infrastructure and to report on:</u>
 - (i) the Measurement Equipment in relation to its compliance with this Section F (Network Design and Specification); and/or
 - (ii) the pipeline infrastructure in relation to its compliance with this Section F
 (Network Design and Specification), provided that where T&SCo is seeking
 access, such rights are for the purposes of paragraph 4.7 of Section J
 (General),

but in each case only to the extent such rights are necessary for the purposes of this Section F (Network Design and Specification), paragraph 4.7 of Section J (General) and subject to the other provisions of this Section F (Network Design and Specification).

- The Party providing access must use all reasonable endeavours to procure that all reasonable arrangements and provisions are made, and revised from time to time, as and when necessary or desirable to facilitate the safe exercise by the other Party, its agents, subcontractors and representatives (including in the case of T&SCo, the Independent Verifier) of any right of access granted pursuant to paragraphs 10.1 and 10.2 with the minimum of disruption, disturbance and inconvenience to such persons.
- The arrangements and provisions referred to in paragraph 10.3 may, to the extent that the same are reasonable, limit or restrict the exercise of such right of access granted pursuant to paragraphs 10.1 and 10.2 and/or provide for the Party providing access to make directions or regulations from time to time in relation to a specified matter.
- 10.5 Matters to be covered by the arrangements and/or provisions referred to in paragraph 10.3 include:
 - (a) provision of a site safety induction;
 - (b) supply of all necessary personal protective equipment;
 - (c) a method of identifying any relevant component of the Measurement Equipment;
 - (d) the particular access routes applicable to the relevant property having particular regard to the weight and size limits on those routes;
 - (e) any limitations on times of exercise of the right of access;
 - <u>any requirements as to prior notification and as to authorisation or security clearance</u> of individuals exercising the right of access and procedures for obtaining the same;
 - (g) the means of communication by the Party granting access (to all individuals exercising the right of access) of any relevant directions or regulations made by such Party;

- (h) the availability of all site personnel that the individuals exercising the right of access may wish to liaise with during the exercise of the right of access granted pursuant to paragraphs 10.1 and 10.2;
- (i) the identification of any arrangements applicable to the individuals exercising the right of access granted pursuant to paragraphs 10.1 and 10.2; and
- (j) <u>disclosure of any known hazards on the site.</u>
- Subject to paragraph 4.2 of Section J (General), the Party requiring access must take all reasonable steps to procure that all personnel, agents, subcontractors and representatives (including in the case of the T&SCo, the Independent Verifier) of it take all reasonable steps in the exercise of any right of access pursuant to paragraphs 10.1 and 10.2, in order to:
 - (a) avoid or minimise damage in relation to any relevant property; and
 - (b) cause as little disturbance and inconvenience as possible to any third party or other occupier of any relevant property,

and that each person makes good any damage caused to such property in the course of the exercise of such rights as soon as practicable.

- 5.2 If and to the extent that the User's site is not owned by the User, the User shall use its best endeavours to Party to which access is being requested, such Party must ensure that the relevant owner grants T&SCothe other Party a right of access in accordance with paragraph 5.110.1.
- 10.8 <u>If:</u>
 - (a) 5.3 If any Measurement Equipment is found to not comply with the requirements of this Section F, Section F (Network Design and Specification); and/or
 - (b) the T&SCo is prevented from exercising its access rights under this paragraph 10,

the T&SCo can may require the User to immediately cease delivery of carbon dioxide at the User's Delivery Point and the User must immediately commence the procedure to cease delivery of carbon dioxide (as set out in the Isolation Procedure).

- 5.4 The User shallmust, at its own cost and expense, repair or replace (any part of) the Measurement Equipment that is not functioning in accordance with the Measurement Requirements or otherwise in accordance with this Section F (Network Design and Specification).
- The User must, at its own cost and expense, ensure that any pipeline infrastructure owned by a User between the Metering Point and the Delivery Point is maintained and inspected by the User acting as a Reasonable and Prudent Operator, and the User must comply with any reasonable requests from the T&SCo for copies of records of any such maintenance or inspection.
- 10.11 The User must ensure that any carbon dioxide flowing through the Metering Point is subsequently delivered by the User at the Delivery Point, and that no carbon dioxide is

off-taken or added by the User or any other party between the Metering Point and the Delivery Point.

11. Measurement Records

- 11.1 The User must record all Measurement Data (including corrected and estimated data) at the intervals and in the format and otherwise as provided in the Measurement Requirements and the Data Transfer Procedures.
- 11.2 The User must prepare and maintain auditable, complete and accurate records relating to the installation, maintenance, testing, operation, calibration, setting and validation of Measurement Equipment, including:
 - (a) the configuration of flow computers and programmable devices forming part of the Measurement Equipment; and
 - (b) the results of all tests and validations carried out in relation to the Measurement Equipment,

and provide a copy of such records to the T&SCo promptly upon request.

- 11.3 The User must retain all records of Measurement Data for a period of no less than ten (10) years after the last Day to which such Measurement Data relates.
- 11.4 The records required to be maintained and retained under this paragraph 11 shall be stored in electronic format.
- 11.5 Records stored in electronic format must be stored so as:
 - (a) to be capable of immediate access or retrieval within a period of not less than twelve (12) months after the Day to which such records relate; and
 - (b) thereafter, to be retrievable from archive within five (5) Business Days after a request to retrieve such data.
- The User must, as soon as reasonably practicable and without charge, provide to the T&SCo on request a copy of the records maintained in accordance with this paragraph 11 (provided that if the T&SCo requires more than one (1) copy, or a copy on more than one (1) occasion, of records relating to the same matter, the User may charge the T&SCo the costs incurred in providing such additional or further copies).
- 11.7 The T&SCo must be entitled at any time, upon giving five (5) Business Days prior notice to the User, to inspect the records of the Measurement Data for any Day or Days within the preceding twelve (12) months.
- 12. 6. Measurement Equipment Destruction, Damage or Errors
- 6.1 Where a T&SCo or a User believes that there has been a Measurement Equipment Error:
 - (a) that Party shall notify the other Party of the same;

- 12.1 The User must, at its cost, take measures to enable it to identify Measurement Equipment Errors in accordance with the standard of a Reasonable and Prudent Operator.
- 12.2 If at any time any component of the Measurement Equipment is destroyed or damaged or the User identifies a Measurement Equipment Error, the User must:
 - (a) promptly notify the T&SCo thereof and include in its notification all relevant information relating to nature of the destruction, damage or Measurement Equipment Error; and
 - (b) without prejudice to paragraph 12.14, unless otherwise agreed by the T&SCo (acting as a Reasonable and Prudent Operator), immediately commence the procedure to cease delivery of carbon dioxide (as set out in the Isolation Procedure) through the User's Delivery Point until such time as is specified by the T&SCo under paragraph 12.10(a).

12.3 The User must:

- (a) investigate the cause of such destruction, damage or Measurement Equipment Error within two (2) Business Days of becoming aware of the same; and
- (b) as soon as possible provide the T&SCo with all relevant information relating to such cause(s).
- 12.4 If the User employs a third party agent or representative to investigate the relevant component of the Measurement Equipment under paragraph 12.3, the User must procure such investigation takes place within five (5) Business Days of becoming aware of the destruction, damage or Measurement Equipment Error.
- The User must use all reasonable endeavours to rectify the destruction, damage or

 Measurement Equipment Error, including by repairing or replacing any defective component
 so as to ensure that such component is back in service and is operating in accordance with
 the Measurement Requirements as soon as reasonably practicable, and in any event within
 ten (10) Business Days of the date on which the destruction, damage or Measurement
 Equipment Error is discovered by or notified to the User.
- 12.6 If the destruction, damage or Measurement Equipment Error has not been rectified within the required ten (10) Business Day period, the User must notify the T&SCo on the Business Day immediately following such ten (10) Business Day period with a proposal setting out how and the time period within which it intends to rectify the matter.
- 12.7 The User must notify the T&SCo within two (2) Business Days of rectifying the relevant destruction, damage or Measurement Equipment Error.
- 12.8 The T&SCo shall be entitled to attend any investigation of the destruction, damage or

 Measurement Equipment Error and the reasonable costs of such attendance shall be borne
 by the User.
- <u>Where a T&SCo believes that there has been a Measurement Equipment Error (or a Measurement Equipment Error has been identified as part of the validation carried out under</u>

- paragraph 6.1), the T&SCo must notify the User of the nature of such error within one (1)
 Business Day of becoming aware of the same.
- <u>12.10</u> <u>Without prejudice to paragraphs 12.2 to 12.8, where a notification has been made under paragraph 12.2 or 12.9:</u>
 - (a) the T&SCo must, as soon as reasonably practicable and at the latest within (1) hour of being notified of a rectification under paragraph 12.7, notify the User of when the User will be entitled to recommence delivery of carbon dioxide;
 - (b) the Parties shallmust meet to discuss the matter; and
 - (c) the Parties shall seek (if required) consult together in good faith and use their reasonable endeavours to agree any Measurement Adjustment.
- 12.11 6.2 If the Parties are unable to reach agreement under paragraph 6.1 12.10 as to any Measurement Adjustment within thirty (30) Days then, subject to paragraph 7.2 13.2, either Party shall be entitled to have (any part of) the Measurement Equipment inspected by the Independent Verifier, upon reasonable notice to the other Party, and the other Party shall be entitled to have its employees, agents or representatives present during any such inspection.
- 12.12 6.3 The parties shall Parties must accept the result of any inspection by the Independent Verifier as a final determination of any Measurement Equipment Error (including the Independent Verifier's decision regarding any Measurement Adjustment).
- 12.13 6.4 The costs of the inspection inunder paragraph 6.2 12.11 shall be borne by the party Party requiring the inspection, unless the inspection demonstrates that (the relevant part of) the Measurement Equipment was not functioning in accordance with the Measurement Requirements or otherwise in accordance with this Section F (Network Design and Specification), in which case the costs shall be borne by the User.
- 12.14 Where a Measurement Equipment Error has been notified under this paragraph 12, the

 T&SCo may require the User to immediately commence the procedure to cease delivery of

 carbon dioxide (as set out in the Isolation Procedure) at the User's Delivery Point until such

 time as a Measurement Adjustment has been completed.
- 12.15 The T&SCo must notify the User no later than one (1) hour after receiving a notification under paragraph 12.2(a) or 12.9 of its decision to either accept or reject carbon dioxide through the User's Delivery Point in whole or at a partially reduced flow, it being understood that:
 - (a) such acceptance may be subject to certain conditions and will be for the limited period of time specified by the T&SCo; and
 - (b) if the carbon dioxide is accepted in part, the remaining part shall be deemed to have been rejected.
- Any carbon dioxide accepted by the T&SCo under this paragraph 12 shall be deemed to comply with all relevant Entry Provisions and shall not be Non-Compliant CO₂ with effect from the time such carbon dioxide was delivered to the T&S Network, provided that where such acceptance was subject to certain technical or operational conditions, only that part of the

- carbon dioxide stream that complies with such conditions shall be deemed to comply with all relevant Entry Provisions and carbon dioxide which is not compliant with such conditions shall be rejected.
- 12.17 The T&SCo will not be deemed to accept any Non-Compliant CO₂ delivered to a User's

 Delivery Point following a notification under paragraph 12.2(a) or 12.9 where the T&SCo has

 not acknowledged such notification and accepted such carbon dioxide within the required

 period set out in paragraph 12.15.
- 12.18 If and to the extent that the T&SCo has accepted carbon dioxide under paragraph 12.15 (whether or not subject to certain conditions), the User shall be entitled to resume the delivery of carbon dioxide at the Delivery Point to enable the accepted carbon dioxide to enter the T&S Network for the period of time specified by the T&SCo in accordance with paragraph 12.15.

13. **7.** Flow Meter Errors

- 7.1 Where any Measurement Adjustment in respect of a Flow Meter Error is agreed or determined under paragraph 6.1 12.10 or 6.2 12.11, such agreement or determination shallmust set out:
 - (a) the period within which any adjustments to the flow rate measurement are to be made ("Adjustment Period");
 - (b) the amount by which it is agreed or determined that the Flow Meter has over- or under-registered the volume of CO₂quantity of carbon dioxide delivered at the relevant Delivery Point;
 - (c) the adjusted raw data for the Adjustment Period; and
 - (d) the adjusted Daily Quantity for the Adjustment Period.
- 7.2 Where a Flow Meter Error results from a total failure of the Flow Meter such that there is no available Flow Rate data for a period, provided that the Parties have failed to reach agreement as to the Measurement Adjustment to be applied during such period under paragraph 6.1(c)13.1(b), the Measurement Adjustment for the relevant period shall substitute the Flow Rate data for the relevant period with the User's Registered Capacity for that period.
- 13.3 7.3-T&SCo will provide the results of any Flow Meter Adjustment to the CDS for the purposes of paragraph 11.8 of Section H 19.1 of Section H (Charges, Invoicing and Payment), within two (2) Business Days of such Flow Meter Adjustment being determined.

14. 8. Re-use Service

Nature of Re-use Service

- 14.1 8.1 Carbon dioxide re-use (the "Re-use Service") involves aan Onshore User:
 - (a) taking delivery of carbon dioxide from the T&S Network, at a Re-use Delivery Point; and

(b) subject to all other provisions of the Code relating to the User's rights to deliver carbon dioxide at the User's Delivery Point, re-delivering an equivalent quantity of carbon dioxide into the T&S Network at the User's Delivery Point,

where the taking delivery of the carbon dioxide pursuant to paragraph (a) above 14.1(a) is:

- (c) for a Re-use Purpose; and
- (d) only temporary,

on the basis that the User's objective is that any carbon dioxide it has taken delivery of should be returned into the T&S Network in accordance with paragraph (b)14.1(b).

User right to request Re-use Service

- 8.2 An Onshore User (including a prospective Onshore User applying for a connection under Section C (Connection)) may at any time request the T&SCo to provide a Re-use Service to the User.
- 8.3-Where a User wishes to request the T&SCo to provide the Re-use Service, the User shallmust provide the request to the T&SCo in writing, setting out:
 - (a) the technical and operational reasons why it is requesting the Re-use Service-; and
 - (b) the date on which it requires the Re-use Service to commence.

Circumstances where T&SCo shallmust offer the Re-use Service

- 8.4 Where T&SCo has received Within twenty (20) Business Days of receiving a request from a User in accordance with paragraph 8.314.3, the T&SCo shall must notify the User of its initial view as to whether the T&SCo will be able to offer the Re-useRe-Use Service to the.
- Where the T&SCo has notified the User that it is of the view that it is not able to offer the Re-Use Service to the User, the T&SCo must provide reasons to the User.
- 14.6 <u>The T&SCo must offer the Re-Use Service if requested by a User where the T&SCo, acting</u> as a Reasonable and Prudent Operator, considers that:
 - (a) it is technically possible to install the required Re-use Service Infrastructure and provide the Re-use Service to the User; and
 - (b) the installation of the Re-use Service Infrastructure and provision of the Re-use Service to the User will not adversely impact the operation of the T&S Network and/or the ability of other Users to deliver carbon dioxide at their Delivery Points.
- Where the T&SCo has notified the User that it is of the view that it will be able to offer the Re-Use Service to the User, the T&SCo must:
 - (a) advise the User what additional technical information it requires; and

- (b) within twenty (20) Business Days of the date of its notification to the User under paragraph 14.4, make an offer to the User in writing setting out details of:
 - (i) <u>the Re-Use Service Infrastructure and owners of such infrastructure;</u>
 - (ii) the Re-Use Service Infrastructure Works and responsibilities of the Parties for carrying out such works;
 - (iii) the estimated capital cost of the Re-Use Service Infrastructure Works; and
 - (iv) the proposed programme to carry out the Re-Use Infrastructure Works.

Carrying out of Worksworks

- 8.5 Where a User has requested the Re-use Service in accordance with paragraph 8.314.3 and the T&SCo has offered to provide the Re-use Service, then the User shall be responsible for the cost of the Re-use Service Infrastructure and the Re-use Service Infrastructure Works.
- 14.9 8.6-The carrying out of the Re-use Service Infrastructure Works shall be governed by the Construction Agreement and the parties Parties shall agree which parts of the Re-use Service Infrastructure Works shall be:
 - (a) carried out by <u>the T&SCo</u>, and therefore be <u>"T&SCo Works"</u> for the purposes of the Construction Agreement; and
 - (b) carried out by the User, and therefore the "User Works" for the purposes of the Construction Agreement.
- 4.10 8.7 Where any Re-use Service Infrastructure Works are agreed to be "T&SCo Works" for the purposes of the Construction Agreement, then:
 - (a) the Construction Agreement shall include a provision requiring the User to pay for the cost of the T&SCo carrying out such Re-use Service Infrastructure Works; and
- <u>Where the User has requested the provision of the Re-use Service after the construction and commissioning of the Delivery Point to which the Re-user Service relates:</u>
 - (a) the Parties shall enter into a deed of amendment of the User's existing Construction

 Agreement or enter into a further Construction Agreement (as may be appropriate in the circumstances) to govern the construction and commissioning works, as provided for in paragraph 14.9; and
 - (b) the Parties shall enter into a deed of amendment of the User's existing Connection Agreement to document the matters provided for in paragraph 15.3.

15. 9. Terms for provision of the Re-Use Service

Application of Code to the Re-use Service

- 9.1 Where the Re-use Service is being provided to a User, the provisions of the Code shall apply as follows:
 - (a) title and risk (including carbon dioxide compositional risk) in carbon dioxide delivered to the User Facility at a Re-use Delivery Point shall pass to the User at that Re-use Delivery Point upon delivery;
 - the provisions of Section J (General) relating to liabilities between the parties,
 default and suspension shall apply;
 - (c) the provisions of Section F Section F (Network Design and Specification) relating to Measurement Equipment (including the application of the Measurement Requirements, validation of Measurement Equipment and the process for resolving Measurement Equipment Errors) shall apply to the Re-use Meter at the Re-use Delivery Point in the same way as the Flow Meter at the User's Delivery Point;
 - the quantity of carbon dioxide being delivered back to the User at the Re-use Delivery Point shall be measured using the Re-use Meter in accordance with Section F (Network Design and Specification) and that quantity, expressed in tCO₂/Day, is referred to as the "Re-use Daily Quantity";
 - (e) when the Onshore Flow Charge and the Offshore Flow Charge payable by the User are calculated in accordance with Section H, the User's Re-use Daily Quantity shall be subtracted from the User's Daily Quantity;
 - (f) the User shall pay for the operating costs of the provision of the Re-use Service by paying the Re-use Service Charge, which shall be calculated by multiplying the Re-use Service Charge Rate by the User's Re-use Daily Quantity; and
 - (e) (g)-when the User takes delivery of a quantity of carbon dioxide at the Re-use Delivery Point and then re-delivers an equivalent quantity of carbon dioxide at the User's Delivery Point, the User must have sufficient Registered Capacity for that quantity and must ensure that quantity is included in the Daily Nominations and Re-nominations made by the User in accordance with Section E. Section E (Network Use and Capacity);

Re-use Service Charge Rate

9.2 The methodology for calculating the Re-use Service Charge Rate shall be on a on a cost reflective basis (applied consistently to all Users).

Note: the nature of the Re-use Service Charge Rate is subject to further review.

the User will be required to pay the Re-use Service Charge in accordance with Section H (Charges, Invoicing and Payment); and

- (g) the provisions of Section I (Data) relating to the CDS Contract shall apply to enable the CDS to take into account any quantities of carbon dioxide delivered to the User Facility at a Re-use Delivery Point for the purposes of preparing the Processed Flow Meter Data Statement.
- 15.2 The T&SCo and the User must agree on the operational procedures (the "Re-use Service

 Operational Procedures") that are to govern the arrangements for the User to take

 deliveries of carbon dioxide at the Re-use Delivery Point, such as:
 - (a) any restrictions on the quantities of carbon dioxide that the User may take delivery of; and
 - (b) any forecasts and nominations that must be provided by the User to give the T&SCo notice of when quantities of carbon dioxide will be required by the User.

Connection Agreement

- 9.3 Where the Re-use Service is being provided to the User, the following details shall be set out in Schedule 4 Schedule 4 of the User's Connection Agreement:
 - (a) the description of the Re-use Delivery Point, including a schematic showing the location of the Re-use Delivery Point;
 - (b) details of the Re-use Meter;
 - (c) <u>details of the Re-use Service Charge Rate Operational Procedures agreed by the Parties;</u>
 - (d) any details of the size of the Re-use Delivery Point-and any restrictions on how muchcan be taken out for re-use and the process for notifying T&SCo of when the Userintends to take delivery of carbon dioxide; and
 - (e) where any part of the equipment/infrastructure associated with the Re-use Service is located at a site owned or controlled by the T&SCo, provision for the T&SCo to grant, upon reasonable notice and subject to any site safety or other rules regarding access, the User and its agents, subcontractors and representatives access to such equipment/infrastructure to inspect it.

CCS Network Code

Section G Industrial Section G - Common Interface Procedures

1. Key Principles

Key Principles

1.1 <u>Each T&SCo must:</u>

- (a) establish, maintain and from time to time modify the Common Interface Procedures in accordance with this Section G (Common Interface Procedures); and
- (b) ensure that each User at all times has an up-to-date version of such Common Interface Procedures.
- 1.2 <u>The scope of Common Interface Procedures are procedures that:</u>
 - (a) impact both the T&SCo and User(s) across defined site boundaries; and/or
 - (b) require a co-ordinated action or response between the T&SCo and User(s); and/or
 - <u>require the T&SCo and/or User(s) to communicate operational or emergency events,</u> issues or actions that may impact one another.
- 1.3 The Industrial Each T&SCo must ensure that the Common Interface Procedures shall for its applicable T&S Network at all times conform to the following key principles:
 - (a) the Industrial Common Interface Procedures shall be designed to:
 - (i) ensure a uniform approach is taken in relation to procedures that:
 - (A) <u>affect more than one (1) User; and</u>
 - (B) are related to topics which are likely to impact a User's cost base, maintenance or other operational processes;
 - (ii) (ii) ensure safe conduct of all operations;
 - (iii) (iii) protect of minimise any detrimental impact to the environment;
 - (iv) (iii) minimise disruption to the operation of the T&S Network;
 - (v) (iv) meet the needs of both the T&SCo and Users; and
 - (vi) support the <u>safe</u> economic <u>and</u> efficient <u>and effective</u> capture, transport and storage of CO2carbon dioxide;
 - (b) the <u>Industrial Common Interface</u> Procedures <u>shall must</u> comply with <u>applicable law and regulation</u> Legal Requirements;
 - (c) the Industrial Common Interface Procedures shall must be:

- (i) transparent; and
- (ii) non-discriminatory; and as between Users and in accordance with paragraph 6.4 of Standard Condition B6 (Conduct of T&S Business) of the Licence;
- (d) the Industrial Common Interface Procedures shall must be aligned with the aims of the CCUS Network Code and HM Government ambitions for CCUS carbon capture and storage in the UK. United Kingdom; and
- (e) the Common Interface Procedures must be consistent with the provisions of the Code.
- 1.4 <u>For the purposes of this Section G (Common Interface Procedure), references to Users shall</u> be treated as including references to Eligible Applicants.

2. Compliance

2.1 All Users and the T&SCos are required to comply with the Industrial Common Interface

Procedures to be developed in accordance with the provisions of this section Section G

(Common Interface Procedures).

3. List of Procedures

- 3.1 The following procedures shall comprise the Industrial Common Interface Procedures:
 - (a) Emergency Procedure;
 - (b) Commissioning; and Start-up and Shut-down Procedure;
 - (c) Isolation- Procedure; and
 - (d) <u>CO₂ Quality Monitoring Procedure.</u>
- 3.2 The Implementation Date for each Industrial Common Interface Procedure is as follows:
 - (a) Emergency Procedure: Scheduled Commercial Operations Date;
 - (b) Commissioning: HandoverStart-up and Shut-down Procedure: Scheduled Commercial
 Operations Date; and
 - (c) Isolation Procedure: Scheduled Commercial Operations Date-; and
 - (d) CO₂ Quality Monitoring Procedure: the date that is three (3) months after the date that the T&SCo has become a Party to the Code, or such earlier or later date as the T&SCo and the Users on its T&S Network agree, acting reasonably.

4. **Preparation**

Drafting Principles

4.1 Each T&SCo shallmust develop, finalise, maintain and from time to time review, the Industrial and implement the Common Interface Procedures listed in paragraph 3.2 for the applicable T&S Network.

- 4.2 Each T&SCo must engage in sharing best practice with other T&SCos and Prospective

 T&SCos when developing and finalising the Common Interface Procedures in accordance
 with paragraph 4.1.
- 4.3 4.2 In developing the Industrial Common Interface Procedures in accordance with paragraph 4.1, each T&SCo is required to ensure that the Industrial Common Interface Procedures:
 - (a) comply with the key principles in set out in paragraph 1; and
 - (b) comply with the Terms of Reference for each IndustrialCommon Interface Procedure as set out in Annexure F to Annexure H. Annexure D (CO₂ Quality Monitoring

 Procedure Terms of Reference), Annexure F (Emergency Procedure Terms of Reference), Annexure G (Start-up and Shut-down Procedure Terms of Reference)

 and Annexure H (Isolation Procedure Terms of Reference).
- 4.4 4.3 No later than 6eighteen (18) months prior to the Implementation Date for each Industrial Procedure Scheduled Commercial Operations Date, the T&SCo shallmust deliver to:
 - (a) each User; and
 - (b) the AuthorityRegulator,

the <u>first</u> draft <u>Industrial Common Interface</u> Procedures required to be developed pursuant to paragraph <u>3.14.1</u> (other than the CO₂ Quality Monitoring Procedure which shall be delivered <u>in accordance with paragraph 4.5</u>) for review and, <u>in the case of each User</u>, comment in accordance with paragraphs <u>6.15.1</u> to <u>6.65.6</u>.

4.5 As soon as reasonably practicable after the date that the T&SCo has become a Party to the Code, the T&SCo must deliver to:

5. New Industrial Procedures

- 5.1 Where, after the Code Implementation Date, a T&SCo wishes to introduce a further procedure that:
 - (a) is likely to affect more than one each User in the relevant T&S Network;
 - (b) places obligations onto Usersthe Regulator; and
 - (c) is likely to materially impact a User's cost base, maintenance or other operational processes the Secretary,

the first draft CO₂ Quality Monitoring Procedure required to be developed pursuant to paragraph 4.1 for review and, in the case of each User, comment in accordance with paragraphs 5.1 to 5.6.

(in each case, a Proposed Industrial Procedure),

paragraph 6.1 shall also apply to any Proposed Industrial Procedures, save that:

(d) the T&SCo shall also include the proposed Implementation Date;

- (e) the relevant time to submit drafts of any such procedure to Users will be 6 calendar months before the proposed Implementation Date; and
- 5.2 Provided that any comments on the Proposed Industrial Procedure are resolved in accordance with the process set out in paragraph 6.6, the T&SCo which introduced the procedure shall be entitled to raise a Self-Governance Modification Proposal pursuant to paragraph 1.26 of Section B (Governance) to update the list of Industrial Procedures in paragraph 3.1.
- 5.3 A User may, at any time after the Code Implementation Date, request a T&SCo to introduce a Proposed Industrial Procedure under paragraph 5.1 and the T&SCo shall, give reasonable consideration to all such requests.

5. 6. Review Procedure

- 6.1 As soon as reasonably practicable and in any event within twenty (20) Business Days of receipt of any draft Industrial Procedure or Proposed Industrial Common Interface Procedure, each User shallmust notify the T&SCo of any comments it has on the draft Industrial Procedure or Proposed Industrial Common Interface Procedure.
- 5.2 6.2 When providing comments to the T&SCo in accordance with paragraph 6.15.1, each User shall be required to provide the following:
 - (a) <u>information about the User:</u>
 - <u>(i)</u> <u>including but not limited to the company name and the address of its registered</u> <u>office; and</u>
 - (ii) (a) information about the User (company, registered office, etc.) and about at least one reference person (the name, telephone number, fax number, and e-mail address, etc.) (as applicable) of at least one (1) reference person who can be contacted with respect to the User comments;
 - (b) the reasons why the User believes the User comments on the draft Common Interface

 Procedure should be incorporated; and
 - (c) any documentation (<u>such as any applicable</u> analyses, and reports, etc.) to support the request.
- 6.3 Before preparing the final version of any Industrial Procedure or Proposed Industrial any updated draft of a Common Interface Procedure, the T&SCo shallmust convene a meeting with Users in order to discuss any User comments provided in accordance with paragraph 6.15.1, the date of which shallmust be communicated to Users with not less than ten (10) Business Days' notice.
- 5.4 6.4 In preparing the final version of any Industrial Procedure or Proposed Industrial updated draft of any Common Interface Procedure, the T&SCo shallmust:
 - (a) give reasonable consideration to any comments provided by a User pursuant to paragraphs 6.15.1 and 6.35.3; and

- (b) ensure that the draft:
 - <u>provides for and does not have an adverse impact on the safe and efficient operation of the T&S Network or the User Facility; and</u>
 - (ii) balances the commercial interests between the T&SCo and Users with respect to the cost and risk of operation of the T&S Network and User Facility; and
- (c) (b) prepare a report to be circulated to all Users collectively prior to delivery of the final version of the Industrial Procedure or Proposed Industrial updated draft of the Common Interface Procedure pursuant to paragraph 6.65.6 detailing:
 - (i) where the T&SCo has taken on board or incorporated any comments provided by Users pursuant to paragraphs 6.15.1 and 6.35.3; and/or
 - (ii) where the T&SCo does not take on board or incorporate any comments provided by a User pursuant to paragraphs 6.15.1 and 6.35.3, an explanation to the User of the reasons for this decision; and
 - <u>(iii)</u> an explanation of the criteria taken into account in developing the updated draft Common Interface Procedure.
- 6.5 For the avoidance of doubt, it the shall be reasonable for the T&SCo to not incorporate comments made by a User in relation to a draft Industrial Procedure or Proposed Industrial Common Interface Procedure where such comments, if incorporated, would affect the safe operation of the T&S Network.
- 5.6 Subject to paragraph 8.2, once 7.2, within twenty (20) Business Days of all comments have having been received and considered pursuant to the process set out in paragraphs
 6.15.1 to 6.55.5, the T&SCo shall must deliver to:
 - (a) each User; and
 - (b) the Authority Regulator,

the updated draft of the Common Interface Procedure.

- 5.7 Subject to paragraph 5.8 and 5.9, the T&SCo may review and/or revise any draft Common Interface Procedure previously issued to Users and the Regulator:
 - (a) when the T&SCo considers it is necessary to do so;
 - (b) upon request by a User; or
 - (c) <u>in order to resolve a dispute under paragraph 7,</u>

provided that any such revision is in accordance with the procedure set out in this paragraph 5.

5.8 No later than six (6) months prior to the Scheduled Commercial Operations Date, the T&SCo must finalise and deliver to:

- (a) each User; and
- (b) the Regulator,

the final version of each Common Interface Procedure (other than the CO₂ Quality Monitoring Procedure which shall be delivered in accordance with paragraph 5.9), along with confirmation of the Implementation Date.

- 5.9 No later than three (3) months after the date that the T&SCo has become a Party to the Code or such earlier or later date as the T&SCo and the Users on its T&S Network agree, acting reasonably, the T&SCo must finalise and deliver to:
 - (a) <u>each User;</u>
 - (b) the Regulator; and
 - (c) the Secretary,

the final version of the Industrial Procedure or Proposed Industrial CO₂ Quality Monitoring Procedure, as the case may be, along with the confirmation of the Implementation Date.

- 5.10 Subject to paragraph 7.2, once the final version of the CO₂ Quality Monitoring Procedure has been delivered in accordance with paragraph 5.9:
 - (a) the Code shall be modified by insertion of the final version of the CO₂ Quality

 Monitoring Procedure as a new section of the part of the Annexure C Measurement

 Requirements that are applicable to the T&S Network to which the CO₂ Quality

 Monitoring Procedure relates and such Modification shall not be subject to the

 Modification Rules; and
 - (b) the Secretary must publish such final version by notice on the CCS Network Code Website.
- 6. **7.** Revisions
- 6.1 7.1 The T&SCo may review and/or revise the Industrial Procedures final version of any
 Common Interface Procedure (other than the CO2 Quality Monitoring Procedure which shall only be modified in accordance with paragraph 6.4):
 - (a) when the T&SCo considers it is necessary to do so;
 - (b) upon request by a User; or
 - (c) in order to resolve a dispute under paragraph 87,
 - 7.2 provided that any such revision shall be subject to is in accordance with the procedure set out in paragraph 65, and subject always to paragraph 7.26.2.
- 6.2 The T&SCo shall not be required to follow the procedure set out in paragraph <u>5</u> in relation to revisions to <u>IndustrialCommon Interface</u> Procedures where any such revision <u>relates to the CO₂ Quality Monitoring Procedure or is:</u>

- (a) of a routine nature;
- (b) does not place new obligations onto Users; and
- (c) does not have a material impact on a User's cost base, maintenance or other operational processes.
- 7.3-Revisions or additions to the list of Industrial Common Interface Procedures set out in paragraph 3.1, shall be addressed as a Self-Governance (and any associated terms of reference) (as distinct from revisions to the text of the Common Interface Procedures themselves, other than the CO₂ Quality Monitoring Procedure), shall be raised as a Modification Proposal via pursuant to the Modification Procedure in Section B set out in Section B (Governance).
- 6.4 Revisions to the CO₂ Quality Monitoring Procedure shall be raised as a Modification Proposal pursuant to the Modification Procedure set out in Section B (Governance).
- Where the T&SCo revises a Common Interface Procedure in accordance with paragraph 6.2, it must deliver to the Regulator and each User the final version of the revised Common Interface Procedure.
- 7.4-Where a majority of Users consider that a revision made by the T&SCo in accordance with paragraph 7.16.1 should not have been made then, such matter may be referred by any one of the Users who considers that the revision should not have been made to dispute resolution in accordance with the provisions of paragraph 8 of Section B (Governance), including to an Expert for determination in accordance with the provisions of paragraph 51 of Section B (Governance).

7. 8. Disputes

7.1 **8.1** Where:

- (a) a User comment provided pursuant to paragraphs 6.1 and 6.2 of this section 5.1

 and/or 5.3 of this Section G (Common Interface Procedures) was not included by the T&SCo in the final Supplementary Industrial version of any Common Interface

 Procedure; and a majority of Users consider that such comment should have been incorporated by the T&SCo; or
- (b) a majority of Users considers that such comment should have been incorporated consider that the final Common Interface Procedure delivered by the T&SCo in accordance with paragraph 5.8 or 5.9 does not reflect the principles set out in paragraph 5.4(b),

then such matter may be referred by the any User who made the original comment considers that such comment should have been incorporated by the T&SCo or that the principles set out in paragraph 5.4(b) have not been reflected to dispute resolution in accordance with the provisions of Section B (Governance), including to an Expert for determination in accordance with paragraph 8 of Section B51 of Section B (Governance).

- 7.2 8.2-Where any matter is referred to dispute resolution under paragraph 8.16.6 or 7.1, the T&SCo shallmust not finalise and implement the Supplementary IndustrialCommon Interface Procedure until such dispute has been resolved in accordance with the provisions of paragraph 8 of Section B Section B (Governance).
- 7.3 The Parties acknowledge and agree that notwithstanding paragraph 17.1 of Section J
 (General), the Contracts (Rights of Third Parties) Act 1999 shall apply to the rights conferred
 on Eligible Applicants and Prospective T&SCos under this Section G (Common Interface
 Procedure).
- 8. CCS Network Code provisions to prevail
- 8.1 In the event that any Common Interface Procedure is inconsistent with the provisions of the Code (notwithstanding paragraph 1.3(e)), the provisions of the Code shall prevail, subject to, in all cases, the rights and obligations of T&SCo to prevent imminent danger to the health and safety of any persons or imminent and avoidable physical damage or environmental harm.

CCS Network Code

Section H – Charges, Invoicing and Payment

Part 1 – Use of System Charging Methodology

- 1. <u>Introduction to Use of System Charging Methodology</u>
- 1.1 Part 1 of this Section H (Charges, Invoicing and Payment) sets out the Use of System
 Charging Methodology that is required to be maintained by each T&SCo pursuant to Special
 Condition H25 (Use of System Charging Methodology) of the Licence, for determining Use of
 System Charges payable by Users under the Code.
- 1.2 The Use of System Charges are intended, subject to the Mutualisation Cap, to recover the Allowed Revenue of the T&SCo, as determined by the Regulator in accordance with the conditions of the T&SCo's Licence.
- 1.3 Any costs incurred by the T&SCo relating to the connection of new Users to the T&S

 Network, or relating to the modification of existing Delivery Points, are also to be recovered through the Use of System Charges, instead of separate connection charges.
- 2. 4. Structure of Use of System Charges

Payment of charges

1.1 This Section H sets out the Charging Methodology deemed established by the T&SCos pursuant to the Licence for determining the Charges.

Note: any changes to the Charging Methodology will need to be made in accordance with the Modification Rules in Section B.

- 1.2 The User agrees to pay to T&SCo the Charges calculated in accordance with this Section H.
- 2.1 1.3 The "Use of System Charges shall be calculated on" comprise the basis of following:
 - (a) the Onshore Flow Charge, calculated using the rate determined in accordance with paragraph 2.110.1;
 - (b) the Offshore Flow Charge, calculated using the rate determined in accordance with paragraph2.2 11.1;
 - (c) the Re-use Service Charge, calculated using the rate determined in accordance with paragraph 12.1;
 - (d) (e) the Onshore Capacity Charge, calculated using the rate determined in accordance with paragraph 3.113.1;
 - (e) (d) the Offshore Capacity Charge, calculated using the rate determined in accordance with paragraph 3.214.1;

- (f) (e) the Onshore Network Charge, calculated using the rate determined in accordance with paragraph 4.115.1; and
- (g) (f) the Offshore Network Charge, calculated using the rate determined in accordance with paragraph 4.516.1.

3. Payment of Use of System Charges

- 3.1 Each User agrees to pay to the T&SCo the Use of System Charges calculated in accordance with this Section H (Charges, Invoicing and Payment).
- 3.2 A User is not liable to pay Use of System Charges until the Commencement Date (as defined in the User's Connection Agreement), which is the date when commissioning has been completed in accordance with the Construction Agreement.
- 3.3 Where a User's Commencement Date or User Discontinuance Date arises part way through a Charging Year, the Use of System Charges payable by the User must be adjusted on a prorata basis.

4. Flow Charges

- 4.1 4.4 A User shallmust pay to deliver carbon dioxide to the T&S Network each Day, at the User's Delivery Point, on the following basis:
 - (a) an Onshore User shallmust pay (in relation to each Day of delivery):
 - (i) the Onshore Flow Charge Rate multiplied by the User's Daily Quantity for that Day (the "Onshore Flow Charge"); and
 - (ii) the Offshore Flow Charge Rate multiplied by the User's Daily Quantity for that Day (the "Offshore Flow Charge"); and
 - (b) an Offshore User shallmust pay (in relation to each Day) the Offshore Flow Charge Rate multiplied by the User's Daily Quantity for that Day (the "Offshore Flow Charge").
- 4.2 4.5 A User's **Daily Quantity** is the quantity of carbon dioxide that the User delivers each Day at the User's Delivery Point expressed in tCO₂:
 - (a) less the User's Re-use Daily Quantity (if any);
 - (b) (a) as measured in accordance with Section F Section F (Network design Design and specification); and
 - (c) (b) as set out in the Processed Flow Meter Data Statement prepared by the CDS for each Billing Period.

5. **Re-use Service Charge**

5.1 Where an Onshore User is being provided with the Re-use Service by the T&SCo, in accordance with the provisions of Section F (Network Design and Specification), the User

must pay to take delivery of carbon dioxide from the T&S Network, at the User's Re-use Delivery Point, on the following basis:

- (a) the Onshore User must pay (in relation to each Day of delivery), the Re-use Service

 Charge Rate multiplied by the Onshore User's Re-use Daily Quantity for that Day (the

 "Re-use Service Charge"); and
- (b) the Onshore User's "Re-use Daily Quantity" is the quantity of carbon dioxide being delivered back to the Onshore User at the User's Re-use Delivery Point expressed in tCO₂:
 - <u>(i)</u> <u>as measured in accordance with Section F (Network Design and Specification);</u> and
 - (ii) as set out in the Processed Flow Meter Data Statement prepared by the CDS for each Billing Period.

6. Capacity Charges

- 6.1 4.6-A User shallmust pay for its Registered Capacity on the following basis:
 - (a) an Onshore User shallmust pay:
 - (i) the Onshore Capacity Charge Rate multiplied by the User's Registered Capacity (the "Onshore Capacity Charge"); and
 - (ii) the Offshore Capacity Charge Rate multiplied by the User's Registered Capacity (the "Offshore Capacity Charge"); and
 - (b) an Offshore User shall must pay the Offshore Capacity Charge Rate multiplied by the User's Registered Capacity (the "Offshore Capacity Charge").
- 1.7 A User's Registered Capacity is as defined in Section E (Network use and capacity allocation).
- 6.2 1.8 Unless otherwise provided in this the Code or agreed, the a User must pay Capacity Charges shall be payable by a User irrespective of whether the User utilises its Registered Capacity by delivering carbon dioxide at its Delivery Point.

Note: send or pay principle applies to Capacity Charges but not Flow Charges.

1.9 A new User connecting to the T&S Network shall not be liable to pay for its Registered Capacity until Commissioning has been completed in accordance with its Construction Agreement.

7. Network Charges

- 7.1 4.10 A User shallmust pay charges in respect of certain residual amounts of the proportion of the Allowed Revenue which is recoverable from the User, but which is not recovered under the Flow Charges or Capacity Charges, on the following basis:
 - (a) an Onshore User shallmust pay:

- (i) the Onshore Network Charge Rate multiplied by the User's Delivery Point Size ("Onshore Network Charge"); and
- (ii) the Offshore Network Charge Rate multiplied by the User's Delivery Point Size ("Offshore Network Charge"); and
- (b) an Offshore User shallmust pay the Offshore Network Charge Rate multiplied by the User's Delivery Point Size (the Offshore Network Charge).
- 7.2 A "User's Onshore Proportionate Network Charge" is the Onshore Proportionate Network

 Charge Rate multiplied by the User's Delivery Point Size.
- 7.3 A "User's Offshore Proportionate Network Charge" is the Offshore Proportionate Network Charge Rate multiplied by the User's Delivery Point Size.
- 1.11 A User's "Delivery Point Size" is, in relation to a User's Delivery Point, the maximum possible throughput of carbon dioxide at that Delivery Point, based on the for which the T&SCo's installed Delivery Point assets have been designed, as set out in the User's Connection Agreement on the date it becomes effective or as amended following acceptance of a separate Connection Offer relating to modification of the User's Delivery Point in accordance with Section C (Connection) through a subsequent Selection Process, expressed in tCO₂/hour.
- 7.5 Unless otherwise provided in the Code, a User must pay Network Charges irrespective of whether the User utilises its Registered Capacity by delivering carbon dioxide at its Delivery Point.
- 8. Charges relating to the Storage ComplexSite
- 8.1 1.12 For the avoidance of doubt, the The Offshore Capacity Charge, the Offshore Flow Charge and the Offshore Network Charge cover any costs relating to the Storage ComplexSite.
- 9. Onshore Users and Offshore Users
- 9.1 1.13 When a prospective User applies for a connection under Section C (Connection), the T&SCo shallmust determine, in accordance with the Code, whether the User is an Onshore User or an Offshore User, and this shallmust be recorded in the User's Connection Agreement.
- 2. Charging Methodology Flow Charges
- 10. Calculation of Onshore Flow Charge Rate
- 10.1 2.1 The "Onshore Flow Charge Rate" for a Charging Year t is calculated as follows:

 $\frac{AOnFR_t}{TFOnF_t}$

where:

(a) "AOnFR_t" is the <u>"Allowed Onshore Flow Revenue</u> being the amount allowed for Variable Opex for the Onshore Transportation System within the determination of the Allowed Revenue <u>by the Regulator</u> under the Licence for <u>the-Charging Year t</u>; and

Note: these costs are intended to be those costs that reflect the operational costs imposed by a User in using the Onshore Transportation System to transport and store each unit of carbon dioxide.

(b) "TFOnF_t" is the total forecast flow of carbon dioxide to be delivered to and transported through the Onshore Transportation System in the Charging Year t using the forecasts-provided by Users in accordance with paragraph 5 Charges Forecast for Charging Year t,

expressed as £[x]/tCO2.

- 11. <u>Calculation of Offshore Flow Charge Rate</u>
- 11.1 2.2 The "Offshore Flow Charge Rate" for a Charging Year t is calculated as follows:

 $\frac{AOfFR_t}{TFOfF_t}$

where:

(a) "AOfFR_t" is the <u>"Allowed Offshore Flow Revenue</u> being the amount allowed for Variable Opex for the Offshore Transportation and Storage System within the determination of the Allowed Revenue <u>by the Regulator</u> under the Licence for the Charging Year t; and

Note: these costs are intended to be those costs that reflect the operational costs imposed by a User in using the Offshore Transportation and Storage System to transport and store each unit of carbon dioxide.

(b) "TFOfF_t" is the total forecast flow of carbon dioxide to be delivered to and transported through the Offshore Transportation and Storage System in the Charging Year tusing the forecasts provided by Users in accordance with paragraph 5 Charges Forecast for Charging Year t,

expressed as £[x]/tCO2.

3. Charging Methodology - Capacity Charges

- 12. <u>Calculation of the Re-use Service Charge Rate</u>
- 12.1 The "Re-use Service Charge Rate" for Charging Year t is calculated as follows:

 $\frac{(\text{Add})SRt_t}{TFReUSF_t}$

where:

- (a) "AReUSRt" is the "Allowed Re-use Service Revenue" being the amount allowed for the Re-use Service Opex within the determination of the Allowed Revenue by the Regulator under the Licence for Charging Year t;
- (b) "Re-use Service Opex" means that part of the Opex (as that term is defined in the Licence) which relates to the provision of the Re-use Service by the T&SCo; and
- (c) "TFReUSF_t" is the total forecast flow of carbon dioxide to be taken from the Onshore
 Transportation System by Users using the Re-use Service in Charging Year t, using
 the Charges Forecast for Charging Year t,

expressed as £[x]/tCO2.

13. <u>Calculation of Onshore Capacity Charge Rate</u>

3.1 The "Onshore Capacity Charge Rate" for a Charging Year t is calculated as follows:

(Mod)R_t

where:

(a) "AOnCRt" is the <u>"Allowed Onshore Capacity Revenue</u> being the amount allowed for Depreciation <u>Building Block</u> and WACC for the Onshore Transportation System within the determination of the Allowed Revenue <u>under the by the Regulator in accordance</u> <u>with the T&SCo's</u> Licence for <u>the Charging Year t</u>; and

Note: these costs are intended to be those capital costs that reflect the costs imposed by a User in reserving capacity in the Onshore Transportation System.

- (b) "MRCEOn;" is the capacity of the Most Restricted Communal Element of OnRNC;" is the relevant Network Capacity for the T&S Network for the relevant Charging Year expressed as tCO2/hour, t, which shall be the greater of:
 - (i) the ONC Maximum Annual Cumulative Flow divided by the number of hours in Charging Year t, expressed as tCO₂/hour; and
 - (ii) the aggregate Registered Capacity of all Onshore Users for Charging Year t, expressed as tCO₂/hour,

expressed as £[x]/tCO₂/hour.

14. <u>Calculation of Offshore Capacity Charge Rate</u>

3.2-The "Offshore Capacity Charge Rate" for a-Charging Year t is calculated as follows:

 $\frac{(\text{Mod})R_{t}}{\textit{MRCEO}_{t}f_{t}}$

where:

(a) "AOfCRt" is the <u>"Allowed Offshore Capacity Revenue</u> being the amount allowed for the Depreciation <u>Building Block</u> and WACC for the Offshore Transportation and Storage System within the determination of the Allowed Revenue under the by the Regulator in accordance with the T&SCo's Licence for the Charging Year t; and

Note: these costs are intended to be those capital costs that reflect the costs imposed by a User in reserving capacity in the Offshore Transportation and Storage System.

- (b) "OfRNC_t" is the relevant Network Capacity for the T&S Network for Charging Year t, which shall be the greater of:
 - (i) the ONC Maximum Annual Cumulative Flow divided by the number of hours in Charging Year t, expressed as tCO₂/hour; and
 - (ii) (b) "MRCEOf_t" is the capacity of the Most Restricted Communal Element of the Offshore Transportation and Storage System for the relevant the aggregate Registered Capacity of all Users for Charging Year t, expressed as tCO₂/hour,

expressed as £[x]/tCO₂/hour.

4. Charging Methodology - Network Charges

Note: the Network Charge is proposed to collect the remainder of a User's share of Allowed Revenue provided such charge does not exceed the relevant caps.

- 15. Calculation of Onshore Network Charge Rate
- 15.1 4.1 The **"Onshore Network Charge Rate"** for **a** Charging Year **t** is calculated as follows:

 $OnPNCR_t + OnMNCR_t$

where:

- (a) "OnPNCR_t" is the **"Onshore Proportionate Network Charge Rate"** for the Charging Year t, calculated in accordance with paragraph 4.215.2; and
- (b) "OnMNCR_t" is the **"Onshore Mutualised Network Charge Rate**" for the Charging Year t_calculated in accordance with paragraph 4.3 paragraphs 15.3 to 15.13.

Calculation of Onshore Proportionate Network Charge Rate

4.2 The **"Onshore Proportionate Network Charge Rate"** for Charging Year t is calculated as follows:

 $\frac{(\text{Mod})CR_t}{\textit{OmRIDP}_t}$

where:

(a) <u>"AOnNCRt"</u> is the Allowed Onshore Network Charge Revenue for the Charging Year beingt, calculated as follows:

(Mod)ed Onshore Revenue for Ather Charging & GRE-ADORR & EAOnCRt

where:

- (i) the <u>"Allowed Onshore Revenue for the Charging Year"</u> is the amount of Allowed Revenue for the Onshore Transportation System <u>for Charging Year t</u> as determined by the <u>AuthorityRegulator</u> in accordance with <u>the T&SCo's Licence</u>;
- (ii) "AOnFR_t" <u>is the Allowed Onshore Flow Revenue and</u> has the meaning given in paragraph 2.1(a)10.1(a);
- (iii) "AReUSRt" is the Allowed Re-use Service Revenue and has the meaning given in paragraph 12.1(a); and
- (iii) "EAOnCRtAOnCRt" is the amount of Allowed Onshore Capacity Revenue expected to be recovered from Users on the basis of total Registered Capacity calculated as follows:and has the meaning given in paragraph 13.1(a); and
- (b) "OnRDP_t" is the relevant Network Capacity for the T&S Network for Charging Year t, which shall be the greater of:
 - (i) the ONC Maximum Annual Cumulative Flow divided by the number of hours in Charging Year t, expressed as tCO₂/hour; and
 - <u>(ii)</u> the aggregate of the Delivery Point Sizes of all Delivery Points to the Onshore Transportation System, expressed as tCO₂/hour,

expressed as £[x]/tCO₂/hour.

Onshore T&S Network Mutualisation Check

- 15.3 The calculation of the Onshore Mutualised Network Charge Rate for each User is subject to an iterative process which is set out in paragraphs 15.4 to 15.13 (inclusive).
- The "Onshore T&S Charge" for the T&S Network which will be recovered from all Onshore

 Users of the T&S Network through the application of the Onshore Flow Charge, the Re-use

 Service Charge, the Onshore Capacity Charge and the User's Onshore Proportionate

 Network Charge for each User (OnRt) for Charging Year t is calculated as follows:

 $(Mod)t * REOGRC_t + OnCCt + OnPNCt$

where:

- (a) "OnFCt" is the total of all Users' forecast Onshore Flow Charges;
- (b) "ReUSCt" is the total of all Users' forecast Re-use Service Charges;
- (c) "OnCCt" is the total of all Users' Onshore Capacity Charges; and
- (d) "OnPNCt" is the total of all Users' Onshore Proportionate Network Charges,

using each Users' Charges Forecast for Charging Year t and expressed as £[x]/tCO2.

15.5 If the Onshore T&S Charge is less than the Allowed Onshore Revenue, the T&SCo must carry out mutualisation, starting with calculation of the Onshore Network Charges Cap in

accordance with paragraph 15.6, followed by calculation of the Onshore User T&S Charge in accordance with paragraph 15.7 and calculation of the Provisional Onshore Mutualised Network Charge in accordance with paragraph 15.9.

Onshore Network Charges Cap Calculation

- 15.6 The "Onshore Network Charges Cap" is equal to the Onshore Proportion of the Mutualisation Cap, where:
 - (a) "Onshore Proportion", expressed as a percentage, is calculated as follows:

 $\frac{\text{(Add)'ed Onshore Revenue for Charging Year } t}{\textit{Allowed Revenue for Charging Year } t} \times 100$

where the Allowed Onshore Revenue is the amount of Allowed Revenue for the

Onshore Transportation System as determined by the Regulator in accordance with
the T&SCo's Licence; and

(b) "Mutualisation Cap" for Charging Year t means a cap, expressed in £[x]/tCO₂, equal to the average of the daily average price observed, as at 31 October of Charging Year t-1, for the March t-1, December t, and March t UK Emissions Trading Scheme futures contract (UKA Futures Contract), published by ICE Futures Europe, over the preceding three (3) month period.

Onshore User T&S Charge Calculation

15.7 The "Onshore User T&S Charge" for each User for Charging Year t is calculated as follows:

 $\frac{(\mathrm{Add})Rt}{UFOnFt}$

where:

(a) "UonRt" is the total revenue for Charging Year t which will be recovered from the User through the application of the Onshore Flow Charge, the Re-Use Service Charge, the Onshore Capacity Charge and the User's Onshore Proportionate Network Charge, to be calculated as follows:

(Add)Ct + UReUSCt + UOnCCt + UOnPNCt

where:

- (i) (A)-"OnCctUOnFCt" is the <u>User's forecast</u> Onshore Capacity Charge Rate for the Charging Year calculated in accordance with paragraph 3.1 Flow Charge; and
- (ii) "UReUSCt" is the User's forecast Re-use Service Charge;
- (iii) "UOnCCt" is the User's Onshore Capacity Charge; and

(iv) (B) "TFOnRC_t" is the total Registered Capacity in the UOnPNCt" is the User's Onshore Transportation System in the Charging Year; and Proportionate Network Charge,

calculated using the User's Charges Forecast for Charging Year t; and

(b) "UFOnFt" is a User's forecast flow of carbon dioxide to be delivered by the User to and transported through the Onshore Transportation System, calculated using that Users' Charges Forecast for Charging Year t,

Note: the difference in fees on the basis of actual Registered Capacity and the fees actually recovered will form part of the adjustments to determine the Allowed Revenue in Year t+2 pursuant to T&SCo's Licence.

Note: the Capacity Charges for Year t will be calculated on the basis of actual Registered Capacity. Because capacity will only be allocated during annual Capacity Application Windows, it will not need to be forecast for Year t.

(b) "OnNDP" is the aggregate of the Delivery Point Sizes of Delivery Points to the Onshore
Transportation System that the Onshore Transportation System can accommodate,
where the aggregate of the Delivery Point Sizes of Delivery Points to the Onshore
Transportation System that the Onshore Transportation System can accommodate is
equal to MRCEOnt when there are no interruptible or flexible connection agreements,

expressed as £[x]/tCO₂/hour.

Onshore User T&S Charge Check

- 15.8 If the Onshore Network Charges Cap for Charging Year t:
 - <u>is below the Onshore User T&S Charge for a User, the Onshore Mutualised Network</u>

 Charge Rate shall be deemed to be zero (0) for that User; or
 - (b) is above the Onshore User T&S Charge for a User, the Onshore User's Mutualised

 Network Charge Rate shall be calculated in accordance with paragraphs 15.9 to

 15.11.

Calculation of Provisional Onshore Mutualised Network Charge Rate Calculation

4.3-The <u>"Provisional Onshore Mutualised Network Charge Rate"</u> for a Charging Year <u>t</u> is the lesser of the Onshore Network Charges Cap and the amount calculated as follows:

 $\frac{(\text{Mod}) \textit{NER}_t}{\textit{OOHDBO}_t}$

where:

(a) "UAOnNCRtUAOnRt" is the Allowed Onshore Network Charge Revenue for the Charging Year which will not be recovered through the application of the Onshore

Flow Charge, the Re-use Service Charge and the revenue from the Onshore Capacity

Charge and User's Onshore Proportionate Network Charge Rate that the T&SCo

expects to recover from Users, to be calculated as follows:

 $(\text{Mod}) CR_t AO (ORRENCAR) OUR UDR (OnCRET x TFO) - (OnPNCRT x OnUDPt)$

where:

- (i) "AOnR_t" is the Allowed Onshore Revenue and has the meaning given in paragraph 15.2(a);
- (ii) "AOnFRt" is the Allowed Onshore Flow Revenue and has the meaning given in paragraph 10.1(a):
- (iii) "AReUSR_t" is the Allowed Re-use Service Revenue and has the meaning given in paragraph 12.1(a);
- (iv) (i) "AOnNCR_tOnCCRt" is the Allowed Onshore Network Charge Revenue Capacity Charge Rate for Charging Year t calculated in accordance with paragraph 4.2(a)13.1;
- (v) "TFOnRCt" is the total Registered Capacity in the Onshore Transportation System in Charging Year t;
- (vi) (ii) "OnPNCR_t" is the Onshore Proportionate Network Charge Rate calculated in accordance with paragraph 4.215.2 (or Onshore Network Charge Rate, as applicable in subsequent rounds of mutualisation); and
- (vii) "OnUDPt" is the aggregate of the Delivery Point Sizes of Onshore Users'Delivery Points to for which the Onshore Transportation System in the
 Charging Year, Network Charges Cap is above each of those Users' Onshore
 User T&S Charge (or Updated Onshore User T&S Charge, as applicable in subsequent rounds of mutualisation); and
- (b) "OnUDP_t" has the meaning given to it in paragraph 15.9(a)(vii),

expressed as £[x]/tCO₂/hour.

Updated Onshore User T&S Charge Calculation

After the first round of mutualisation is carried out in accordance with paragraphs 15.8(b) and 15.9, the T&SCo must re-calculate the Onshore User T&S Charge for each relevant User to include the Provisional Onshore Mutualised Network Charge Rate that was calculated in accordance with paragraph 15.9 (the "Updated Onshore User T&S Charge"), to be calculated as follows:

 $\frac{(Add)nRt}{UFOnFt}$

where:

(a) "UMOnRt" is the onshore revenue for Charging Year t which will be recovered from the User through the application of the Onshore Flow Charge, the Re-Use Service

<u>Charge, the Onshore Capacity Charge and the Onshore Network Charge, to be calculated as follows:</u>

(Add)Ct + UReUSCt + UOnCCt + UOnNCt

where:

- (i) "UOnFCt" is the User's forecast Onshore Flow Charge;
- (ii) <u>"UReUSCt" is the User's forecast Re-use Service Charge;</u>
- (iii) "UOnCCt" is the User's Onshore Capacity Charge; and
- (iv) <u>"UOnNCt" is the User's Onshore Network Charge,</u>

calculated using the User's Charges Forecast for Charging Year t; and

(b) "UFOnFt" is a User's forecast flow of carbon dioxide to be delivered by the User to and transported through the Onshore Transportation System, calculated using that User's Charges Forecast for Charging Year t,

expressed as £[x]/tCO2.

Onshore Mutualised Network Charge Rate Calculation

- 15.11 If the Provisional Onshore Mutualised Network Charge Rate for a User calculated in accordance with paragraph 15.9, results in the Updated Onshore User T&S Charge calculated in accordance with paragraph 15.10 for that User:
 - (a) exceeding the Onshore Network Charges Cap, the Onshore Mutualised Network

 Charge Rate for that User will be set at a level that results in the Onshore User T&S

 Charge equalling the Onshore Network Charges Cap; or
 - (b) not exceeding the Onshore Network Charges Cap, the Onshore Mutualised Network
 Charge Rate for that User will be set at the Provisional Onshore Mutualised Network
 Charge Rate.

T&S Network Mutualisation – Further Rounds

- After the first round of mutualisation, the T&SCo must re-calculate the Onshore T&S Charge to include the Updated Onshore User T&S Charge for each relevant User (the "Updated Onshore T&S Charge").
- 15.13 <u>If, after the first round of mutualisation:</u>
 - (a) the Allowed Onshore Revenue less the Updated Onshore T&S Charge is zero (£0/tCO₂), no further mutualisation is required; or
 - (b) <u>the:</u>
 - (i) Allowed Onshore Revenue less the Updated Onshore T&S Charge is above zero (£0/tCO₂); and

(ii) Updated Onshore User T&S Charge remains below the Onshore Network
Charges Cap for at least one (1) User,

the T&SCo must proceed to repeat the process for Users with an Updated Onshore

<u>User T&S Charge that remains below the Onshore Network Charges Cap at</u>

paragraphs 15.7 to 15.13 in iterative rounds until the earlier of:

- (A) the Allowed Onshore Revenue is recovered; or
- (B) 4.4 Theall Users' Onshore Network User T&S Charges Cap is are equal to the Onshore Proportion of the Mutualisation Network Charges Cap, where:
- (a) Onshere Proportion, expressed as a percentage, is calculated as follows:

 $\frac{\text{(Del)}\textit{ved Onshore Revenue for Charging Year }t}{\textit{Allowed Revenue for Charging Year }t} \times 100$

where the **Allowed Onshore Revenue** for the Charging Year is the amount of Allowed Revenue for the Onshore Transportation System as determined by the Authority in accordance with T&SCo's Licence; and

- (b) Mutualisation Cap for Charging Year t means a cap, expressed in £[x]/tCO₂/hour, equal-to the average of the daily average price observed for the March t-1, December t, and March t-UK Emissions Trading Scheme futures contract (UKA Futures Contract), published by ICE Futures Europe, over the preceding three-month period.
- 16. Calculation of Offshore Network Charge Rate
- 16.1 4.5 The "Offshore Network Charge Rate" for a Charging Year t is calculated as follows:

 $OfPNCR_t + OfMNCR_t$

where:

- (a) "OfPNCR_t" is the <u>"Offshore Proportionate Network Charge Rate</u> for <u>the-Charging</u> Year <u>t, calculated in accordance with paragraph 4.616.2</u>; and
- (b) "OfMNCR_t" is the <u>"Offshore Mutualised Network Charge Rate</u> for the Charging Year t, calculated in accordance with <u>paragraph4.7</u> paragraphs 16.3 to 16.13.
- 16.2 4.6 Calculation of The "Offshore Proportionate Network Charge Rate" for Charging Year t is calculated as follows:

 $\frac{(\text{Mod})CR_t}{\text{OffRIDH}t}$

where:

(a) "AOfNCR_t" is the Allowed Offshore Network Charge Revenue for the Charging Year beingt, calculated as follows:

 $(Mod)ed\ Off shore\ Revenue\ for Alle Ellargin Off all the AOFR_t-EAOFCR_t$

where:

- (i) the <u>"Allowed Offshore Revenue for the Charging Year"</u> is the amount of Allowed Revenue for the Offshore Transportation and Storage System <u>for</u> <u>Charging Year t</u> as determined by the <u>AuthorityRegulator</u> in accordance with the T&SCo's Licence;
- (ii) "AOfFR_t" <u>is the Allowed Offshore Flow Revenue and</u> has the meaning given in paragraph 2.2(a)11.1(a);and
- (iii) "EAOfCR_tAOfCR_t" is the amount of Allowed Offshore Capacity Revenue expected to be recovered from Users on the basis of total forecast Registered Capacity calculated as follows: and has the meaning given in paragraph 14.1(a); and
- (b) "OfRDPt" is the relevant Network Capacity for the T&S Network for Charging Year t, which shall be the greater of:
 - (i) the ONC Maximum Annual Cumulative Flow divided by the number of hours in Charging Year t, expressed as tCO₂/hour; and
 - (ii) the aggregate of the Delivery Point Sizes of all Delivery Points to the T&S

 Network, expressed as tCO₂/hour,

expressed as £[x]/tCO2/hour.

Offshore T&S Network Mutualisation Check

- 16.3 The calculation of the Offshore Mutualised Network Charge Rate for each User is subject to an iterative process which is set out in paragraphs 16.4 to 16.13 (inclusive).
- 16.4 The "Offshore T&S Charge" for the T&S Network which will be recovered from all Offshore
 Users of the T&S Network through the application of the Offshore Flow Charge, the Offshore
 Capacity Charge and the User's Offshore Proportionate Network Charge for each User
 (OnRt) for Charging Year t is calculated as follows:

 $(Mod)_t * TFOCRG_t Of PNCt$

where:

- (a) "OfFCt" is the total of all Users' forecast Offshore Flow Charges;
- (b) "OfCCt" is the total of all Users' Offshore Capacity Charges; and
- (c) "OfPNCt" is the total of all Users' Offshore Proportionate Network Charges,

using each Users' Charges Forecast for Charging Year t and expressed as £[x]/tCO2.

16.5 If the Offshore T&S Charge is less than the Allowed Offshore Revenue, the T&SCo must carry out mutualisation, starting with calculation of the Offshore Network Charges Cap in

accordance with paragraph 16.6, followed by calculation of the Offshore User T&S Charge in accordance with paragraph 16.7 and calculation of the Provisional Offshore Mutualised Network Charge in accordance with paragraph 16.9.

Offshore Network Charges Cap Calculation

- 16.6 The "Offshore Network Charges Cap" is equal to the Offshore Proportion of the Mutualisation Cap, where:
 - (a) "Offshore Proportion", expressed as a percentage, is calculated as follows:

 $\frac{\text{(Add)}\textit{ved Off shore Revenue for Charging Year }t}{\textit{Allowed Revenue for Charging Year }t} \times 100$

where the "Allowed Offshore Revenue" for Charging Year t is the amount of Allowed Revenue for the Offshore Transportation and Storage System as determined by the Regulator in accordance with the T&SCo's Licence; and

(b) "Mutualisation Cap" for Charging Year t means a cap, expressed in £[x]/tCO₂, equal to the average of the daily average price observed, as at 31 October of Charging Year t-1, for the March t-1, December t, and March t UK Emissions Trading Scheme futures contract (UKA Futures Contract), published by ICE Futures Europe, over the preceding three (3) -month period.

Offshore User T&S Charge Calculation

16.7 The "Offshore User T&S Charge" for each User for Charging Year t is calculated as follows:

 $\frac{(\text{Add})Rt}{UFOfFt}$

where:

(a) "UofRt" is the total revenue for Charging Year t which will be recovered from the User through the application of the Offshore Flow Charge, the Offshore Capacity Charge and the User's Offshore Proportionate Network Charge, to be calculated as follows:

$$(Add)Ct + UOfCCt + UOfPNCt$$

where:

- (i) (A) "OfCC: UofFCt" is the User's forecast Offshore CapacityFlow Charge-Rate-calculated in accordance with paragraph 3.2; and
- (ii) (B)-"TFOfRC_t" is the total Registered UOfCCt" is the User's Offshore Capacity in the Offshore Transportation and Storage System in the Charging Year Charge; and

Note: The difference in fees on the basis of actual Registered Capacity and the fees actually recovered will form part of the adjustments to determine the Allowed Revenue in Year t+2 pursuant to T&SCo's Licence.

- (b) "OfNDP" is the aggregate of the Delivery Point Sizes of Delivery Points to the Offshore
 Transportation and Storage System that the T&S Network can accommodate, where
 the aggregate of the Delivery Point Sizes of Delivery Points to the Offshore
 Transportation and Storage System that the Offshore Transportation and Storage
 System can accommodate is equal to MRCEOft when there are no interruptible or
 flexible connection agreements,
 - (iii) "UOfPNCt" is the User's Offshore Proportionate Network Charge, calculated using the User's Charges Forecast for Charging Year t; and
- (b) "UFOfFt" is a User's forecast flow of carbon dioxide to be delivered by the User to and transported through the Offshore Transportation and Storage System, calculated using that Users' Charges Forecast for Charging Year t,

expressed as £[x]/tCO₂/hour.

Offshore User T&S Charge Check

- 16.8 If the Offshore Network Charges Cap for Charging Year t:
 - (a) is below the Offshore User T&S Charge for a User, the Offshore Mutualised Network

 Charge Rate shall be deemed to be zero (0) for that User; or
 - (b) is above the Offshore User T&S Charge for a User, the Offshore User's Mutualised

 Network Charge Rate shall be calculated in accordance with paragraphs 16.9 to

 16.11.

Calculation of Provisional Offshore Mutualised Network Charge Rate Calculation

4.7 The <u>"Provisional Offshore Mutualised Network Charge Rate"</u> for a Charging Year <u>t</u> is the lesser of the Offshore Network Charges Cap and the amount calculated as follows:

 $\frac{(\text{Mod}) \mathbf{N} \mathbf{C} \mathbf{R}_t}{\mathbf{O} \mathbf{P} \mathbf{P} \mathbf{D} \mathbf{P} \mathbf{P}_t}$

where:

(a) "UAOfNCRtUAOfRt" is the Allowed Offshore Network Charges Revenue for the Charging Year which will not be recovered through the application of the Offshore Flow Charge and the revenue from the Offshore Capacity Charge and the User's Offshore Proportionate Network Charges Ratethat the T&SCo expects to recover from Users, to be calculated as follows:

 $(Mod)_{CR_t}AOf(CR_tPNCR_tCCR_tCCR_tDUDPQ)_{RCt}) - (OfPNCR_t \times OfUDPt)$

where:

(i) "AOfR_!" is the Allowed Offshore Revenue and has the meaning given in paragraph 16.2(a);

- (ii) "AOfFRt" is the Allowed Offshore Flow Revenue and has the meaning given in paragraph 11.1(a);
- (iii) (i) "AOfNCR₁OfCCRt" is the Allowed-Offshore Network Charges-RevenueCapacity Charge Rate for Charging Year t, calculated in accordance with 4.2(a)paragraph 14.1;
- <u>"TFOfRCt" is the total Registered Capacity in the Offshore Transportation</u>

 System in Charging Year t;
- (v) (ii) "OfPNCR_t" is the Offshore Proportionate Network Charges Rate calculated in accordance with paragraph 4.616.2 (or Offshore Network Charge Rate, as applicable in subsequent rounds of mutualisation); and
- (vi) (iii) "OfUDPt" is the aggregate of the Delivery Point Sizes of the Users' Delivery Points to the Onshore Transportation System and the Offshore Transportation and Storage System in the Charging Year, Offshore Users for which the Offshore Network Charges Cap is above each of those Users' Offshore User T&S Charge (or Updated Offshore User T&S Charge, as applicable in subsequent rounds of mutualisation; and
- (b) "OfUDP_t" has the meaning given to it in paragraph 16.9(a)(vi),

expressed as £[x]/tCO₂/hour.

Updated Offshore User T&S Charge Calculation

After the first round of mutualisation is carried out in accordance with paragraphs 16.8(b) and 16.9, the T&SCo must re-calculate the Offshore User T&S Charge for each relevant User to include the Provisional Offshore Mutualised Network Charge Rate that was calculated in accordance with paragraph 16.9 (the "Updated Offshore User T&S Charge"), to be calculated as follows:

$$\frac{(\text{Add})fRt}{UFOfFt}$$

where:

(a) "UMOfRt" is the offshore revenue for Charging Year t which will be recovered from the

User through the application of the Offshore Flow Charge, the Offshore Capacity

Charge and the Offshore Network Charge, to be calculated as follows:

$$(Add)Ct + UOfCCt + UOfNCt$$

where:

- (i) <u>"UOfFCt" is the User's forecast Offshore Flow Charge;</u>
- (ii) "UOfCCt" is the User's Offshore Capacity Charge; and
- (iii) "UofNCt" is the User's Offshore Network Charge,

calculated using the User's Charges Forecast for Charging Year t; and

(b) "UFOfFt" is a User's forecast flow of carbon dioxide to be delivered by the User to and transported through the Offshore Transportation and Storage System, calculated using that User's Charges Forecast for Charging Year t,

expressed as £[x]/tCO2.

Offshore Mutualised Network Charge Rate Calculation

- 16.11 If the Provisional Offshore Mutualised Network Charge Rate for a User calculated in accordance with paragraph 16.9, results in the Updated Offshore User T&S Charge calculated in accordance with paragraph 16.10 for that User:
 - (a) exceeding the Offshore Network Charges Cap, the Offshore Mutualised Network

 Charge Rate for that User will be set at a level that results in the Offshore User T&S

 Charge equalling the Offshore Network Charges Cap; or
 - (b) not exceeding the Offshore Network Charges Cap, the Offshore Mutualised Network
 Charge Rate for that User will be set at the Provisional Offshore Mutualised Network
 Charge Rate.

T&S Network Mutualisation – Further Rounds

- 16.12 After the first round of mutualisation, the T&SCo must re-calculate the Offshore T&S Charge to include the Updated Offshore User T&S Charge for each relevant User (the "Updated Offshore T&S Charge").
- 16.13 If, after the first round of mutualisation:
 - (a) the Allowed Offshore Revenue less the Updated Offshore T&S Charge is zero (£0/tCO₂), no further mutualisation is required; or
 - (b) <u>the:</u>
 - (i) Allowed Offshore Revenue less the Updated Offshore T&S Charge is above zero (£0/tCO₂); and
 - (ii) Updated Offshore User T&S Charge remains below the Offshore Network
 Charges Cap for at least one (1) User,

the T&SCo must proceed to repeat the process for Users with an Updated Offshore

User T&S Charge that remains below the Offshore Network Charges Cap at

paragraphs 16.7 to 16.13 in iterative rounds until the earlier of:

- (A) the Allowed Offshore Revenue is recovered; or
- (B) 4.8 Theall Users' Offshore Network User T&S Charges Cap is are equal to the Offshore Proportion of the Mutualisation Network Charges Cap, where:

17. Forecasting for the purposes of calculation of Use of System Charges

(a) Offshore Proportion, expressed as a percentage, is calculated as follows:

 $\frac{\text{(Del)}\textit{ved Off shore Revenue for Charging Year t}}{\textit{Allowed Revenue for Charging Year t}} \times 100$

where the Allowed Offshore Revenue for the Charging Year is the amount of Allowed Revenue for the Offshore Transportation and Storage System as determined by the Authority in accordance with T&SCo's Licence; and

(b) Mutualisation Cap for Charging Year t means a cap, expressed in £[x]/tCO₂/hour, equalto the average of the daily average price observed for the March t-1, December t, and-March t UK Emissions Trading Scheme futures contract (UKA Futures Contract), published by ICE Futures Europe, over the preceding three-month period.

5. Charging Methodology: forecasting

- 17.1 5.1 The T&SCo will use forecasts provided by Users in relation to:
 - (a) the massquantity of carbon dioxide to be delivered to the T&S Network through the User's Delivery Point; and
 - (b) the User's expected and/or booked Registered Capacity;
 - (b) <u>for Users being provided with the Re-use Service, the quantity of carbon dioxide to be</u> taken from the T&S Network through the User's Re-use Delivery Point; and
 - (c) the User's Registered Capacity (including any Network Capacity that is due to become the User's Registered Capacity in Charging Year t),

to calculate the <u>Use of System</u> Charges, in accordance with the this Use of System Charging Methodology, for each <u>Charging Year (Year t)</u>.

- For the purpose of Users providing the forecasts set out at paragraph 17.1, the T&SCo will prepare a template document setting out the information to be provided by each User Type for the relevant T&S Network (the "Forecasting Pro Forma").
- 17.3 5.2 By Subject to paragraph 17.4, by 1 October of Charging Year t-1, the T&SCo shall must send the Forecasting Pro Forma to Users.
- The T&SCo must send the Forecasting Pro-Forma, for the first time, to Users by 1 October of the Charging Year prior to the Charging Year in which the T&SCo's Scheduled Commercial Operations Date falls.
- 17.5 <u>5.3 Users shall Each User must</u>:
 - (a) complete the Forecasting Pro Forma by providing the required information for Charging Year t, Charging Year t+1 and Charging Year t+2; and

- (b) send the completed Forecasting Pro Forma to <u>the</u> T&SCo by 31 October of <u>Charging</u> Year t-1 ("Charges Forecast for Charging Year t").
- 17.6 The T&SCo must review each of the Charges Forecast for Charging Year t to check for any readily apparent anomalies or errors and may submit requests for further information from Users.
- 17.7 5.4-Users shall<u>must</u> use reasonable endeavours to provide accurate forecasts when providing information to T&SCo under this paragraph 5 and all such forecasts must be provided an accurate Charges Forecast for Charging Year t and must provide such forecast on a bona fide basis.
- 17.8 5.5 Users shall must respond promptly and within ten (10) Business Days to any queries from the T&SCo regarding such forecasts Charges Forecast for Charging Year t and shall submit a revised Charges Forecast for Charging Year t where reasonably requested by the T&SCo.
- A User must provide its first Charges Forecast for Charging Year t by 31 October in the Charging Year prior to the Charging Year in which that User's Scheduled Commencement Date falls.
- 18. 6-Preparation and publication of Charges Use of System Charging Statement
- 18.1 6.1 For anyeach Charging Year (Year t), the T&SCo shallmust prepare a statement of:
 - (a) the rates for each of the <u>Use of System Charges</u> that apply in <u>Charging Year</u> t, as calculated in accordance with the <u>Use of System Charging Methodology (Charges Statement). -; and</u>
- 6.2 The Charges Statement shall be prepared using a template approved by the Authority.
 - (b) the calculations used to determine each of the rates referred to in paragraph 18.1(a) for each of the Use of System Charges that apply in Charging Year t using the Use of System Charging Methodology, including the Charges Forecast for Charging Year t which have been used by the T&SCo for the purposes of such calculations,
 - (the "Use of System Charging Statement").
- 18.2 The T&SCo must prepare the Use of System Charging Statement in a form approved by the Regulator.
- 18.3 6.3-The T&SCo must calculate Use of System Charges payable by Userseach User during each Charging Year shall be calculated by reference tot using the rates set out in the Charges Use of System Charging Statement prepared and published by the T&SCo in accordance with its Licence and this paragraph 618.
 - **Note:** it is only the rates for calculating the Charges that will be publicly published. The Charges Statement will not include the actual Charges payable by individual Users.
- 18.4 6.4 The T&SCo shallmust prepare and publish the Charges Use of System Charging Statement by 30 November of Charging Year t-1.

- 18.5 6.5 The T&SCo shall also must prepare and publish a draft Charges Use of System Charging Statement for Charging Years t+1 and t+2, by 30 November of Charging Year t-1, which shall must:
 - (a) set out estimates of the rates for each of the <u>Use of System</u> Charges that apply in Charging Years t+1 and t+2, <u>as</u>-calculated in accordance with the <u>Use of System</u> Charging Methodology; and
 - (b) be subject to revision and <u>is beingshall be</u> provided to Users for information purposes only.
- 18.6 The T&SCo must prepare and publish its first Use of System Charging Statement by:
 - (a) 30 November in the Charging Year prior to the Charging Year in which that T&SCo's Scheduled Commercial Operations Date falls; or
 - (b) as soon as reasonably practicable following the Regulator's calculation of the T&SCo's

 Allowed Revenue under Special Condition G16.8 (Post Construction Review) of the

 Licence,

whichever is the later.

Part 2 – Payments and invoicing

- 19. Introduction to payments and invoicing
- 19.1 Part 2 of this Section H (Charges, Invoicing and Payment) sets out provisions relating to the obligation of Users to pay Use of System Charges, including security requirements, payment obligations and invoicing.
- The amounts payable by Users to the T&SCo and by the T&SCo to Users in accordance with the Code will be invoiced and payable in accordance with this Section H (Charges, Invoicing and Payment).
- 20. 7. Required Security
- 20.1 7.1 The User must provide the Required Security to the T&SCo by 31 January of each year Charging Year t-1 and thereafter maintain the Required Security.

Note: the obligation to maintain the security shall include the circumstances in which T&SCo-has had recourse to the existing security (i.e. the User is required to replace the security insuch circumstances).

- 20.2 7.2 The "Required Security shall be a" must:
 - (a) be in the form of:
 - (i) (a) a Letter of Credit; or
 - (ii) (b) a Deposit Deed;

(b) be for the Required Security Amount, as calculated in accordance with paragraph 20.3; and

which:

- (c) is for the value of the aggregate of the forecast two highest value Invoice Documents for the relevant Charging Year; and
- (c) (d) provides provide for the rights of the T&SCo required by paragraph 4229.
- 20.3 The "Required Security Amount" is the amount of financial security required to be provided by each User to the T&SCo for Charging Year t, calculated as:

(Add)l + UoSC2

where:

- (a) "UoSC1" is the value of the Use of System Charges expected to be payable by a User for the Billing Period in which the quantity of carbon dioxide to be delivered to the T&S

 Network through the User's Delivery Point is the highest for Charging Year t, as set out in the Charges Forecast for Charging Year t; and
- (b) "UoSC2" is the value of the Use of System Charges expected to be payable by a User for the Billing Period in which the quantity of carbon dioxide to be delivered to the T&S Network through the User's Delivery Point is the second highest for Charging Year t, as set out in the Charges Forecast for Charging Year t.
- 20.4 The T&SCo must notify each User of the Required Security Amount by 30 November of Charging Year t-1.
- If at any time a User fails to comply with paragraph 7.1–20.1, the T&SCo may issue a notice ("Security Default Notice") requiring the User to rectify such non-compliance within 5ten (10) Business Days.
- 20.6 7.3 Where a User has not rectified its non-compliance with paragraph 20.1 within five (5) Business Days of its receipt of a Security Default Notice, the T&SCo shall be entitled to reject or refuse to accept all or any of the following by the relevant User:
 - (a) delivery of carbon dioxide at the User's Delivery Point; er
 - (b) taking delivery of carbon dioxide at the User's Re-use Delivery Point; or
 - (c) (b) an application for Registered Capacity or increased Registered Capacity at any Delivery Point under Section E (Network use and capacitySection C (Connection),

with effect until such time as the relevant User has provided the Required Security.

20.7 7.4 A User shall remainremains liable to pay any Capacity Charges and Network Charges during any period in which the T&SCo has exercised its rights under paragraph 7.4 applies 20.6.

21. 8. Issue of Invoice Document

21.1 8.1 The parties parties agree and acknowledge that the T&SCo shallmust use the Processed Flow Meter Data set out in the Processed Flow Meter Data Statement, prepared by the CDS in accordance with the CDS Contract, for the purposes of calculating the Onshore Flow Charge and, the Offshore Flow Charge and the Re-Use Service Charge.

Note: provision will be made for fixed decimalisation and rounding in the CDS Contract and the Data Transfer Procedures. This is subject to determining the levels of metering accuracy.

- 21.2 8.2 Within [tenNo later than five (105) Business Days] of a filling Period, the T&SCo must submit to each User:
 - (a) the Processed Flow Meter Data Statement, setting out the Processed Flow Meter Data which:
 - (i) relates to the actual quantity of carbon dioxide:
 - (i) relates to the actual mass of carbon dioxide delivered by the User into the T&S Network during that Billing Period; and
 - (B) where the User is being provided with the Re-use Service, taken from the T&S Network during that Billing Period; and
 - (ii) has been used to calculate the Onshore Flow Charge-and, the Offshore Flow Charge and the Re-use Service Charge (if applicable) payable by the User in relation to that Billing Period; and
 - (b) the Invoice Document.

Note: the timing for the issue of the Invoice Document remains subject to review in the context of both the CDS Contract process and User Requirements.

- 21.3 8.3-Where a User delivers carbon dioxide at more than one (1) Delivery Point then the T&SCo shallmust issue to that User a separate Invoice Document in relation to each Delivery Point.
- 22. 9. Content of Invoice Document
- 22.1 9.1 For each Billing Period, the Invoice Document submitted by the T&SCo to each User under paragraph 8.2 shall21.2 must separately identifying identify the following:
 - (a) the Onshore Flow Charge (if applicable);
 - (b) the Offshore Flow Charge;
 - (c) (b) Onshore Capacitythe Re-use Service Charge (if applicable);
 - (d) (e) the Onshore Network Capacity Charge (if applicable);
 - (d) Offshore Flow Charge;

- (e) the Offshore Capacity Charge;
- (f) the Onshore Network Charge (if applicable);
- (g) (f) the Offshore Network Charge; and
- (b) (g) any other charges or amounts that may become due and payable under the Code or any Ancillary Agreement.
- 22.2 9.2 Each Invoice Document shallmust be issued set out on the T&SCoSCo's letterhead and shall specifyit must include the following information:
 - (a) the identity of the User;
 - (b) the Delivery Point to which the Invoice Document relates;
 - (c) the Re-use Delivery Point to which the Invoice Document relates (if applicable);
 - (d) (e) the Billing Period to which the Invoice Document relates;
 - (e) (d) the Invoice Amount;
 - (f) (e) the unique number by which the Invoice Document may be identified;
 - (g) (f) the amount of VAT Value Added Tax (if any) payable;
 - (h) (g) the VATValue Added Tax registration number of both the User and the T&SCo;
 - (i) (h) the date of the Invoice Document;
 - (j) (i) the credit terms and beneficiary bank details; and
 - (k) (j) the currency of the Invoice Amount.
- 22.3 9.3 An Invoice Document may contain an adjustment by way of a credit or a debit in respect of:
 - (a) an Invoice Amount in another Invoice Document; and
 - (b) any other amount which is agreed or determined to be payable by a User to the T&SCo or by the T&SCo to a User under or in connection with this the Code (including any adjustment required as a result of any reduction in Capacity Charges and Network Charges payable by a User in accordance with paragraph 45.2 of Section E (Network Use and Capacity).
- 22.4 9.4-Any amounts payable by the T&SCo to a User under or in connection with this the Code, as set out in paragraph 9.3(b)22.3(b), shall be paid by way of an adjustment to the next Invoice Document due to be submitted after the amount becomes payable, and where any balance is still payable to the User, then that amount shall be paid by the T&SCo to the User when that amount is due or at the same time as the Invoice Document is submitted to the User, whichever is the later.

- 22.5 9.5 Invoice Documents will must be submitted to Users through the T&S Network Portal.
- 22.6 An Invoice Document will not be invalid solely by reason of it not being accompanied by any particular item of supporting data, but any lack of supporting data may be queried through the Invoice Dispute process.
- 23. Value Added Tax
- 23.1 9.6 All amounts expressed as payable by a User or the T&SCo pursuant to the Code shall be exclusive of VATare exclusive (unless expressly stated otherwise)otherwise stated) of any applicable Value Added Tax, and accordingly Value Added Tax must be paid by the paying party where payable in respect of any such amount.
- 24. 40. Invoice payment
- 10.1 The Invoice Amount under the Invoice Document shall be paid by the User to T&SCo:
- Where an Invoice Document has been submitted to a User by the T&SCo in accordance with this Section H (Charges, Invoicing and Payment), the User must pay the Invoice Amount set out in the Invoice Document to the T&SCo:
 - (a) on or before the Invoice Due Date; and
 - (b) by wire transfer of immediately available funds.
- 24.2 10.2 The "Invoice Due Date-shall be" is the date falling thirty-five twenty five (3525) Business

 Days after from the enddate of the Billing Period to which the Invoice Document relates.
- 24.3 40.3 Amounts payable under the Code shallmust be paid:
 - (a) free and clear of any restriction, reservation or condition; and
 - (b) except to the extent (if any) required by law, without deduction or withholding in respect of tax or on account of any amount due or to become due to the paying party, whether by way of set off counterclaim or otherwise, other than by way of adjustment as set out in paragraph 9.3(b) and 9.4paragraphs 22.3(b) and 22.4.
- 25. **Termination and insolvency**
- Notwithstanding paragraphs 21 and 24, the T&SCo may, at any time (whether before or after the User Discontinuance Date) after the T&SCo submits to a User a Termination Notice under Section J (General), submit to that User any Invoice Document in respect of any Billing Period or part of a Billing Period ending at or before the time at which the T&SCo submits such Invoice Document.
- Where the T&SCo has submitted a Termination Notice to a User, all amounts payable by that
 User to the T&SCo or by the T&SCo to the User (whether the Invoice Document in which
 such amounts are shown was submitted before or after the date of the Termination Notice)
 shall be immediately payable notwithstanding paragraph 24.
- 25.3 Notwithstanding paragraphs 21 or 25.1, the T&SCo may, at any time on or following the occurrence of an Insolvency Default in relation to any User, submit to that User any Invoice

- <u>Document in respect of any Billing Period or part of a Billing Period ending at or before the time at which the T&SCo submits such Invoice Document.</u>
- 25.4 In relation to a User, on the occurrence of an Insolvency Default, all amounts payable to the T&SCo (whether the Invoice Document in which such amounts are shown was submitted before or after the date of the occurrence of the Insolvency Default) shall be immediately payable notwithstanding paragraph 24.1.
- 26. 41. Invoice Disputes and amended Invoice Documents

 Resolution of Invoice Disputes
- 26.1 11.1 For the purposes of this Section H an An "Invoice Dispute" is any question or dispute as to the proper calculation of any amount shown as payable by a User under an Invoice Document or as to whether any such amount is or was properly payable, where that question or dispute is not a CDS Data Dispute or in relation to a Flow Meter Error.
- 26.2 41.2 References to the amount of an Invoice Dispute are to the amount by which the User submitting the Invoice Dispute considers the Invoice Amount to be incorrect.
- 26.3 41.3 Where an Invoice Dispute has been raised:
 - (a) the User shallmust provide to the T&SCo in writing all relevant details of the Invoice Dispute, including the amount of the Invoice Dispute; and
 - (b) <u>the T&SCo</u> and the relevant User <u>willmust</u> endeavour to resolve the Invoice Dispute by agreement.
- 26.4 11.4 Without prejudice to any obligation of Users pursuant to this Section, Section H (Charges, Invoicing and Payment), the T&SCo agrees that it will make available such resources as are in its reasonable opinion reasonably adequate to deal reasonably promptly with the Invoice Dispute.
- 26.5 Here a User raises an Invoice Dispute it shallmust pay the amount of the Invoice Dispute on the Invoice Due Date notwithstanding the Invoice Dispute.
- 26.6 <u>Hand to resolve the Invoice Dispute within 30 twenty (20)</u>
 Business Days of the Invoice Dispute being raised, the parties either of them shall be entitled to refer to the Invoice Dispute for resolution byto an Expert Determination for determination, in accordance with Section B paragraph 51 of Section B (Governance).
- 26.7 41.7 As soon as reasonably practicable after any Invoice Dispute is resolved, and in any event within twenty (20) Business Days, the T&SCo willmust prepare and submit to the relevant User an appropriate Invoice Document in respect of the amount (if any) agreed or determined to be payable or repayable by the T&SCo or the relevant User, in accordance with paragraph 8 of this Section H21.
- 27. Consequences of resolution of Flow Meter Errors and CDS Data Disputes
- 27.1 11.8 Where:

- (a) a CDS Data Dispute has been finally resolved in accordance with Expert
 Determination under paragraph 9 of Section B (Governance); or
- (b) a Flow Meter Error has been finally resolved in accordance with Section F Section F
 (Network design Design and specification Specification),

the T&SCo shallmust instruct the CDS to re-process the applicable data (including any new data provided pursuant to paragraphs 6.3, 7.1 or 7.2 of Section F 11.1, 13.1 or 13.2 of Section F (Network Design and Specification)) for the relevant Billing Period and re-issue the Processed Flow Meter Data Statement for the relevant Billing Period(s) within ten (10) Business Days of being notified of the outcome of the CDS Data Dispute or the Flow Meter Adjustment, as the case may be.

- 27.2 11.9 Upon being re-issued with a Processed Flow Meter Data Statement pursuant to paragraph 11.827.1, the T&SCo shallmust, within 10 Business Days, re-submit the Invoice Document(s) to the User in accordance with paragraph 8 of Section H21.
- 28. Payment of Invoice Amounts which may be impacted by another dispute
- 28.1 11.10 Where a User considers that the Invoice Amount payable under an Invoice Document may be incorrect because of matters which are the subject of a CDS Data Dispute or because of a Flow Meter Error, the User shall must pay the Invoice Amount on the Invoice Due Date notwithstanding the CDS Data Dispute or the alleged Meter Flow Meter Error (as the case may beapplicable).
- 29. 42. Late payment
- 29.1 42.1 Where any amount payable under an Invoice Document is not paid on or before the 5th fifth Business Day after the Invoice Due Date, the T&SCo shall be entitled to issue a notice of demand for payment of the outstanding amount ("Late Payment Notice").
- 29.2 Here a partyParty has not made payment of an amount that is the subject of a Late Payment Notice within five (5) Business Days of such Late Payment Notice:—, the T&SCo is entitled to have recourse to the Required Security to discharge the debt due, following which the User must, within ten (10) Business Days following such drawdown, ensure that the Required Security is replenished to the Required Security Amount.
 - (a) without prejudice to any other rights of T&SCo under the Code, T&SCo shall be entitled to reject or refuse to accept all or any of the following by the relevant User:
 - (i) delivery of carbon dioxide at the User's Delivery Point; or
 - (ii) an application for Registered Capacity or increased Registered Capacity at any Delivery Point under Section E (Network use and capacity),

with effect from the day after the due date for payment until such time as the relevant-User has paid the amount due for payment in full; and

(b) T&SCo shall be entitled to have recourse to the Required Security.

- 12.3 A User shall remain liable to pay Capacity Charges and Network Charges during any period in which paragraph 12.2 applies.
- 29.3 12.4 Where any amount payable under an Invoice <u>Document</u> is not paid on or before the Invoice Due Date, the User <u>shallmust</u> pay interest, <u>after as well as before judgment</u>, at the Applicable Interest Rate, on the unpaid amount from the Invoice Due Date until the Day on which payment is made.
- 29.4 12.5 The "Applicable Interest Rate" is the rate of interest.:
 - (a) expressed as a percentage rate per annum, payable in respect of amounts overdue for payment, or subject to repayment, under the Code-; and
 - (b) 12.6 The Applicable Interest Rate shall be the rate of interest-set for the relevant period as the statutory interest rate for the purposes of the Late Payment of Commercial Debts (Interest) Act 1998.

CCS Network Code

Section I – Data

1. **Information sharing**

1.1 Where any provision of the Code requires or entitles a T&SCo (or the T&SCosSCo's) to "publish" any information or document, the T&SCo willmust provide such information or document to each Party, any other person to whom the T&SCo may (pursuant to its Licence or any other Legal Requirement or any provision of the Code) be required to provide such information and (subject to any confidentiality restrictions under the Code) such other persons as the T&SCo shall thinkthinks fit.

2. Liability

Save as Unless expressly provided elsewhere otherwise in the Code, no Party shall have any liability to any other Party in the event that any data or information exchanged or provided pursuant to this the Code is found to be inaccurate, in the absence of wilful misconduct by the Party providing such information or data.

3. **Definitions and interpretation**

3.1 "Measurement Data" means all data which (as the context may require) is, has been or is required to be measured by Measurement Equipment installed pursuant to paragraph 5 of Section F (Network Design and Specification) and provided to or accessed by the T&SCo pursuant to the Data Transfer Procedures, and includes data calculated or otherwise derived by the operation of any component of the Measurement Equipment, which is not User-Specific Data.

3.2 "Protected Information" means:

- (a) for the purposes of the T&SCo's obligations under paragraph 4.1:
 - any information relating to the affairs of a User or another T&SCo which is obtained by the T&SCo pursuant to or in the course of implementation or performance of the Code, the Code Agreement or any Ancillary Agreement to which that User is party; and
 - (ii) the terms of any Ancillary Agreement; and
- (b) for the purposes of a User's obligations under paragraph 4.2:
 - any information relating to the affairs of the T&SCo or of another User which is obtained by the User pursuant to or in the course of the negotiation, implementation or performance of the Code, the Code Agreement or any Ancillary Agreement to which the User and (in relation to another User) that other User are party; and
 - (ii) the terms of any Ancillary Agreement to which that User is party.

- 3.3 "Receiving Party" and "Protected Party" shall be construed as follows:
 - (a) for the purposes of the T&SCo's obligations under paragraph 4.1, the Receiving Party is the T&SCo and the Protected Party is the Party (either the User or another T&SCo) to whose affairs any Protected Information relates; and
 - (b) for the purposes of a User's obligations under paragraph 4.2, the Receiving Party is the User and the Protected Party is the Party (either the T&SCo or another User) to whose affairs any Protected Information relates.
- 3.4 "T&S Network Data" means all data related to the T&S Network which is not User Specific

 Data or Measurement Data.
- 3.5 <u>"User-Specific Data"</u> means all data:
 - (a) held by the T&SCo in relation to a specific User, including:
 - (i) <u>User's Registered Capacity;</u>
 - (ii) accepted and rejected Nominations and Renominations;
 - (iii) <u>Invoice Documents;</u>
 - (iv) metered quantities;
 - (v) <u>outages affecting specific Users;</u>
 - (vi) information required to be provided by the User to a Revenue Support Contract counterparty (pursuant to its Revenue Support Contract) or any other body pursuant to Legal Requirements;
 - (vii) the User's Measurement Equipment;
 - (viii) the User Type; and
 - (ix) the User's contact details; and
 - (b) <u>held by a User about User networks and/or User Facilities, including:</u>
 - (i) details of planned maintenance/outages in relation to User Facilities;
 - (ii) <u>forecast flow data;</u>
 - (iii) actual flow data;
 - (iv) Nominations and Renominations;
 - (v) capacity bookings; and
 - (vi) any information which is required to be provided by the T&SCo to the Regulator (pursuant to its Licence), the North Sea Transition Authority (pursuant to its licence/permit) or any other body pursuant to Legal Requirements.

4. 3. Information and confidentiality

T&SCo obligation

4.1 3.1 The T&SCo shall must preserve the confidentiality of Protected Information and ensure that Protected Information is not:

- (a) disclosed, revealed, reported or published or transferred to any person other than:
 - (i) an officer or employee of the T&SCo or its Affiliates or Related Undertakings who is required to have access to the information to carry outfor the purposes of their duties; or
 - (ii) a professional adviser, <u>agent</u> of or consultant <u>or contractor</u> to the T&SCo<u>or</u> those of its Affiliates or Related Undertakings; or
 - (iii) operators appointed by the T&SCo to operate the T&S network-
 - who, before the Protected Information had been disclosed to them by the T&SCo, had been informed of the nature and effect of this paragraph 4.1; and
- (b) used by the T&SCo for anythe purpose of carrying on any activities other than carrying out its activities in relation to the T&S Network in accordance with the Code and the Licence-

except in the circumstances set out in paragraph 4.8.

User obligation

- 4.2 Seach User shall must preserve the confidentiality of Protected Information and ensure that Protected Information is not:
 - (a) disclosed, revealed, reported, published or transferred to any person other than:
 - (i) an officer or employee of the User or its Affiliates or Related Undertakings who
 is required to have access to the information to carry outfor the purposes of
 their duties; or
 - (ii) a professional adviser, <u>agent</u> of or consultant <u>or contractor</u> to <u>that</u>the User <u>or</u>
 <u>those of its Affiliates or Related Undertakings</u>; or
 - (b) used by such User for any purpose other than one expressly contemplated by the Code or any Ancillary Agreement to which such User is party.

except in the circumstances set out in paragraph 4.8.

Protected Information

4.3 3.3 In this section Protected Information means The Receiving Party shall:

(a) immediately notify the Protected Party in writing if it suspects or becomes aware of any unauthorised access, copying, use or disclosure in any form of any Protected Information; and

- (a) for the purposes of the T&SCo's obligations under paragraph 3.1:
 - (i) any information relating to the affairs of a User which is obtained by T&SCopursuant to or in the course of the negotiation, implementation or performance of the Code, the Code Agreement or any Ancillary Agreement to which that User is party;
 - (ii) the terms of any Ancillary Agreement;
- (b) for the purposes of a User's obligations under paragraph 3.2: use reasonable endeavours to recover and limit further onward dissemination of such Protected Information.
 - (i) any information relating to the affairs of T&SCo or of another User which is obtained by the User pursuant to or in the course of the negotiation, implementation or performance of the Code, the Code Agreement or any Ancillary Agreement to which the User and (in relation to another User) that other User are party;
 - (ii) the terms of any Ancillary Agreement to which that User is party.

Terms of permitted disclosure

- 3.4-Where Protected Information is disclosed by the T&SCo as permitted under paragraph 3.1(a)4.1 or by a User as permitted under paragraph 3.2(a)4.2, the Disclosing Receiving Party shall (without prejudice to its obligations under paragraph 3.14.1 or 3.24.2) take all reasonable steps to secure that the person to whom the information Protected Information is disclosed:
 - (a) is aware of the Disclosing Receiving Party's obligations under paragraph 3.1 or 3.2 in relation thereto4.1 or 4.2; and
 - (b) does not use or disclose the <u>informationProtected Information</u> other than as is permitted for <u>such</u>the <u>Receiving</u> Party in accordance with paragraph <u>3.14.1</u> or <u>3.24.2</u>.
- 4.5 The Receiving Party shall adopt procedures within its organisation for ensuring compliance with paragraphs 4.1 and 4.2. These procedures shall ensure that:
 - (a) the Protected Information will be disseminated to persons within the Receiving Party only on a "need to know" basis;

Exceptions

3.5 For the purposes of this section, **Disclosing Party** and **Protected Party** shall be construed as follows:

(a) for the purposes of T&SCo's obligations under paragraph 3.1, the Disclosing Party is

T&SCo and the Protected Party is the User to whose affairs any Protected Information relates;

- (b) for the purposes of a User's employees, officers, directors, agents, consultants, contractors and professional advisers of the Receiving Party or those of its Affiliates or Related Undertakings in receipt of Protected Information will be made fully aware of the Receiving Party's obligations under paragraph 3.2, the Disclosing Party is such User and the Protected Party is the Party (either T&SCo or another User) to whose affairs any Protected Information relates.4.1 and 4.2; and
- (c) any copies of the Protected Information, whether in hard copy or computerised form, will clearly identify the Protected Information as confidential.
- The Receiving Party shall procure that its Affiliates, Related Undertakings, agents, consultants, contractors and professional advisers observe the restrictions set out in this paragraph 4 (as if references to "User" or "T&SCo" were references to such Affiliates, Related Undertakings, agents, consultants, contractors and professional advisers) and shall be responsible under the Code for any failure by such persons to observe such restrictions.
- 4.7 Protected Information which any Party is permitted or obliged to disclose or publish to any other Party pursuant to the Code shall not solely be regarded as being in the public domain by reason of being so disclosed or published.

Exceptions

- 4.8 $\frac{3.6}{1.0}$ Nothing in paragraph $\frac{3.14.1}{1.0}$ or $\frac{3.24.2}{1.0}$ shall apply:
 - (a) to the disclosure or use by the Receiving Party of Protected Information to the extent otherwise expressly permitted by the Code;
 - (b) (a)-to the disclosure or use by the Disclosing Receiving Party of Protected Information to which the Protected Party has consented in writing;
 - (c) (b) to any Protected Information which:
 - (i) before it is obtained by the Disclosing Receiving Party is in the public domain; or
 - (ii) after it is obtained by the <u>Disclosing Receiving</u> Party enters the public domain_
 - (iii) in either case <u>otherwiseother</u> than as a result of a breach by the <u>DisclosingReceiving</u> Party of its obligations under paragraph <u>3.14.1</u> or <u>3.24.2</u>;
 - (c) to the disclosure of any Protected Information to any person if and to the extent that the Disclosing Receiving Party is required to make such disclosure to such person:
 - (i) in compliance with the duties of the <u>DisclosingReceiving</u> Party under the Act or any other requirement of a Competent Authority; or
 - (ii) in compliance with the conditions of the Licence held by the

 Disclosing Receiving Party or any document referred to in such Licence with

- which the <u>DisclosingReceiving</u> Party is required by virtue of the Act or such <u>licenceLicence</u> to comply; <u>or-</u>
- (iii) in compliance with any other Legal Requirement; or Requirements;
- (iv) in response to a requirement of any stock exchange or regulatory authority or the Panel on Take-overs and Mergers; or
- (v) pursuant to any judicial or other arbitral process or tribunal having jurisdiction in relation to the <u>DisclosingReceiving</u> Party;
- (e) to any Protected Information to the extent that the Receiving Party is expressly permitted or required to disclose that Protected Information under the terms of any agreement or arrangement with the Party to whose affairs such Protected Information relates;
- (d) to the disclosure of Protected Information to any lending or other financial institution or investment entity proposing to provide or arrange the provision of finance to the Disclosingor equity investment to the Receiving Party or an Affiliate or Related Undertaking of the Receiving Party, where and to the extent that the disclosure of such information is reasonably required for the purposes of the provision or arrangement of such finance, or equity investment and provided that the person to whom the information is disclosed undertakes in writing to, and in terms reasonably satisfactory to the Protected Party-to, maintain the confidentiality of such informationProtected Information;
- (g) (e) to the disclosure of any Protected Information to the Regulator, where the Disclosing Receiving Party considers in good faith that the Protected Party may be in breach of a condition of the Licence, to the extent reasonably necessary to draw such possible breach to the attention of the Regulator;
- (f) to the disclosure of any Protected Information to any person proposing to make a connection to the T&S Network, where and to the extent that the disclosure of such information is reasonably required for the purposes of supporting the T&SCo's charges Use of System Charges or requirement to allow such proposed connection to the T&S Network to be made and provided that the person to whom the information is to be disclosed is informed in writing of the confidentiality of such information and that prior to disclosure, such person has entered into a confidentiality agreement with the T&SCo which prohibits the use or disclosure of such Protected Information inon terms no less onerous than those contained set out in this section. Section I (Data); or
- where Protected Information is provided by the Receiving Party to its Affiliates or Related Undertakings or to the employees, directors, agents, consultants, contractors and professional advisors of the Receiving Party or those of its Affiliates or Related Undertakings, in each case on the basis set out in paragraph 4.6.

4.9 The T&SCo may use all and any information or data supplied to or acquired by it, from or in relation to Users and other T&Scos in performing its activities in relation to the T&S Network in accordance with the Code and the Licence, including for the following purposes:

- (a) the operation and planning of the T&S Network; and
- (b) the calculation of Use of System Charges and preparation of offers of terms for connection to or use of the T&S Network.
- 4.10 Without prejudice to the other provisions of this paragraph 4, the Receiving Party shall procure that any additional copies made of the Protected Information whether in hard copy or computerised form, will clearly identify the Protected Information as confidential.

Survival

4.11 3.7 The provisions of paragraphs 3.14.1 to 3.64.10 shall continue, for a period of three (3) years after the User Discontinuance Date, to bind a Discontinuing User and (as respects the Discontinuing User) the T&SCo and each other User, notwithstanding that the Discontinuing User has ceased to be a User and irrespective of the reason for such cessation.

Licence

- 4.12 3.8 Nothing in the Code, the Code Agreement or any Ancillary Agreement shall be construed as requiring the T&SCo to disclose or use any information in breach of any requirement of the T&SCo's Licence.
- 5. 4. Specific disclosure requirements
- 5.1 Without prejudice to the generality of paragraph 34, the parties acknowledge that a User may be required to make disclosures of Protected Information, User-Specific Data and T&S Network Data to the LCCC pursuant to the User Requirements.
- 5.2 The T&SCo must make available to Users any information that they may reasonably request to enable them to comply with their reporting obligations under their User Requirements.

5. Categories of Data

User-Specific Data

- 5.1 In this Section, User-Specific Data means all data:
 - (a) held by a T&SCo in relation to a specific User, including:
 - (i) User's Registered Capacity;
 - (ii) accepted and rejected Nominations;
 - (iii) invoices;
 - (iv) metered quantities;
 - (v) outages affecting specific Users;

(vi) information required to be provided by the User to a Support Contract counterparty (pursuant to its Support Contract) or any other body pursuant to legal requirements; and

- (vii) User's Measurement Equipment;
- (viii) User Type; and
- (ix) User contact details.
- (b) held by a User about User networks including:
 - (i) details of planned maintenance/outages in relation to User facilities;
 - (ii) forecast flow data;
 - (iii) actual flow data;
 - (iv) nominations;
 - (v) capacity bookings; and
 - (vi) any information which is required to be provided by T&SCo to the Authority (pursuant to its licence), the North Sea Transition Authority (pursuant to its licence/permit) or any other body pursuant to legal requirements.

Measurement Data

5.2 In this section, **Measurement Data** means all data generated by Measurement Equipment installed pursuant to paragraph 5 of Section F which is not User-Specific Data.

T&S Network Data

- 5.3 In this Section, T&S Network Data means all data related to the CCS Network which is not User-Specific Data or Measurement Data.
- 6. Access to Data

Data Transfer Procedures

- 6.1 Each T&SCo shallmust establish, document, implement and maintain written procedures to address exchange of and access to User-Specific Data and Measurement Data between the T&SCo and Users (for their respective T&S Networks ("Data Transfer Procedures").
- The Users and each T&SCo shall be required to must comply at all times with the Data Transfer Procedures applicable to the relevant T&S Network.
- The Data Transfer Procedures shall be established and published by the T&SCos on the T&S

 Network Portal no later than fewelve (12) months prior to the Scheduled Commercial

 Operations Date for each T&S Network and shall, as a minimum coverset out the following content in relation to different User-Specific Data and Measurement Data:

- (a) methods of data acquisition (e.g. SCADA, email accounts, web-based platforms etc);
- (b) template format for data provision and reporting;
- (c) minimum data transfer intervals;
- (d) level of granularity within datasets;
- (e) naming protocols for specific datasets; and
- (f) means of transfer (e.g. email, telemetry, application programming interface, etc).

Minimum content requirements

The Data Transfer Procedures shall at all times comply with the requirements for format, naming, and minimum <u>data</u> transfer intervals in relation to the User-Specific Data and the Measurement Data set out in <u>Annexure I (Minimum Documentation Requirements)</u>.

Preparation of Data Annexures

- No later than 42 twenty four (24) months prior to the earliest Scheduled Commercial Operations Date, the T&SCoSCos shall jointly develop and deliver to:
 - (a) each User; and
 - (b) the AuthorityRegulator,

the proposed draft of Annexure I and Annexure J Annexure I (Minimum Documentation Requirements), Annexure J (Information Publication Requirements) and Annexure K (Retained Data) (together, the "Draft Data Annexures") for review and comment in accordance with paragraphs 6.6 to 6.8-below.

- As soon as reasonably practicable and in any event within twenty (20) Business Days of receipt of the Draft Data Annexures, each User shallmust notify the T&SCos of any comments on the Draft Data Annexures. Such comments may include further requested content or revisions to existing content, including principles regarding data aggregation or collation to inform the content of the Data Transfer Procedures.
- 6.7 When providing comments to the T&SCos in accordance with paragraph 6.6, each User shall be required tomust provide the following:
 - (a) information about the User (company, registered office, etc.) and about at least one
 (1) reference person (name, telephone number, fax number, e-mail address, etc.) who can be contacted with respect to the Userthat User's comments;
 - (b) the reasons why the User believes the <u>UserUser's</u> comments should be incorporated and, to the extent any specific content is requested, the text of such content; and
 - (c) any documentation (analyses, reports, etc.) to support the request.

Before preparing the final version of Annexure I (Minimum Documentation Requirements),

Annexure J (Information Publication Requirements) and Annexure K (Retained Data) (the

"Data Annexures"), the T&SCos must convene a meeting with Users in order to discuss any

User comments provided in accordance with paragraph 6.6, the date of which must be

communicated to Users with not less than ten (10) Business Days notice.

- 6.9 6.8 In preparing the final versions of Annexure I and Annexure Jthe Data Annexures, the T&SCos shall:
 - (a) give reasonable consideration to any comments provided by a User pursuant to paragraph 6.6; and
 - (b) prepare a report to be circulated to all Users collectively prior to delivery of the final versions of the Data Annexures pursuant to paragraph 6.11 detailing:
 - where the T&SCos have taken on board or incorporated any comments provided by Users pursuant to paragraph 6.6; and/or
 - (ii) (b) where the T&SCos do have not take taken on board or incorporate incorporated any comments provided by a User pursuant to paragraph 6.6, provide an explanation to the User of the reasons for this decision.
- 6.10 6.9 For the avoidance of doubt, it shall be reasonable for the T&SCos to not incorporate comments made by a User in relation to the Draft Data Annexures into the final version of the Data Annexures where such comments, if incorporated, would affect the safe operation of the T&S Network.
- 6.11 Subject to paragraph 6.146.17, once all comments have been received and considered pursuant to the process set out in paragraph 6.8, no later than eighteen (18) months prior to the earliest Scheduled Commercial Operations Date, the T&SCos shall deliver to:
 - (a) each User; and
 - (b) the Regulator; and
 - (c) (b) the Authority Secretary,

the final version of Annexure I and Annexure Jversions the Data Annexures and shall proceed to develop the Data Transfer Procedures for the relevant T&S Network in accordance with paragraph 6.1 based on the final versions of the Data Annexures.

- 6.12 Once the final versions of the Data Annexures have been delivered to each User, the

 Regulator and the Secretary in accordance with paragraph 6.11, the Code shall be modified

 by insertion of the final version of the Data Annexures and such Modification shall not be

 subject to the Modification Rules.
- 6.13 Once the Secretary has received the final versions of the Data Annexures, the Secretary must publish such final versions by notice on the CCS Network Code Website.

Revisions

6.14 Once the final versions of Annexure I and Annexure Jthe Data Annexures have been delivered pursuant to paragraph 6.106.11, the T&SCos may review and/or revise the contents of Annexure I or Annexure Jthe Data Annexures:

- (a) when the T&SCos consider it is necessary to do so;
- (b) upon request by a User; or
- (c) in order to resolve a dispute under paragraph 6.146.17,

provided that any such revision shall be subject to the procedure set out in paragraph 6.6 to 6.10, and subject always to paragraph 6.14 Modification Rules.

- 6.12 The T&SCos shall not be required to follow the procedure set out in paragraph 6.11 in relation to revisions to Annexure I and/or Annexure J where any such revision is:
 - (a) of a routine nature; and
 - (b) does not have a material impact on a User's cost base, or other operational processes.
- 6.15
 6.13 Any revisions or amendments to the Data Transfer Procedures for a T&S Network will be implemented by the T&SCo upon reasonable notice to enable each User of that T&S

 Network to adjust its systems (if required) following a Code Modification to reflect the revised Data Annexures.

Disputes

6.16 6.14 Where:

- (a) a User comment provided pursuant to paragraph 6.6 of this Section I-was not included by the T&SCos in the final Annexure I or final Annexure Jversions of the Data Annexures; and
- (b) a majority of Users considers that such comment should have been incorporated by the T&SCo; or.
- (c) a majority of Users consider that a revision made by the T&SCo in accordance with paragraph 6.12 should not have been made,

then such matter may be referred by a User to dispute resolution in accordance with the provisions of paragraph 8 of Section B (Governance).

6.17 6.15 Where any matter is referred to dispute resolution under paragraph 6.146.16, the T&SCo shallmust not finalise the Data Transfer Procedures for the relevant T&S Network until such dispute has been resolved in accordance with the provisions of paragraph 8-of Section B48 of Section B (Governance).

T&S Network Data

6.18 6.16 Each Subject to paragraph 6.21, each T&SCo shallmust procure that amongst others, the following T&S Network Data is <u>publicly</u> accessible to <u>Users</u> at all times via the T&S Network Portal <u>once established</u>, subject to operational availability, and in case of its unavailability the T&SCo shall endeavour to provide this information by other means available to the T&SCo:

- (a) existing connections and new connections applications Connection Applications;
- (b) Network Capacity available (in accordance with the requirements of paragraph 3.2 of Section E (network use and capacity 6.1(a) of Section E (Network Use and Capacity));
- (c) annual Maintenance Programme;
- (d) information on Capacity Constraints, use of the Enhanced Response Service, Flow Tolerance Requests, Exceedance Events, Zero Nomination Low Flow Events and Minimum flow Deficits (in accordance with the requirements of paragraph 12.1(a) of Section E (network use and capacityparagraphs 36.4 and 37.1(a) of Section E (Network Use and Capacity));
- (e) network expansion and development information;
- (f) process and timelines for capacity Network Capacity releases;
- (g) aggregate Registered Capacity of all Users;
- (h) daily aggregate physical flow (gross);
- (i) aggregate metered quantities of all Users per calendar month;
- (j) <u>Use of System Charges; and</u>
- (k) certain historic data in respect of the above.
- 6.19 6.17 The publication of T&S Network Data in accordance with paragraph 6.66.18 shall comply with the formatting and frequency requirements set out in Annexure J (Information Publication Requirements).
- 6.20 Prior to establishment of the T&S Network Portal or in the case of unavailability of the T&S Network Portal, the T&SCo must endeavour to provide the information set out in paragraph 6.18 to Users and the LCCC by other means available to the T&SCo.
- 6.21 The T&SCo must not make any T&S Network Data publicly accessible where to do so would contravene any confidentiality obligations under the Code, the T&SCo's Licence or any relevant Legal Requirements.
- 6.22 Where there is only one (1) T&SCo, references in this paragraph 6 to "the T&SCos" shall be treated as references to the sole T&SCo that is a Party to the Code until such time as another T&SCo accedes to the Code.
- 6.23 For the purposes of this paragraph 6, references to Users shall be treated as including references to Eligible Applicants.

6.24 The Parties acknowledge and agree that notwithstanding paragraph 17.1 of Section J

(General), the Contracts (Rights of Third Parties) Act 1999 shall apply to the rights conferred on Eligible Applicants under this paragraph 6.

7. Storage of Data

Each The T&SCo shall must keep records of all relevant data and information, including the information as listed in Annexure KAnnexure K (Retained Data), for at least ten (10) years.

8. **Data ownership**

- 8.1 Subject to paragraph 8.2(a), all data which is processed by a T&SCo shall belong for the purposes of the Code to the T&SCo which owns or operates the T&S Network (or part of the T&S Network) to which such data relates; and subject to paragraph 8.2(b), the T&SCo may, but without prejudice to paragraph 3 or any other requirement of the Code, use and deal with such data as it thinks fit.
- Where pursuant to the Code, a User provides or arranges for the provision of data to a T&SCo:
 - (a) such data (as provided to the T&SCo by the User) shall belong to the User and all intellectual property relating to such data shall at all times vest in the User; and
 - (b) the User hereby grants to the T&SCo (its successors, assigns, agents and contractors) a perpetual, non-exclusive, royalty-free licence (which shall survive the User Discontinuance Date) in respect of such data and all intellectual property rights therein to use, copy, adapt and deal with such data for the purposes of performance and implementation of the Code and for other purposes contemplated by the Code, but not otherwise.

9. Central Data Service

Procurement

- 9.1 The T&SCos shall be required must, acting jointly:
 - (a) to procure an independent Central Data Service Provider ("CDS") under a contract ("CDS Contract")-; and
 - (b) to ensure that the CDS is appointed <u>under the CDS Contract</u> by the date which is <u>no</u> <u>later than six (6)</u> months prior to the earliest <u>Scheduled</u> Commercial Operations Date.

Note: the timing of this will need to be reviewed once the scheduled COD dates for each of the T&SCos have been confirmed.

- 9.2 The scope of the CDS Contract shall include:
 - (a) provision of a data processing and management service for the processing of rawdata from the boundary flow meter Flow Meter and the Re-use Meter (if applicable) at each User's Delivery Point or Re-use Delivery Point (as applicable);

- (c) coverage across all T&S Networks.
- 9.3 The CDS shall act on its own account and not as agent of any Party in the provision of the services under the CDS Contract.
- 9.4 Each T&SCo must provide the CDS with such information relating to their respective T&S

 Networks as it reasonably requires for the purpose of performing its functions under the CDS

 Contract and such assistance as it may reasonably require in interpreting such information.
- 9.5 9.3-The terms of the CDS Contract must comply with the terms of reference set out in Annexure LAnnexure L (CDS Terms of Reference).
- 9.6 9.4 The T&SCos shall be required to:
 - jointly manage the CDS Contract and performance by the CDS of its obligations under the CDS Contract;
 - (b) bear the fees incurred under the CDS Contract equally between each T&SCo; and
 - (c) ensure that a CDS is in place at all times.
- 9.7 Each T&SCo undertakes that it will act and exercise its rights and powers under the CDS

 Contract in relation to the CDS so as jointly to control and govern the CDS on an economic, efficient and effective basis.
- 9.8 9.5-Where the CDS contract expires or is otherwise terminated in accordance with its terms, the T&SCos shall ensure that any replacement CDS is procured in accordance with paragraphs 9.2 and 9.3 above and so as to maintain continuous coverage in accordance with the principle set out in paragraph 9.4(c)9.6(c).
- 9.9 The Parties acknowledge and agree, in connection with the services to be performed by the CDS under the CDS Contract, that:
 - (a) the CDS Terms of Reference set out or describe the functions of the CDS as obligations of the CDS (as if the CDS were a party to the Code);
 - (b) the CDS is not a Party, and is not bound by any Code Agreement to comply with the Code; and
 - (c) <u>accordingly the CDS Contract operates to bind the CDS to perform those functions</u> assigned to it under the CDS Terms of Reference.
- 9.10 The rights and obligations of the Parties under the Code shall not be affected by any failure or delay on the part of the CDS in performing the services under the CDS Contract, so far as such rights and obligations are capable of being construed and determined notwithstanding

- such failure or delay, but without prejudice to any provision of the Code which expressly addresses any such failure or delay or the consequences thereof.
- 9.11 In the event of any conflict between the provisions of the Code and the provisions of the CDS Contract:
 - (a) <u>as between the Parties, the provisions of the Code shall prevail; and</u>
 - (b) any Party may propose (in accordance with paragraph 12) an amendment to the CDS Terms of Reference to remove such conflict).
- 9.12 Each User authorises the CDS to provide the Processed Flow Meter Data Statement to the T&SCo and LCCC in accordance with the CDS Contract.
- 9.13 Where there is only one (1) T&SCo, references in this paragraph 9 to "the T&SCos" shall be treated as references to the sole T&SCo that is a Party to the Code until such time as another T&SCo accedes to the Code.

CDS Data Disputes

- 9.14 9.6 Where a Party wishes to dispute the content of the Processed Flow Meter Data
 Statement ("CDS Data Dispute"), such dispute shall be referred to Expert Determination to shall be entitled to refer the dispute to an Expert for determination, in accordance with paragraph 9 of Section B51 of Section B (Governance).
- 10. Independent Verifier

Procurement

- 10.1 Each T&SCo shall be required must:
 - (a) to-procure an independent third party verifier (the "Independent Verifier") under a contract (the "Independent Verifier Appointment") to carry out the validation activities for each respective T&S Network required by paragraphs 4.56.1 and 6.2 of Section F 12.11 of Section F (Network Design and Specification) with the aim of providing a verification report that concludes with reasonable assurance that the Measurement Equipment at each Delivery Point (and, if applicable, the Re-Use Meter at the Re-Use Delivery Point) in the applicable T&S Network complies with the Measurement Requirements; and
 - (b) to ensure that the Independent Verifier is appointed by the date which is no later than six(6) months prior to itsthe Scheduled Commercial Operations Date for the applicable T&S Network.
- 10.2 The terms of the Independent Verifier Appointment must comply with the terms of reference set out in Annexure MAnnexure M (Independent Verifier Terms of Reference).
- 10.3 Each T&SCo shall be required to must:
 - (a) manage the Independent Verifier Appointment and performance by the Independent Verifier of its obligations under the Independent Verifier Appointment;

(b) subject to paragraph 6.4 of Section F 12.13 of Section F (Network design and Specification) Design and Specification), bear the fees incurred under the Independent Verifier Appointment; and

- (c) ensure that an Independent Verifier is in place at all times.
- 10.4 Where the Independent Verifier Appointment expires or is otherwise terminated in accordance with its terms, the T&SCos shall ensure that any replacement independent verifier is procured in accordance with paragraph 10.1 above and so as to maintain the principle of continuous coverage in accordance with paragraph (e)10.3(c).
- 11. T&S Network Portal

Procurement

- 11.1 Each T&SCo shall be required must:
 - (a) to procure a T&S Network Portal provider (the T&S Network Portal Provider) for its T&S Network under a contract (the Network Portal Service Contract); and
 - (a) at its cost establish and maintain a web-based portal to publish information, issue communications and receive communication relating to the operation of its T&S Network (the "T&S Network Portal"); and
 - (b) to ensure that the its T&S Network Portal Provider is appointed by the date which is no later than is fully operational six (6) months prior to its the Scheduled Commercial Operations Date for its T&S Network.
- 11.2 The terms of the Network Portal Service Contract must comply with the terms of reference setout in Annexure N (Network Portal Terms of Reference).
- 11.2 <u>11.3 Each The T&SCo shall be required to must ensure that its T&S Network Portal:</u>
 - (a) manage the Network Portal Service Contract;
 - (b) bear the fees incurred under the Network Portal Service Contract; and
 - (a) (c) ensure that a Network Portal is in place at all times.; and
- 11.4 Where the Network Portal Service Contract expires or is otherwise terminated in accordance with its terms, the T&SCos shall ensure that any replacement Network Portal Service Contract is procured in accordance with paragraph 11.1 above and so as to maintain continuous coverage in accordance with paragraph 11.3(c).
 - (b) has at least the features and functionalities set out in the T&S Network Portal Minimum Specification.
- 12. Revisions to Terms of Reference

Revisions to Terms of Reference

12.1 Where a Party (the **"Proposing Party"**), wishes to revise:

- (a) the CDS Terms of Reference; or
- (b) the Independent Verifier Terms of Reference; or,

(c) the Network Portal Terms of Reference

any such revision shall be subject to the procedure set out in paragraph paragraphs 12.2 to 12.9 below.

- 12.2 The Proposing Party shall develop and deliver to:
 - (a) each other Party;
 - (b) the LCCC; and
 - (c) the <u>Authority:</u> <u>Regulator</u>,

the proposed amendments to the applicable Terms of Reference ("ToR Proposal") for review and comment in accordance with paragraphs 12.3 to 12.5-below.

- 12.3 As soon as reasonably practicable and in any event within twenty (20) Business Days of receipt of thea ToR Proposal, each Party shall notify the Proposing Party of any comments on the ToR Proposal.
- When providing comments to the Proposing Party in accordance with paragraph 12.3, each Party shall be requiredmust to provide the following:
 - (a) information about the Party (company, registered office, etc.) and about at least one
 (1) reference person (name, telephone number, fax number, e-mail address, etc.) who can be contacted with respect to that Party's comments;
 - (b) the reasons why that Party believes its comments should be incorporated; and
 - (c) any documentation (analyses, reports, etc.) to support the request.
- 12.5 The Proposing Party shall:
 - (a) give reasonable consideration to any comments provided by a Party pursuant to paragraph 12.3; and
 - (b) where the Proposing Party does not take on board or incorporate any comments provided by a Party pursuant to paragraph 12.3, provide an explanation to the Party of the reasons for this decision.
- Once all comments have been received and considered pursuant to the process set out in paragraphs 12.3 to 12.5, the Proposing Party shall deliver to:
 - (a) each other Party;
 - (b) the LCCC; and
 - (c) the Authority Regulator,

the final draft of the revised Terms of Reference ("Revised Terms of Reference").

12.7 Subject to paragraph 12.8, any Revised Terms of Reference, shall be implemented as a Self-Governance agreed under this paragraph 12, shall be put forward by the Proposing Party as a Modification Proposal viaunder the Modification Procedure in Section B (Governance).

Disputes

12.8 Where:

- (a) a PartyParty's comment provided pursuant to paragraph 12.3 of this Section I was not included in the Revised Terms of Reference; and
- (b) a majority of other Parties <u>considers</u> consider that such comment should have been incorporated by the Proposing Party,

then such matter may be referred by the Party who made the original comment to dispute resolution in accordance with the provisions of paragraph 8 of Section B (Governance).

12.9 Where any matter is referred to dispute resolution under paragraph 12.8, the Proposing Party shall not raise a Self-Governance Modification Proposal until such dispute has been resolved in accordance with the provisions of paragraph 848 of Section B (Governance).

CCS Network Code

Section J – General CCS Network Code

Section J – General

1. Notices and communications

1.1 Except where the means by which a particular Communication is to be given is specified in the Code, such Communication shall Communications must be given by Conventional Notice.

1.2 <u>"Conventional Notice"</u>:

- (a) 1.2 References in this paragraph to "Conventional Notice" are to includes any Communication or other notice to be given by one (1) Party to another under the Code or an Ancillary Agreement, other than one which is given by means of the T&S Network Portal or by telephone, which satisfies the requirements of paragraph 1; and
- (b) 1.3 Any Conventional Notice shallmust be in writing and shallmust be addressed to the recipient Party at the recipient Party's address or e-mail address notified pursuant to paragraph 1.4 1.3 (and where there is more than one (1) address and/or representative specified pursuant to paragraph 1.6, all such addresses and representatives) and marked for the attention of the relevant representative (identified by name or title) in accordance with paragraph 1.41.3.
- 1.3 1.4 The initial address, e-mail address and details of the representative for whose attention notices are to be marked shall be as specified in a Connection Agreement and any changes shallmust be communicated in writing by each Party to the other Party before taking effect.
- 1.4 1.5 Any Conventional Notice given by delivery shall be given by letter must be delivered by hand, and any notice Conventional Notice given by post shall must be sent by first class prepaid post (airmail if overseas).
- 1.5 4.6-Any Conventional Notice shall be deemed to have been received:
 - (a) in case of delivery by hand, when delivered; or
 - (b) in the case of first class prepaid post, on the second Day following the Day of posting (or, if sent airmail overseas or from overseas, on the fifth Day following the Day of posting); or
 - (c) in the case of e-mail, subject to paragraph 1.8, shall be deemed to have been received 1.7, one (1) hour after being sent in the absence of any undeliverable return receipt received bybeing sent to the sender during that period.
- 1.6 1.7 A Party may specify different addresses and representatives pursuant to paragraph 1.3 for the purposes of Conventional Notices of different kinds or relating to different matters.
- 1.7 4.8 If the time at which any Conventional Notice sent by e-mail is deemed to have been received falls after 1700 hours on a day Day, the Conventional Notice shall be deemed to have been received at the start of the next Business Day.

Section J – General CCS Network Code

1.8 4.9 Where a Conventional Notice is sent by e-mail, the Party giving the Conventional Notice shall (but without prejudice to paragraph 4.6(c)1.5(c)) if requested by the recipient Party, resend as soon as reasonably practicable the Conventional Notice by email.

- 1.9 Where any provision of the Code or any Ancillary Agreement specifies any requirement to be complied with by any Party in respect of any specific Communication, such requirement shall be in addition to (and, to the extent inconsistent, in substitution for) the provisions of this paragraph 1.
- 2. **Communication by telephone**
- <u>2.1</u> <u>For the purposes of enabling Communications to be given (where required or permitted to be so given) by telephone:</u>
 - (a) each Party must provide to the other Party not more than three (3) (or such other number as they may agree) telephone numbers and details (by name or title) of the representative(s) to whom the Party giving such a Communication should speak;
 - (b) each Party must use reasonable endeavours to ensure that a Party seeking to give such Communication will at any time be able to contact a representative (of the first Party) by means of one (1) of such telephone numbers; and
 - (c) the Parties must, if either of them shall so request, establish such further procedures as may be reasonable and appropriate for the purposes of ensuring:
 - <u>(i)</u> that a Communication being given by telephone may be identified by the recipient as such; and/or
 - (ii) that such Communication may be given securely, without delay and effectively.
- 2.2 Where a Party seeking to give a Communication by telephone is unable to contact a representative of the receiving Party, such Party must give the Communication by e-mail and the Communication will not be deemed to have been received except in accordance with paragraph 1.5(c).
- 2.3 <u>Unless otherwise agreed between the relevant Parties, a telephone notice may not be given</u> as a message recorded on a telephone answering device.
- 2.4 Where a Communication is given by telephone:
 - (a) the Party giving the Communication will promptly after the telephone Communication is completed, make and keep a record in which the time and content of the telephone Communication is logged, but may do so by recording the telephone Communication where it has notified the recipient Party on the occasion or on a standing basis, of its intention to do so; and
 - (b) the Communication shall be treated as given at the time at which the telephone Communication is completed.

Section J – General CCS Network Code

2.5 A Party may specify different telephone numbers and representatives pursuant to paragraph
2.1 for the purposes of receiving by telephone Communications of different kinds or relating to different matters.

3. 2. Alternative means of communication

- - (a) the Code specifies that any Communication or other notice is to be provided by a User to the-T&SCo, or by the-T&SCo to the User, by means of the T&S Network Portal; and
 - (b) as a result of a technical fault or other <u>circumstancescircumstance</u> the T&S Network Portal is not functioning or one <u>(1)</u> of the <u>partiesParties</u> is, through unforeseen circumstances, unable to access the T&S Network Portal,

then the partyParty experiencing the difficulties specified in paragraph 2.1(b)3.1(b) shall make the relevant Communication or notice by means of telephone or e-mail (depending on which is most appropriate in the circumstances) using the other party's Party's Alternative Operational Communication Contact Details.

- 3.2 2.2 The T&SCo shallmust provide to Users, and each User shallmust provide to the T&SCo in writing, the "Alternative Operational Communication Contact Details", which shallmust consist of:
 - (a) an e-mail address; and
 - (b) a telephone number,

which will be closely monitored by the <u>partiesParties</u> for the purposes of receiving any Communications or other notices made in accordance with paragraph <u>2.13.1</u>.

- 3.3 2.3 Any changes to a party's Party's Alternative Operational Communication Contact Details must be notified to the other party Party or parties as soon as is reasonably practicable.
- 4. 3. Liability between the T&SCo and a User
- 4.1 3.1 For the purposes of this paragraph 34, each of:
 - (a) the T&SCo; and
 - (b) the User that is a counterparty to the T&SCo under a Connection Agreement,
 - (c) an Eligible Applicant; and
 - (d) a Prospective T&SCo,

shall be a "Counterparty".

Liability

4.2 Subject to the remainder of this paragraph 34 and any other express provision to the contrary, no Party shall be liable to its Counterparty for any claims, losses, damages, costs,

expenses or liabilities arising from any default, negligence or breach of duty (whether statutory or otherwise) or any breach of this the Code or an Ancillary Agreement (or the Party's performance of its obligations under this the Code or an Ancillary Agreement), except for any such claims, losses, damages, costs, expenses or liabilities directly resulting from such default, negligence or breach and which was reasonably foreseeable as not unlikely to occur in the ordinary course of events from such default, negligence or breach in respect of:

- (a) physical damage to the property of the Counterparty; and/or
- (b) the liability (in law) of the Counterparty to any third party for loss related to physical damage to the property of that third party.
- 4.3 3.3 Where the T&SCo elects to accept delivery of carbon dioxide which does not comply with the Entry Provisions for delivery into the T&S Network-pursuant to paragraph 3.1(b) of Section F4 or 12 of Section F (Network design and specification), then Design and Specification), it thereby takes responsibility for the impact of such carbon dioxide on the T&S Network.
- 4.4 Note: this provision Paragraph 4.3 is without prejudice to any liability at law between the Parties where there is no such acceptance of a User to a T&SCo where the T&SCo does not accept delivery of carbon dioxide which does not comply with the Entry Provisions pursuant to paragraph 4 or 12 of Section F (Network Design and Specification).
- 4.5 3.4 The amount or amounts for which a Party may be liable to a Counterparty pursuant to paragraph 3.24.2 in respect of any one (1) event or circumstance shall not exceed:
 - (a) in respect of the liability of the T&SCo to any one (1) User or of any one (1) User to the T&SCo, £[20twenty million] pounds sterling (£20,000,000); or
 - (b) in respect of the liability in aggregate of the T&SCo to more than one 1 User, figure 100,000,000,000.

Exclusions and exceptions

- 4.6 3.5 No Subject to paragraph 4.7, no Party shall in any circumstances circumstance be liable in respect of any breach of this the Code or anany Ancillary Agreement (or the Party's performance of its obligations under this the Code or any Ancillary Agreement) to any Counterparty for:
 - (a) any loss of profit, loss of revenue, loss of use, loss of contract, loss of goodwill, or increased cost of working; or
 - (b) any indirect or consequential loss, including any ETS Liabilities; or
 - (c) except as provided in paragraphs 3.2(b)4.2(b) and 3.64.8, loss resulting from the liability of any Counterparty to any other person howsoever and whensoever arising.
- 4.7 Where:

<u>a fault or damage in the infrastructure owned by a User between the Metering Point and the Delivery Point arises:</u>

- (b) such fault or damage results in carbon dioxide losses by way of leakage and/or venting;
- (c) <u>a T&SCo ETS Liability arises as a result of such leakage and/or venting; and</u>
- (d) the T&SCo establishes (pursuant to its investigation and audit rights under Section F

 (Network Design and Specification)) that such fault or damage arose on the infrastructure on the side of the Delivery Point owned by the User,

such User shall be liable to compensate the T&SCo for all costs associated with such ETS Liabilities.

- 4.8 3.6 Nothing in this the Code or any Ancillary Agreement excludes or limits the liability of any Party:
 - (a) for death or personal injury resulting from the negligence of such Party; or
 - (b) where such a limitation or exclusion would be contrary to applicable law.
- 4.9 3.7 Paragraph 3.24.2 is without prejudice to any provision of the Code or any Ancillary Agreement which provides for a Party to make a payment to a Counterparty.
- 4.10 3.8 Nothing in this the Code or any Ancillary Agreement shall prevent any T&SCo or User from seeking injunctive relief, specific performance or other equitable relief.
- 4.11 3.9 The Subject to paragraph 4.4, the rights and remedies of the Parties pursuant to the Code and any Ancillary Agreement exclude and are in place of any rights or remedies of any Party in tort (including negligence and nuisance) or misrepresentation in respect of the subject matter of the Code or such Ancillary Agreement and accordingly, but without prejudice to paragraphs 3.64.8, each Party (to the fullest extent permitted by law):
 - (a) waives any <u>such</u> rights or remedies; and
 - (b) releases the Counterparty from any duties or liabilities arising in tort or misrepresentation in respect of the subject matter of the Code or such Ancillary Agreement.
- 4.12 3.10-Without prejudice to paragraph 3.94.11, where any provision of the Code or any Ancillary Agreement provides for any amount to be payable by a Party upon or in respect of that Party's breach of any provision of the Code or any Ancillary Agreement, each Party agrees and acknowledges that the remedy conferred by such provision is exclusive of and is in substitution for any remedy in damages in respect of such breach or the event or circumstance giving rise thereto.
- 3.11-Nothing in this paragraph 34 shall prevent any Party from or restrict it in enforcing any obligation owed to it under or pursuant to the Code, the Code Agreement or any Ancillary Agreement.

4.14 Each provision of this paragraph 4 shall be construed as a separate and severable contract term, and shall as respects any Discontinuing User, survive that Discontinuing User ceasing to be a User.

- 5. 4-Liabilities between Users, between T&SCos and cross-network liabilities
- 4.1 Unless otherwise provided, nothing in the Code or any Ancillary Agreement creates contractual rights or liabilities:
 - (a) between Users inter se;
 - (b) between T&Scos SCos inter se; or
 - (c) between Users of a T&S Network and the T&SCo of another T&S Network.
- 5.2 4.2 Each:
 - (a) T&SCo:
 - (i) waives any rights or remedies against any:
 - (A) other T&SCo; and
 - (B) User other than the User which is its Connection Agreement counterparty; and
 - (ii) releases:
 - (A) each other T&SCo; and
 - (B) each User other than the User which is its Connection Agreement counterparty; and
 - (b) User:
 - (i) waives any rights or remedies against any:
 - (A) other User; and
 - (B) T&SCo other than the T&SCo which is its Connection Agreement counterparty; and
 - (ii) releases:
 - (A) each other User; and
 - (B) T&SCo other than the T&SCo which is its Connection Agreement counterparty,

from of-any duties or liabilities arising in tort (including negligence and nuisance) or misrepresentation in respect of the subject matter of the Code or any Ancillary Agreement.

6. 5. Warning notices

- 6.1 Where:
 - (a) a User is in material or persistent breach of the Code; and
 - (b) such breach is not a Payment Default or Insolvency Default,

the T&SCo may give a warning notice of such material or persistent breach to the User (the "Warning Notice") which:

- (i) requires the User to desist from failing to comply with the Code; and
- (ii) includes details of how the User has materially or persistently failed to comply with the requirements of the Code-; and
- requires the User to take steps to remedy the breach in all material respects, where the breach is capable of remedy within the grace period set out in paragraph 6.2.
- 6.2 (c) the The User shall be permitted a "grace period" of fourteen (14) Days to implement measures to assist the User in complying with the Code; and from the date of a Warning Notice to take steps to remedy the relevant breach in all material respects.
- 6.3 (d) for the avoidance of doubt, any Any non-compliance with the Code by the User during the grace period referred to in paragraph (e)6.2 shall not entitle the T&SCo to serve another Warning Notice on the User.
- 6.4 Regardless of whether a breach is remedied during the grace period referred to in paragraph
 6.2, any Warning Notice issued under paragraph 6.2 shall continue to apply for the purposes
 of paragraph 7.1.
- 6.5 For the purposes of paragraph 6.1, the following breaches are excluded:
 - <u>a breach which results from a breach by the T&SCo of the Code or an Ancillary</u>

 Agreement; or
 - <u>a breach other than a wilful breach of a provision of the Code where the Code</u>

 <u>specifically provides some other remedy for such breach and such other remedy may reasonably be considered to be adequate in the circumstances.</u>

7. 6. Default and remedy

6.1 Where:

- 7.1 (a) Where, having received at least three (3) Warning Notices within the preceding twelve (12) months, the material or persistent breach identified in such Warning Notices remains unremedied in any material respect by the User; and
- (b), the such breach is not a Payment Default or Insolvency Default, then

T&SCo may give notice of such breach to the User ("Default Notice").

- 7.2 6.2 Within fourteen (14) Days of such notice receipt of a Default Notice, the User must:
 - (a) where the breach is reasonably capable of remedy within such period of fourteen (14) Days, remedy the breach in all material respects;
 - (b) where the breach is capable of remedy but not reasonably within such period of fourteen (14) Days, provide to the T&SCo a remediation plan setting out the steps to be taken by the User and the timetable for taking such steps for the remedy of the breach as soon as is reasonably practicable; or
 - (c) where the breach is not capable of remedy, provide a remediation plan setting out:
 - (i) the steps to be taken by the User and the timetable for taking such steps to mitigate the impact of the breach so far as is reasonably practicable; and
 - (ii) the steps to be taken to prevent the recurrence or re-occurrence of the relevant breach so far as is reasonably practicable.
- 7.3 Where, notwithstanding the reasonable diligence of the User, it is not reasonably practicable for the User to remedy the breach in accordance with a remediation plan provided pursuant to paragraph 7.2(b), the User may provide to the T&SCo a revised remediation plan setting out the steps to be taken by the User and the timetable for taking such steps for the remedy of the breach as soon as is reasonably practicable (provided that the User shall have the opportunity to provide only one revised remediation plan to address the breach).
- 7.4 Where, notwithstanding the reasonable diligence of the User, it is not reasonably practicable for the User to comply in all material respects with a remediation plan provided under paragraph 7.2(c), the User may provide to the T&SCo a revised remediation plan setting out the matters in paragraph 7.2(c) (provided that the User shall have the opportunity to provide only one (1) revised remediation plan to address the breach).
- 6.3 For the purposes of paragraphs 6.1 and 6.2 the following breaches are excluded:
 - (a) a breach which results from a breach by T&SCo of the Code or an Ancillary Agreement;
 or
 - (b) a breach other than a wilful breach of a provision of the Code where the Code specifically provides some other remedy for such breach and such other remedy may reasonably be considered to be adequate in the circumstances.
- 8. **7.** General Default
- 8.1 A "General Default" occurs when a Default Notice has been issued and:
 - (a) paragraph 6.2(a)7.2(a) applies and the breach remains unremedied in any material respect after the expiry of seven (7) Days after a further notice by the T&SCo to the User to the effect that the User has not complied with paragraph 6.2(a)7.2(a); or
 - (b) paragraph 6.2(b) 7.2(b) or 7.3 applies and the User does not:

(i) remedy the breach in all material respects with all reasonable diligence and so far as reasonably practicable in accordance with the remediation plan provided under 6.2(b)paragraph 7.2(b) or a revised remediation plan pursuant to paragraph 7(b)(ii) below; or

- (ii) 7.3 (where a revised remediation plan has been provided) where notwithstanding the reasonable diligence of the User it is not reasonably practicable for the User to remedy the breach in accordance with that remediation plan, provide to T&SCo a revised remediation plan,
- and such failure remains unremedied in any material respect after the expiry of seven (7)

 Days after a further notice by the T&SCo to the User to the effect that the User has not complied with this paragraph 7(b)8.1(b);
- (c) paragraph $\frac{6.2(c)}{7.2(c)}$ or $\frac{7.4}{2}$ applies and:
 - (i) the User does not:
 - (i) (A) the User does not comply in all material respects with the remediation plan provided under 6.2(c) paragraph 7.2(c) or a revised remediation plan pursuant to paragraph below; or
 - (B) 7.4 (where a revised remediation plan has been provided) where notwithstanding the reasonable diligence of the User it is not reasonably practicable for the User to comply in all material respects with that remediation plan, provide to T&SCo a revised remediation plan.
 - and such failure remains unremedied in any material respect after the expiry of seven (7) Days after a further notice by the T&SCo to the User to the effect that the User has not complied with this paragraph 7(c)8.1(c); or
 - (ii) at any time within the period of twelve (12) months following T&SCo'sthe
 Default Notice, there occurs a further material breach by the User of the same
 provision of the Code; and the T&SCo has given a notice of such further
 breach to the User and a period of seven (7) Days has expired following such
 notice.
- 9. 8. Insolvency Default
- 9.1 8.1 An An "Insolvency Default" occurs, save if and to the extent the provisions of section 233B of the Insolvency Act 1986 apply, when a User:
 - (a) is dissolved (other than pursuant to a solvent consolidation, amalgamation or merger) or becomes insolvent or unable to pay its debts within the meaning of section 123 of the Insolvency Act 1986;
 - (a) is unable to pay its debts (within the meaning of Section 123(1) or (2) of the Insolvency
 Act 1986, but subject to paragraph 8.2, or any voluntary arrangement is proposed inrelation to it under Section 1 of that Act or it enters into any composition or scheme of

- arrangement (other than for the purpose of a bona fide solvent reconstruction or amalgamation); or
- (b) has a receiver (which expression shall include an administrative receiver within the meaning of Section 29 of the Insolvency Act 1986) of the whole or any material part of its assets or undertaking appointed; or
- (c) has an administration order under Section 8 of the Insolvency Act 1986 made in relation to it; or
- (b) (d) passes any has an order made or resolution passed for its winding-up or liquidation (other than for the purpose of a bona fide pursuant to a solvent reconstruction erconsolidation, amalgamation or merger); or
- (e) becomes subject to an order by the High Court for winding-up; or
- (f) becomes subject to a bankruptcy order; or
- (c) (g) becomes subject to anany event made in a with respect to it which, pursuant to the applicable laws of any jurisdiction outside England and Wales, equivalent or, has an analogous effect to any one or more of those the events listed in paragraphs 8.1(a) to 8.1(f) above specified in paragraph 9.1(a) or 9.1(b).
- 8.2 For the purposes of paragraph 8.1(a), Section 123(1)(a) of the Insolvency Act 1986 shall have effect as if for '£750' there was substituted £10,000, and the User shall not be deemed to be unable to pay its debts for the purposes of that paragraph if any such demand as is mentioned in the said Section is being contested in good faith by the User with recourse to all appropriate measures and procedures.

10. 9. Payment Default

- A **"Payment Default"** occurs where in relation to anyan amount (or amounts in aggregate) of not less than ten thousand pounds sterling (£10,000 which) has become due for payment by the User under the Code and:
- (a) the T&SCo has issued a notice of demand for payment of the outstanding amount (a

 Late Payment Notice); and for the outstanding amount and the User has not paid such
 amount in full by the thirtieth Day after the date of the Late Payment Notice; or
- (b) the <u>T&SCo has issued a notice of non-payment in respect of such amount (where it relates to an amount that is not the subject of a Late Payment Notice) (a <u>"Non-Payment Notice") and the</u> User has not paid such amount in full by the <u>thirtieth</u> (30th) Day after the date of the <u>Late Payment Non-Payment Notice</u>.</u>

11. 40. Right to suspend

11.1 40.1 Without prejudice to any other rights of the T&SCo under the Code, during any period in which a Default Notice or Late Payment Notice or Non-Payment Notice has been issued and such breach of the Code or failure to pay is subsisting, or during any period while an

Insolvency Default is subsisting, the T&SCo shall be entitled to reject or refuse to accept all or any of the following by the relevant User:

- (a) delivery of carbon dioxide at the User's Delivery Point; er
- (b) <u>taking delivery of carbon dioxide at the User's Re-Use Delivery Point; or</u>
- (c) (b) an application for Registered Capacity or increased Registered Capacity at any Delivery Point under Section E (network use and capacitySection C (Connection).
- 11.2 10.2 A User shall remain liable to pay Capacity Charges and Network Charges during any period in which paragraph 911.1 applies.
- 12. 11. Default Termination
- 12.1 11.1 Upon Following the occurrence of a User Default, the T&SCo may give notice (a "Termination Notice") to the defaulting User to the effect that the User shallwill cease to be a User of or in relation to the T&S Network with effect from the date (which may be any date on or after the date on which the notice is given) specified in the notice.specified in the Termination Notice (which must be on or following the date of the Termination Notice in respect of an Insolvency Default or be no less than seven (7) Days after the date of issue of the Termination Notice in respect of any other User Default).
- 12.2 Where the T&SCo gives a Termination Notice to a User, with effect from the date specified in the notice Termination Notice, the User will cease to be a User of the T&S Network and paragraph 13.214.2 shall apply.
- 12.3 11.3 The giving of a Termination Notice and the application of paragraph 11.2 12.2 shall not affect the rights and obligations of the T&SCo and the User under the Code or any Ancillary Agreement (including rights and obligations in respect of the User Default, and in respect of amounts including interest payable by either Party, and rights and obligations arising pursuant to any provision of the Code in respect of the User's User ceasing to be a User) accrued up to the date referred to in paragraph 11.2 12.2, which shall continue to be enforceable notwithstanding that paragraph 12.2.
- Where the T&SCo has given a Termination Notice it shall be entitled to notify such persons as it thinks fit (including another T&SCo) that it has done so and it shall notify the CDS that it has done so.
- 13. 12. User discontinuationexit
- 13.1 12.1 ASubject to the satisfaction of the requirements of paragraph 13.2, a User may by giving notice (Discontinuance a "User Exit Notice") to the T&SCo apply to cease to be a User of or in relation to the T&SCo Network:
 - (a) where it no longer intends to hold Registered Capacity;
 - (b) while a Construction Longstop Default affecting the User is subsisting in respect of a failure of the T&SCo to complete the relevant T&SCo Works; or

where a Capacity Constraint Default affecting the User is subsisting in respect of a Capacity Constraint.

Note: the definition of "Capacity Constraint Default" is still being developed, but it is intended to reflect a prolonged outage.

- 13.2 12.2 A User may not cease to be a User under this paragraph 1213 until such time as:
 - (a) where paragraph 13.1(a) applies, the earlier of:
 - (i) the date that the User's Registered Capacity has expired;
 - (ii) where paragraph 14.3(a)(i) applies, the later of:
 - (A) the date that is three (3) years after the date of a User Exit Notice; and
 - (B) the date that the User has paid all Capacity Charges and Network
 Charges due in respect of the three (3) year period after the date of issue of the User Exit Notice in full; or
 - (iii) (a)-where paragraph 12.1(a)14.3(c) applies, the date that the User's Registered Capacity has expired been surrendered and registered to another User;
 - (b) subject to paragraph <a href="https://dx.da/.nlm.nih.google-new-red-color: blue-new-red-color: blue-new-red-color: blue-new-new-red-color: blue-new-red-color: blu
 - (c) any requirements under any Ancillary Agreement in respect of termination have been complied with; and
 - (d) any outstanding breach, being a breach capable of remedy and of which the T&SCo has given notice to the User, by the User of any provision of the Code or any Ancillary Agreement shall have has been remedied.
- 13.3 Here a User has given notice under paragraph 12.1a User Exit Notice, the User and the T&SCo shall remain bound by the Code and any Ancillary Agreement to which the User is party until the requirements of paragraph 12.2paragraphs 13.2 and 13.4 are satisfied.
- 13.4 Where a User has given notice under paragraph 12.1a User Exit Notice after the satisfaction of the last of the requirements of paragraph 12.213.2 to be satisfied:
 - (a) with effect from the 5th fifth Business Day following such satisfaction, the User will cease to be a User; and
 - (b) without prejudice to paragraph <u>12.5_13.5</u>, <u>the T&SCo willmust</u> as soon as reasonably practicable (and where possible before such date) <u>informnotify</u> the User of the date on which it ceases to be a User under paragraph <u>(a)13.4(a)</u>.

13.5 12.5 Notwithstanding paragraph 12.3 13.3, the T&SCo or (as the case may be) the User shall remainremains liable, subject to and in accordance with the Code, to the other and after the User Discontinuance Date:

- (a) for any amount which was or becomes payable under the Code or any Ancillary Agreement in respect of any period before the User Discontinuance Date; and
- (b) in respect of any outstanding breach of any provision of the Code or any Ancillary Agreement where such breach was not (for the purposes of paragraph 12.2(d)13.2(d)) capable of remedy or (notwithstanding that paragraph) was capable of remedy but was not remedied.

14. **13.** Discontinuing Users and termination

- 14.1 A User may cease to be a User of or in relation to a T&S Network pursuant to paragraph 11 or 12 and for the purposes of the Codeor 13. A:
 - (a) -a_"Discontinuing User" is a User who so ceases to be a User; and the-
 - (b) <u>"User Discontinuance Date"</u> is the date with effect from which (in accordance with paragraph <u>11 or 12 or 13</u>) a Discontinuing User ceases to be a User.
- 14.2 <u>13.2-Upon a User's ceasing to be athe</u> User <u>Discontinuance Date</u>:
 - (a) subject to any provision of the Code expressed to survive termination and to paragraph 41.312.3, the Code Agreement shall cease to bind the Discontinuing User and (as respects solely in connection with the Discontinuing User) the T&SCo; and
 - (b) each Ancillary Agreement to which a Discontinuing User is party shall, unless otherwise provided in such Ancillary Agreement, terminate as respects that User (but without prejudice to the continuance of that Agreement as respects any other User(s) party thereto) with effect from the User Discontinuance Date.
- 14.3 A Discontinuing User's Registered Capacity shall not be reduced or cancelled (and the User will remain liable for payment of Capacity Charges in respect thereof but mayof its

 Registered Capacity but may (subject to agreement with the T&SCo) elect to make prepayment thereof of the Capacity Charges in full discharge of such liability) other than:
 - (a) Long-term Registered Network Capacity after the date which is three (3) years from the Discontinuance Date;
 - (i) date of a User Exit Notice; or
 - (ii) <u>User Discontinuance Date where a Termination Notice has been issued and this paragraph 14.3(a)(ii) shall survive the User Discontinuance Date;</u>
 - (b) Registered Capacity from the <u>User Discontinuance Date</u>, where a <u>Discontinuance User Exit Notice</u> has been issued under paragraph <u>12.1(b)</u>13.1(b) or <u>[0]</u>-13.1(c); or

- (c) by agreement with the T&SCo, where a full surrender of Registered Capacity has occurred under paragraph 27 of Section E (Network Use and Capacity) and such capacity has become registered to another User.
- 14.4 <u>13.4-</u>A Discontinuing User will remain liable for payment of Network Charges for the period in which it is liable for Capacity Charges in accordance with paragraph <u>13.314.3</u>.
- 14.5 Hollowing the User Discontinuance Date, each party Party shall bear responsibility for:
 - (a) the safe decommissioning of the connection assets owned by that partyParty; and
 - (b) all decommissioning costs and liabilities (including any such liabilities arising under relevant legal requirements Legal Requirements) associated with the decommissioning of such assets,

based on the boundaries of ownership set out in the Connection Agreement between the T&SCo and the Discontinuing User.

15. 44. Governing law

The Code, the Code Agreement, every Code Accession Agreement, every CDS Accession Agreement and every Ancillary Agreement, and any dispute, controversy, proceedings proceeding or claim of whatever nature arising out of or in any way relating to the Code, the Code Agreement, any Code Accession Agreement, any CDS Accession Agreement or any Ancillary Agreement or their formation (including any non-contractual disputes or claims), shall be governed by and construed in accordance with English law.

16. 45. Jurisdiction

- Subject and without prejudice to the provisions providing for mediation and expert determination in Section B (Governance), all the Parties irrevocably agree that the courts of England shall have exclusive jurisdiction to hear, settle and/or determine any dispute, controversy or claim (including any non-contractual dispute, controversy or claim) arising out of or in connection with the Code, the Code Agreement, any Code Accession Agreement, any CDS Accession Agreement or any Ancillary Agreement, including any question regarding its existence, validity, formation or termination. For these purposes, each Party irrevocably submits to the jurisdiction of the English courts.
- Any User which is not a company incorporated under the Companies Act 1985 shall provide to each T&SCo an address in England or Wales for service of process on its behalf in any proceedings.

17. 46. Third party rights

Unless expressly provided otherwise, nethe Parties do not intend that any term of any Code
Document is intended to shall be enforceable by third parties and the solely by virtue of the
Contracts (Rights of Third Parties) Act 1999 by any person who is not a Party and, subject to
paragraph 47.5 of Section B (Governance), paragraphs 1.3 and 11.3 of Section C
(Connection), paragraph 3.13 of Section F (Network Design and Specification), paragraph 7.3
of Section G (Common Interface Procedures) and paragraph 6.24 of Section I (Data), the
operation of the Contracts (Rights of Third Parties) Act 1999 is hereby excluded. Where an

obligation is expressly stated to be enforceable by a third party, the consent of that third partywill not be required to rescind or vary the relevant term.

Notwithstanding any express provision of the Code pursuant to which paragraph 17.1 is disapplied in relation to a term of any Code Document, the Parties may, to the extent otherwise permitted under the Code, rescind, vary, waive, release, assign, novate or otherwise dispose of all or any of their respective rights or obligations under the Code or a Code Document without the consent of any person who is not a Party.

18. 47. Severance

- 18.1 47.1 If any provision of this the Code, the Code Agreement, any of the Code Accession
 Agreements, any of the CDS Accession Agreements or any of the Ancillary Agreements (the
 "Code Documents") is or becomes (whether or not pursuant to any judgment or otherwise)
 invalid, illegal or unenforceable in any respect under the law of any jurisdiction:
 - the validity, legality and enforceability under the law of that jurisdiction of any other provision; and
 - (b) the validity, legality and enforceability under the law of any other jurisdiction of that or any other provision,

shall not be affected or impaired in any way thereby.

18.2 17.2 If any provision of any Code Document shall be held to be void or declared illegal, invalid or unenforceable for any reason whatsoever, such provision shall be divisible from the Code Document and shall be deemed to be deleted from the Code Document and the validity, legality and enforceability of the remaining provisions shall not be affected.

19. **18.** Entire agreement

- Each of the Code Documents represents the entire understanding, and constitute constitutes the whole agreement of the Parties in relation to their the subject matter and supersede anythereof and supersedes any and all previous agreementagreements or understandings between the Parties with respect thereto and, to the fullest extent practicable under the relevant law, and without prejudice to the generality of the foregoing, exclude any warranty, condition or other undertaking implied at law or by custom.
- 19.2 Each Party acknowledges that in entering into any Ancillary Agreement, any Code
 Agreement, any Code Accession Agreement or any CDS Accession Agreement, it does not
 rely on any representation, warranty or other understanding not expressly contained in the
 Code, any Ancillary Agreement, any Code Agreement, any Code Accession Agreement or
 any CDS Accession Agreement.
- Nothing contained in a document (other than an Ancillary Agreement) referred to in the Code, beyond what is expressly contemplated by the Code as being contained in such document or is necessary for the purposes of giving effect to a provision of the Code, shall modify or have any effect for the purposes of the Code or be construed as relevant to the interpretation of the Code.

19.1 A waiver of any term, provision or condition of, or consent granted under, any Code Documentshall be effective only if given in writing and signed by the waiving or consenting party and then only in the instance and for the purpose for which it is given.

- 20.1 19.2-No failure or delay on the part by or omission of any partyParty in exercising any right, power or privilege or remedy under any Code Document Documents shall operate to impair such right, power, privilege or remedy or be construed as a waiver thereof, nor shall any.
- 20.2 <u>Any</u> single or partial exercise of any <u>such</u> right, power<u>er</u>, privilege <u>or remedy shall not</u> preclude any other or <u>further future</u> exercise thereof or the exercise of any other right, powerer, privilege <u>or remedy</u>.
- 20.3 49.3 No breach of any provision of any Code Document shall be waived or discharged except with the express written consent of the partiesParties.
- 20.4 19.4 The rights and remedies herein provided are cumulative with and not exclusive of any rights or remedies provided by law.

21. 20. Language

Every communication or notice to be given by one Party to another under any Code Document shall be in the English language.

22. 21. Assignment and transfer

- 22.1 21.1 Neither Party shall assign or transfer its rights and/or obligations under this the Code and any Ancillary Agreement except in accordance with the provisions of clauses 20.2 paragraphs 22.2 and 20.322.3.
- 22.2 21.2 The User shall only be entitled to assign or transfer its rights and or obligations under the Code and any Ancillary Agreement to a third party-where:
 - where the User assigns all (but not part only) of its rights and benefits under the Code by way of security to or in favour of a Lender; or
 - (b) where:
 - (i) (a) the User <u>also</u> assigns <u>or transfers</u> the ownership and operation of the User Facility to the third party;
 - (ii) (b) the User has obtained the prior written consent of the T&SCo, which shall not be unreasonably withheld; and
 - (iii) (c) the third party becomes bound by the Code by entering into a Code Accession Agreement.
- 22.3 21.3 A T&SCo shall sonly be entitled to assign or transfer its rights and/or obligations under this the Code and any Ancillary Agreement to a third party where:

(a) <u>the T&SCo also</u> assigns <u>or transfers</u> the ownership and operation of the T&S Network to <u>athe</u> third party <u>holdingwho also holds</u> a Licence; <u>and</u>

- (b) the third party becomes bound by the Code by entering into a Code Accession Agreement—; and
- (c) the third party becomes bound by the CDS Contract by entering into a CDS Accession Agreement.
- Where a Party transfers its obligations under the Code or any Ancillary Agreement to a person, that Party shall be released from its obligations under the Code and any such Ancillary Agreement arising after the time at which the transfer is effective, but shall remain liable for any obligations accruing up to such time.
- 22.5 Any reference in the Code or any Ancillary Agreement to any Party shall include a reference to that Party's successors and assigns.

CCS Network Code

Section K - Glossary

In this the Code, the following words and expressions shall have the following meanings, unless the context otherwise requires:

Term	Meaning
Act	means the Energy Act <u>2023</u> .
Adjustment Period	has the meaning given in paragraph 7.1 of Section F (network design and specification 13.1 of Section F (Network Design and Specification).
Affected User	means a User likely to be affected by Programmed Maintenance or Reactive Maintenance.
<u>Affiliate</u>	has the meaning given to in paragraph 6.6 in Section A (Introduction, Structure and Interpretation).
<u>Agenda</u>	means an agenda detailing (amongst other things) the nature of the matters and materials to be discussed at the meeting of the Modification Panel to which the agenda relates.
Allowed Onshore Offshore Capacity Revenue	has the meaning given in <u>in accordance with</u> paragraph 3.1(a) of Section H (Charges, invoicing Invoicing and payment Payment).
Allowed Onshore Offshore Flow Revenue	has the meaning given in paragraph 4.1(a) of Section H11.1(a) of Section H (Charges, invoicing Invoicing and payment Payment).
Allowed Offshore Revenue	has the meaning given in paragraph 16.2(a)(i) of Section H (Charges, Invoicing and Payment).
Allowed Offshore Onshore Capacity Revenue	has the meaning given in in accordance with paragraph 3.2(a) of Section H (Charges, invoicing Invoicing and payment Payment).
Allowed Offshore Onshore Flow Revenue	has the meaning given in paragraph 4.6(a) of Section H10.1(a) of Section H (Charges, invoicing Invoicing and payment Payment).
Allowed Onshore Revenue	has the meaning given in paragraph 15.2(a)(i) of Section H (Charges, Invoicing and Payment).
Allowed Re-use Service Revenue	has the meaning given in paragraph 12.1(a) of Section H (Charges, Invoicing and Payment).
Allowed Revenue	means the amount that T&SCo is entitled to recover in accordance with its Licence, as defined in the Licence.
Alternative Operational Communication Contact	means the alternative contact details to be provided in accordance with paragraph 3.2 of Section J (General).

Term	Meani	ng		
<u>Details</u>				
Ancillary Agreements	means	s the Cor	nstruction	n Agreement and the Connection Agreement.
Ancillary Agreement Templates			ng given nterpreta	in paragraph 5.4 of Section A (Introduction, ation).
Annual Forecast				in paragraphs 5.8 and 5.9 of Section E <u>9.2 of</u> <u>Use</u> and capacity Capacity).
Annual Maintenance Meeting				in paragraph 10.5(b) of Section E <u>31.1(b) of</u> <u>Use</u> and capacity Capacity).
Annual Network Capacity			<mark>k Capac</mark> i arging Y	ity allocated by T&SCo on an annual basis for ear.
Appeal	means an appeal made by an Appealing Party to the Modification Panel of a determination by the Modification Panel under paragraph [2.69] of Section B (Governance) in respect of a Self-Governance Modification Proposal;			
Appeal Criteria	means the following criteria in respect of a Regulator Appeal: (a) the Regulator Appeal has been made within the time specified in the Code;		Appeal has been made within the time e Code;	
	<u>(b)</u>	in the	opinion (of the Regulator:
		<u>(i)</u>	prejud non-im	pealing Party is, or is likely to be, unfairly iced by the implementation or applementation of the Self-Governance cation Proposal;
		<u>(ii)</u>	ground	egulator Appeal has been made on the ds that the Appealing Party reasonably es that:
			<u>(A)</u>	where the Modification Panel has made the determination to implement the proposal, the proposal does not better facilitate the achievement of at least one (1) of the Relevant Objectives; or
			<u>(B)</u>	where the Modification Panel has made the determination not to implement the proposal, the proposal does better facilitate the achievement of at least one (1) of the Relevant Objectives; and
		<u>(iii)</u>		egulator Appeal has not been made for ns that are trivial, frivolous or vexatious; and
		(iv)	the Re	egulator Appeal has a reasonable prospect of

Term	Meaning		
	success.		
Appealing Party	Meansmeans in relation to an Appeal or an Authoritya Regulator Appeal, athe User, the T&SCo or the Third Party Participant making such appeal.		
Applicable Interest Rate	has the meaning given in paragraph 12.5 of Section H29.4 of Section H (Charges, invoicing Invoicing and payment Payment).		
Applicable Standards	has the meaning given in the Measurement Requirements.		
Appointment Period	has the meaning given in paragraph 1.31 of Section B8.1 of Section B (Governance).		
Authority Associate	means the economic regulator appointed under section [x] of the Aethas the meaning given in the Licence, provided that for the purposes of the Code, a reference in that definition to "Licensee" shall be construed as a reference to the relevant organisation under paragraph 10.4 of Section B (Governance).		
Authority Appeal Availability Adjustment	means an appeal to the Authority made by the Appealing Party under paragraph 3.59 of Section B (Governance)adjustment to the T&SCo's Allowed Revenue under the Licence which is intended to incentivise the T&SCo to maintain availability of the T&S Network.		
Available Capacity	has the meaning given in paragraph 38.1 of Section E (Network Use and Capacity).		
Available Registered Capacity	means the Registered Capacity of a User, as reduced pursuant to paragraph 8.4(b) or paragraphs 38.1 to 41 of Section E (Network Use and Capacity).		
Backstop Lead Time	means where the AuthorityRegulator Decision Date is later than the last Proposed AuthorityRegulator Decision Date, the proposed period of time (commencing on the AuthorityRegulator Decision Date) required to enable the Modification to be implemented.		
Balancing Mechanism	means the Balancing Mechanism within the meaning of the Balancing and Settlement Code, as may be replaced by any equivalent mechanism requiring electricity generators to respond to the electricity system operator at short notice.		
Billing Period	means a calendar month.		
Breach Notice	has the meaning given in paragraph 12.1 of Section J (General).		
Budget Forecast	means a budget forecast that meets the requirements of paragraph 1.18 of Section B7.11 of Section B (Governance).		
Business Day	has the meaning given in paragraph 7(b) of Section A (Introduction, Structure and Interpretation).		
Business Interruption	has the meaning given in the Licence.		

Term	Meaning		
Proceeds			
Calendar Week	has the meaning given in paragraph 7(c) of Section A (Introduction, Structure and Interpretation).		
Capacity Applicant	has the meaning given in paragraph 3.3 of Section E6.2 of Section E (Network use Use and capacity Capacity).		
Capacity Charges	means the Onshore Capacity Charge and/or the Offshore Capacity Charge.		
Capacity Constraint	has the meaning given to that term in paragraph 11.1 of Section <u>E35.1 of Section E</u> (Network <u>useUse</u> and <u>capacityCapacity</u>).		
Capacity Constraint Default Capacity Offer	(a) in the case of a Supported User, the occurrence of a T&SCo Prolonged Unavailability Event (as such term is defined under that User's Revenue Support Contract), which has resulted in termination of its Revenue Support Contract; and (b) Note: this definition is still being developed, but it is intended to reflect a prolonged outage in the case of any other User, the occurrence of a T&SCo Prolonged Unavailability Event which has not been remedied within three (3) years of that occurrence or which the Regulator determined will not be capable of remedying by the T&SCo (or any replacement of the T&SCo).		
carbon dioxide	E (Network use Use and capacity Capacity). unless the context otherwise requires, means a stream consisting overwhelmingly of carbon dioxide (CO ₂) molecules, and other components and as such, references to carbon dioxide being delivered by a User to the T&S Network assume that the total mass being delivered is carbon dioxide.		
Carbon Dioxide Specifications	has the meaning given in paragraph 2.2 of Section F 3.2 of Section F (Network design Design and specification Specification).		
Casting Vote	means a vote exercisable by the Panel Chairperson in favour of or against any matter to be determined by the Modification Panel, except the making of a recommendation under paragraph 33.1(b) or 35.6(a) of Section B (Governance).		
CCS Network Code or Code Website	means this CCS Network Code, including all Uniform Provisions and Specific Provisions, as may be applicable the website to be established by the Secretary under paragraph 7.7(a) of Section B (Governance).		

Term	Meaning		
ccus	means carbon capture, use and storage.		
CCUS Cluster Sequencing Process	means the programme(s) by that name implemented by $\ensuremath{^{\underline{\text{HM}}}}$ Government to select:		
	(a) prospective T&S Networks and prospective Users of those T&S Networks; and/or		
	(b) prospective or existing Users for allocation of Network Capacity in existing or prospective T&S Networks.		
CDS	has the meaning given in paragraph 9.1 of Section I (Data).		
CDS Accession Agreement	means an agreement in the form set out in Annexure DExhibit D pursuant to which a T&SCo accedes to the CDS Contract.		
CDS Contract	Note: this will be a short-form standard accession agreement has the meaning given in paragraph 9.1 of Section 19.1(a) of Section 1 (Data).		
CDS Data Dispute	has the meaning given in paragraph 9.6 of Section I 9.14 of Section I (Data).		
CDS Terms of Reference	has the meaning given in paragraph 9.3 of Section I (Data).		
Charges Change in Scope Approval	means the approval of the Regulator under the Change in Scope (as that term is defined in the T&SCo's Licence) mechanism, where that approval is required by the T&SCo to proceed with an expansion or enhancement to the T&SCo's T&S Network which:		
	(a) is required to accommodate a new Delivery Point (or a modification to an existing Delivery Point) in accordance with Section C (Connection); and		
	(b) means each of the charges listed in paragraph 1.3 of Section H (Charges, invoicing and payment).which if granted by the Regulator, is sufficient to enable the T&SCo to proceed with the relevant Connection Application.		
Charges -Statement	means the statement to be published in accordance with each of the charges listed in paragraph 6 of Section H2.1 of Section H (Charges invoicing Invoicing and payment).		
Charges Forecast for Charging Methodology Year t	means has the methodology for determining the Charges in accordance with Section H (Charges, invoicing and payment meaning given in paragraph 17.5(b) of Section H (Charges, Invoicing and Payment).		
Charging Year	means the period from 1 April in any year until and including 31 March in the following year.		

Term	Meaning		
Claimant	has the meaning given in paragraph 51.4(a) of Section B (Governance).		
CO ₂ Quality Monitoring Procedure	means a Common Interface Procedure established and maintained in accordance with Section G (Common Interface Procedures), as more particularly described in that Section.		
<u>Code</u>	has the meaning given in paragraph 1.1 of Section A (Introduction, Structure and Interpretation).		
Code Accession Agreement	means an agreement in the form set out in Annexure AExhibit A pursuant to Awhich a User or a T&SCo accedes to the Code Agreement.		
	Note: this will be a short-form standard accession agreement.		
Code Agreement	means the agreement which makes the Code binding on a T&ScoSCo or a User (in the form set out as an exhibit to the Code Accession Agreement).		
Code Document	has the meaning given in paragraph 17.1 of Section J 18.1 of Section J (General).		
Code Implementation Date	means the effective date of the Code Agreement (specified in the Code Agreement) when the CCS Network Code is given legal effect.		
<u>Commencement Date</u>	means, in relation to each User, the Commencement Date determined in accordance with that User's Connection Agreement.		
Commercial Operations Date	has the meaning given in the T&SCo's Licence with respect to a particular T&SCo.		
CommunicationCommission ing Programme	Note: to be developed – this will include different categories of communications under the Code, such as nominations, curtailment notices, etchas the meaning given to that term in the Construction Agreement, being a commissioning programme agreed by the parties to the Construction Agreement.		
Common Interface Procedures	means the list of procedures set out at paragraph 3 of Section G (Common Interface Procedures).		
Communication	means any operational communication from a User to the T&SCo, or from the T&SCo to a User, including any forecasts, reports, Nominations, Renominations, notifications of Capacity Constraints, Minimum Flow Deficits and Emergencies.		
Competent Authority	means the AuthorityRegulator, or any local, national or supra-national agency, authority, department, inspectorate, minister, official, court, tribunal or public or statutory person (whether autonomous or not) of the United Kingdom (or the government thereof) which has jurisdiction over the T&SCo or a User or the subject matter of the Code.		

Term	Meaning		
Confirmation Close Time	has the meaning given in paragraph 16.2 of Section E (Network Use and Capacity).		
Confirmed Nominated Quantity	has the meaning given in paragraph 6.11 of Section E 16.4 of Section E (Network use Use and capacity Capacity).		
Conflicting Appointment	has the meaning given in paragraph 10.4 of Section B (Governance).		
Connected Dispute	has the meaning given in paragraph 49.1(a) of Section B (Governance).		
Connection Agreement	means an agreement between a T&SCo and a User <u>substantially</u> in the form set out in <u>Exhibit CExhibit C</u> .		
	Note: in the interests of consistency it is expected that T&SCos and Users will enter into a Connection Agreement in a prescribed form, except for some User specific details.		
Connection Application	has the meaning given to that term in paragraph 6.1 of Section C (Connection).		
Connection Offer	means a connection offer that complies with the requirements of paragraph 9.3 of Section C (Connection), in a written format determined by T&SCo.		
Consolidation Request	has the meaning given in paragraph 48.2 of Section B (Governance).		
Constrained Capacity Optimisation Principles	has the meaning given in paragraph 12.7 of Section E41.2 of Section E (Network use Use and capacity Capacity).		
Constrained Registered Capacity	has the meaning given in paragraph 12.10 of Section E42.1 of Section E (Network use Use and capacity Capacity).		
Constrained User	has the meaning given in paragraph 12.3(a) of Section E37.1(b) of Section E (Network use Use and capacity Capacity).		
Construction Agreement	means an agreement between a T&SCo and a User <u>substantially</u> in the form set out in <u>Exhibit BExhibit B</u> .		
	Note: in the interests of consistency it is expected that T&Scos and Users will enter into a Construction Agreement in a prescribed form, except for some User-specific details.		
Construction Longstop Default	means a failure by T&SCo to complete the T&SCo Works or the User to complete the User Works by the Longstop Date (as extended in accordance with the Construction Agreement) in accordance with paragraph 13clause 17 or clause 11 of Schedule 6 of the Construction Agreement (as applicable).		
Construction Programme	has the meaning given to that term in the Construction Agreement, being a construction programme agreed by the parties to the Construction Agreement.		

Term	Meaning		
Consultation	means the consultation process described in paragraph 3.34 and 3.35 of Section B (Governance).		
Conventional Notice	has the meaning given in paragraph 1.2 of Section J (General).		
Counterparty	has the meaning given in paragraph 3.1 of Section J (General).		
Credit Rating Agency	means:		
	(a) Fitch Ratings _τ <u>i</u>		
	(b) Meedy's Moody's Investment Service and Standard; and/or		
	(c) Poor's Poor's Rating Group and any of their subsidiaries.		
Curtailed Nomination	has the meaning given in paragraph 6.13 of Section Eparagraphs 17.2 and 19.3 of Section E (Network use Use and capacity Capacity).		
CURC	has the meaning has the meaning given in paragraph 38.1 of Section E (Network Use and Capacity).		
Daily Nomination	has the meaning given in paragraphs 6.2 and 6.4 of Section E 13.5 of Section E (Network use Use and capacity Capacity).		
Daily Quantity	has the meaning given in paragraph 1.5 of Section H4.2 of Section H (Charges, invoicing Invoicing and payment Payment).		
Day Data Annexures	means has the period from 00:00 hours on one day until 11:59 on the following daymeaning given in paragraph 6.8 of Section I (Data).		
Data Transfer Procedures	has the meaning given in paragraph 6.1 of Section I (Data).		
<u>Day</u>	has the meaning given in paragraph 7(a) of Section A (Introduction, Structure and Interpretation).		
Default Notice	has the meaning given in paragraph 6.1 of Section J7.1 of Section J (General).		
<u>Delivery Period</u>	has the meaning given in paragraph 2.2 of Section E (Network Use and Capacity).		
Delivery Point	means the point of connection between a User Facility and the T&S Network at which point a User will deliver carbon dioxide into the T&S Network.		
Delivery Point Size	has the meaning given in paragraph 1.11 of Section H7.4 of Section H (Charges, invoicing Invoicing and payment Payment).		
Deposit Deed	means an agreement that is Enforceable and in such form as provided to the User from time to time by T&SCo enabling the deposit of cash as security.		
Depreciation Building Block	has the meaning given in the Licence.		

Term	Meaning		
Deputy Panel Chairperson	means the person appointed by the Members pursuant to paragraph 2.1(b) of Section B (Governance) to take on the Panel Chairperson's duties if the Panel Chairperson is absent for any reason.		
Deselected	means that an Eligible Applicant, having first been Selected, has been advised by the Secretary of State through the Selection Process that it is no longer eligible for:		
	(a) award of an Emittera Revenue Support Contract; or		
	(b) connection to a T&S Network; or		
	(c) modification of its existing Delivery Point.		
Designated Non-Code Party	means a Non-Code Party that the Regulator designates in accordance with Standard Condition B5.8 (CCS Network Code) of the Licence as being a Non-Code Party to whom copies of documents relating to Modification Proposals should be sent.		
Directive	means any present or future directive, request, requirement, instruction, code of practice, direction or rule of any Competent Authority (but only, if not having the force of law, if it is reasonable in all the circumstances for it to be treated as though it had legal force), and any modification, extension or replacement thereof.		
Disclosing PartyDiscontinuing User	has the meaning given in paragraph 3.5 of Section I (Data 14.1(a) of Section J (General).		
<u>Dispute</u>	means any dispute, controversy or claim (including any non-contractual dispute, controversy or claim) arising out of or in connection with the Code, the Code Agreement, any Code Accession Agreement, any CDS Accession Agreement or any Ancillary Agreement, including any question regarding its existence, validity, formation or termination.		
Discontinuance Dispute Notice	has the meaning given in paragraph 13.1 of Section J (General 18.7 of Section B (Governance).		
<u>Dispute Parties</u>	means the T&SCo(s) and/or the User(s) and/or any third party who is a party to a Dispute.		
Dispute Resolution Procedure	means the procedure to be followed in paragraph 48 of Section B (Governance).		
Discontinuing User Draft Data Annexures	has the meaning given in paragraph 13.1 of Section J (General 6.5 of Section I (Data).		
Draft Statement of Expenses	has the meaning given in paragraph 1.23 of Section B7.14 of Section B (Governance).		
Eligible Applicant	has the meaning given to that term in paragraph 5.1 of Section- <u>CSection C</u> (Connection).		

Term	Meaning		
Emergency	means a situation where T&SCo in its judgment considers that actimust be taken without delay to:		
	(a) avert or reduce danger to life or property; or		
	(b) secure the safety of the T&S Network or a part of the T&S Network or the safe transportation of carbon dioxide by it or reducing the risk to it.		
Emitter Revenue Support Contract Emergency Procedure	means a revenue support contract granted to an emitter pursuant to-Part 2 of the [Act] or a Contract for Difference granted to an emitter-pursuant to Chapter 2 of Part 2 of the Energy Act 2013Common_Interface Procedure established and maintained in accordance with Section G (Common Interface Procedures), as more particularly described in that Section.		
Enforceable	means T&SCo (acting reasonably) is satisfied that the instrument of security or surety is legally enforceable and in this respect, where surety is provided by a company registered outside of England and Wales, the country of residence of such company must have a sovereign credit rating of a Qualifying Company (where such ratings conflict, the lower of the ratings will be used) and the User shall-must at its own expense- provide , provide such legal opinion as the T&SCo may reasonably require.		
Enhanced Response Service	has the meaning give in paragraph 20.2 of Section E (Network Use and Capacity).		
Enhanced Response Service Cap	has the meaning given in paragraph 20.4(f) of Section E (Network Use and Capacity).		
Entry Provisions	has the meaning given in paragraph 2.1 of Section F3.1 of Section F (Network design Design and specification Specification).		
ETS Liabilities	means any costs or liabilities of a Party relating to that Party's liabilities or obligations under the UK Emissions Trading Scheme established under the Greenhouse Gas Emissions Trading Scheme Order 2020.		
Exceedance Event	has the meaning given in paragraph 25.2 of Section E (Network Use and Capacity).		
Expenses Dispute Notification	has the meaning given in paragraph 1.24 of Section B7.15 of Section B (Governance)		
Expert	means any person appointed in accordance with the Expert Determination Procedure to determine an Expert Dispute.		
Expert Appointment Date	means the date on which an Expert is appointed to determine an Expert Dispute by means of an appointment letter entered into by such Expert and each of the Parties.		

Term	Meanir	ng		
Expert Determination	means the process for determination of a dispute by Experts in accordance with paragraph 9 of Section B 51 of Section B (Governance).			
Expert Determination Notice	has the	has the meaning given in paragraph 51.1 of Section B (Governance).		
Expert Determination Procedure			es, obligations and procedures set out in paragraph 51 Governance).	
Expert Dispute		<u> </u>	ute which is referred for determination in accordance t Determination Procedure.	
Expert Referral Date	has the		ng given in paragraph 51.7(a) of Section B	
Fast Track Self-Governance	means	that a p	proposal, if implemented:	
Procedure	(a)	would	meet the Self-Governance Criteria; and	
	(b)	•	perly a housekeeping modification required as a result me error or factual change, including-but not limited to:	
		(i)	updating names or addresses listed in the uniformnetwork code Code;	
		(ii)	correcting minor typographical errors;	
		(iii)	correcting formatting and consistency errors, such as paragraph numbering; or	
		<u>(iv)</u>	updating out of date references to other documents or paragraphs.	
Final Confirmation Notice	means a notice issued by the Secretary of State as part of a Selection Process:			
	(a) to an Eligible Applicant, applying for a connection to a T&S Network through the Selection Process; or (b) to a User applying for a modification of an existing Delivery Point through the Selection Process,		•	
	confirm	ing the	following details:	
	<u>(c)</u>		Ill necessary due diligence and any other steps that are tof the Selection Process have been completed;	
	<u>(d)</u>	User	mount of Network Capacity being allocated to that or Eligible Applicant, to be its Registered Capacity at elevant Delivery Point; and	
	(e)		onditions precedent specified by the Secretary of State ation to the Final Confirmation Notice taking effect.	
Final Confirmed Nominated	Hashas the meaning given in paragraph 6.2224.1 of Section ESection E (Network useUse and capacityCapacity).			

Term	Meaning		
Quantity			
Final Statement of Expenses	has the meaning given in paragraph 1.25 of Section B7.16 of Section B (Governance).		
First Submission	has the meaning given in paragraph 51.7(b) of Section B (Governance).		
First Submission Deadline	has the meaning given in paragraph 51.7(b) of Section B (Governance).		
Fixed Implementation Date	means the proposed implementation date of a Modification, such date being included in accordance with paragraph 3.4(c) of Section—B24.1(c) of Section B (Governance) and, except where the Modification Proposal is a Self Governance Self-Governance Modification Proposal, paragraph [2.31(f)] of Section B 23.1(c) of Section B (Governance).		
Flow Charges	means the Onshore Flow Charge and/or the Offshore Flow Charge.		
Flow Meter	Note: this definition is to be developed on the basis of the Measurement Requirementsmeans one (1) or more flow meter(s) (including ancillary instrumentation) for measuring the mass flow rate of carbon dioxide delivered by a User, expressed in (tCO ₂ /hour) to the Delivery Point.		
Flow Meter Error	means a Measurement Equipment Error relating to the Flow Meter.		
Flow Meter Adjustment	means a Measurement Adjustment made in accordance with paragraph 7.1 or paragraph 7.2 of Section F 13.1 or 13.2 of Section F (Network design Design and specification Specification).		
Flow Meter Error	means a Measurement Equipment Error relating to the Flow Meter.		
Flow Rate	means the total instantaneous flow rate, in the form of carbon dioxide, expressed in tCO ₂ /hour, that is being delivered at the Delivery Point.		
Flow Tolerance Request	has the meaning given in paragraph 21.1 of Section E (Network Use and Capacity).		
<u>Forecast</u>	has the meaning given in paragraph 8.2 of Section E (Network Use and Capacity).		
Forecasting Pro-Forma	means the pro-forma document issued by the T&SCo identifying the information to be provided by Users in accordance with paragraph 5.2 of Section H 17 of Section H (Charges, invoicing Invoicing and payment Payment).		
Forecast	has the meaning in paragraph 5.2 of Section E (Network use and capacity).		

_	
Term	Meaning
General Default	has the meaning given in paragraph 7 of Section J 8 of Section J (General).
Handover Date	has the meaning given in T&SCo's Licence.
Hour	means a period of 60 minutes, with the first Hour of each Day starting at 00:00 hours.
Implementation Date	means the date upon which an Industriala Common Interface Procedure is to be implemented.
Independent Verifier	has the meaning given in paragraph 10.1 of Section I (Data).
Independent Verifier Appointment	has the meaning given in paragraph 10.1 of Section I Section I (Data).
Independent Verifier Terms of Reference	has the meaning given in paragraph 10.2 of Section I Section I (Data).
Industrial Procedures	has the meaning given in paragraph 3.1 of Section H (Industrial Procedures).
Initial Offer	means an initial offer that complies with the requirements of paragraph 9.2 of Section C9 of Section C (Connection), in a written format determined by T&SCo.
Initial User	means a User or prospective Useran Eligible Applicant who was Selected through the Track-1 Cluster Sequencing Process and-
	(a) has not become a User. became a User on or before the Code Implementation Date; or
	(b) as at the Code Implementation Date, has not yet become a User,
	but is in the process of reaching financial investment decision and has already applied to connect to a T&S Network, and has not been- Deselected.
Insolvency Default	has the meaning given in paragraph 8.1 of Section J 9.1 of Section J (General).
Insured Event	has the meaning given in paragraph 45.2(c) of Section E (Network Use and Capacity).
Insured Risk	has the meaning given in the Supplemental Compensation Agreement.
Interim Period	has the meaning given in the Energy Bill Act.
Invoice Amount	means the amount shown as payable by the User or T&SCo in respect of that item under the relevant Invoice Document.
Invoice Dispute	has the meaning given in paragraph 11.1 of Section H26.1 of Section

Term	Meaning
	H (Charges, invoicing Invoicing and payment Payment).
Invoice Document	means an invoice document submitted by T&SCo to a User pursuant to Section HSection H (Charges, invoicing Invoicing and paymentPayment).
Invoice Due Date	has the meaning given in paragraph 10.2 of Section H24.2 of Section H (Charges, invoicing lnvoicing and payment Payment).
Isolation Procedure	means a Common Interface Procedure established and maintained in accordance with Section G (Common Interface Procedures), as more particularly described in that Section.
Late Payment Notice	has the meaning given in paragraph 12.1 of Section H29.1 of Section H (Charges, invoicing lnvoicing and payment Payment).
LCCC	means the Low Carbon Contracts Company or any successor or equivalent body.
Local Requirements LCIA	has <u>means</u> the meaning given in paragraph 2.9 of Section F (Network design and specification)London Court of International Arbitration.
Legal Requirement	means any Act of Parliament, regulation, licence or Directive of a Competent Authority.
<u>Lender</u>	means any bank or financial institution (excluding any direct or indirect shareholder of the User) which provides debt financing or refinancing in relation to the User Facility.
Letter of Credit	means an unconditional irrevocable standby letter of credit in such form as reasonably provided to the User from time to time by

Term	Meaning
	containing in relation to the relevant Planning Period the information specified in paragraph 10.2 of Section E (Network useUse and capacityCapacity).
Materially Maximum Annual Cumulative Flow	has the meaning given in paragraph 6.26 of Section E (Network use and capacity)means the maximum quantity of carbon dioxide (i.e. a cumulative flow cap) that the T&SCo is permitted to inject into a Storage Site in any period of twelve (12) consecutive months, in accordance with the Storage Permit issued in relation to that Storage Site, as may be amended by the NSTA from time to time.
Maximum Annual Cumulative Flow Constraint	has the meaning given in paragraph 23.6 of Section E (Network Use and Capacity).
Maximum Annual Cumulative Flow Mitigation Measures	has the meaning given in paragraph 23.7 of Section E (Network Use and Capacity).
Maximum Ramp Rate	means the maximum unconstrained Ramp Rate that is technically achievable by the User Facility, as set outspecified by the User and recorded in the User's Connection Agreement.
Measurement Adjustment	means any action or adjustment which is required to correct any Measurement Equipment Error.
Measurement Data	has the meaning given in <u>5.23.1</u> of <u>Section I Section I</u> (Data).
Measurement Equipment	has the meaning given in paragraph 4.1 of Section F 5.1 of Section F (Network design Design and specification Specification).
Measurement Equipment Error	means where the Measurement Equipment (or any part of it) is or has been:
	(a) <u>faulty:</u>
	(b) (a)-registering erroneously; or (c) (b)-not functioning in accordance with the Measurement Requirements or otherwise in accordance with Section-F-Section F (Network designDesign and specificationSpecification).
Measurement Requirements	has the meaning given in paragraph 2.5 of Section F 3.5 of Section F (Network design Design and specification Specification).
Member	means any individual for the time being appointed to the Modification Panel.
Metering Point	means the point, co-located with the User's Delivery Point or located between the User Facility and the Delivery Point, where the Measurement Equipment is located.

Term	Meaning
Minimum Flow Deficit	has the meaning give in paragraph 22.1 of Section E (Network Use and Capacity).
Minimum Flow Rate	means the minimum flow required for the reliable and safe operation of the T&S Network, as set out in the Technical Parameters for that T&S Network.
Minimum Ramp Rate	means the minimum unconstrained Ramp Rate that is technically achievable by the User Facility, as set outspecified by the User and recorded in the User's Connection Agreement.
Minimum Turndown Rate	means the User's Flow Rate, in tCO ₂ /hour (as <u>specified by the User and as</u> set out in the User's Connection Agreement), when the User is operating the User Facility at its lowest operational level, to represent the lowest amount of <u>network capacity Network Capacity</u> that the User can make use of when its Registered Capacity is constrained in accordance with <u>Section E of the Code Section E</u> (Network Use and Capacity).
Modification	means any modification of the CCS Network-Code made pursuant to the Modification Rules.
Modification Panel	means the panel comprised in accordance with paragraph 1.2 of Section B 1.3 of Section B (Governance) which is to perform the functions described in paragraph 1 of Section B 1.2 of Section B (Governance).
Modification Procedures	means the provisions relating to Modifications and Modification Proposals set out in paragraphs 1 and 2 to 7 of Section B (Governance) and, as the context may require, any of those provisions.
Modification Proposal	means a proposal for a Modification which meets the requirements of the Code.
Modification Report	means a draft, final or amended final report prepared in accordance with paragraphs 3.40 to 3.43 of Section Bparagraph 35.1 of Section B (Governance).
Modification Rules	means the rules set out in paragraphs 1 and 2 to 7 of Section B 47 of Section B (Governance).
Monthly Forecast Modification Terms of Reference	has the meaning given in paragraphs 5.12 and 5.13 of Section E (Network use and capacity)means those terms of reference in relation to a Modification Proposal referred to a Workgroup by the Modification Panel, pursuant to paragraph 28.2(d) of Section B (Governance), finalised by the Secretary under paragraphs 28.1 and 39 of Section B (Governance) and as may be amended pursuant to paragraph 40.3 of Section B (Governance).
Most Restricted Communal	has the meaning given by reference to the following concepts:

Term	Meaning
Element Mutualisation Cap	(a) "Most Restricted", which means the part of the T&S Network (or part of the T&S Network) through which the lowest mass of carbon dioxide can instantaneously flow, determined by the lower of any physical or regulatory limits; and
	(b) "Communal Element", which refers to any part of the T&S-Network (or part of the T&S Network) that all Users' carbon dioxidemust flow through, excluding any contingent infrastructure. has the meaning given in paragraph 15.6(b) and 16.6(b) of Section H (Charges, Invoicing and Payment).
Network Capacity	has the meaning given in paragraph 1.3 of Section E2.1(a) of Section E (Network use Use and capacity Capacity).
Network Charges	means the Onshore Network Charge and/or the Offshore Network Charge.
Network Expansion Works	means works for the expansion, reinforcement or extension of a T&S Network, including:
	(a) works in relation to existing parts of a T&S Network; and
	(b) works required to connect a User or a prospective User to the T&S Network,
	(including taking any part of a T&S Network out of service, whether on a provisional or permanent basis) to enable such works to be carried out.
Network Responsiveness Review	has the meaning give in paragraph 48.4(b) of Section E (Network Use and Capacity).
Network Portal Service ContractNew T&SCo	has the meaning given in paragraph 11.1 of Section I (Data 4.1(b) of Section A (Introduction, Structure and Interpretation).
Network Portal Terms of Reference New Code Party	has the meaning given in paragraph 11.2 of Section I (Data 4.1 of Section A (Introduction, Structure and Interpretation).
New User	has the meaning given in paragraph 4.1(a) of Section A (Introduction, Structure and Interpretation).
Nominated Quantity	has the meaning given in paragraph 6.4(d) of Section E13.5(d) of Section E (Network use Use and capacity Capacity).
Nomination	means a nomination by a User in respect of a quantity of carbon dioxide to be delivered to the T&S Network on a Day.
Nomination Close Time	has the meaning given in paragraph 6.2 of Section E13.2 of Section E (Network use Use and capacity Capacity).
Nominations Non-conformity	has the meaning <u>givengive</u> in paragraph <u>6.24 of Section E48.4(c) of Section E</u> (Network <u>use Use</u> and <u>capacity Capacity</u>).

Term	Meaning
Notice Review	
Non-Code Party	means any person that is not a party to the Code, other than a Third Party Participant, the Regulator or the Secretary of State.
Non-Compliant CO ₂	means carbon dioxide which does not comply with the requirements regarding the specification of carbon dioxide under the Entry Provisions, as set out in the Measurement Requirements.
Non-Delivery Non-Payment Notice	Hashas the meaning given in paragraph 6.25(a) of Section E. (Network use and capacity 10(b) of Section J (General).
<u>NSTA</u>	means the North Sea Transition Authority, which is the business name of the Oil and Gas Authority, which is the regulatory authority for offshore carbon dioxide storage, responsible for permitting and licensing.
Obligated Network Capacity	has the meaning given in paragraph 1.5 of Section E2.4 of Section E (Network use Use and capacityCapacity).
Offshore Capacity Charge	has the meaning given in paragraph 1.6(a)(ii) of Section H6.1(a)(ii) of Section H (Charges, invoicing Invoicing and payment Payment).
Offshore Capacity Charge Rate	means the rate determined in accordance with paragraph 3.2 of Section H 14.1 of Section H (Charges, invoicing Invoicing and payment Payment).
Offshore Flow Charge	has the meaning given in paragraph 1.4(a)(ii) of Section H4.1(b) of Section H (Charges, invoicing Invoicing and payment Payment).
Offshore Flow Charge Rate	means the rate determined in accordance with paragraph 2.2 of Section H 11.1 of Section H (Charges, invoicing Invoicing and payment Payment).
Offshore Mutualised Network Charge Rate	means the rate determined in accordance with paragraph 4.7 of Section H 16.3 to 16.12 of Section H (Charges, invoicing Invoicing and paymentPayment).
Offshore Network Charge	has the meaning given in paragraph 1.10(a)(ii) of Section H7.1(a)(ii) of Section H (Charges, invoicing Invoicing and payment Payment).
Offshore Network Charges Cap	has the meaning given in paragraph 16.6 of Section H (Charges, Invoicing and Payment).
Offshore Network Charge Rate	means the rate determined in accordance with paragraph 4.5 of Section H 16.1 of Section H (Charges, invoicing Invoicing and payment Payment).
Offshore Proportion	has the meaning given in paragraph 16.6(a) of Section H (Charges, Invoicing and Payment).
Offshore Proportionate	means the rate determined in accordance with paragraph 16.2 of Section H (Charges, Invoicing and Payment).

Term	Meaning
Network Charge Rate	
Offshore Pipeline Infrastructure	means that part of the T&S Network which is offshore, including being the pipelines and related infrastructure from the pig trap(s) at the Onshore Transportation Systemoutlet(s) of the Terminal(s) to the pig trap(s) or manifold entry flange(s) at the inletentry to the Storage Complex, but excluding the Storage Complex and including the TerminalSite.
Offshore Proportionate Network T&S Charge Rate	means has the rate determined in accordance with meaning given in paragraph 4.6 of Section H 16.4 of Section H (Charges, invoicing Invoicing and payment Payment).
Offshore Transportation and Storage System	means: means (a) the Offshore Pipeline Infrastructure and any Storage Complex.; (b) the Storage Site(s); and
	(c) the Terminal(s).
Offshore User	means a User with a Delivery Point which connects to the T&S Network at the Offshore Transportation and Storage System.
Offshore User T&S Charge	has the meaning given in paragraph 16.6 of Section H (Charges, Invoicing and Payment).
ONC Maximum Annual Cumulative Flow	means the Maximum Annual Cumulative Flow defined (in terms of the quantity) by the T&SCo's Obligated Network Capacity as set out in the T&SCo's Licence (or established pursuant to the Licence).
ONC Maximum Annual Cumulative Flow Constraint	means a Maximum Annual Cumulative Flow Constraint to the extent that any action taken by the T&SCo pursuant to paragraph 23.4(b) of Section E (Network Use and Capacity) is taken in relation to an ONC Maximum Annual Cumulative Flow.
Onshore Capacity Charge	has the meaning given in paragraph 1.6(a)(i) of Section H6.1(a)(i) of Section H (Charges, invoicing Invoicing and payment Payment).
Onshore Capacity Charge Rate	means the rate determined in accordance with paragraph 3.1 of Section H (Charges, invoicing Invoicing and paymentPayment).
Onshore Flow Charge	has the meaning given in paragraph 1.4(a)(i) of Section H4.1(a)(i) of Section H (Charges, invoicing Invoicing and payment Payment).
Onshore Flow Charge Rate	means the rate determined in accordance with paragraph 2.1 of Section H 10.1 of Section H (Charges, invoicing Invoicing and payment Payment).
Onshore Mutualised Network Charge Rate	means the rate determined in accordance with paragraph 4.3 of Section H 15.3 to 15.12 of Section H (Charges, invoicing Invoicing and paymentPayment).

Term	Meaning
Onshore Network Charge	has the meaning given in paragraph 1.10(a)(i) of Section H7.1(a)(i) of Section H (Charges, invoicing Invoicing and payment Payment).
Onshore Network Charges Cap	has the meaning given in paragraph 4.4 of Section H (charges, invoicing and payment 15.6 of Section H (Charges, Invoicing and Payment).
Onshore Network Charge Rate	means the rate determined in accordance with paragraph 4.1 of Section H 15.1 of Section H (Charges, invoicing Invoicing and payment Payment).
Onshore Proportion	has the meaning given in paragraph 15.6(a) of Section H (Charges, Invoicing and Payment).
Onshore Proportionate Network Charge Rate	means the rate determined in accordance with paragraph 4.2 of Section H (Charges, invoicing Invoicing and payment Payment).
Onshore Transportation System	means thatthose part(s) of the T&S Network which isare located onshore, including the pipelines and related infrastructure from the emitter User(s) boundary fence to thea pipeline entry pig trap at the entry to thea Terminal, but excluding any part(s) of the T&S Network which constitutes constitute the Offshore Transportation and Storage System.
Onshore T&S Charge	has the meaning given in paragraph 15.4 of Section H (Charges, Invoicing and Payment).
Onshore User	means a User with a Delivery Point which connects to the T&S Network at the Onshore Transportation System, including any User with an emergency shutdown valve upstream of any high pressure compression or pumping inlet header at the Terminal.
Original Selection Details	means the final commercial and technical details which formed the basis on which an Eligible Applicant has been Selected (through a Selection Process) such as: (a) the location and key characteristics (including User Type) of the User Facility; (b) the proposed location of the Delivery Point; and the amount of Network Capacity that will be required by the User.
Over Variation Limit	means the percentage set out in the Technical Parameters for a T&S Network.
Panel Chairperson	means the chairperson of the Modification Panel, appointed under paragraph 1.3 of Section B (Governance).
Panel Constituency	has the meaning given in paragraph 1.3 of Section B (Governance).

Term	Meaning
Panel Majority	has the meaning given in paragraph 1.47 12.2 of Section B (Governance).
Payment DefaultParty	has the meaning given in paragraph 9 of Section J (General 2.1(c) of Section A (Introduction, Structure and Interpretation).
Persistently Payment Default	has the meaning given in paragraph 6.26 of Section E (Network use and capacity10 of Section J (General).
Planning Period	means a period of five (5) Charging Years commencing on 1 April.
Preference NoticeProcessed Flow Meter Data	has the meaning given in paragraph 9.9 of Section B (Governance) means, in relation to each User and that User's Flow Meter, data from that Flow Meter that has been processed by the CDS.
Preference Number	has the meaning given in paragraph 9.9 of Section B (Governance).
Processed Flow Meter Data Statement	has the meaning given in paragraph <u>9.29.2(b)</u> of <u>Section ISection I</u> (Data).
Programmed Maintenance	means maintenance performed in accordance with the prevailing Maintenance Programme.
Project-Specific Document	has the meaning given to it in the T&SCo's Licence.
Proposer	means the person that submitted a Modification Proposal.
Proposed <u>AuthorityRegulator</u> Decision Date	means the proposed date by which the AuthorityRegulator shall give notice to the Secretary in accordance with paragraph [2.46(b24.1(c)(ii) of Section B (Governance)]- of its determination to implement the Modification, such date being included in accordance with paragraph 3.4(c)(ii)24.1(c)(ii) and, except where the Proposal is a Self GovernanceSelf-Governance Modification Proposal, paragraph [2.42(g)(ii)]36.1(f)(ii).
Proposed Industrial Procedure	has the meaning given in paragraph 5.1 of Section G (Industrial procedures).
Proposing Party	has the meaning given in paragraph 12.1 of Section I (Data).
Pro-rata Reduction	has the meaning given in paragraph 12.4 of Section E38.1 of Section E (Network use Use and capacity Capacity).
Prospective T&SCo	means a T&SCo who:
	(a) has been selected through the CCUS Cluster Sequencing Process (or another Selection Process) to build and operate a new T&S Network; and
	(b) will become a party to the Code upon being awarded a

Term	Meaning	
	Licence and acceding to the Code Agreement.	
Protected Information	has the meaning given in paragraph 3 of Section I 3.2 of Section I (Data).	
Protected Party	has the meaning given in paragraph 3.3 of Section I (Data).	
Provisional Offshore Mutualised Network Charge Rate	means the rate determined in accordance with paragraph 16.9 of Section H (Charges, Invoicing and Payment).	
Provisional Onshore Mutualised Network Charge Rate	means the rate determined in accordance with paragraph 15.9 of Section H (Charges, Invoicing and Payment).	
Qualifying Company	means:	
	in the case of a company registered in England and Wales a public or private company within the meaning of section 1(3) of the Companies Act 1985 with a long-term debt rating of [at least A]- provided by a Credit Rating Agency (where such ratings conflict, the lower of the ratings will be used); or	
	(b) in the case of an entity registered outside of England and Wales, such equivalent entity to paragraph (ia) above that is acceptable to the T&SCo, acting reasonably.	
Ramp Rate	means the rate (expressed in tCO ₂ /hour per minute) at which the User is able to increase or decrease its Flow Rate.	
Reactive Maintenance	means any maintenance which is not Programmed Maintenance but which T&SCo considers, acting as a Reasonable and Prudent Operator, needs to be scheduled before the next update to the Maintenance Programme to avoid an Emergency or Capacity Constraints of greater magnitude than the Capacity Constraints caused by the carrying out of the maintenance itself.	
Reasonable and Prudent Operator	means a person seeking, in good faith, to perform its contractual obligations and, in so doing and in the general conduct of its undertaking, exercising that degree of skill, diligence, prudence and foresight which would reasonably and ordinarily be expected from a skilled and experienced operator complying with all applicable Legal Requirements engaged in the same type of undertaking in similar circumstances and conditions.	
Reduced CapacityReceiving Party		
Recoverable	means recoverable by the T&SCo acting in compliance with Special Condition H27 (Insurance) of the Licence.	

Term	Meaning
Registered Capacity	has the meaning given in paragraph 1.6 of Section E3.1 of Section E (Network useUse and capacityCapacity).
Registered Long-term Network Capacity <u>Financial</u> Security	means Long-term has the meaning given in paragraph 7.3 of Section E (Network Use and Capacity that is registered as held by a User).
Regulator	means the Office of Gas and Electricity Markets, in its capacity as the economic regulator pursuant to Section 1 of the Act.
Regulator Appeal	means an appeal to the Regulator made by the Appealing Party under paragraph 40.6 of Section B (Governance).
Regulatory Requirements	means various laws, regulations, permits and licences relating to the operation of the T&S Network, as well as orders and directions issued by relevant regulators under these instruments, including (but not limited to) the

Term	Meaning
Response Submission	has the meaning given in paragraph 51.7(c) of Section B (Governance).
Re-use Daily Quantity	has the meaning given in paragraph 15.1(d) of Section F (Network Design and Specification).
Re-use Delivery Point	means the point of connection between a User Facility and the T&S Network at which point a User, who is being provided with the Re-use Service, will take delivery of carbon dioxide from the T&S Network, on a temporary basis.
Re-use Meter	means any additional Flow Meter required to measure the flow of carbon dioxide being delivered to a User at the User's Re-use Delivery Point.
Re-use Purpose	means the use of carbon dioxide by the User at the User Facility for the re-pressurisation of an LP compressor during a plant start up sequence.
Re-use Service	has the meaning given in paragraph 8.1 of Section F 14.1 of Section F (Network design Design and specification Specification).
Re-use Service Charge	has the meaning given in paragraph 5.1(a) of Section H (Charges, Invoicing and Payment).
Re-use Service Charge Rate	has the meaning given in paragraph 12.1 of Section H (Charges, Invoicing and Payment).
Re-use Service Infrastructure	means the infrastructure and equipment required to allow the Re-use Service to be offered to the User, including any additional pipework, valves, compressors and Re-use Meters.
Re-use Service Infrastructure Works	means all the work required for the provision of the Re-use Service, including the installation of the Re-use Service Infrastructure, but not including any Works (as that term is defined in the Construction Agreement) that are required to connect the User Facility to the T&S Network.
Re-use Service Operational Procedures	has the meaning given in paragraph 15.2 of Section F (Network Design and Specification).
Re-use Service Opex	has the meaning given in paragraph 12.1(b) of Section H (Charges, Invoicing and Payment).
Revenue Support Contract	means a revenue support contract granted to a transport and storage network user pursuant to Part 2 of the Act or a Contract for Difference granted to a transport and storage network user pursuant to Chapter 2 of Part 2 of the Energy Act 2013.
Revised Terms of Reference	has the meaning given in paragraph 12.6 of Section I Section I (Data).

Term	Meaning
Rolling QuarterlyThree-month Forecast	has the meaning given in paragraphs 5.10 and 5.11 of Section Eparagraph 10.2 of Section E (Network use Use and capacity Capacity).
RURC	has the meaning given in paragraph 38.1 of Section E (Network Use and Capacity).
Scheduled Commencement Date	has the meaning given in each User's Connection Agreement.
Scheduled Commercial Operations Date	means, in relation to the T&S Network of a T&SCo, Scheduled COD (as that term is defined in that T&SCo's Licence).
Secretary	means the individual appointed under paragraph 1.15 of Section-B7.1 of Section B (Governance).
Secretary Expenditure	means the costs and expenditure incurred by the Secretary in the carrying out of the various tasks and duties assigned to it in Section-BSection B (Governance).
Secretary of State	means the Secretary of State, acting in that capacity, unless otherwise expressly stated or the context otherwise requires.
SecretarySecretary's Expenses Claim	has the meaning given in paragraph $\frac{1.22 \text{ of Section B}}{2.13 \text{ of Section}}$ B (Governance).
Secretary of StateSection E Review	meanshas the Secretary of State for Energy Security and Net Zeromeaning give in paragraph 48.1 of Section E (Network Use and Capacity).
Security Default Notice	has the meaning given in paragraph 7.3 of Section H20.5 of Section H (Charges, invoicing Invoicing and payment Payment).
Selected	means that where a User or a prospective User has applied, through the Selection Process, for the potential:
	(a) award of an Emittera Revenue Support Contract; or
	(b) connection to a T&S Network; or
	(c) modification of its existing Delivery Point,
	that prospective User has been notified by the Secretary of State that it has been selected for the final stage of the process, subject only to any final requirements specified as part of the Selection Process and acknowledging that such status shall end on the date that the matters in paragraph 13 of Section C (Connection) are satisfied in respect of that application, and the word Selection shall be construed accordingly.
Selection Process	Note: this definitions remains under review and development means a Government-led process for the selection of new Users to connect

Term	Meani	Meaning		
	to T&S	Netwo	rks, the selection of existing Users who wish to modify	
	their existing connection and/or the allocation of Network Capacity,			
	which shall be held from time to time as determined by HM			
	Government.			
Self-Governance Criteria		means a proposal that a Modification Proposal, if implemented:		
	(a)	(a) is unlikely to have a material effect on:		
		<u>(i)</u>	existing or future Users;	
		<u>(ii)</u>	(i) competition in the <u>pipeline</u> transportation <u>orand</u> storage of carbon dioxide conveyed through pipes or any commercial activities connected with the <u>pipeline</u> transportation <u>orand</u> storage of carbon dioxide-conveyed through pipes; and	
		(ii) th	e operation of one or more pipe-line system(s); and	
		(iii)	matters relating to sustainable development, safety or security of supply, or the management of market or network the operation of the T&S Network, including emergencies; and	
		(iv)	the Code governance procedures or the Code Modification procedures Rules; and	
	(b)		ikely to discriminate between different classes of es to the Code.	
Self-Governance Determination Date	means the date on which the Modification Panel makes a determination whether to implement or not to implement a Self-Governance Modification Proposal.			
Self-Governance	means a Modification Proposal which:			
Modification Proposal	(a)	Self-G Secre Self-G by the	Iodification Panel has determined satisfies the Governance Criteria and in respect of which the etary has submitted to the AuthorityRegulator a Governance Statement (which has not been rejected a AuthorityRegulator by the Self-Governance fication Proposal Determination Date); or	
	(b)	Self-0	uthorityRegulator has determined satisfies the Governance Criteria under paragraph [2.24] of Section 12 of Section B (Governance).	
Self-Governance Modification Proposal Determination Date	means the proposed date on which the Modification Panel is to make a determination whether to implement or not to implement a Self-Governance Modification Proposal.			
Self-Governance Statement			ment complying with the requirements of the Code s the proposed Self-Governance Determination Date.	

Term	Meaning	
Senior Representatives	means one (1) or more senior employees or officers selected by a Party to represent it in relation to paragraph 50 of Section B (Governance).	
Senior Representatives Settlement	has the meaning given in paragraph 50.1(a) of Section B (Governance).	
Significant Code Review	has the meaning given in the Standard Conditions in Part II of the Licence.	
Significant Code Review Modification Proposal	means a Modification Proposal made by: (a) the Authority Regulator in respect of a Significant Code Review; or	
	(b) a T&SCo pursuant to a direction of the Authority Regulator which is made further to a Significant Code Review.	
Significant Code Review Phase	Note: definition to be developed by reference to the Licence condition dealing with the CCS Network Code has the meaning given to that term in the Standard Conditions.	
Significant Code Review Suspended Modification Proposal	means, unless paragraph 3.12 of Section B 27.3 of Section B (Governance) applies a Modification Proposal in respect of which the application of the Modification Procedures is suspended pursuant to paragraph 3.15 of Section B 27.2 of Section B (Governance) until the end of the Significant Code Review Phase.	
SoS Modification Proposal	has the meaning given in paragraph 7.1 of Section B47.3 of Section B (Governance).	
Specific Provisions Special Conditions	means the provisions of the Code which only apply to a particular T&S Network, or part of a T&S Network, as set out in the Annexures special conditions of the Licence.	
Specific Provisions	has the meaning given in paragraph 5.3 of Section A (Introduction, Structure and Interpretation).	
Standard Licence Conditions	means the standard licence conditions of the Licence.	
Statement of Expenses Standard Response Service	means has the statement produced by the Secretary pursuant to meaning given in paragraph 1.17 of Section B (Governance 18.2 of Section E (Network Use and Capacity).	
Start-up and Shut-down Procedure	means a Common Interface Procedure established and maintained in accordance with Section G (Common Interface Procedures), as more particularly described in that Section.	
Storage Permit	means a storage permit issued to the T&SCo by the NSTA in relation to a Storage Site that forms part of the T&SCo's T&S Network.	

Term	Meaning	
Storage ComplexSite	has the meaning given to it in regulations in respect of the storage of carbon dioxide under the Energy Act 2008in the Storage of Carbon Dioxide (Licensing etc.) Regulations 2010 and, for the purposes of the Code, is the Storage Site that forms part of the T&S Network, and unless the context otherwise requires, where the T&S Network of a T&SCo includes more than one (1) Storage Site, the term "Storage Site" shall be interpreted to refer to all those Storage Sites.	
Supplemental Compensation Agreement	has the meaning given in the Licence.	
Supported User	means a User who has been awarded an Emittera Revenue Support Contract.	
Surplus Available Capacity	has the meaning given in paragraph 12.5 of Section E38.3 of Section E (Network use Use and capacity Capacity).	
T&SCo	means has the holder of a Licence, who operates a T&S Network meaning given in paragraph 2.1(a) of Section A (Introduction, Structure and Interpretation).	
T&S Network	means a pipeline system for the transportation and storage of carbon dioxide, which may include the Onshore Transportation System and the Offshore Transportation and Storage System.	
T&S Network Data	has the meaning given in paragraph 5.3 of Section I 3.4 of Section I (Data).	
T&S Network Portal	has the meaning given in paragraph 11.1 of Section I (Data).	
T&S Network Portal Minimum Specification	means a web portal established and maintained by T&SCo to publish and provide access to up to date information. Note: it is assumed that T&SCos will have the ability to maintain a portal which will facilitate various processes with Users such asforecasts, nominations and notifications of constraints. means the specification set out in Annexure N (T&S Network Portal Minimum Specification).	
T&S Network Portal ProviderSCo Prolonged Unavailability Event	 in the case of a Supported User, a T&SCo Prolonged Unavailability Event (as such term is defined in that User's Revenue Support Contract); 	
	(b) in the case of any other User: (i) a delay in achievement of T&SCo Commercial Operations Date which has been continuing for a continuous period of at least six (6) Months;	
	(ii) an event or circumstance affecting the relevant T&S	

Term	Meaning	
	<u>(iii)</u>	Network which prevents the User Facility from accessing and exporting any captured carbon dioxide to such T&S Network (including as a result of technical feasibility) which has not arisen out of or in connection with any act, omission, breach or default by the User and which has been continuing for a continuous period of at least six (6) Months; or the occurrence of any of the following: (A) a notice of discontinuation is issued by the Secretary of State to the relevant T&SCo pursuant to a discontinuation agreement entered into between the relevant T&SCo and the Secretary of State; or (B) has the meaning given in paragraph 11.1
T&SCo Representative		of Section I (Data).the relevant T&SCo's Licence to operate the relevant T&S Network is: (i) revoked; and (ii) is not transferred to a substitute T&SCo, such that the relevant T&S Network ceases to operate or the User is no longer able to connect to that T&S Network. sentative of a T&SCo or T&SCos appointed to the
+00-	Modification Pa	
Technical Parameters	means the Tec	of carbon dioxide. chnical Parameters for each individual T&S Network, earts for each T&S Network in Annexure E (Technical T&S Network).
Terminal	pig trap(s) at the the pipeline pig Infrastructure, Transportation	onshore terminal facility and all ancillaries between the ne outlet of the Onshore Transportation System(s) and trap(s) at the Onshore inlet(s) of the Offshore Pipeline which is deemed to form part of the Offshore and Storage System and not form part of the sportation System.
Termination Notice	has the meanir <u>J</u> (General).	ng given in paragraph 11.1 of Section J 12.1 of Section
Terms of Reference		cure D – CO ₂ Quality Monitoring Procedure Terms of ence; and/or

Term	Meaning		
	(b) Annexure F – Emergency Procedure Terms of Reference; and/or		
	(c) Annexure G – Start-up and Shut-down Terms of Reference; and/or		
	(d) Annexure H – Isolation Procedure Terms of Reference; and/or		
	(e) Annexure L – CDS Terms of Reference; and/or		
	(f) Annexure M – Independent Verifier Terms of Reference.		
Third Party Participant	has the meaning given in paragraph 2.1 of Section B22.1 of Section B (Governance).		
ToR Proposal	has the meaning given in paragraph 12.2 of Section ISection I		
Track-1 Cluster Sequencing Process	means the CCUS Cluster Sequencing Process comprising Phase 1, launched on 7 May 2021 and Phase 2, launched on 8 November 2021.		
<u>Unanimously</u>	means in respect of a meeting of the Modification Panel, a determination made with the agreement of all Voting Members participating in the meeting.		
<u>Under Variation Limit</u>	means the percentage set out in the Technical Parameters for a T&S Network.		
Uniform Provisions	means all the provisions of the Code, other than the Specific Provisions, which apply to each T&S Network or part of a T&S Network, unless otherwise specified has the meaning given in paragraph 5.2 of Section A (Introduction, Structure and Interpretation).		
Updated Onshore T&S Charge	has the meaning given in paragraph 15.12 of Section H (Charges, Invoicing and Payment).		
Updated Onshore User T&S Charge	has the meaning given in paragraph 15.10 of Section H (Charges, Invoicing and Payment).		
Updated Offshore T&S Charge	has the meaning given in paragraph 16.12 of Section H (Charges, Invoicing and Payment).		
Updated Offshore User T&S Charge	has the meaning given in paragraph in 16.10 of Section H (Charges, Invoicing and Payment).		
Urgent Modification	means a Modification made pursuant to an Urgent Modification Proposal.		
Urgent Modification Proposal	means a Modification Proposal which the <u>AuthorityRegulator</u> agrees or determines should be treated as an Urgent Modification Proposal.		

Term	Meaning		
User Use of System Charges	means a person other than a T&SCo who is for the time being bound by the Codehas the meaning given in paragraph 2.1 of Section H (Charges, Invoicing and Payment).		
Use of System Charging Methodology	means the methodology for the calculation of Use of System Charges, as set out in Part 1 of Section H (Charges, Invoicing and Payment).		
Use of System Charging Statement	has the meaning given in paragraph 18.1 of Section H (Charges, Invoicing and Payment).		
<u>User</u>	has the meaning given in paragraph 2.1(b) of Section A (Introduction, Structure and Interpretation).		
User Default	means a: (a) (e)-General Default; (b) (d)-Payment Default. (c) (e)-Insolvency Default; or (d) (f)-Construction Longstop Default relating to the User's failure to complete the User Works by the Longstop Date (as extended in accordance with the Construction Agreement) in accordance with paragraph 13 of the Construction Agreement.		
User Discontinuance Date	means the date determined under paragraph 19.1 of Section I 14.1(b) of Section J (General).		
User Exit Notice	has the meaning given in paragraph 13.1 of Section J (General).		
User Facility	means a single facility or several facilities (where such facilities share a single Delivery Point) operated by a User connected to the T&S Network at a Delivery Point has the meaning given paragraph 1.3 of Section F (Network Design and Specification).		
User's Offshore Proportionate Network Charge	has the meaning given in paragraph 7.3 of Section H (Charges, Invoicing and Payment).		
User's Onshore Proportionate Network Charge	has the meaning given in paragraph 7.2 of Section H (Charges, Invoicing and Payment).		
User Representative	means a representative of a User or Users appointed to the Modification Panel.		
User Requirements	means the requirements that arise from legal or contractual obligations of the User including requirements under support contracts and, in the case of a Supported User, its Revenue Support		

Term	Meaning	
	Contract.	
User-Specific Data	has the meaning given to in 5.1 of Section lin paragraph 3.5 of Section I (Data).	
User Type	means: [x]	
	means a User falling into one (1) (and not more than one (1)) of the following categories:	
	(a) hydrogen production; (b) industry (other than waste):	
	(b) industry (other than waste);	
	(d) waste; and	
	(d) <u>dispatchable power,</u> Note: to include list of types of years as dispatchable power.	
	Note: to include list of types of users eg dispatchable power, industrial, hydrogen etc. or any other category identified as part of the Selection Process.	
Valid Connection Application	has the meaning given in paragraph 7.1 of the Section C (Connection).	
<u>Validation Report</u>	has the meaning given in paragraph 6.5 of Section F (Network Design and Specification).	
Variable Opex	means that part of the Opex (as that term is defined in the Licence) which varies with each tonne of CO2carbon dioxide delivered into the T&S Network.	
Variation Request	has the meaning given in paragraph 31.1(c) of Section B (Governance).	
Voting Member	means a T&SCo Representative or User Representative appointed as a Voting Member, and the AuthorityChair, where the AuthorityChair elects to exercise a casting vote under paragraph-1.46 of Section BCasting Vote.	
WACC	has the meaning given in the Licence.	
Warning Notice	has the meaning given to that term in paragraph 5.1 of Section E. (Network use and capacity6 of Section J (General).	
Weekly Forecast	has the meaning given in paragraphs 5.14 and 5.15 of Section Eparagraph 11.2 of Section E (Network use Use and capacity Capacity).	
Workgroup	means a group constituted of T&SCos and Users and Third Party Participants which is convened for the general purposes of consideration and discussion of matters relating to the Code or a Modification Proposal in accordance with its Terms of Reference.	
Workgroup Assessment	means an assessment of a Modification Proposal by a Workgroup in	

Term	Meaning
	accordance with paragraph 3.16 to 3.19 of Section Bparagraphs 28.1 to 28.5 of Section B (Governance).
Workgroup Report	means the report of a Workgroup in relation to a Modification Proposal referred to it by the Modification Panel prepared pursuant to paragraph 3.17 of Section B28.2 of Section B (Governance).
Zero Nomination Exceedance Event	has the meaning given to it in paragraph 25.7(a) of Section E (Network Use and Capacity).
Zero Nomination Low Flow Event	has the meaning given to it in paragraph 25.7(b) of Section E (Network Use and Capacity).

CCS Network Code Exhibit A Code Accession Agreement

[T&SCo]

and

[New User / New T&SCo]

Contents

<u>1</u>	<u>Interpretation</u>	<u>268</u>
<u>2</u>	Admission as a Party	<u>269</u>
<u>3</u>	<u>Counterparts</u>	<u>269</u>

Schedules

<u>1.</u> [Ne	w T&SCo's Carbor	n Dioxide Specifications	s, Measurement Requ	<u>iirements</u> <u>270</u>
and	d Technical Parame	eters]		

THIS CODE ACCESSION AGREEMENT is made on [●] 20[●]

BETWEEN:

- (1) [T&SCO] (Company no. [●]) whose registered office is at [●], (the "T&SCo" which expression includes its legal successors in title and permitted assignees); and
- (2) [NEW T&SCO/NEW USER] (Company no. [●]) whose registered office is at [●], (the "User" which expression includes its legal successors in title and permitted assignees),

each a "Party" and together, the "Parties".

Exhibit A Code Accession Agreement

The Code Agreement will be signed by the Original T&SCo and the Initial Users, to give legal effect to the Code from the Code Implementation Date. Any User or T&SCo that is required to become a Party to the CCS Network Code after the Code Agreement is signed (i.e. after the Code Implementation Date) will become Parties to the Code Agreement by signing an Accession Agreement.

1. Parties

The parties to the Code Accession Agreement shall be:

- (a) a T&SCo that is already a party to the Code Agreement; and
- (b) a New T&SCo or a New User (as those terms are defined in the Code Agreement).

2. RECITALS:

- (A) 2.1 The Code Agreement dated [date to be inserted], attached to this Code Accession Agreement as an exhibit, gives legal effect to the CCS Network Code and makes the Parties legally bound by it.
- (B) 2.2 The [New User/New T&SCo] wishes to be admitted as a Party under the Code Agreement.
- (C) [The T&SCo is already a Party under the Code Agreement and has been nominated by the Regulator and authorised by the existing Parties to the Code Agreement to enter into this Code Accession Agreement with the New T&SCo.]
- (D) [The T&SCo is already a Party under the Code Agreement and, being the T&SCo that owns and operates the T&S Network (or part of it) that the New User is connecting to, has been authorised by the existing Parties to enter into this Code Accession Agreement with the New User.]

THE PARTIES AGREE AS FOLLOWS:

- 1. 3. Interpretation
- 3.1 Capitalised terms used in this Code Accession Agreement shall have the meaning given to them in the Code Agreement and the CCS Network Code.
- 1.2 3.2 Paragraphs [14 to 20] of Section J 15 to 22 of Section J (General) of the CCS Network Code apply to this Code Accession Agreement as if they were set out in this Code Accession Agreement.

- 2. 4. Admission as a Party
- 2.1 4.1 The T&SCo, acting on its own behalf and on behalf of each of the other existing Parties to the Code Agreement, admits the [New User/New T&SCo] as a Party under the Code Agreement.
- 4.2-The [New User/New T&SCo] accepts its admission as a Party and undertakes to perform and to be bound by the Code Agreement as a Party from the date of this <u>Code Accession</u> Agreement.
- 2.3 4.3 This <u>Code Accession</u> Agreement and the Code Agreement shall be read and construed as one (1) document and references (in or pursuant to the Code Agreement) to the Code Agreement should be read and construed as references to the Code Agreement and this Code Accession Agreement.
- 4.4 [The Carbon Dioxide Specifications and Measurement Requirements and Technical Parameters that apply to the New T&SCo's T&S Network, and that will become new annexures to the CCS Network Code upon the New T&SCo becoming a Party to the CCS Network Code, are set out in the Schedule to this Code Accession Agreement.]

Note: paragraph 4.4 provides for a mechanism for the T&S Network-specific Carbon Dioxide Specifications and Measurement Requirements to be included in the Code.

3. **Counterparts**

This Code Accession Agreement may be executed in any number of counterparts and all of those counterparts taken together will be deemed to constitute one and the same instrument.

THIS CODE ACCESSION AGREEMENT has been entered into on the date first above written.

[Schedule]Signatures

Schedule 1

[New T&SCo's Carbon Dioxide Specifications and, Measurement Requirements and Technical Parameters]

Exhibit 1 Exhibit to Code Accession Agreement

CODE AGREEMENT

It is intended that the T&SCos and Users who first achieve a Financial Investment Decision will sign a Code Agreement to give legal effect to the CCS Network Code from the Code Implementation Date.

Subsequent parties — new Users and new T&SCos — will become bound by the Code Agreement by signing a Code Accession Agreement.

1. Parties

The parties to the Code Agreement shall be:

(a) T&SCo (referred to as the "Original T&SCo"); and

Note: depending on timing and process, it may be that more than one T&SCo signsthe Code Agreement on the Code Implementation Date.

(b) the Initial Users.

CCS Network Code Exhibit B Construction Agreement

[Name of T&SCo]

[Name of User]

Contents

<u>1.</u>	<u>Definitions and Interpretation</u>	<u>278</u>
<u>2.</u>	Commencement and Term	<u>283</u>
<u>3.</u>	T&SCo WorksT&SCo Works	<u>283</u>
<u>4.</u>	<u>User Works</u>	<u>283</u>
<u>5.</u>	Design of the Works	<u>284</u>
<u>6.</u>	Consents for the Works	<u>284</u>
<u>7.</u>	Carrying out the Works	<u>285</u>
<u>8.</u>	<u>Liaison</u>	<u>286</u>
<u>9.</u>	Shared access	<u>286</u>
<u>10.</u>	Construction Programme	<u>287</u>
<u>11.</u>	Commissioning Programme	<u>287</u>
<u>12.</u>	Commissioning	<u>287</u>
<u>13.</u>	<u>Delays</u>	<u>287</u>
<u>14.</u>	Consequences of T&SCo delays	<u>287</u>
<u>15.</u>	Consequences of User delays	<u>288</u>
<u>16.</u>	Payment of Liquidated Damages	<u>288</u>
<u>17.</u>	Longstop date	<u>289</u>
<u>18.</u>	Requirements in preparation for commissioning	<u>289</u>
<u>19.</u>	Becoming operational	<u>289</u>
<u>20.</u>	T&S Network Commissioning	<u>289</u>
<u>21.</u>	Works relating to Re-use Service	<u>290</u>
<u>22.</u>	<u>Termination Events</u>	<u>290</u>
<u>23.</u>	General provisions	<u>290</u>
<u>24.</u>	<u>Variations</u>	<u>29</u> 2
<u>25.</u>	Conditions Precedent	<u>29</u> 2
<u>26.</u>	[Anti-bribery and corruption]	<u>29</u> 2

27. [Human rights] 292

Schedules

<u>1.</u>	T&SCo Works	<u>294</u>
<u>2.</u>	<u>User Works</u>	<u>295</u>
<u>3.</u>	<u>Construction Programme</u>	<u>296</u>
<u>4.</u>	Commissioning Programme	<u>298</u>
<u>5.</u>	<u>Liquidated Damages</u>	<u>299</u>
<u>6.</u>	T&S Network Commissioning	<u>300</u>
<u>7.</u>	Works Relating to Re-Use Service	<u>306</u>
<u>8.</u>	Conditions Precedent	<u>307</u>

THIS AGREEMENT/DEED is made on [●] 20[●]

BETWEEN:

- (1) [INSERT T&SCO] (Company no. [●]) whose registered office is at [●], (the "T&SCo" which expression includes its legal successors in title and permitted assignees); and
- (2) [INSERT USER] (Company no. [●]) whose registered office is at [●], (the "User" which expression includes its legal successors in title and permitted assignees),
 - each a 'Party' and together, the 'Parties'.

2. RECITALS:

- (A) Pursuant to Standard Condition B5 (CCS Network Code) of the Licence, the T&SCo is required to maintain a CCS Network Code setting out the terms of the arrangements for connection to and use of the T&S Network.
- (B) The User has applied for connection to and use of the T&S Network and pursuant to Standard Condition B3 (Access to T&S Network) of the Licence, the T&SCo is required to offer terms in accordance with the Code in this respect.
- 2.1 The Original T&SCo holds a Licence.
- 2.2 The Original T&SCo, with input from industry, prepared a document as the CCS Network Codewhich the Original T&SCo and all other T&SCos are required to do so pursuant to their Licence.
- (C) 2.3 The Original T&SCo is entering into this Code Agreement for the purpose of giving effect to and binding itself to the CCS Network Code.and the User are parties to the [Code Agreement/Accession Agreement] (being an agreement by which the Code is made contractually binding between the parties).
- 2.4 The Initial Users are the first Users to connect to a T&S Network and are entering into this Code.

 Agreement for the purposes of binding themselves to the CCS Network Code.
- (D) Certain works are required to be carried out and performed by the T&SCo and the User to connect the User Facility to the T&S Network.
- (E) <u>Certain works are also required to commission the Works.</u>
- (F) This Construction Agreement is entered into pursuant to the terms of the Code.

THE PARTIES AGREE AS FOLLOWS:

- 1. 3. Definitions and Interpretation
- 3.1 Capitalised terms used in the Code Agreement shall have the meaning given to them in the CCS-Network Code and:
- 1.1 Unless the subject matter or context otherwise requires or is inconsistent therewith, terms and expressions defined in Section K (Glossary) of the Code and in the Connection

Agreement have the same meanings, interpretations or construction in this Construction Agreement.

1.2 <u>The following terms shall have the following meanings:</u>

Commissioned, in relation to the Works, means that all works and activities required to be carried out in accordance with the Commissioning Programme or T&S Network

Commissioning Programme (as applicable) have been carried out;

Commissioning Programme means the sequence of operations and/or tests necessary to connect the User Facility and Works to the T&S Network as set out in Schedule 4 for the purpose of making the Delivery Point operable;

<u>Commissioning Programme Commencement Date</u> means the date for the <u>commencement of the Commissioning Programme as specified in the Commissioning Programme in Schedule 4 or any substituted date fixed under the terms of this Construction Agreement;</u>

Completion Date means the date for completion of the Works (including that the Works are Commissioned) as specified in the Construction Programme in Schedule 3 or any substituted date fixed under the terms of this Construction Agreement;

Conditions Precedent means the conditions precedent set out in Schedule 8;

Consents means:

- (a) Accession Requirements means: all such planning and other statutory consents; and
 - (i) for New Users, the requirements to be complied with by a New User pursuant to the CCS Network Code before the New User may accede to this Agreement; and
 - (ii) for New T&SCos, the requirement that the New T&SCo holds a Licence;
- (b) CCS Network Code means the CCS Network Code prepared by the Original T&SCopursuant to its Licences, as modified;
- (b) <u>all wayleaves, easements, rights over or interests in land or any other property</u> consent; or
- <u>permission of any kind as shall be necessary for the construction of the Works and for commencement and carrying on of any activity proposed to be undertaken at or from such Works when completed;</u>

Construction Longstop Default has the meaning given in clause 17;

Construction Programme means the agreed programme for the Works to be carried out by or on behalf of the T&SCo and the User as set out in Schedule 3 or as amended from time to time in accordance with the express terms of this Construction Agreement;

Force Majeure Event means:

<u>any event or combination of events or circumstance (other than the lack, nonreceipt or unavailability of financial resources or funds) which is beyond the control of a Party</u>

- acting and having acted as a Reasonable and Prudent Operator, and which prevents that Party from or causes hindrance, delay or impediment to that Party in fulfilling all or any of its obligations under this Construction Agreement; unforeseen adverse weather conditions shall not constitute a Force Majeure Event; and
- (b) any failure by the Party to perform its obligations hereunder to the extent that such failure was caused by any change in the law or cancellation of any consent, approval or licence rendering it unlawful for a Party to comply with its obligations hereunder after the date of this Construction Agreement unless such change or cancellation would not have occurred except for any act or omission of the Party concerned in relation to such law, consent, approval or licence unless itself caused by the Force Majeure Event;

Liquidated Damages means the sums specified in or calculated pursuant to Schedule 5;

(c) Code Implementation Longstop Date means [date to be inserted agreed on a case by case basis];

Safety Rules means the rules relating to occupational health and safety that seek to ensure that persons working on plant and/or apparatus owned or operated by the T&SCo or User (as applicable) to which the rules apply are safeguarded from hazards arising from such plant or apparatus;

- (d) Effective Date means:
 - (i) with respect to each of the Original T&SCo and Initial Users, the Code-Implementation Date;
 - (ii) with respect to any New User or New T&SCo who is admitted as an additional Party to this Agreement after the date of this Agreement, the date of the relevant Code Accession Agreement;
- (e) **New T&SCo** means a T&SCo who is admitted as an additional Party to this Agreement after the date of this Agreement;
- (f) ew User" means a User who is admitted as an additional Party to this Agreement after the date of this Agreement;
- (g) Party means:
 - (i) the Original T&SCo;
 - (ii) the Initial Users;
 - (iii) each New T&SCo; and
 - (iv) each New User.
- 3.2 Paragraphs [14 to 20] of Section J (General) of the CCS Network Code apply to this Agreement as if they were set out in this Agreement.

4. CCS Network Code

- 4.1 The CCS Network Code is hereby given effect and made binding upon each Party with effect from the Effective Date.
- 4.2 With effect from the Effective Date each Party undertakes to each other Party to comply with and to perform its obligations in accordance with and subject to the CCS Network Code.

5. New T&SCo

- 5.1 Where a New T&SCo has complied with all the relevant Accession Requirements, the New T&SCo shall become a Party to this Agreement by signing a Code Accession Agreement;
- 5.2 Each Party that has already been admitted as a Party to the Code Agreement:
 - (a) authorises a T&SCo (that is already a Party) nominated by the Authority to sign a Code-Accession Agreement with a New T&SCo being admitted as a Party; and
 - (b) [agrees that upon a New T&SCo signing a Code Accession Agreement, the Carbon-Dioxide Specifications and Measurement Requirements that apply to that New T&SCo's T&S Network shall be added as new annexures to the CCS Network Code];
- 5.3 A copy of each Code Accession Agreement shall be provided to the Authority within 28 days after the Code Accession Agreement is made.

6. New Users

6.1 Where a New User is connecting to a T&S Network, the New User: Commissioning Programme has the meaning set out in Schedule 6;

(a) shall satisfy the T&SCo that owns and operates that T&S Network, or part of a T&S Network, that the New User is connecting to, that it has complied with all the relevant Accession Requirements; and

<u>T&SCo Commissioning Activities</u> has the meaning given to 'Commissioning Activities' in the T&SCo's Licence;

T&SCo Conditions Precedent has the meaning set out in Schedule 8;

(b) provided that T&SCo is already a Party, shall enter into a Code Accession Agreement with that Works means those Works specified in the Schedule 1 for which the T&SCo- is responsible;

6.2 Each Party that has already been admitted as a Party to the Code Agreement authorises the T&SCo referred to in paragraph 6.1 to sign a Code Accession Agreement with a New Userbeing admitted as a Party.

7. Ceasing to be Party to this Agreement

- 7.1 A Party that is a User will cease to be a Party where it is a Discontinuing User.
- 7.2 A Party that is a T&SCo will cease to be a Party where it no longer holds a Licence.

User Facility means [●];

User Conditions Precedent has the meaning set out in Schedule 8;

<u>User Commissioning Activities</u> means the operations and/or tests necessary to commission the connection of the User Facility to the T&S Network, as set out in the construction agreement relating to the User Facility;

<u>User Works</u> means those Works specified in Schedule 2 for which the User is responsible; and

Works means the works required to be carried out and performed by the T&SCo and the User to connect the User Facility to the T&S Network.

Exhibit B Construction Agreement

1. Parties

The parties to the Construction Agreement shall be:

- (a) **T&SCo** (being the identified party to the Agreement which, in relation to each T&S Network or part of a T&S Network, shall be the entity that is the holder of a Licence authorising it to operate that T&S Network or part of a T&S Network); and
- (b) **User** (being the identified party to the Agreement which is the person other than T&SCowho is bound by the provisions of the CCS Network Code and delivers carbon dioxideinto the T&S Network at a Delivery Point).
- 2. ScopeCommencement and Term
- 2.1 Subject to clause 25, this Construction Agreement becomes effective on the date of its execution.
- 2.2 This Construction Agreement shall continue to be in force until it is terminated in accordance with the provisions of Section J (General) of the Code.

The Construction Agreement relates to the:

- (a) carrying out of work required to be performed by T&SCo (**T&SCo Works**) and the User (**User Works**) to connect the User Facility to the T&S Network (**the Works**); and
- (b) commissioning of the Works.

Note: the scope of the works will be specifically described in Schedules 1 and 2 on an agreement by agreement basis.

Note: for initial users, commissioning of the connection will be conditional upon-commissioning of the T&S Network itself. For this reason, Schedule 7, dealing with the interdependencies between the two, will apply to users who enter into a Construction-Agreement with T&SCo before the T&S Network achieves commercial operation. Schedule 7-may also be relevant in other instances where works to connect a User are tied to substantial expansion of an existing T&S Network.

3. T&SCo Works

- 3.1 The T&SCo Works, to be carried out by the T&SCo, are those works required to connect the User Facility to the T&S Network, up to the boundary of the User Facility, as more particularly described in Schedule 1Schedule 1.
- 3.2 The T&SCo shall be responsible for all costs associated with carrying out the T&SCo Works.
- 4. User Works
- 4.1 The User Works, to be carried out by the User, are those works required to connect the User Facility to the T&S Network, within boundary of the User Facility, as more particularly described in Schedule 2Schedule 2.

- 4.2 The User shall be responsible for all costs associated with carrying out the User Works.
- 5. Carrying out of Design of the Works
- <u>Without prejudice to the Parties' obligations to ensure that the Works comply with the requirements of this Construction Agreement:</u>
 - (a) the User must, at regular intervals, and in any case, no less than once every three (3) months from the date of this Construction Agreement until the User's designs for the User Works are finalised, provide the T&SCo with the opportunity to review the User's designs for the User Works; and
 - (b) the T&SCo must, at regular intervals, and in any case, no less than once every three
 (3) months from the date of this Construction Agreement until the T&SCo designs for the T&SCo Works are finalised, provide the User with the opportunity to review the T&SCo's designs for the T&SCo Works.
- 5.1 The T&SCo shall have the right to comment on the designs submitted to it under clause
 5.1 where it considers, acting as a Reasonable and Prudent Operator-and subject to the
 terms and conditions of this Agreement, carry out the T&SCo Works in accordance with the
 Construction Programme., that such designs may prevent the User Works from complying
 with the requirements of this Construction Agreement or the Code.
- 5.3 The User must incorporate any comments provided by the T&SCo under clause 5.2 within two (2) calendar months of receipt of the same, and resubmit the designs to the T&SCo, indicating how they have addressed the T&SCo comments.
- 5.2 The User shall, acting as a Reasonable and Prudent Operator and subject to the terms and conditions of this Agreement, carry out the User Works in accordance with the Construction-Programme.
- 5.3 The User shall, in carrying out the User Works, comply with the User Work Obligations and any reasonable directions issued by T&SCo.
- 5.4 T&SCo shall, in carrying out the T&SCo Works, comply with any operational and health and safety requirements relating to the User Facility, which have been communicated by the User to T&SCo, and any reasonable directions issued by the User.

6. Design of the Works

Without prejudice to the User's obligation to ensure that the Works comply with the requirements of this Construction Agreement, the User shall provide T&SCo with the opportunity to regularly review the User's designs for the User Works and comment where it considers that such design may prevent the User Works from complying with the requirements of this Agreement or the Code.

- 6. **7.** Consents for the Works
- 6.1 7.1 Following the effective date of the Subject to clauses 6.2 and 6.3, following the date of this Construction Agreement the T&SCo shallmust use its best endeavours to obtain in relation to the T&SCo Works, and the User shallmust use its best endeavours to obtain in relation to the User Works, all outstanding Consents. Each partyParty shall give advice and assistance to the other to the extent reasonably required by the other in the furtherance of these obligations. Further, each Party shall, so far as it is legally able to do so, grant to the other all such wayleaves, easements, servitude rights, rights over or interests (but not estates as

regards land in England and Wales and not heritable or leasehold interests as regards land in Scotland) in land or any other Consents reasonably required by the other in order to enable, without prejudice to clause 9, the Works to be expeditiously completed and to enable that other to carry out its obligations to the other under this Construction Agreement and in all cases subject to such terms and conditions as are reasonable.

Note: it is expected that all substantive consents will be in place prior to commencement of the Construction Agreement.

- 6.2 7.2 Each partyParty shall provide copies of any relevant correspondence and keep the other partyParty regularly updated in writing or by such other means as the parties may agree as to the progress made from time to time in the obtaining of relevant Consents.
- 6.3 The following additional provisions shall apply in respect of the Consents and the Works:
 - (a) all dates specified in this Construction Agreement are subject to the T&SCo obtaining
 Consents for the T&SCo Works and the User obtaining Consents for the User Works
 in a form acceptable to it within the time required to carry out the T&SCo Works or the
 User Works (as applicable) in accordance with the Construction Programme;
 - (b) in the event of:
 - (i) the Consents not being obtained by the required date;
 - (ii) the Consents being subject to conditions which affect the dates; or
 - (iii) the T&SCo or the User wishing to amend the T&SCo Works, the User Works, the T&SCo Commissioning Activities, the User Commissioning Activities or the Commissioning Programme to facilitate the granting of the Consents,

either Party shall be entitled to raise a request for this Construction Agreement to be varied in accordance with clause 24 to revise the T&SCo Works or User Works (as applicable) and/or the dates specified in this Construction Agreement.

7. **Carrying out the Works**

- Forthwith following the date of this Construction Agreement, the User and the T&SCo must agree the Safety Rules to apply to the User Works and the Safety Rules to apply to the T&S Works during the Construction Programme and Commissioning Programme. Failing agreement within three (3) months of the date of this Construction Agreement, the Parties shall be entitled to refer the dispute to an Expert for determination, in accordance with paragraph 51 of Section B (Governance).
- 7.2 The T&SCo must, acting as a Reasonable and Prudent Operator and subject to the terms and conditions of this Construction Agreement, carry out the T&SCo Works in accordance with the Construction Programme.
- 7.3 The User must, acting as a Reasonable and Prudent Operator and subject to the terms and conditions of this Construction Agreement, carry out the User Works in accordance with the Construction Programme.
- 7.4 Each Party shall, in carrying out the User Works and the T&SCo Works (as applicable), comply with the applicable Safety Rules and any reasonable directions issued by the other Party.

- Parties shall be entitled to contract or sub-contract for the carrying out of their respective parts of the Works. The User or any contractor on its behalf shall be responsible for commencing and for carrying out the User Works to such stage of completion as shall render them capable of being Commissioned and the T&SCo works to such stage of completion as shall render them capable of being Commissioned.
- 7.6 It is hereby agreed and declared for the purposes of the Construction (Design and Management) Regulations 2015 that the User is the only client in respect of the User Works and the T&SCo is the only client in respect of the T&SCo Works and each of the User and the T&SCo must accordingly discharge all the duties of clients under the said Regulations.
- 7.7 If at any time prior to the Completion Date it is necessary for a Party to make any addition to or omission from or amendment to the Works in relation to that Party as a result of an order or direction made in accordance with any Regulatory Requirements, that Party shall be entitled to raise a request for this Construction Agreement to be varied in accordance with clause 24.

8. Liaison

- 8.1 The parties shall continuously liaise throughout the Construction Programme and the Commissioning Programme and each partyParty shall provide to the other all information relating to:
 - (a) in the case of the User, the User Works; and
 - (b) in the case of the T&SCo, the T&SCo Works,

as is reasonably necessary to assist the other Party in performance of that ether's other
Party's part of the Works, and shall use all reasonable endeavours to coordinate and integrate their respective part of the Works.

- There shall be on-site meetings between representatives of the parties at intervals to be agreed between the parties.
- 8.3 Each <u>partyParty</u> shall deliver to the other <u>partyParty</u> a written report of progress during each calendar quarter within seven (7) Days of the end of that quarter.

9. Shared access

- 9.1 During the <u>period of and at the times and as otherwise provided in the Construction</u>
 Programme and the Commissioning Programme, the User <u>shallmust</u> provide <u>the T&SCo</u> with <u>necessary</u> access to the site of the User Works, and <u>the T&SCo</u> <u>shallmust</u> provide the User <u>with necessary</u> access to the site of the T&SCo Works (and each case that access rights shall be extended to that <u>partyParty'</u> employees, agents, suppliers and contractors) but not so as to disrupt or delay the construction and completion of the <u>other'sother's</u> Works on the said sites or the operation of the <u>other'sother's</u> plant and apparatus located thereon.
- 9.2 Any access under paragraph clause 9.1 shall be exercised only:
 - (a) where it is reasonably necessary for the purposes of carrying out of the Works;
 - (b) with the prior approval of the other party (which should not be unreasonably withheld); and
 - (c) in accordance with any reasonable requirements (including safety procedures the Safety Rules) of the owner of the site relating to such access.

10. Construction Programme

An agreed Construction Programme for the Works is set out in Schedule 3 Schedule 3, subject to any changes to that Construction Programme in accordance with this Construction Agreement.

Note: where Schedule 6 applies, clauses 11, 12, 13, 14, 15, 17, 18 and 19 will not apply.

11. Commissioning Programme

An agreed Commissioning Programme for the Works is set out in Schedule 5Schedule 4, subject to any changes to that Commissioning Programme in accordance with this Construction Agreement.

12. **Commissioning**

- 12.1 Each <u>partyParty</u> shall give written notice to the other declaring its readiness to commence the Commissioning Programme.
- The Commissioning Programme shall commence forthwith once both parties Parties Parties have given written notice to the other under paragraphclause 12.1.
 - 12.3 The Works shall be deemed to have been Commissioned on the date that the T&SCo certifies in writing to that effect.

13. **Delays**

- 13.1 If either partyParty shall have reason to believe that it is being delayed or will be delayed in carrying out that party's Party's Works for any reason (whether it is one entitling it to an adjustment to that the Completion Date and/or the fixing of a new date under partyParty in writing of the circumstances giving rise to the delay and of the extent of the actual and/or anticipated delay.
- 13.2 If prior to the Completion Date a partyParty (in this paragraphclause 13.2 "the "Affected Party") shall be delayed in carrying out any of the Affected Party's Party's Works (including their commissioning) by reason of:
 - (a) any act, default or omission on the part of the other Party (in this paragraph the "Defaulting Party") or the Defaulting Party's employees, agents, contractors or sub-contractors; or
 - (b) a Force Majeure Event,

the Affected Party shall be entitled to have any dates in the Construction Programme and/or the Commissioning Programme (including the Completion Date and the Commissioning Programme Commencement Date) and the Longstop Date (as the case may be) amended as may be fair and reasonable in the circumstances provided that it shall have notified the other Party in writing of such event within twenty-eight (28) Days of it becoming aware of the occurrence giving rise to the delay together with an estimate of the proposed delay which it will cause the Affected Party.

14. Consequences of T&SCo delays

Delayed commencement of commissioning

14.1 In the event that the actual date of commencement of the Commissioning Programme is later than the Commissioning Programme Commencement Date, the T&SCo (if and to the extent that it is responsible for delayed commissioning beyond the Commissioning Programme

- Commencement Date, shall be liable to pay the User Liquidated Damages for each day Day that the actual date of commencement of the Commissioning Programme is later than the Commissioning Programme Commencement Date.
- 14.2 Such Liquidated Damages shall cease to be payable in respect of any period after the date of actual commencement of the Commissioning Programme.

Delayed completion of commissioning

- In the event that the actual date on which the T&SCo Works are Commissioned in accordance with the Commissioning Programme is later than the Completion Date the T&SCo (if and to the extent that it is responsible for delayed completion beyond the Completion Date) shall be liable to pay the User Liquidated Damages for each day that the actual date on which the T&SCo Works are Commissioned is later than the Completion Date.
 - **14.4**-Such Liquidated Damages shall cease to be payable in respect of any period after completion of the T&SCo Works.

15. Consequences of User delays

Delayed commencement of commissioning

- In the event that the actual date of commencement of the Commissioning Programme is later than the Commissioning Programme Commencement Date, the User (if and to the extent that it is responsible for delayed commissioning beyond the Commissioning Programme Commencement Date,), shall be liable to pay the T&SCo Liquidated Damages for each day that the actual date of commencement of the Commissioning Programme is later than the Commissioning Programme Commencement Date.
- 15.2 Such Liquidated Damages shall cease to be payable in respect of any period after the date of actual commencement of the Commissioning Programme.

Delayed completion of commissioning

- 15.3 In the event that the actual date on which the User Works are Commissioned is later than the Completion Date, the User (if and to the extent that it is responsible for delayed completion beyond the Completion Date) shall be liable to pay the T&SCo Liquidated Damages for each day that the actual date on which the User Works are Commissioned is later than the Completion Date.
 - 45.4 Such Liquidated Damages shall cease to be payable in respect of any period after completion of the User Works.

16. **Payment of Liquidated Damages**

- Liquidated Damages payable under this Construction Agreement shall accrue on a daily basis but shall be payable monthly.
- On or before the fifteenth Day of each month the Party entitled to receive the payment of the Liquidated Damages shall send to the other Party a statement of the Liquidated Damages which have accrued in the previous calendar month.
- 16.3 The Party receiving such statement shall in the absence of manifest error pay the Liquidated Damages shown on the statement within twenty eight (28) Days of the date upon which the statement is received.
- 16.4 <u>The payment or allowance of Liquidated Damages pursuant to:</u>

- (a) clauses 14 and 15 shall be in full satisfaction of each Party's liability for failure to perform its obligations by the Commissioning Programme Commencement Date and/or the Completion Date as appropriate; or
- (b) clauses 8 and 9 of Part C to Schedule 6 shall be in full satisfaction of each Party's
 liability for failure to perform its relevant obligations by the corresponding dates from
 which Liquidated Damages begin to accrue under clauses 8 and 9 of Part C to
 Schedule 6,

(as applicable).

17. 16. Longstop date

In the event that the T&SCo or the User shall-have failed, in circumstances not entitling it to the fixing of a new Longstop Date pursuant to paragraph clause 13.2, to complete the T&S Works or User Works (as applicable) by the Longstop Date, then such failure shall constitute a Construction Longstop Default and Section J (general) paragraph 12 or paragraph 13 (as applicable) of Section J (General) of the Code shall apply.

18. 47. Requirements in preparation for commissioning

- Not later than three (3) months prior to the expected Commissioning Programme

 Commencement Date or by such other time as may be agreed between the parties parties, the parties Parties shall provide to each other:
 - (a) any technical drawings and other details relating to each other's Works;
 - (b) contact details of key personnel to be involved in the carrying out of the Commissioning Programme; and

(c) in the case of the User, all information that the User is required to provide to the T&SCo pursuant to the Code and the Industrial Common Interface Procedures in order for the User to be able to commence delivery of carbon dioxide at the Delivery Point.

19. 48. Becoming operational

19.1 Once:

- (a) <u>the T&SCo</u> has certified that the Works have been Commissioned pursuant to paragraph clause 12.3; and
- (b) the User has complied with the requirements of paragraph 17(c)clause 18.1(c),

the T&SCo shallmust forthwith notify the User in writing that the connection at the Delivery Point has become operational.***

20. 49. T&S Network Commissioning

20.1 Schedule 7Schedule 6 has effect under this Construction Agreement.

Drafting instruction: Schedule 7Schedule 6 applies (i) where this Construction Agreement is being entered into prior to System Acceptance of the T&S Network, where System Acceptance means when the AuthorityRegulator has confirmed, in accordance with the T&SCo's Licence, that the T&SCo has completed T&S Network commissioning as required in order to be able to commence commercial operation of the T&S Network. Where Schedule 7 and (ii) completion of construction and commissioning of the User Facility and User Works is

anticipated to align with System Acceptance, such that the User will be a commissioning User of the T&S Network. Where Schedule 6 is not applicable, then this paragraph 19 clause 20 should be deleted.

- 21. 20. Works relating to Re-use Service
- 21.1 Schedule 8 Schedule 7 has effect under this Construction Agreement.

Drafting instruction: Schedule 8 Schedule 7 applies where this Construction Agreement also covers any Works relating to the Re-use Service which may be provided to some Users under Section F (Network Design and Specification) of the Code, and where details, such as the payment by the User for any part of the T&SCo Works relating to the Re-use Service, need to be set out in the Construction Agreement. Where Schedule 8 Schedule 7 is not applicable, then this paragraph 20 clause 21 should be deleted.

- 22. 21. Termination Events
- This Construction Agreement the associated Connection Agreement and the Code
 Agreement (in respect of the User) shall be co-terminus coterminous such that upon
 termination of this Construction Agreement, the associated Connection Agreement and the
 Code Agreement shall also terminate.
- 23. **22.** General provisions
- 23.1 Paragraphs [14 to 20] of Section J (General) of the CCS Network Code apply to this Construction Agreement as if they were set out in this Construction Agreement.

Note: this refers to the general boilerplate provisions in Section J of the CCS Network Code.

- 24. 23. Assignment Variations
- 23.1 Neither party shall assign its rights and obligations under this Agreement except in accordance with the provisions of clauses 23.2 and 23.3.
- 23.2 The User shall only be entitled to assign its rights and obligations under this Agreement to a thirdparty where:
 - (a) the User also assigns the ownership and operation of the User Facility to the third party;
 - (b) the User has obtained the prior written consent of T&SCo, which shall not beunreasonably withheld; and
 - (c) the third party becomes bound by the Code and this Agreement.
- 23.3 T&SCo shall only be entitled to assign its rights and obligations under this Agreement to a thirdparty where:
- 24.1 Subject to clause 24.4, no variation to this Construction Agreement shall be effective unless made in writing and signed by or on behalf of both the T&SCo and the User.
- Where a Party has raised a request for this Construction Agreement to be varied in accordance with clause 6.3 or clause 7.7, the other Party shall be required to review and consider in good faith any such request and (subject to clause 24.3) act reasonably in its consideration of whether to accept the proposed variation.

- 24.3 The Parties acknowledge and agree that in considering whether to accept a proposed variation to this Construction Agreement under clause 24.2, it will be reasonable to refuse such a request where any such variation would adversely impact the reviewing Party's ability to deliver the Works and/or where such variation would cause one of the Parties to be in breach of the terms of any contract or agreement that it is a party to at the time such variation is requested.
- 24.4 The T&SCo and the User must effect any amendment required to be made to this

 Construction Agreement as a result of a Modification to the Code, including as a result of any
 modifications to or directions issued by the Regulator in connection with the Licence or any
 Project-Specific Document referred to in the T&SCo's Licence.
- 24.5 The User authorises and instructs the T&SCo to make any amendment required under clause 24.4 on its behalf and undertakes not to withdraw, qualify or revoke such authority or instruction at any time.

25. **Conditions Precedent**

- 25.1 The provisions of, and the rights and obligations of the Parties pursuant to, this Construction Agreement (other than this clause 25) are conditional upon the Conditions Precedent being:
 - (a) <u>fulfilled by the T&SCo also assigns the ownership and operation of that part, in the case</u> of the T&S Network that includes the Delivery Point to a third party holding a Licence; and SCo Conditions Precedent;
 - (b) <u>fulfilled by the User, in the case of the User Conditions Precedent; or</u>
 - (c) waived by the Party for whose benefit the relevant Condition Precedent has been included (as specified in Schedule 8).
- 25.2 The T&SCo must use reasonable endeavours to fulfil or procure the fulfilment of the T&SCo Conditions Precedent as soon as reasonably practicable, and in any event no later than the date specified in Schedule 8,
- 25.3 The User must use reasonable endeavours to fulfil or procure the fulfilment of the User Conditions Precedent as soon as reasonably practicable, and in any event no later than the date specified in Schedule 8.
- 25.4 The Parties shall keep each other informed about the status of the Conditions Precedent and each Party shall notify the other Party when a Condition Precedent has been fulfilled.

26. [Anti-bribery and corruption]

- 26.1 [The Parties acknowledge that the intent behind this Construction Agreement is to ensure consistency with Legal Requirements relating to:
 - (a) the prevention of corruption and bribery ("ABC Laws"); and
 - (b) the third party becomes bound by the Code and this Agreementanti-money laundering ("Anti-money Laundering Laws").

- 26.2 In performing its obligations and activities in accordance with this Construction Agreement and the Code, each Party must comply with all applicable ABC Laws and Anti-money Laundering Laws.
- Each Party represents, warrants and undertakes to the other Party that it has not, directly or indirectly, engaged in any bribery or corruption or other act that may contravene any ABC Laws or Anti-Bribery Laws.]

Drafting instruction: this clause 26 (Anti-Bribery and Corruption) is optional for inclusion by the Parties. It is anticipated that the Parties may agree alternative provisions regulating the subject matter contained within this clause 26.

- 27. 24. Definitions [Human rights]
- 27.1 Capitalised terms used in the In performing its obligations and activities in accordance with this Construction Agreement have the meaning given to them in and the Code and, each Party must:

Commissioning Programme Commencement Date means the date for the commencement of the Commissioning Programme as specified in the Commissioning Programme in Schedule 5 or any substituted date fixed under the terms of this Construction-Agreement;

Commissioned, in relation to the Works, means that all works and activities required to becarried out in accordance with the Commissioning Programme have been carried out;

Completion Date means the date for completion of the Works as specified in the Construction Programme in Schedule 3 or any substituted date fixed under the terms of this Construction Agreement;

Consents means:

- (a) all such planning and other statutory consents comply with all applicable Human Rights

 Laws; and
- (b) take reasonable steps to ensure its contractors and consultants comply with all applicable Human Rights Laws in performing any obligations and activities in connection with this Construction Agreement and the Code,

where "Human Rights Laws" means all Legal Requirements regarding respecting human rights, the use of labour (including the terms and conditions of employment), prevention of smuggling and trafficking of human beings and modern slavery and immigration.]

Drafting instruction: this clause 27 (Human Rights) is optional for inclusion by the Parties. It is anticipated that the Parties may agree alternative provisions regulating the subject matter contained within this clause 27.

(b) all wayleaves, easements, rights over or interests in land or any other consent; or

(c) permission of any kind as shall be necessary for the construction of the Works and forcommencement and carrying on of any activity proposed to be undertaken at or fromsuch Works when completed.

Construction Longstop Default has the meaning given in paragraph 16;

Commissioning Programme means the sequence of operations/tests necessary to connect the User Facility and Works to the T&S Network, as set out in Schedule 5 for the purpose of making the Delivery Point operable;

Construction Programme means the agreed programme for the Works to be carried out by or on behalf of T&SCo and the User as set out in Schedule 3 to the Construction Agreement or as amended from time to time in accordance with the express terms of the Construction Agreement;

Force Majeure Event means:

- (a) any event or combination of events or circumstance (other than the lack, nonreceipt or unavailability of financial resources or funds) which is beyond the control of a party acting and having acted as a Reasonable and Prudent Operator, and which prevents that party from or causes hindrance, delay or impediment to that party in fulfilling all or any of its obligations under the Agreement; For the avoidance of doubt unforeseen adverse weather;
- (b) any failure by the party to perform its obligations hereunder to the extent that such failure was caused by any change in the law or cancellation of any consent, approval or licence rendering it unlawful for a party to comply with its obligations hereunder after the date of Acceptance unless such change or cancellation would not have occurred except for any act or omission of the party concerned in relation to such law, consent, approval or licence unless itself caused by the Force Majeure Event;

Liquidated Damages means the sums specified in or calculated pursuant to Schedule 5 to this Construction Agreement;

Longstop Date means [to be agreed on a case by case basis];

Note: where Schedule 7 applies, the Longstop Date relating to the Works will be linked to the Longstop Dates relating to the User Facility and the T&S Network.

T&SCo Works means those Works specified in the Construction Agreement for which T&SCo is responsible;

User Works means those Works specified in the Construction Agreement for which the User is responsible.

Schedule 1 Schedule 1

T&SCo Works

Note: details of the T&SCo Works, including design specifications to be included. <u>The T&SCo Works</u> are intended to include any physical works for which the T&SCo is responsible and which are required to be carried out by the T&SCo on its side of the Delivery Point boundary to facilitate the connection of the User Facility to the T&S Network.

Schedule 2 Schedule 2

User Works

Note: details of the User Works, including design specifications to be included. The User Works are intended to include any physical works for which the User is responsible and which are required to be carried out by the User on its side of the Delivery Point boundary to facilitate the connection of the User Facility to the T&S Network.

Schedule 3 Schedule 3

Construction Programme

Key dates of Construction Programme

Completion Date: [agreed date to be inserted]

Note: other dates and details to be inserted.

Schedule 4 Schedule 4

User Works Obligations

Note: any general technical/operational/safety requirements that will apply to all User Works to be inserted.

Schedule 5

Commissioning Programme

Key dates of Commissioning Programme

Commissioning Programme Commencement Date: [agreed date to be inserted]

Note: other dates and details to be inserted.

Schedule 5 Schedule 6

Liquidated Damages

- 1. User Liquidated Damages $\mathfrak{L}[x]$
- 2. T&SCo Liquidated Damages $\pounds[x]$

Drafting instruction: in relation to any Construction Agreement entered into in the First Regulatory Period under T&SCo's Licence, the User Liquidated Damages and T&SCo Liquidated Damages shall be set at £0.

Schedule 6 Schedule 7

T&S Network Commissioning

Drafting instruction: this schedule is only to be included where applicable - see paragraph 19.

Part A Scope of this Schedule 6

- 1. Scope of this Schedule 7Schedule 6
- 1.1 This Schedule 7 Schedule 6 applies where this Construction Agreement has been entered into prior to System Acceptance of the T&S Network, where "System Acceptance" means when the AuthorityRegulator has confirmed, in accordance with the T&SCo's Licence, that the T&SCo has completed T&S Network commissioning as required in order to be able to commence commercial operation of the T&S Network, and the User will be participating in commissioning of the T&S Network prior to System Acceptance.
- 1.2 Notwithstanding any other provisions of this Construction Agreement, the provisions set out in Part B and Part 2C of this Schedule 6 shall apply.
- 1.3 The provisions set out in <u>Part B and Part 2C</u> of this <u>Schedule 7Schedule 6</u> govern the interaction between:
 - (a) the construction and commissioning of the User Facility and the User Works as required to enable the construction and commissioning of the T&S Network and the T&SCo workWorks (including, where applicable, the supply of carbon dioxide for the commissioning of the T&S Network); and
 - (b) the construction and commissioning of the T&S Network in accordance with the T&SCo's Licence and the T&SCo Works.

Part B 2-T&S Network Commissioning Definitions

- 1. 2.1 Definitions
- 1.1 In this Schedule 7Schedule 6, the following definitions apply:

Commercial Operations Date has the meaning given in the Licence;

Emitter Revenue Support Contract means a revenue support contract granted to an emitter a transport and storage network user pursuant to Part 2 of the [Energy Bill] Act 2023 or a Contract for Difference granted to an emitter a transport and storage network user pursuant to Chapter 2 of Part 2 of the Energy Act 2013;

Scheduled COD has the meaning given in the Licence;

Supported User means a User who has been awarded an Emittera Revenue Support Contract:

T&S Network Commissioning Plan Programme means the programme established by the T&SCo pursuant to its Licence, for the commissioning of the T&S Network;

T&S Network Commissioning Programme Commencement Date means the date specified in the T&S Network Commissioning Programme for the commencement of the T&SCo Commissioning Activities or any substituted date fixed under the terms of this Construction Agreement;

T&S Network Longstop Date means the longstop date Longstop Date, as such term is defined under the T&SCo's Licence;

T&SCo Delay means any delay in the construction or commissioning of the T&S Network and/or the T&SCo Works;

User Delay means any delay in the construction or commissioning of the User Facility and/or the User Works; and

User Facility Longstop Date means:

- (a) in the case of a Supported User, the longstop date under the User's Emitter Revenue Support Agreement Contract (where applicable), or
- (b) in the case of a User which is not a Supported User, [to be agreed on a case by case basis].
- 1.2 Defined terms used in this Schedule 6 and not listed in clause 1.1 of Part B of this Schedule 6 have the meaning given to them in clause 1 of this Construction Agreement.

Part C 2.2 Operative Specific provisions relating to be agreed by User and T&SCoS Network Commissioning

- 1. **Effect of existing provisions**
- 1.1 Where this Schedule 6 applies, the following clauses of this Construction Agreement (and any definitions used only in such clauses) shall cease to have effect:
 - (a) <u>clause 11;</u>
 - (b) <u>clause 12;</u>
 - (c) clause 13;
 - (d) clause 14;
 - (e) clause 15;
 - (f) clause 17;
 - (g) clause 18; and
 - (h) <u>clause 19.</u>
- 2. **Provision of User information**
- 2.1 The User must liaise with the T&SCo throughout the Construction Programme and provide all information relating to the User Facility and the User Works as is reasonably necessary to

assist the T&SCo in preparing the T&S Network Commissioning Programme, including any specific requirements relating to commissioning.

- 3. **Provision of T&S Network Commissioning Programme**
- 3.1 Not later than six (6) months prior to the T&S Network Commissioning Programme

 Commencement Date, the T&SCo must provide the User with a draft T&S Network

 Commissioning Programme for the T&SCo Commissioning Activities which shall incorporate the information provided by the User pursuant to clause 2.1 of Part C to this Schedule 6.
- 3.2 The User must, as quickly as practicable and in any event within two (2) months of receipt of the draft T&S Network Commissioning Programme, notify the T&SCo of any comments on the draft T&S Network Commissioning Programme, including any changes or variations to the draft T&S Network Commissioning Programme recommended by the User.
- 3.3 The T&SCo must give reasonable consideration to any comments provided by the User under clause 3.2 of Part C to this Schedule 6 and incorporate such comments where possible, provided that it shall be reasonable for the T&SCo not to incorporate such comments where such changes or variations may be reasonably likely to cause the T&SCo to be in breach of its Licence.
- 3.4 Within one (1) month of receipt of the comments provided by the User under clause 3.1 of Part C to this Schedule 6, the T&SCo must deliver to the User the final version of the T&S Network Commissioning Programme.
- 3.5 The T&S Network Commissioning Programme shall be implemented by the Parties and their sub-contractors in accordance with its terms.
- 4. Activities and responsibilities of parties
- 4.1 **Drafting instruction**: the T&SCo and User are to agree and set out here provisions dealing with the following matters:
 - (a) T&S Network Commissioning Plan
 - Details of the T&S Network Commissioning Plan.
- <u>◆ The</u> activities and responsibilities of <u>the</u> T&SCo and the User under the T&S Network Commissioning <u>PlanProgramme</u>.
- 5. **Commissioning**
- 5.1 The Works shall be deemed to have been Commissioned on the date that the T&SCo certifies in writing to that effect.
- 6. (b) Notification of delay
- 6.1 If the User has reason to believe that there is or will be a User Delay for any reason (whether it is one entitling it to an adjustment to the User Facility Longstop Date or any other date under this Construction Agreement or not) it shall forthwith notify the T&SCo in writing of the circumstances giving rise to the delay and of the extent of the actual and/or anticipated delay.
- 6.2 If the T&SCo has reason to believe that there is or will be a T&SCo Delay for any reason (whether it is one entitling it to an adjustment to the T&SCo Network Longstop Date or any other date under this Construction Agreement or not) it shall forthwith notify the User in writing of the circumstances giving rise to the delay and of the extent of the actual and/or anticipated delay.

7. <u>Impact of delays</u>

- 7.1 Drafting instruction: the T&SCo and User to agree the process for rescheduling the activities and responsibilities of the T&SCo and the User under the Construction Programme and/or the T&S Network Commissioning Programme, provided that any such process shall be without prejudice to any decision of the Regulator with respect to such delays.
- 8. <u>Liquidated damages payable to the User</u>

Delayed commencement of T&S Network commissioning

- 8.1 In the event that the actual date of commencement of the T&S Network Commissioning
 Programme is later than the T&S Network Commissioning Programme Commencement
 Date, the T&SCo (if and to the extent that it is responsible for delayed commencement of the
 T&S Network Commissioning Programme beyond the T&S Network Commissioning
 Programme Commencement Date), shall be liable to pay the User Liquidated Damages for
 each day that the actual date of commencement of the T&S Network Commissioning
 Programme is later than the T&S Network Commissioning Programme Commencement
 Date.
- 8.2 Such Liquidated Damages shall cease to be payable in respect of any period after the date of actual commencement of the T&S Network Commissioning Programme.

Delayed completion of T&S Network commissioning

- 8.3 In the event that the actual date on which the Commercial Operations Date is achieved is later than Scheduled COD, the T&SCo (if and to the extent that it is responsible for the Commercial Operations Date being delayed beyond Scheduled COD) shall be liable to pay the User Liquidated Damages for each day that the actual date on which the Commercial Operations Date is achieved is later than Scheduled COD.
- 8.4 Such Liquidated Damages shall cease to be payable in respect of any period after the Commercial Operations Date.
- Notification Liquidated damages payable to the T&SCo of User Delay and notification to User of T&SCo Delay.

Delayed commencement of T&S Network commissioning

- 9.1 In the event that the actual date of commencement of the T&S Network Commissioning
 Programme is later than the T&S Network Commissioning Programme Commencement
 Date, the User (if and to the extent that it is responsible for delayed commencement of the
 T&S Network Commissioning Programme beyond the T&S Network Commissioning
 Programme Commencement Date) shall be liable to pay the T&SCo Liquidated Damages for
 each day that the actual date of commencement of the T&S Network Commissioning
 Programme is later than the T&S Network Commissioning Programme Commencement
 Date.
- 9.2

 Process for rescheduling the activities and responsibilities of T&SCo and User under Such
 Liquidated Damages shall cease to be payable in respect of any period after the T&S

 Network Commissioning Plan Programme Commencement Date.

(c) Liquidated damages

• Provision for liquidated damages to be set in accordance with Schedule 5 in response to User Delay and T&SCo Delay.

Delayed completion of T&S Network commissioning

- 9.3 In the event that the actual date on which the Commercial Operations Date is achieved is later than Scheduled COD, the User (if and to the extent that it is responsible for the Commercial Operations Date being delayed beyond Scheduled COD) shall be liable to pay the T&SCo Liquidated Damages for each day that the actual date on which the Commercial Operations Date is achieved is later than Scheduled COD.
- 9.4 Such Liquidated Damages shall cease to be payable in respect of any period after the Commercial Operations Date.

10. Costs of rescheduling

10.1

Provision for T&SCo and the User to In the event of User Delay and/or T&SCo Delay, each Party shall be responsible for its own costs of rescheduling of activities and responsibilities under the T&S Network Commissioning PlanProgramme.

11. (d) Longstop date

- Provision for the Longstop Date under this Construction Agreement to be set by reference to the T&SCo Network Longstop Date (as extended under T&SCo's Licence) and the User Facility Longstop Date (as extended under the User's Emitter Revenue Support Contract, where applicable).
- 11.1 Where the T&S Network Longstop date has been extended, the T&SCo must promptly notify the User in writing of the same.
- 11.2 Where the User Facility Longstop Date has been extended, the User must promptly notify the T&SCo in writing of the same.
- 11.3 In the event that the T&SCo has failed to achieve the Commercial Operations Date by the T&S Network Longstop Date, such failure shall constitute a Construction Longstop Default and paragraph 13 of Section J (General) of the Code shall apply.
- 11.4 In the event that the User has failed to complete the User Facility and/or User Works by the
 User Facility Longstop Date, such failure shall constitute a Construction Longstop Default and
 paragraph 12 of Section J (General) of the Code shall apply.
 - ◆ Provision Drafting instruction: Users and T&SCos to provide for the impact of extension of the T&S Network Longstop Date and /or the User Facility Longstop Date on the activities and responsibilities of the T&SCo and the User under the T&S Network Commissioning Plan.

12. **Becoming operational**

12.1 Once:

- (a) the T&SCo achieves the Commercial Operations Date and the T&SCo has certified that the Works have been Commissioned pursuant to clause 5 of Part C to this Schedule 6; and
- (b) the User has certified to the T&SCo that the User Facility is complete,

the T&SCo must forthwith notify the User in writing that the connection at the Delivery Point has become operational.

Schedule 7 Schedule 8

Works relating to Re-use Relating to Re-Use Service

Drafting instruction: this schedule is only to be included where applicable. If the User is required to pay for any T&SCo Works to the extent they relate to the Re-use Service, then this should be set out here.

Schedule 8

Conditions Precedent

<u>Drafting instruction</u>: any conditions precedent, as contemplated by Section C of the Code, are to be set out here.

T&SCo Conditions Precedent

<u>Drafting instruction</u>: these are intended to be any conditions precedent for the benefit of the T&SCo, and to be fulfilled by the T&SCo, such as the approval of the Regulator under the Change in Scope mechanism under its Licence. The date by which they are to be satisfied should also be included here.

User Conditions Precedent

Drafting instruction: these are intended to be any conditions precedent for the benefit of the User, and to be fulfilled by the User. The date by which they are to be satisfied should also be included here.

Signatures

CCS Network Code Exhibit C Connection Agreement

[Name of T&SCo]

[Name of User]

20

Contents

<u>1.</u>	<u>Definitions and Interpretation</u>	<u>311</u>
<u>2.</u>	Commencement and Term	<u>312</u>
<u>3.</u>	<u>Scope</u>	<u>312</u>
<u>4.</u>	Use of T&S Network	<u>313</u>
<u>5.</u>	Entry Provisions	<u>313</u>
<u>6.</u>	Installation of Measurement Equipment	<u>314</u>
<u>7.</u>	Provision of Security	<u>314</u>
<u>8.</u>	Registered Capacity	<u>314</u>
<u>9.</u>	Use of System Charges	<u>315</u>
<u>10.</u>	Re-use Service	<u>315</u>
<u>11.</u>	<u>Variations</u>	<u>315</u>
<u>12.</u>	General provisions	<u>315</u>
<u>13.</u>	Conditions Precedent	<u>316</u>
<u>14.</u>	[Anti-bribery and corruption]	<u>316</u>
<u>15.</u>	[Human rights]	<u>317</u>

Schedules

<u>1.</u>	Key User Connection Details	<u>318</u>
<u>2.</u>	Entry Provisions	<u>320</u>
<u>3.</u>	Measurement Equipment	<u>321</u>
<u>4.</u>	Re-use Service	<u>322</u>
<u>5.</u>	Conditions Precedent	<u>323</u>

THIS AGREEMENT is made on [●] 20[●]

BETWEEN:

- (1) [INSERT T&SCO] (Company no. [●]) whose registered office is at [●], (the "T&SCo" which expression includes its legal successors in title and permitted assignees); and
- (2) [INSERT USER] (Company no. [●]) whose registered office is at [●], (the "User" which expression includes its legal successors in title and permitted assignees),

each a "Party" and together, the "Parties".

RECITALS:

- (A) Pursuant to its Licence, the T&SCo maintains the CCS Network Code setting out the terms of the arrangements for connection to and use of the T&S Network by Users.
- (B) The User has applied for connection to and use of the T&S Network and pursuant to its Licence, the T&SCo is required to offer terms in accordance with the Code in this respect.
- <u>The T&SCo and the User are parties to the Code Agreement (being an agreement by which the Code is made contractually binding between the parties to the Code).</u>
- (D) This Connection Agreement documents details relating to the User's connection to and use of the T&S Network and is entered into pursuant to the Code and shall be read as being governed by it.

THE PARTIES AGREE AS FOLLOWS:

- 1. **Definitions and Interpretation**
- 1.1 Unless the subject matter or context otherwise requires or is inconsistent with the subject matter or context:
 - (a) capitalised terms used in the Connection Agreement have the same meaning given to them in Section K (Glossary) of the Code and in the Construction Agreement and, where relevant, are more particularly described in the Schedules to this Connection Agreement; and
 - (b) the interpretation provisions set out in the Code shall apply to the interpretation of this Connection Agreement.
- 1.2 <u>The following terms shall have the following meanings:</u>

Exhibit C Agreement Date means the date of this Connection Agreement;

1. Parties

The parties to the Connection Agreement shall be:

<u>Commencement Date</u> means the date when Commissioning has been completed in accordance with the Construction Agreement;

Conditions Precedent means the conditions precedent set out in Schedule 5;

Construction Agreement means the Construction Agreement entered into between the T&SCo and the User in relation to the Works required to connect the User Facility to the T&S Network at the Delivery Point;

Key User Connection Details means details set out in Schedule 1;

Scheduled Commencement Date means the date anticipated to be the Commencement

Date for this Connection Agreement in accordance with the Construction Programme and/or

Commissioning Programmes set out in the User's Construction Agreement, and which is
specified in the Key User Connection Details;

(a) T&SCo (being means the identified party to the identified in this Connection Agreement as such, which, in relation to each T&S Network or part of a T&S Network, shall be the entity that is the holder of a Licence authorising it to operate that T&S Network or part of a T&S Network); and

(b) User (being means the identified party to the identified in this Connection Agreement as such, which is the personentity other than T&SCo who that is bound by the provisions of the CCS Network Code and delivers carbon dioxide into the T&S Network at a Delivery Point).

- 2. Recitals Commencement and Term
- 2.1 T&SCo and the User are parties to the Code Agreement (being an agreement by which the CCS Network Code is made contractually binding between the parties to the CCS Network Code).
- 2.1 Subject to clause 13, this Connection Agreement becomes effective on the date of its execution.
- 2.2 This Connection Agreement is entered into pursuant to the CCS Network Code and shall be read as being governed by itshall continue to be in force until it is terminated in accordance with the provisions of Section J (General) of the Code.
- 3. Scope
- 3.1 The This Connection Agreement relates to the connection of the User Facility to the T&S Network at the Delivery Point and the User's rights to deliver carbon dioxide at the Delivery Point.
- 3.2 Schedule 1 sets out key User-specific details, which define the User's rights under the Code and this Connection Agreement.
- 3.2 The Key User Connection Details define the parameters of the User's connection to and use of the T&S Network and in particular describe the following:
 - (a) the User's contact details, as required for the purposes of Section J (General) of the Code;
 - (b) the T&S Network to which the User is connected at the Delivery Point;
 - <u>(c)</u> <u>the Delivery Point at which the User is entitled to deliver carbon dioxide; in accordance with and subject to the provisions of the Code;</u>

- (d) the User's Scheduled Commencement Date;
- (e) the nature of the User's User Facility which is connected to the T&S Network at the Delivery Point;
- (f) the User Type of the User, for the purposes of Section B (Governance) of the Code;
- (g) the Delivery Point Size of the Delivery Point;
- (h) whether the User is an Onshore User or an Offshore User;
- (i) the Registered Capacity allocated to the User (expressed in tCO₂/hour, over a set number of years), which represents the Registered Capacity held by the User as at the date of this Connection Agreement but which may subsequently change in accordance with the provisions of the Code; and
- (j) the Minimum Turndown Rate, the Minimum Ramp Rate and the Maximum Ramp Rate of the User Facility, which the T&SCo will have regard to pursuant to Section E of the Code.

4. Use of T&S Network

- 4.1 Subject to and in accordance with the terms of this Connection Agreement and the Code, from the Commencement Date the User has the right to use the T&S Network by delivering carbon dioxide from the User Facility at the Delivery Point.
- 4.2 3.3 Except as expressly provided otherwise in this <u>Connection</u> Agreement <u>and the Code</u>, each <u>partyParty</u> will perform its duties under this <u>Connection</u> Agreement <u>and the Code</u> in accordance with the standard of a Reasonable and Prudent Operator.
- 3.4 If there are any changes to the User contact details, Minimum Turndown Rate, the Minimum Ramp Rate and the Maximum Ramp Rate specified in Schedule 1, the User shall inform T&SCo as soon as is reasonably practicable.

4. Commencement and Term

- 4.1 The Connection Agreement commences on the Commencement Date specified in Schedule 1, which is the date when Commissioning has been completed in accordance with the Construction Agreement.
- 4.2 This Connection Agreement shall continue to be in force until it is terminated in accordance with the Provisions of Section J (General) of the CCS Network Code.

5. Entry Provisions

- 5.1 The Entry Provisions that apply to the User are specified in Schedule 2.
- 5.1 To the extent that any of the Entry Provisions specified in the Code are specific to an individual T&S Network, Schedule 2 identifies (by reference to the Code) those T&S Network-specific Entry Provisions that are applicable to the User's Delivery Point.
- 5.2 The Entry Provisions shall include any Local Requirements where, in accordance with the provisions of Section F (Network Design and Specification) of the Code, such Local Requirements are required to deal with any of the User's User Requirements.
- 5.3 5.2 The User recognises and acknowledges that:

- (a) the Entry Provisions are designed to protect the T&S Network and to ensure that the T&SCo can:
 - (i) safely transport and <u>or</u> store carbon dioxide within the T&S Network in compliance with its Legal Requirements.; and
 - (ii) measure the quality and quantity of carbon dioxide delivered by the User,
 - in compliance with the Code, all Regulatory Requirements and any contractual requirements; and
- (b) the User is required to fully comply with the Entry Provisions at all times in accordance with the Code and any failure to do so may constitute a breach of the Code.

6. Installation of Measurement Equipment

- 6.1 In accordance with the requirements of the Code, Measurement Equipment has been installed at the Delivery User's Metering Point to:
 - (a) determine mass flow of carbon dioxide for fiscal purposes; and
 - (b) measure and monitor the quantity and the critical components, quality of the and characteristics of carbon dioxide being delivered by the User at the Delivery Point as against the Carbon Dioxide Specifications for integrity purposes.
- The details and schematics of the Measurement Equipment that has been installed by the User, as specified and approved by T&SCo, in accordance with the requirements of the Code (including the Entry Provisions) are set out in Schedule 3. and Metering Point that are referred to in paragraph 6.1 are set out in Schedule 3.

7. **Provision of Security**

- 7.1 The User shallmust provide the Required Security:
 - (a) the Registered Capacity Financial Security, in accordance with Section H Section E
 (Network Use and Capacity) of the Code, for an amount calculated by the T&SCo in accordance with Section H Section E (Network Use and Capacity) of the Code; and
 - (b) the Required Security, in accordance with Section E Section H (Charges, Invoicing and Payment) of the Code, for an amount calculated by the T&SCo in accordance with Section E Section H (Charges, Invoicing and Payment) of the Code.
- The User must ensure that the Registered Capacity Financial Security and the Required Security are in place before the User is entitled to start delivering carbon dioxide at the Delivery Point, and in any case within fourteen (14) Days of the Commencement Date.

Drafting instruction: the level of financial security under Section E has been set at zero.

- 7.2 The amount of the Required Security to be provided by the User as the Commencement Date is set out in Schedule 1, as calculated in accordance with the Code, but this shall be revised from time to time in accordance with the Code.
- 8. Registered Capacity Allocation
- 8.1 The User shall acquire Registered Capacity to use the T&S Network in accordance with Section Eof the CCS Network Code.

- 8.1 8.2 The Registered Capacity that has been allocated to the User pursuant to the Code and that the User holds as at the date of entering into this Connection Agreement is set out in Schedule 1 Commencement Date is set out in the Key User Connection Details.
- 8.2 If the User wishes to increase or decrease the amount of Registered Capacity held by the User, the User must apply to do so in accordance with the provisions of Section E (Network Use and Capacity) of the Code.
- 9. Use of System Charges
- 9.1 From the Commencement Date, the User must pay all Use of System Charges in accordance with the Code.
- 10. **Re-use Service**
- 10.1 This clause 10 applies where:
 - <u>the T&SCo has agreed to provide the Re-use Service to the User in accordance with Section F (Network Design and Specification) of the Code; and </u>
 - (b) <u>details of the terms for the provision of that Re-use Service are set out in Schedule 4.</u>
- 10.2 The User shall pay all Charges Subject to and in accordance with the CCS Network.

 Code terms of this Connection Agreement and the Code, from the Commencement Date the User has the right to take delivery of carbon dioxide from the T&S Network at the User's Re-use Delivery Point.
- 10.3 From the Commencement Date, the User must pay the Re-use Service Charge for the Re-use Service provided by the T&SCo to the User.
- 11. 40. Variations
- Subject to clause 11.2, no variation to this Connection Agreement shall be effective unless made in writing and signed by or on behalf of both the T&SCo and the User.
- 11.2 10.1 The T&SCo and the User shallmust effect any amendment required to be made to this Connection Agreement as a result of a modification Modification to the CCS Network Code or the T&S Licence, or, including as a result of an order or direction made in accordance with the Regulatory Requirements any modifications to or directions issued by the Regulator in connection with the Licence or any Project-Specific Document referred to in the T&SCo's Licence.
- 11.3 10.2 The User authorises and instructs the T&ScoSCo to make any such amendment required under clause 11.2 on its behalf and undertakes not to withdraw, qualify or revoke such authority or instruction at any time.

11. Re-use Service

- 11.1 This paragraph 11 applies where T&Sco has agreed to provide the Re-use Service to the User in accordance with Section F of the Code.
- 11.2 Details of the Re-use Service that T&Sco has agreed to provide to the User are set out in Schedule 4.
- 12. General provisions
- Paragraphs [14 to 20] of Section J (General) of the CCS Network Code apply to this Connection Agreement as if they were set out in this Connection Agreement.

Note: this refers to the general boilerplate provisions in Section J of the CCS Network Code.

- 13. Assignment Conditions Precedent
- 13.1 The provisions of, and the rights and obligations of the parties pursuant to, this Connection Agreement (other than this clause 13) are conditional upon the Conditions Precedent being:
 - (a) fulfilled by the T&SCo, in the case of the T&SCo Conditions Precedent;
- 13.1 Neither party shall assign its rights and obligations under this Connection Agreement except in accordance with the provisions of clauses 13.2 and 13.3.
- 13.2 The User shall only be entitled to assign its rights and obligations under this Agreement to a third-party where:
 - (a) the User also assigns the ownership and operation of the User Facility to the third party;
 - (b) <u>fulfilled by</u> the User-has obtained the prior written consent of T&Sco, which shall not be unreasonably withheld; and, in the case of the User Conditions Precedent; or
 - (c) the third party becomes bound by the Code and this Agreement waived by the party for whose benefit the relevant Condition Precedent has been included (as specified in Schedule 5).
- 13.3 T&Sco shall only be entitled to assign its rights and obligations under this Agreement to a thirdparty where:
 - (a) T&Sco also assigns the ownership and operation of that part of the T&S Network that includes the Delivery Point to a third party holding a Licence; and
- The T&SCo must use reasonable endeavours to fulfil or procure the fulfilment of the T&SCo
 Conditions Precedent as soon as reasonably practicable, and in any event no later than the
 date specified in Schedule 5.
- 13.3 The User must use reasonable endeavours to fulfil or procure the fulfilment of the User Conditions Precedent as soon as reasonably practicable, and in any event no later than the date specified in Schedule 5.
- 13.4 The parties shall keep each other informed about the status of the Conditions Precedent and each party shall notify the other party when a Condition Precedent has been fulfilled.
- 14. [Anti-bribery and corruption]
- 14.1 [The Parties acknowledge that the intent behind this Connection Agreement is to ensure consistency with Legal Requirements relating to:
 - (a) the prevention of corruption and bribery ("ABC Laws"); and
 - (b) the third party becomes bound by the Code and this Agreementanti-money laundering ("Anti-money Laundering Laws").
- 14.2 In performing its obligations and activities in accordance with this Connection Agreement and the Code, each Party must comply with all applicable ABC Laws and Anti-money Laundering Laws.

Each Party represents, warrants and undertakes to the other Party that it has not, directly or indirectly, engaged in any bribery or corruption or other act that may contravene any ABC Laws or Anti-Bribery Laws.]

Drafting instruction: this clause 14 (Anti-Bribery and Corruption) is optional for inclusion by the Parties. It is anticipated that the Parties may agree alternative provisions regulating the subject matter contained within this clause 14.

- 15. **14. Definitions**[Human rights]
- 15.1 [In performing its obligations and activities in accordance with this Connection Agreement and the Code, each Party must:
 - (a) comply with all applicable Human Rights Laws; and
 - (b) take reasonable steps to ensure its contractors and consultants comply with all applicable Human Rights Laws in performing any obligations or activities in connection with the Connection Agreement and the Code,

where "Human Rights Laws" means all Legal Requirements regarding respecting human rights, the use of labour (including the terms and conditions of employment), prevention of smuggling and trafficking of human beings and modern slavery and immigration.]

Drafting instruction: this clause 15 (Human Rights) is optional for inclusion by the Parties. It is anticipated that the Parties may agree alternative provisions regulating the subject matter contained within this clause 15.

Capitalised terms used in the Connection Agreement have the meaning given to them in CCS Network Code and, where relevant, are more particularly described in the Schedules to this Connection Agreement.

Schedule 1 Schedule 1

Details relating to the User's Connection Agreement

Key User Connection Details

User	[insert company details]
User contact details	[insert details, for purposes of notice provisions (paragraph 1.4 of Section J under the Code]
T&S Network	[description to be inserted]
User Facility	[description to be inserted]
<u>User Type</u>	[User's User Type, by reference to the definition of User Type in the Code, to be inserted]
Scheduled Commencement Date	[to be inserted]
Delivery Point	[description to be inserted, as well as a schematic showing the location of the Delivery Point and the demarcation of the User Facility and the T&S Network]
Delivery Point Size	[to be inserted, expressed in tCO2/hour]
Type of Onshore/Offshore User by reference to the location of the Delivery Point	[Onshore User/Offshore User]
Commencement Date	[date to be inserted]
Required Security under- Section E	[£ to be inserted] Drafting instruction: the level of financial security under- Section E has been set at zero.
Required Security under Section H	[£ to be inserted]
Registered Capacity as at the date of the Connection Agreement	[to be inserted – note that this will also be recorded in the T&S Network Portal] [this is to be expressed in tCO2/hour]
Period of Registered Capacity	[insert years] from the Commencement Date [note that this will also be recorded in the T&S Network Portal]
Minimum Turndown Rate	[to be inserted]

	Note: further consideration is being given to whether the Minimum Turndown Rate could vary depending on, for example, the season.
Minimum Ramp Rate	[to be inserted]
Maximum Ramp Rate	[to be inserted]

Schedule 2 Schedule 2

Entry Provisions

Carbon Dioxide Specifications	[insert here which part of Annexure B (Carbon Dioxide Specifications set out in the Code apply to the User to be confirmed here) applies to the T&S Network to which this Connection Agreement relates]
Measurement Requirements	[insert here which part of Annexure C (Measurement Requirements set out in the Code apply to the User to be confirmed here) applies to the T&S Network to which this Connection Agreement relates]
Local Requirements	[it is intended that here will be set out any additional or alternative technical requirements that may apply to the User' Delivery Pointfor the User's Delivery Point that are necessary due to the nature of the User Facility or its particular operating processes or which otherwise reasonably warrant a deviation from the Carbon Dioxide Specifications or Measurement Requirements for the T&S Network are included here — e.g. any changes to the Carbon Dioxide Specification that may be agreed in accordance with the provisions of the Codetemperature and pressure requirements or carbon dioxide monitoring requirements]

Schedule 3 Schedule 3

Measurement Equipment

Details of the Measurement Equipment at the Delivery Point, including schematic with location of the Measurement Equipment Metering Point	[details to be inserted and schematic showing location to be attached to this schedule]
Metering Point	[details to be inserted describing the location of the Metering Point, including a schematic/map showing its location]

Schedule 4 Schedule 4

Re-use Service

Re-use Delivery Point	[description to be inserted, as well as a schematic showing the location of the Re-use Delivery Point]
Re-use Delivery Point Size	[to be inserted, expressed in tCO ₂ /hour]
Re-use Meter	[details of meter requirements to be inserted]
Re-use Service Charge Rate	[to be inserted — in £[x[/tCO ₂]] Note: the approach to charging for any- operating costs associated with the Re-use- Service is being further considered, but the- rate is expected to be cost reflective and- calculated on a basis applied to all Users in a- consistent manner.
Access right to inspect equipment/infrastructure associated with Re-use Service	[details of User's access rights to be inserted, where there is equipment/infrastructure on a site owned or controlled by T&SCo]
Re-use Service Operational Procedures	[details to be inserted of the arrangements agreed between the parties, in accordance with Section F of the Code, for offtake of carbon dioxide by the User at the Re-use Delivery Point]

Schedule 5

Conditions Precedent

<u>Drafting instruction</u>: any conditions precedent, as contemplated by Section C of the Code, are to be set out here.

T&SCo Conditions Precedent

<u>Drafting instruction</u>: these are intended to be any conditions precedent for the benefit of the T&SCo, and to be fulfilled by the T&SCo, such as the approval of the Regulator under the Change in Scope mechanism under its Licence. The date by which they are to be satisfied should also be included here.

User Conditions Precedent

<u>Drafting instruction</u>: these are intended to be any conditions precedent for the benefit of the User, and to be fulfilled by the User. The date by which they are to be satisfied should also be included here.

Exhibit D CDS Accession Agreement

Signatures

CCS Network Code Exhibit D CDS Accession Agreement

T&SCo

<u>and</u>

CDS

and

New T&SCo

Accession agreement to the Central Data Service Contract.



Contents

<u>1.</u>	<u>Interpretation</u>	<u>327</u>
<u>2.</u>	Admission as a Party	<u>327</u>
<u>3.</u>	<u>Counterparts</u>	328

THIS CDS ACCESSION AGREEMENT is made on [●] 20[●]

The Note: the CDS Contract will be signed by the Original first T&SCo(s) to appoint the CDS. Any T&SCo that is required to become a Party to the CDS Contract after the CDS Contract is signed will become a Party to the CDS Contract by signing a CDS Accession Agreement.

1. PARTIES BETWEEN:

The parties to the CDS Accession Agreement shall be:

- (1) (a) a **T&SCo** that is already a party to the CDS Contract;
- (2) (b) the CDS; and
- (3) (c) [a **New T&SCo** (as this term is defined in the CDS Contract)].]1

Note: the details of the CDS Accession Agreement are subject to the terms of the CDS Contract itself, and will be further developed once the CDS Contract is developed.

2. RECITALS:

- (A) 2.1-The CDS Contract dated [date to be inserted], attached to this CDS Accession Agreement as an exhibit, appoints the CDS.
- (B) 2.2-The [New T&SCo] is acceding to the CCS Network Code and in accordance with the provisions of the CCS Network Code, the [New T&SCo] wishes to become a Party to the CDS Contract.

THE PARTIES AGREE AS FOLLOWS

- 1. 3. Interpretation
- 1.1 Capitalised terms used in this CDS Accession Agreement shall have the meaning given to them in the CDS Contract and the CCS Network Code.
- 1.2 <u>Note: reference here any boilerplate provisions set out in the CDS Contract.</u>
- 2. 4. Admission as a Party
- 2.1 4.1 The T&SCo(s) and the CDS admit the [New T&SCo] as a Party under the CDS Contract.
- 4.2 The [New T&SCo] accepts its admission as a Party and undertakes to perform and to be bound by the CDS Contract as a Party from the date of this <u>CDS Accession</u> Agreement.
- 4.3-This <u>CDS Accession</u> Agreement and the CDS Contract shall be read and construed as one document and references (in or pursuant to the CDS Contract) to the CDS Contract should be read and construed as references to the CDS Contract and this <u>CDS Accession</u> Agreement.

¹ Note: the details of the CDS Accession Agreement are subject to the terms of the CDS Contract itself, and are to be further developed once the CDS Contract is developed.

3. <u>Counterparts</u>

This CDS Accession Agreement may be executed in any number of counterparts and all of those counterparts taken together will be deemed to constitute one and the same instrument.

THIS CDS ACCESSION AGREEMENT has been entered into on the date first above written.

Signatures

CCS Network Code

Annexures

Annexure A Annexure A - Carbon Dioxide Specifications - General

The carbon dioxide specifications shall as a minimum:

- (a) recognise that >95mol% CO₂ is the industry standard;
- (b) recognise that the combined non-condensable content of <4mol% is the industry standard;
- (c) set the H₂O specification such that, and with sufficient margin, an aqueous phase is never present during any operational scenario, including transient operations;
- (d) ensure that impacts of all impurities in the CO₂-stream at all operational conditions are considered, when determining the maximum value of the saturation pressure;
- (e) consider the impact of lighter impurity components on refrigerated storage where the CO₂ stream itself is being used as the refrigerant;
- (f) take into consideration the impact of lighter components on the potential for running ductile fracture of pipelines carrying a CO₂ stream in dense phase;
- (g) ensure that the impact of the level of H₂ on hydrogen-enhanced crack propagation behaviour is considered, and that sufficient margin to crack growth exists;
- (h) in the event of a release of CO₂-stream, ensure that the local hazard associated by any single impurity component, is always lower, and with sufficient margin, than the hazard associated with the CO₂ itself;
- (i) note that in dense phase the concentration of an impurity in the released CO₂-stream can be different from that in the original fluid, which can result in a more corrosive mixture remaining in the pipeline;
- (j) where an impurity may be liquid or solid on release to the environment, the hazard associated with this release shall be considered;
- (k) where an impurity may accumulate anywhere in the CCS/CCU chain, the hazard associated with this accumulation shall be considered;
- consider the corrosion risk of induced aqueous phases in the specification for hygroscopic components that may be present as liquids in a gaseous CO₂ stream, such as glycols, amines, and methanol;
- (m) consider this corrosion risk in the specification for polar light components that may impact the corrosivity of an aqueous phase induced by a hygroscopic impurity;
- (n) assume that the products of possible chemical reactions, either between different impurities in a CO₂ stream or chemical reactions between impurities and the CO₂, are present in the CO₂ stream, and consider their impact(s);

- (o) keep to a practical minimum the presence of liquids that may accumulate in a gaseous CO₂ stream;
- (p) keep to a practical minimum the presence of solids in a CO₂ stream; and
- (q) consider the impact of solid particles within the CO₂ stream on equipment such as compressors, and on the injection reservoir itself (if appropriate).

Annexure B - Carbon Dioxide Specifications - T&S Networks

Drafting instruction: on accession of a T&SCo to the Code, a new Part will be incorporated into this Annexure B that sets out the Carbon Dioxide Specifications for that T&SCos T&S Network.

Annexure C Annexure C – Measurement Requirements

Drafting instruction: on accession of a T&SCo to the Code, a new Part will be incorporated into this Annexure C that sets out the Measurement Requirements for that T&SCos T&S Network. The CO₂ Quality Monitoring Procedure for each T&S Network (when developed in accordance with Section G) is to be incorporated into the relevant T&SCo's Part of this Annexure C following a Code Modification in accordance with Section G.

Note: this Annexure must include details of how carbon dioxide pressure and temperature is to be monitored by T&SCo and Users, including commonly accepted standards to be applied and details of tolerance thresholds.

Annexure D Annexure D - CO₂ Quality Monitoring Procedure Terms of Reference

1. Purpose

The Common Interface Procedure for the CO₂ Quality Monitoring Procedure will form part of the Measurement Requirements and therefore the Entry Provisions and will:

Establishing T&S Network Monitoring Plan

- (a) establish risk-based plans for monitoring of components in the carbon dioxide stream for each T&S Network, including:
 - (i) for each component, identification of acceptable sampling techniques and testing frequencies to be adopted by Users on that T&S Network; and
 - (ii) requirements for on-site testing and off-site laboratory testing, minimum accreditations, any prohibited analytical techniques, industry standard testing turnaround times and when and how monitoring test results are to be sent to T&SCo;

Defining a non-compliance with the Entry Provisions

- (b) define what constitutes a non-compliance of the Entry Provisions in respect of each component, including:
 - (i) confirming the acceptability of data treatment (including having regard to the expected variability of a User's carbon dioxide stream components), result averaging periods, maximum allowable durations of exceedance, frequency of exceedances (i.e. based on historic data on number of exceedances and level of exceedance above limit) and/or any other measurements reasonably agreed for the determination of whether a component does not comply with the Entry Provisions;
 - (ii) the role of any systems for giving an early warning that carbon dioxide may not be in compliance with the Entry Provisions; and
 - (iii) categorise components by reference to the risk that non-compliance with the Entry Provisions may have on the T&S Network;
 - (iv) the following details:

Note: this table is to be populated in the Common Interface Procedure.

Component	Limit (incl. Units)	Exceedance threshold	Type/frequency of monitoring (e.g. online continuous, offline monthly)
[Drafting instruction:	[Drafting instruction:	[Drafting instruction:	[Drafting instruction:

Component	Limit (incl. Units)	Exceedance threshold	Type/frequency of monitoring (e.g. online continuous, offline monthly)
this column is intended to be populated with the components listed in the T&SCo's Annexure B (Measurement Requirements – T&S Network)]	this column is intended to be populated with the limits and units of measurement for the respective components listed in the T&SCo's Annexure B (Measurement Requirements – T&S Network)]	this column is intended to be populated with details of the exceedance limit and number of occasions such limit is to be exceeded / averaging in respect of each component before carbon dioxide is considered non-compliant with the Entry Provisions]	this column is intended describe how a component will be monitored and the frequency of such monitoring]
[x]	[x]	[x]	[x]

Monitoring response following a non-compliance or suspected non-compliance

- (c) set out the monitoring response following a suspected or confirmed non-compliance with the Entry Provisions by the T&SCo or a User, including:
 - (i) approaches and thresholds on the provision of early warning notices by T&SCo and Users;
 - (ii) subject to paragraph 1(c)(iii), communication and response timeframes which reflect the principle that if a User becomes aware of a non-compliance with the Entry Provisions (as defined at paragraph 1(b)), it must immediately take steps to cease delivery of carbon dioxide in accordance with paragraph 1(b) and the Isolation Procedure;
 - (iii) circumstances where a User is permitted to continue or resume flowing carbon dioxide that is not in compliance with the Entry Provisions (as defined at paragraph 1(b);
 - (iv) the nature and duration of any additional sampling and testing that may be requested by the T&SCo or a User;
 - how compliance of carbon dioxide with the Entry Provisions can be demonstrated and confirmed following a suspected or confirmed non-compliance with the Entry Provisions; and
 - (vi) the role of the CO₂ Quality Monitoring Procedure in complying with the Isolation Procedure, including the role in determining whether full isolation is required, timeframes and the process for reconnection to the T&S Network;

Communication following a suspected or confirmed non-compliance

(d) detail the communication requirements between T&SCo and Users in the event of a suspected or confirmed non-compliance with the Entry Provisions;

Derogation

(e) detail how a User may seek a derogation from the need to monitor for a component by demonstrating that such component would not reasonably be expected to be present in the carbon dioxide stream delivered by that User or would be present but at a level that is below an agreed threshold, acknowledging that any such agreed derogation must be specified in Schedule 2 (Entry Provisions) to the Connection Agreement in order to apply to User;

Periodic review of the CO₂ Quality Monitoring Procedure

(f) define a process for periodically reviewing and if appropriate, requiring update of the CO₂ Quality Monitoring Procedure, to ensure it remains effective, risk based and value for money. This shall include a requirement that the T&SCo provide evidence of and justification for any requirements to alter the monitoring approach set out in the CO₂ Quality Monitoring Procedure (provided that any changes shall only be effected through a Code modification pursuant to Section B (Governance);

Compliance with Entry Provisions

(g) define a process for testing and demonstrating compliance with the Entry Provisions, as part of the User's Commissioning Programme (including any additional sampling and testing above the agreed CO₂ Quality Monitoring Procedure and when/how test results are sent to T&SCo); and

Change in operating process / components

(h) detail the communication requirements in the event of a change in the carbon dioxide composition (including identification of components not specified in the Carbon Dioxide Specifications) and/or in the operating process (or otherwise) of a User that may change the components (including levels) to be delivered to the T&S Network and define a process for T&SCo and Users to engage in a review and update to the CO₂ Quality Monitoring Procedures required for a User (which shall be effected through a variation to Schedule 2 (Entry Provisions) to the Connection Agreement).

2. Additional Requirements

Monitoring requirements set out in the CO₂ Quality Monitoring Procedure will:

(a) in the absence of a commonly accepted standard, not exceed the capabilities of best-in-class sampling, monitoring and analysis technologies (including sampling frequency,

- accuracy/uncertainty and limit of detection) existing at the date of finalisation of the CO₂ Quality Monitoring Procedure; and
- (b) balance the commercial interests between T&SCo and the User with respect to the cost and risk of operation of the T&S Network and User Facility and as a result not place disproportionate cost on a User relative to the operational issues that could arise for the T&S Network from the presence or level of any given component in the carbon dioxide; and
- (c) set out a sampling methodology that provides for samples of carbon dioxide to be as representative as possible of a User's carbon dioxide stream.

3. User specific requirements

The CO₂ Quality Monitoring Procedure will:

- (a) describe actions to be taken and obligations to be met where a User can evidence that an allowable threshold as defined in Annexure B (Carbon Dioxide Specifications – T&S Networks) cannot reasonably be achieved to satisfy paragraph 2(a); and
- (b) acknowledge that to the extent that a User's Connection Agreement contains any derogations from the obligations set out in the CO₂ Quality Monitoring Procedure that are reasonably necessary having regard to the User's operational processes and anticipated carbon dioxide flow with respect to any component, including sampling and testing requirements for any component, the terms of the Connection Agreement shall prevail.

Annexure E Annexure E – Technical Parameters – T&S Networks

Drafting instruction: on accession of a T&SCo to the Code, a new Part will be incorporated into this Annexure F that sets out the Technical Parameters for that T&SCos T&S Network.

Annexure F Annexure F – Emergency Procedure Terms of Reference

1. Purpose

- 1.1 The purpose of a Common Interface Procedure for emergencies is to:
 - (a) provide a means of defining a Network Emergency; and
 - (b) execute a co-ordinated response; and
 - (c) establish a hierarchy or communication across the Transportation & Storage Network; and
 - (d) ensure that T&SCo and User specific Emergency Response Plans (ERPs) effectively include consistent and complimentary response to Network emergencies; and
 - (e) ensure a constant state of readiness to respond to Network emergencies and to clearly define the requirements to carry out emergency response exercises and drills.
- 2. **Definition of a network emergency**

[TBD]

3. Co-Ordinated response to a network emergency

[TBD]

4. Communication hierarchy during a network emergency

[TBD]

5. Emergency response plans – T&SCo and User

[TBC]

6. Ensuring a constant state of readiness

[TBC]

Annexure G Annexure G – Start-up and Shut-down Procedure Terms of Reference

1. Purpose

The Common Interface Procedure for Start-up, Shutdown and Recovery post Interruption Procedure will:

- (a) establish an agreed method for communication between the T&SCo and User(s) in the event of; a process interruption, start-up, shutdown and recovery following an interruption;
- (b) detail the communication protocols and methods between the T&SCo and User(s);and
- (c) detail the hierarchal arrangements (where necessary) for shut off and reintroduction of User(s) from and back into the T&S Network.

Annexure H - Isolation Procedure Terms of Reference

1. Purpose

The Common Interface Procedure for isolations will:

- (a) detail the requirements for the safe isolation of T&SCo and User(s) across defined boundaries; Process/Piping, Electrical, Instrumentation/Control and Mechanical, including:
 - (i) agreed industry isolation standard (e.g. HSG 253);
 - (ii) the requirements for cross-party tagging (Lockout/Tagout LOTO);
 - (iii) the requirement for cross-party proof of isolation integrity. i.e., proving zero energy; and
 - (iv) reinstatement of plant and equipment;
- (b) detail the communication requirements in the event of a planned event that causes loss in data communication systems (hardwired and wireless), loss of control signals, loss of metering/measurement data;

Isolations relating to Non-Compliant CO2

- (c) detail the responsibilities of T&SCo and User(s) relating to isolation in the event that carbon dioxide has been confirmed as Non-Compliant CO₂ in accordance with the Entry Provisions, including setting out the role of the CO₂ Quality Monitoring Procedure in complying with this isolation procedure;
- (d) require the T&SCo to have regard to:
 - (i) the findings of monitoring responses (as defined in the CO₂ Quality Monitoring Procure); and
 - (ii) the timeframes in which the findings are expected to be delivered,

to inform risk based decisions on the timeframes for completion of any isolation and the duration of any temporary consent by the T&SCo for the User to continue to flow carbon dioxide, notwithstanding any suspected or actual Non-Compliant CO₂ being confirmed, pending further monitoring information becoming made available by the User;

- (e) require the T&SCo to have regard to:
 - (i) historical data of a User's carbon dioxide flow; and
 - (ii) the expected variability of a User's carbon dioxide flow,

in determining whether to require full or partial cessation of flow of carbon dioxide through a Delivery Point;

- (f) detail the communication requirements and process relating to carbon dioxide flow where the User has notified the T&SCo that it has reason to believe that there will be a temporary deviation of a component of carbon dioxide that might result in Non-Compliant CO₂, having regard to the consequences of a full or partial cessation of flow of carbon dioxide when compared to the potential impact on the T&S Network and value for money resulting from such carbon dioxide being delivered to the T&S Network, including details of:
 - (i) any requirements for increased monitoring under the CO₂ Quality Monitoring Procedure; and
 - (ii) any other conditions relating to carbon dioxide flow following the issue of such notice,

until compliance with the Entry Provisions is demonstrated in accordance with the CO₂ Quality Monitoring Procedure;

- (g) detail the communication requirements and process relating to resumption of carbon dioxide flow:
 - (i) following confirmation that carbon dioxide is no longer Non-Compliant CO₂ in accordance with the Entry Provisions; and
 - (ii) where a User has requested to resume flow of carbon dioxide to the Delivery Point where the User can demonstrate to the T&SCo's reasonable satisfaction that, having regard to historical data in the absence of confirmation that carbon dioxide is no longer Non-Compliant CO₂, the User's carbon dioxide complies with the Entry Provisions; and
- (h) not require Users to take delivery of carbon dioxide from the T&S Network or vent or purge carbon dioxide that has passed through the Delivery Point and entered the T&S Network;

Isolations relating to results of a Validation Report, failure to grant T&SCo access or Measurement Equipment Error, destruction or damage

- (i) detail the responsibilities of T&SCo and User(s) relating to isolation in the event that:
 - the results of a Validation Report indicate that adjustment or replacement of Measurement Equipment is required under paragraph 6 of Section F (Network Design and Specification);
 - (ii) the T&SCo is prevented from exercising its access rights under paragraph 10 of Section F (Network Design and Specification); or
 - (iii) any component of the Measurement Equipment is destroyed or damaged or the User identifies a Measurement Equipment Error under paragraph 12 of Section F (Network Design and Specification),

including communication requirements and process relating to resumption of carbon dioxide flow where adjustment or replacement of Measurement Equipment has been demonstrated, T&SCo's access has been granted, the relevant destruction, damage or Measurement Equipment Error has been rectified or the Measurement Adjustment has been completed (as applicable) in accordance with Section F (Network Design and Specification) or at such earlier time as agreed by the T&SCo.

Annexure I Annexure I – Minimum Documentation Requirements

Note: it is expected that this Annexure will be populated with templates, naming protocols, transfer intervals and methods of acquisition for what are considered to be key datasets across the network, eg flow rates, nominations, compositional analysis etc.

<u>Annexure J Annexure J – Information Publication Requirements</u>

Note: it is expected that this Annexure will contain minimum requirements for granularity and frequency of data release in relation to information to be published by T&SCos.

Annexure K – Retained Data CCS Network Code

Annexure K Annexure K – Retained Data

Note: it is expected that this Annexure will list the categories of data and information, which is required to be retained for at least ten (10) years.

Annexure L - CDS Terms of Reference

Background

Transport and Storage Companies are required to jointly procure a CDS which will be responsible for providing the Processed Flow Meter Data Statement which sets out the data to be used as inputs to the calculation of the Onshore Flow Charge, the Offshore Flow Charge and the Re-Use Service Charge by both T&SCo (for purposes of invoicing under the CCS Network Code) and the LCCC (for payments under the User revenue support contracts). The CCS Network Code sets out the minimum scope of the CDS Contract within Section I (Data) of the CCS Network Code and through this CDS Terms of Reference.

No	Subject	Terms		
1.	Parties	1.1	The parties to the CDS Contract shall be:	
			(a) the T&SCos and	
			(b) the CDS.	
2.	Scope	2.1	The scope of the CDS Contract shall as a minimum comprise:	
			(a) the scope of paragraph 9.2 of Section I (Data) of the CCS Network Code; and	
			(b) responsibility for aggregating the actual mass of carbon dioxide delivered by each User into their respective T&S Networks.	
3.	Roles & Responsibilities	3.1	The CDS will act independently of any User(s) or T&SCo(s) and shall comply with the scope as laid out in paragraph 2.1.	
4.	General standards of performance	4.1	The T&SCos should ensure alignment, among themselves, when reviewing the operational performance of the CDS, and ensure that the CDS, at all times, meets the minimum requirements of the duties under the CDS Contract.	
		4.2	The CDS shall notify each T&SCo when the CDS is capable of collecting data, following:	
			(a) the successful integration of the CDS; and	
			(b) the CDS confirming that it is operationally ready to performance its responsibilities under the CDS Contract.	

No	Subject	Terms		
		4.3	Each party shall, in connection with its performance under the CDS Contract, comply with applicable laws.	
		4.4	The CDS shall:	
			 (a) provide the services in accordance with the terms of the CDS Contract (including in such way as to meet the agreed performance standards and good industry practice; 	
			(b) ensure that the CDS complies with its specification;	
			(c) obtain and maintain, for the duration of the CDS Contract, all necessary licences, permissions, authorisations, exemptions, permits, necessary consents and approvals;	
			(d) become familiar with the terms of the CCS Network Code, and the Data Transfer Procedures and the role (express or implied) under such documents of the "CDS";	
			(e) review information made available to the CDS by the parties in order to become fully acquainted with the CCUS project including any variations to the CCS Network Code made from time to time (and any other relevant documentation or information referred to in or connected with the CCS Network Code); and	
			(f) follow the API Minimum Measures in accordance with paragraph 14 when using API/s.	
5.	Data Processing	5.1	Without prejudice to the Data Transfer Procedures, the T&SCos shall establish, document, implement and maintain written procedures for CDS activities within the CDS Contract, relating to the data processing and management of data from the boundary flow meter.	
		5.2	The CDS Contract shall set out parameters for the data to be processed and the CDS shall:	
			(a) accept data which is within the parameters defined in the CDS Contract;	
			(b) reject data which is outside of the parameters defined in the CDS Contract (and not include such	

No	Subject	Terms		
				data within any Processed Flow Meter Data Statement); and
			(c)	immediately notify the T&SCo of any such rejection and the nature of the divergence or non-compliance with such parameters once identified.
		5.3	Proce	DS must comply with the Data Transfer dures to the extent necessary to be in a position to out its obligations under the CDS Contract.
		5.4	The C	DS Contract shall contemplate the following ss:
			(a)	the CDS shall receive the data from each User's boundary flow meter via API so as to enable the CDS to have received all relevant data within one (1) Business Day following the relevant Billing Period;
			(b)	the CDS will aggregate the data in accordance with the Data Transfer Procedures of the CCS Network Code and the CDS Contract; and
			(c)	at the end of a Billing Period, the CDS will use the aggregated data to produce a Processed Flow Meter Data Statement for the purposes of paragraph 21.2 of Section H (Charges, Invoicing and Payment) which shall be sent to the LCCC and the T&SCos within two (2) Business Days following the end of the relevant Billing Period.
		5.5	the CI Re-us one (1	e a User is being provided with a Re-use Service, DS shall receive the data from each such User's e Meter via API. The data must be received within Business Day following the end of each relevant Period.
		5.6	detern	DS will process any Flow Meter Adjustment as mined under paragraph 13.3 of Section F (Network n and Specification).
6.	Access to Data	6.1	data d	DS must enable automated collection of metered lirectly from the User's SCADA, or similar including 5, operations systems via push and/or pull API.
		6.2	The C	DS must be capable of data collection from the

No	Subject	Terms	
			User's Measurement Equipment, data collection must be device and system agnostic.
		6.3	The CDS shall provide periodic maintenance in accordance with the maintenance terms set out in the CDS Contract, such terms shall align with good industry practice and shall not impact BAU activities where possible.
		6.4	The CDS Contract shall allow for reasonable downtime of the CDS in the event of an emergency and during the CDS's planned maintenance window, which shall be specified in the CDS Contract. The CDS shall provide to the T&SCos as much prior written notice as possible (having regard to the relevant circumstances) of any unplanned downtime.
		6.5	The CDS shall provide support services in accordance with the maintenance terms set out in the CDS Contract, such terms to align with good industry practice.
		6.6	The CDS is to produce the Processed Flow Meter Data Statement at the end of each Billing Period in accordance with Section H (Charges, Invoicing and Payment). This statement shall be accessible via an agreed communication method to the below bodies:
			(a) the T&SCos
			(b) the Users; and
			(c) the LCCC.
7.	CDS Data Disputes	7.1	Following a CDS Data Dispute and Expert Determination, the CDS will reprocess any data that is found to be subject to rectification.
		7.2	The CDS will follow instruction from the T&SCo to reprocess the Flow Meter Data Statement in accordance with paragraph 27 of Section H (Charges, Invoicing and Payment).
8.	Liability	8.1	The CDS appointment shall contain market standard liability limitation provisions.
9.	Collateral Warranties	9.1	The CDS shall be required to enter into collateral warranties with Users and with the LCCC which provide

No	Subject	Terms		
			as a r	ninimum:
			(a)	warranties and undertakings regarding extending duties of care to such third parties;
			(b)	insurances;
			(c)	restrictions on scope variations without such third party's consent; and
			(d)	provision of access to data and reports.
		9.2		dition to the requirements above, the collateral nty between the CDS and Users shall include:
			(a)	permitted assignment and accession by further Users to the collateral warranty;
			(b)	standing authority from Users to the CDS granting access to the Flow Meter and the Re-use Meter (if applicable);
			(c)	standing authority from the Users to the CDS for the CDS to collect and process that User's mass flow of carbon dioxide (measured in tCO ₂) as laid out in Section F (Network Design and Specification) and Section I (Data);
			(d)	a requirement for the CDS to collect the User's data in accordance with the Data Transfer Procedures; and
			(e)	permitted assignment and accession by further Users to the collateral warranty.
		9.3		dition to the requirements above, the collateral nty between the CDS and the LCCC shall include:
			(a)	a requirement for the CDS to provide a copy of the notice under paragraph 4.2 at the same time it provides such notice to the T&SCos
			(b)	a right for the LCCC to request evidence of the CDS' operational performance from the CDS no later than five (5) Business Days after any such request has been made; and
			(c)	a requirement for a copy of the relevant Processed Flow Meter Data Statement to be sent

No	Subject	Terms			
			to the LCCC and the T&SCos within two (2) Business Days following the end of the relevant Billing Period.		
		9.4	The form of any such collateral warranties shall be annexed to the CDS Contract.		
10.	Insurance	10.1	The CDS shall be required to take out and maintain professional indemnity and public liability insurance to agreed limits.		
		10.2	The CDS shall be required to maintain such insurances with reputable insurers who are acceptable to the T&SCo (such acceptance not to be unreasonably withheld or delayed) for a period of not less than twelve (12) years from the date of completion of the CDS's duties under the CDS Contract, provided that each such insurance is available at reasonable commercial rates.		
		10.3	The CDS will be required to provide evidence of such insurance.		
11.	Business Continuity and Disaster Recovery	11.1	The CDS shall implement and maintain business continuity/disaster recovery plan(s) in relation to the services, in accordance with good industry practice.		
12.	Information Security	12.1	The CDS shall have in place and maintain technical and organisational security measures which comply with good industry practice (which shall include at a minimum ISO 27001 and NIST cybersecurity framework).		
13.	API Minimum Measures	13.1	The CDS shall follow the following measures, as a minimum, when using API/s:		
			(a) Strong Authentication and Authorisation: Use OAuth 2.0, J to ensure only authorised access.		
			(b) Input Validation and Sanitisation: Validate and sanitise all inputs to prevent injection attacks.		
			(c) Encryption: Use HTTPS for secure communication and encrypt sensitive data at rest.		
			(d) Rate Limiting and Throttling: Implement controls to prevent Denial of Service attacks.		
			(e) Error Handling: Provide handling and error messages and log errors.		

No	Subject	Terms			
		(f) Monitoring and Logging: Continuously monitor AP activity and maintain logs for audit and forensic purposes.			
		(g) Regular Security Audits: Conduct regular security assessments and penetration testing to identify and fix vulnerabilities.			
14.	Variations	14.1 The T&SCo shall be entitled to request a variation to the CDS Contract where any modifications are made to these Terms of Reference in accordance with Section B of the CCS Network Code.			
15.	Definitions	"API" means application programming interface.			
		"SCADA" means Supervisory Control and Data Acquisition.			
		"DAHS" means Data Acquisition and Handling System.			

Annexure M – Independent Verifier Terms of Reference

The T&SCos shall ensure that the Independent Verifier appointment entered into with an Independent Verifier in connection with the User's procurement, installation, operation and maintenance of the Measurement Equipment in accordance with Section F of the CCS Network Code shall include the minimum terms and conditions set out in this Annexure M.

No	Subject	Terms		
1.	Parties	1.1	The parties to the Independent Verifier Appointment shall be:	
			(a) the T&SCo and	
			(b) the Independent Verifier.	
2.	Recitals	2.1	The Independent Verifier Appointment shall include recitals to capture the following principles:	
			(a) the T&SCo has entered into a Code Agreement for the purpose of giving effect to and binding itself to the CCS Network Code.	
			(b) in accordance with its obligations under the CCS Network Code, the T&SCo wishes to procure the services of an Independent Verifier which the T&SCo will also make available to Users.	
			(c) the T&SCo now wishes to appoint the Independent Verifier to perform certain verification activities and certain services in connection with the T&S Network to the T&SCo on the terms of this Independent Verifier Appointment.	
3.	Definitions	3.1	Terms in the Independent Verifier Appointment which are capitalised but not otherwise defined shall have the meaning given to such terms in the CCS Network Code.	
4.	Appointment and Conflicts of Interest	4.1	The T&SCo appoints the Independent Verifier under the Independent Verifier Appointment to perform the Services.	
		4.2	The Independent Verifier confirms its acceptance of the appointment and agrees to carry out and complete the Services and all other obligations and tasks which are ascribed to the Independent Verifier under CCS Network Code.	
		4.3	The Independent Verifier shall be required to declare	

No	Subject	Terms		
			immed would perfor	has no existing conflicts of interest and diately notify the T&SCo of any situation that compromise the Independent Verifier's ability to m a wholly independent verification or any other t of the Services.
5.	Key Scope and Responsibilities	5.1	The Services that the Independent Verifier will be required to perform under the Independent Verifier Appointment shall be set out in Schedule 1.	
		5.2		ninimum, Schedule 1 is required to provide for lowing:
			(a)	initial validation of installed Measurement Equipment in accordance with Measurement Requirements in paragraph 6.1(a) of Section F of the CCS Network Code;
			(b)	annual validation of Measurement Equipment in accordance with paragraph 6.1(b) of Section F of the CCS Network Code;
			(c)	preparation of a validation report of each validation carried out in accordance with paragraph (a) and (b);
			(d)	participation in any dispute resolution regarding the content of a validation report;
			(e)	inspection of Measurement Equipment following Measurement Equipment Errors; and
			(f)	determination of Measurement Adjustments following Measurement Equipment Errors.
6.	Accreditations/Licenses	6.1		ndependent Verifier shall be required to hold and ain, as a minimum, the following accreditations:
			(a)	be a competent auditor; and
			(b)	possess an appropriate level of knowledge of metering and technical systems of the same type as the Measurement Equipment.
7.	Standard of Performance	7.1	Verifie exerci diliger proper experi	forming its obligations under the Independent or Appointment, the Independent Verifier shall se the level of reasonable skill, care and note to be expected from a competent and rly qualified professional Independent Verifier denced in providing services of a similar scope, scale and complexity to the Services in relation

No	Subject	Terms	
			to a project of a similar scope, type, scale and complexity as the Project and in accordance with all Applicable Requirements and any other criteria, standards or codes relevant to the performance of the Services.
		7.2	The Independent Verifier acknowledges that the T&SCo and Users have relied on, and shall continue to rely upon, the professional skill and judgement of the Independent Verifier in the performance of the Services.
		7.3	The Independent Verifier shall:
			(a) carry out the Services (including without limitation the exercise of any discretion in the performance of the Services) independently, fairly and impartially to and as between the T&SCo, Users and the LCCC and while the Independent Verifier may take account of any representations made by the T&SCos, Users and/or the LCCC, the Independent Verifier shall not be bound to comply with any representations made by any of them in connection with any matter on which the Independent Verifier is required to exercise its professional judgement.
			(b) comply with all reasonable instructions given to it by the T&SCos and Users, except and to the extent that the Independent Verifier reasonably considers that any such instructions vary or might vary the Services or its authority or responsibilities under the Independent Verifier Appointment or prejudice or might prejudice the exercise by the Independent Verifier of its professional judgement in accordance with paragraph 6.1 or 6.1(a).
		7.4	All instructions to the Independent Verifier shall be given in writing by the T&SCo, and such person shall provide a copy of the same to the other, immediately following issue.
		7.5	The T&SCo agrees to co-operate with, and provide reasonable assistance to, the Independent Verifier to familiarise the Independent Verifier with all necessary aspects of the Measurement Equipment and relevant

No	Subject	Terms	
			facilities to enable the Independent Verifier to carry out its obligations under the Independent Verifier Appointment.
		7.6	The Independent Verifier shall familiarise itself with and shall be deemed to have full knowledge of the provisions of the CCS Network Code and shall be deemed to be aware of and to have taken full account of all the obligations, both expressed and implied, on the part of the T&SCos which are set out in the CCS Network Code.
		7.7	The Independent Verifier shall perform the Services in such manner and at such times that no act, omission or default of the Independent Verifier shall constitute or cause any breach by the T&SCos or Users of any requirements and/or obligations in the CCS Network Code.
		7.8	The Independent Verifier shall promptly and efficiently perform the Services as soon as reasonably practicable consistent with the requirements specified in paragraph 6.1 and [6.1(a).
		7.9	The Independent Verifier shall in providing the Services comply with all Applicable Requirements and, without limiting the generality of the foregoing, shall comply with all lawful and reasonable directions, instructions or requests relating to or in pursuance of the same given or made by and/or on behalf of the T&SCos or Users whilst the Independent Verifier is present on the T&SCo land or User land.
8.	Fee and Invoicing	8.1	The T&SCo must pay to the Independent Verifier the agreed fee, at the agreed intervals, subject to a valid application payment application being made.
		8.2	Each invoice submitted by the Independent Verifier to the T&SCo should clearly indicate the nature of the activities performed in accordance with the Services set out in Schedule 1.
9.	User Data and Data Protection	9.1	The T&SCo shall own and continue to own all data and information (including T&S Network Data and Measurement Data) generated by or on behalf of the T&SCo or Measurement Equipment owned by the Users.
		9.2	Each User must own and continue to own all data and

No	Subject	Terms	
			information (including its relevant User-Specific Data) generated by or on behalf of the User or Measurement Equipment owned by the User.
		9.3	The Independent Verifier shall not access, process, use or share any data provided by or on behalf of the T&SCo or a User other than as permitted under the Independent Verifier Appointment.
		9.4	The Independent Verifier may not restrict or withhold access to data or otherwise refuse to return to the T&SCo or a User (or its nominee) any data belonging to the T&SCo or a User.
		9.5	To the extent that any personal information is processed in connection with the Services, the Independent Verifier Appointment shall include adequate provisions detailing each party's roles and responsibilities in connection with any such processing. Each Party shall comply with applicable data protection laws in relation to the processing of personal information in connection with the Independent Verifier Appointment.
		9.6	The Independent Verifier shall implement and maintain reasonable archiving policies and procedures in relation to data of the T&SCo and Users held by it. The Independent Verifier's policies and procedures shall align with good industry practice.
10.	Disclosure and Confidentiality	10.1	Subject to paragraph 10.3, the Independent Verifier shall be required keep confidential all matters relating to the Independent Verifier Appointment and use all reasonable endeavours to prevent its employees and agents from making any disclosure to any person of any matter relating to the Independent Verifier Appointment.
		10.2	Upon termination of the Independent Verifier Appointment for whatever reason, the Independent Verifier shall be required to deliver up to the T&SCo and Users (as appropriate) all working papers, computer disks and tapes or other material aid copies provided to or prepared by the Independent Verifier in connection with the Independent Verifier Appointment.
		10.3	The Independent Verifier Appointment shall contain market standard carve outs in relation to

No	Subject	Terms	
			confidentiality obligations and permitted disclosures.
11.	Information Security	11.1	The Independent Verifier shall have in place and maintain technical and organisational security measures which comply with good industry practice (which shall include at a minimum ISO 27001 and the NIST cybersecurity framework).
12.	Obligations on the T&SCo	12.1	On reasonable prior notice, and in accordance with its obligations under Section F of the CCS Network Code, the T&SCo must provide (and procure that each User provides) the Independent Verifier with all access to the relevant site and any other areas and relevant information and other written material in the possession of the T&SCo as the Independent Verifier requires to carry out the Services.
		12.2	The T&SCo must appoint one (1) or more appropriate persons to liaise with the Independent Verifier.
		12.3	The T&SCo shall be responsible for ensuring that the Independent Verifier is kept informed of any variations to the CCS Network Code, including any updates to the Data Transfer Procedures or Common Interface Procedures.
		12.4	The T&SCo must ensure that the Independent Verifier is provided with all documents and other evidence the Independent Verifier requires in order to perform the Services.
13.	Limitation of Authority	13.1	The Independent Verifier shall not consent or agree to any waiver or release of any obligation of the T&SCo under the CCS Network Code.
		13.2	The Independent Verifier shall not express an opinion on and shall not interfere with or give any advice, opinion or representation in relation to any matters which are beyond its role and responsibilities under the Independent Verifier Appointment.
14.	Termination	14.1	The Independent Verifier Appointment shall contain immediate termination rights in relation to material breaches which remain unremedied, gross misconduct or delay in providing the services and insolvency.
15.	Assignment	15.1	Neither party shall assign, novate, sub-contract or otherwise transfer its benefits or obligations under the

No	Subject	Terms				
			Independent Verifier Appointment or any part thereowithout the prior written consent of the other party.			
16.	Insurance	16.1	The Independent Verifier shall be required to take ou and maintain professional indemnity and public liabilit insurance to agreed limits			
		16.2	The Independent Verifier shall be required to maintain such insurances with reputable insurers who are acceptable to the T&SCo (such acceptance not to be unreasonably withheld or delayed) for a period of not less than twelve (12) years from the date of the completion of the Independent Verifier's duties under the Independent Verifier Appointment, provided that each such insurance is available at reasonable commercial rates.			
		16.3	The Independent Verifier will be required to provide evidence of such insurances.			
17.	Collateral Warranties	17.1	The Independent Verifier shall be required to enter into Collateral Warranties with the LCCC and Users providing as a minimum:			
			(a) warranties and undertakings regarding extending duties of care to such third parties;			
			(b) insurances;			
			(c) restrictions on scope variations without such third party's consent;			
			(d) permitted assignment and accession by further Users to the collateral warranty;			
			(e) provision of access to data and reports;			
			(f) provision of relevant contractual notices.			
		17.2	The form of any such collateral warranty shall be annexed to the Independent Verifier Appointment.			
18.	Limitation of Liability	18.1	The Independent Verifier Appointment shall contain market standard liability limitation provisions.			
19.	Other	19.1	The Independent Verifier Appointment shall contain boilerplate provisions addressing governing law, severability, entire agreement, dispute resolution etc.			
20.	Schedule 1 (Services)	As a mi	a minimum, Schedule 1 shall address the following:			

No	Subject	Terms				
		Familiarisation of the CCS Network Code				
		20.1	The Ir	ndepen	dent Verifier shall :	
			(a)	CCS I and th role (e	ne familiar with the terms of the Project Network Code, Data Transfer Procedures ne Common Interface Procedures and the express or implied) under the such nents of the "Independent Verifier"; and	
			(b)	Independent of the leave of the	v information made available to the endent Verifier by the Parties in order to the fully acquainted with the CCS project ing any variations to the CCS Network made from time to time (and any other ant documentation or information referred or connected with the CCS Network),	
		to the extent necessary to be in a position to carry out its obligations pursuant to this Deed.				
		Verification Activities				
		20.2 The Independent Verifier shall carry out verification activities including, but not limited, to:		·		
			(a)	the Mo	or checks of the technical specification of easurement Equipment provided by a to ensure conformity with the urement Requirements;	
			(b)		or checks of the measurement tainty of the Measurement Equipment;	
			(c)	standa	s to ensure compliance with the ards required by the Measurement rements;	
			(d)		or checks of all points including, but not d, to the:	
				(i)	labelling of equipment;	
				(ii)	general standard of installation, being the good working practice standard;	
				(iii)	compliance of installation with any relevant manufacturers' requirements and/or guidelines;	
				(iv)	compliance with calibration	

No	Subject	Terms			
				requirements; and	
			(v)	compliance with maintenance requirements.	
		Site Visits 20.3 The Independent Verifier shall as required visit the si of the relevant Measurement Equipment and any other areas required to ensure it is able to discharge the functions which the Independent Verifier Appointment contemplates will be discharged by the Independent Verifier.			
		20.4 The Independent Verifier shall give the T&SCo and Users reasonable advance notice of the dates upon which it intends to visit the sites concerned.			
		20.5	site safety pr	dent Verifier shall comply with all relevant cocedures and processes as required by or User in charge of the relevant site.	
		Liaisor	n with T&SCo	s and Users	
		20.6	and Users in as are neces	dent Verifier shall liaise with the T&SCo anticipation of carrying out such activities sary, including the procurement of any ervices, for it to perform the Services.	
		Partici	pation in Disp	oute Resolution	
		20.7	Measuremer Measuremer	dent Verifier shall provide an opinion on nt Equipment Errors and any required nt Adjustments, where such matters are in accordance with Section F of the CCS de.	
		20.8	Independent Resolution P the extent the which have b	required by the T&SCo or a User, the Verifier shall participate in the Dispute Procedure of the CCS Network Code to at issues under the CCS Network Code peen referred to the Dispute Resolution elate to the Services.	

Annexure N – T&S Network Portal Minimum Specification

1. Publicly available information

- 1.1 The T&S Network Portal must have publicly available pages, to publish all the information that the T&SCo is required, in accordance with the Code, to make available for all Users, prospective Users and the public at large, including but not limited to:
 - (a) the T&SCo's Connection Application template, together with a succinct guide to the information that must be provided with the Connection Application, pursuant to paragraph 14.1 of Section C (Connection);
 - (b) the Registered Capacity of every User, pursuant to paragraph 3.7(b) of Section E (Network Use and Capacity);
 - (c) the required information relating to the Network Capacity specified in paragraph 6.1 of Section E (Network Use and Capacity);
 - (d) information about the quantity of residual Network Capacity that is available in the current and next six (6) Delivery Periods, pursuant to paragraph 21.5 of Section E (Network Use and Capacity);
 - details of each Maximum Annual Cumulative Flow that applies to each individual Storage Site, pursuant to paragraph 23.3 of Section E (Network Use and Capacity);
 - (f) details of how it has determined that the Maximum Annual Cumulative Flow will be exceeded, pursuant to paragraph 23.6(a) of Section E (Network Use and Capacity);
 - (g) an updated Maintenance Programme for the Planning Period ahead, pursuant to paragraph 30.1 of Section E (Network Use and Capacity);
 - (h) a draft of the Maintenance Programme pursuant to paragraph 31.1(a) of Section E (Network Use and Capacity);
 - (i) details of the matters referred to in paragraph 36.3 of Section E (Network Use and Capacity):
 - (j) information about all Capacity Constraints pursuant to paragraph 37.1(a) of Section E (Network Use and Capacity);
 - (k) information about Minimum Flow Deficits pursuant to paragraph 22 of Section E (Network Use and Capacity);
 - (I) a summary of its findings based of the data and records reviewed pursuant to paragraph 48.5(a) of Section E (Network Use and Capacity);
 - (m) the T&S Network Data specified in paragraph 6.19 of Section I (Data); and
 - (n) the Use of System Charging Statement for the current Charging Year and previous Charging Years.

2. User accounts in the T&S Network Portal

- 2.1 The T&S Network Portal must also allow for each User to have its own account, accessible using a log-in, to allow the User to send to the T&SCo and receive from the T&SCo various operational communications, as set out in the Code, including but not limited to:
 - (a) Forecasts, Nominations and Renominations made by the User;
 - (b) responses from the T&SCo to Nominations and Renominations;
 - (c) communications relating to Capacity Constraints, Minimum Flow Deficits and Emergencies;
 - (d) the issuance of Invoice Documents and Processed Flow Meter Data.
- 2.2 The T&S Network Portal must have the following minimum features to allow each User to send to the T&SCo and receive from the T&SCo various operational communications in an effective and efficient manner:
 - (a) security features to protect the integrity of the communications; and
 - (b) automation features to facilitate and allow:
 - (i) all communications to be issued and received in a user-friendly and time efficient manner;
 - (ii) communications from the T&SCo to be brought to the timely attention of Users; and
 - (iii) any operational communications from each User, such as Nominations and Renominations (and any other communications being made pursuant to Section E), to be accurate and to reduce the potential for human error for example, for input fields to automatically recognise details such as a User's Delivery Point and the User's Available Registered Capacity, and for Daily Nominations to be automatically rejected if they do not comply with the relevant parameters for compliance.